ResearchOnline@JCU



This file is part of the following work:

Papavasiliou, Shirley (2024) Access to Australian community palliative care services: the influence of peri-urban location. PhD Thesis, James Cook University.

Access to this file is available from: https://doi.org/10.25903/0yhy%2Dbn26

Copyright © 2024 Shirley Papavasiliou

The author has certified to JCU that they have made a reasonable effort to gain permission and acknowledge the owners of any third party copyright material included in this document. If you believe that this is not the case, please email researchonline@jcu.edu.au

Access to Australian community palliative care services: The influence of peri-urban location

Submitted by Shirley Papavasiliou In fulfilment of the requirements for the degree of Doctor of Philosophy College of Business, Law, and Governance James Cook University

[June, 2024]



Acknowledgements

I have received encouragement, support, and assistance from many people, too many to name individually. I would like to acknowledge and express my gratitude to every one of these individuals who played a role which supported completion of this PhD journey.

I am especially grateful to the clinicians, family caregivers and patients who participated in this study and shared your stories. You recognised the importance of what this research is trying to achieve and took the time to participate. Your contribution cannot be overstated as without your input, there would be no thesis.

I would like to express my sincere gratitude to my primary supervisor Dr Carmen Reaiche for your encouragement, perseverance, support and patience. You were always available when I needed guidance. I cannot image a better supervisor, mentor and friend. Carmen, you instilled in me the belief that I could take on this task and succeed.

Thank you to the supervisory team, Professor Stephen Boyle and Dr Reza Akbari for your patience, support and guidance throughout the research process. I value the different perspectives you each brought to the research and your insightful feedback. Thank you for giving so generously of your time. Without your support this thesis would not exist.

Special thanks to Dr Andrea Gregory for stepping in at short notice and assisting with the specialist psychologist input needed for this research.

A special thanks also to my editor, Tina Morganella. Your assistance with all the work you have done with editing my thesis and answering all my questions, is greatly appreciated.

I appreciate and acknowledge the assistance provided by Australian Government Research Scholarship and Stipend with funding for this research.

Finally, to my family, thank you for living through the PhD journey with me. You all recognised how much the subject of this research means to me personally and that many years of work, upskilling, study and research have led to this moment. To my husband Nick and son Jake, you tolerated the endless reams of journals and paperwork covering the dining table, and the frequent voicing of frustration at hurdles encountered on the way. To my daughters Chloe and Samantha and their partners Scott and Levi, thank you for the patience and understanding you demonstrated when responding to my seemingly endless cries for assistance and for arranging everything I needed to commence this journey.

Statement of the Contribution of Others

Nature of Assistance	Contribution	Names, Titles and Affiliations of Co-Contributors
Intellectual support and supervision	Supervisory team provided advice and guidance on study design, methodology, analysis, reporting and editorial assistance within the remit of their roles.	Primary Supervisor Associate Professor Carmen Reaiche, Secondary Supervisors Professor Stephen Boyle and Doctor Mohammadreza Akbari College of Law, Business and Governance, James Cook University
Financial support	 The researcher acknowledges the support received through: Australian Government Research Training Program Scholarship, and Research Training Program Stipend (RTPS) 	Australian Government
Data collection	Research assistance with distribution of link for participant recruitment.	Palliative Care Australia (PCA)
Editing support	Copy editing and proof-reading services.	Tina Morganella, The Editing Suite

Statement of the use of Generative AI

Generative AI technology was not used in the preparation of any part of this thesis.

Statement of Originality

I declare that this thesis is my own work and has not been submitted in any form for another degree or diploma at any university; and to the best of my knowledge and belief this thesis does not contain any material previously published or written by another person except where due reference is made in the text.

Abstract

Background

Evidence supports the benefits of palliative care provision for people diagnosed with a life-limiting illness and their family caregivers (Borbasi, 2017). Palliative care with a personcentred approach aims to optimise quality of life by prioritising the needs and values of individuals in every interaction with healthcare services. Ensuring that every person has access to the healthcare they need, when and where they need it, is recognised as a basic human right (World Health Organization [WHO], 2021). Inequities of access to healthcare services in rural and remote areas of Australia are well known (Australian Institute of Health and Welfare [AIHW], 2019). Peri-urban areas, in particular, offer a rural lifestyle in proximity to urban centres and have witnessed a surge in popularity and unprecedented population growth, impacting the provision of healthcare services. This research suggests that barriers to accessing community palliative care services in peri-urban locations are less understood and require exploration. This dissertation will argue that current classifications of location used to determine palliative care service policy, methods of delivery and allocation of resources, fail to consider the heterogeneity and complexities of peri-urban locations.

Methodology

An exploratory, inductive and interpretative approach to the research was adopted to develop and administer an online web-based Qualtrics survey for data collection. The aim was to understand the barriers to access from the perceptions of providers and consumers of community palliative care in peri-urban locations. Thematic analysis was adopted to find the participant meaning in the qualitative data, while descriptive statistical analysis tools were used to analyse the quantitative data to support the identified reality of participants. Participants included health practitioners (n = 50), family caregivers (n = 91) and patients (n = 30) patients.

Findings

This research identified a number of key issues influencing community palliative care provision in peri-urban locations. The barriers to access, known to exist in rural Australia, were also identified in peri-urban areas, with the home location viewed by both practitioners and caregivers as an obstacle to accessing required services. The distance and time involved in travel influenced decision-making and choice of location for care and place of death. Practitioners described the physical and emotional risks they suffered from their perception of an inability to provide appropriate care to clients. Caregivers voiced frustration and described increased carer stress that they attributed to unmet needs. Therefore, this research contributes

to knowledge by identifying the need for service providers to consider some of these complexities of peri-urban locations in policy.

Research Limitations

The voluntary participation in the survey could represent respondent bias, potentially contaminating results; participants may have been motivated by a desire to voice grievances.

Significance

The risk for physical and emotional distress is real for those individuals seeking palliative care but facing accessibility barriers. The removal of personal choice and preferences in care, treatment and locations for care and death has the potential to impede quality of life, wellbeing, and the grieving process. The demand and need for community palliative care services is increasing and policy recommendations play a critical role in removing barriers due to geographic location.

Keywords

Palliative care, Community care, Peri-urban, End-of-Life, barriers to access

Glossary of Key Terms

Access: Levesque et al.'s (2013b) conceptualisation of access to healthcare was used to support this research. It recognises access as a complex framework with integrated barriers identified by 5 dimensions from both the provider (supply side) and the individual (demand side). The 5 dimensions of accessibility are influenced by the corresponding dimensions or determinants of abilities of the population to utilise healthcare services. Access is viewed as the possibility to identify healthcare needs; seek, obtain or use services; reach resources; and be offered appropriate health services to fulfil needs (Levesque et al., 2013b).

Anticipatory grief: This refers to the distress a person may feel prior to their own death or the death of a loved one. It's also called preparatory grief (Johns, 2015).

Anticipatory prescribing: These are measures and/or medication put in place by clinicians to meet predicted acute events during the palliative journey (Ding et al., 2019).

Bereavement: This broad term encompasses the entire experience of family members and friends in the anticipation and adjustment following the death of a loved one (Christ et al., 2003).

Choice: In healthcare, this refers to the concept that individuals can make informed decisions about their care and services. Choice is a cornerstone of person-centred care and is essential for promoting autonomy, satisfaction and quality-of-life by provision of care tailored to meet individual needs, wishes and preferences, including choice of provider and location for care (Kozikowski et al., 2022; Zolkefli, 2017)

Community palliative care: Community or home-based palliative care is care provided to individuals diagnosed with a life-limiting illness and their families, outside the hospital setting (Kamal et al., 2013). The alternative setting, or place of residence might include owned or rented personal residences, retirement villages and other communal living arrangements such as mobile homes, caravans, cars and other locations as appropriate for people who are experiencing homelessness. For the purpose of this research, residences which generally have access to clinical support 24/7 will be excluded, including residential aged care facilities, correctional facilities, remand centres and detention centres, group homes and, other specialist housing for people with a disability or those experiencing severe mental illness (PCA, 2018b).

Complex grief: This is a syndrome which is chronic and debilitating, and results in significant distress and impairment following the death of a loved one. Symptoms consist of a sense of disbelief regarding the death, persistent intense longing and separation distress, a sense of

meaninglessness, trouble accepting the reality of the loss, and yearning for and preoccupation with the deceased (Holland & Neimeyer, 2010). It is also referred to as prolonged, pathological, complicated or traumatic grief.

Controlled medicine: This is also known as a controlled drug, drug of dependence, scheduled medicine, controlled poison or narcotic substance. It is a medication that is scheduled under the national classification system (Poisons Standard) according to the level of regulatory control over the availability of the medicine (Caring@home, 2020).

End-of-life care: Generally, this refers to the last few weeks of life in which a patient is rapidly approaching death. The needs of the patient and their caregivers are higher at this time (PCA, 2015).

Family: The concept of family encompasses the biological family, the family acquired by marriage/contract, and the family of choice (including pets) and friends. 'Family caregivers' refers to those identified as closest to the patient in knowledge, care and affection. The patient defines who will be involved in their care (Canadian Hospice Palliative Care Nursing Standards Committee, 2014).

Family/caregiver: A relative, friend or partner who has a significant personal relationship and provides assistance (physical, social and/or psychological) to a person with a life-threatening illness. They may or may not reside with the person receiving care and may be primary or secondary family caregivers (Becqué et al., 2021).

Greenfield areas: These are areas on the urban fringe that are zoned for future urban development. In these areas, infrastructure, policies and actions are identified that require resolution prior to planned residential development (Attorney-General's Department, 2021).

Healthcare professional: Aso referred to as a clinician, for the purpose of this research, the term healthcare professional covers a wide range of professions, including doctors, nurses, physiotherapists, pharmacists, occupational therapists, dietitians, volunteers, carers and more. Their involvement and level of expertise varies depending on the complexity of patients' needs (KPMG, 2020). Some specialise in palliative care as a full-time role and others will have palliative care as a part of their daily work.

Health inequity: This is the disparities in health between population groups that are avoidable, unfair and unjust (Dahlgren & Whitehead, 2006).

Imprest stock: This refers to the emergency medication supplied to establish and maintain a stock of medications for use elsewhere (Caring@home, 2020)

Life-limiting illness: This is an overarching term that encompasses all conditions appropriate for hospice or palliative care, sometimes termed terminal illness. It refers to any illness that can reasonably be expected to result in death within a foreseeable future (Macauley, 2019). It includes both malignant and non-malignant illnesses.

Model of service delivery: This describes the mode or method in which healthcare is delivered - either face-to-face, remotely via digital health, or a combination of both. It is related to the model of care utilised (Australian Commission on Safety and Quality in Health Care (ACSQHC), 2021c).

Model of care: This is a multifaceted concept which broadly defines the way health services are delivered to ensure best practice patient care (Agency for Clinical Innovation (ACI), 2013)

Multidisciplinary team (MDT): This refers to the team approach to palliative care. Individuals have clearly defined responsibilities within a team of professionals from multiple disciplines, who function independently with discipline-specific goals and treatment (Bowen, 2014).

Palliative approach: This is when the primary focus shifts from life-prolonging treatments towards symptom management and improving quality of life. Underlying the philosophy of a palliative approach is a positive and open attitude towards death and dying (Royal Australian College of General Practitioners (RACGP), 2023).

Palliative care: This is the holistic care provided to a person diagnosed with a life-limiting illness and their family members and caregivers. It is also known as comfort care, as the aim is not to provide a cure but to optimise quality of life and reduce suffering of patients and their families (WHO, 2020a).

Peri-urban: For the purpose of this research, the term peri-urban will be used to describe the outer metropolitan fringe space between metropolitan and rural surrounds, which has experienced increased residential development.

Person-centred care: This refers to healthcare which actively includes the patient and the family in the decision-making process for care and treatment, recognising the right of patients to make decisions and choices. It is also known as patient-centred care and person and family centred care and respects individual preferences and diversity (ACSQHC, 2023b)

Place: This refers to the specific location or setting where treatment and care is provided, or death occurs. It reflects the personal wishes and physical needs of the person involved. The term is inclusive of hospital and hospice settings, owned or rented personal residences,

retirement villages and other living arrangements such as mobile homes, caravans and care facilities (Gu & Wang, 2020).

Preterminal: This generally refers to the phase of relative stability or minor decline, prior to the rapid decline that ends with death (Cohen-Mansfield et al., 2017).

Primary care: This refers to the health services and staff that have a primary or 'first contact' relationship and may undertake an ongoing role in the support of patients (F. Gardiner et al., 2020). The term is inclusive of general practitioners, community nurses, and specialist services including oncologists, renal, cardiac or respiratory physicians and acute care staff.

Psychosocial care: This type of care is concerned with the psychological and emotional wellbeing of patient and family/caregivers, including issues of self-esteem, social functioning, insight into an adaption to illness and its consequences (Hudson et al., 2010).

Specialist palliative care service: In this research, the term refers to the specialist multidisciplinary practice of palliative care and the designated service system developed and funded to deliver that care which differs by organisation, state and territory (Michael et al., 2016).

Terminal phase: Often termed end-of-life, the terminal phase is generally the last few weeks of life in which a patient is rapidly approaching death. The terminal phase refers to the specific phase of the last days or hours when the patient is recognised as imminently dying. The needs of the patient and their carers are higher at this time, requiring increased support and services (PCA, 2015).

Urban: This is generally defined by population density and size, and accessibility to goods and services using statistical data. It may be referred to as metropolitan or suburban or as a town or city area (ABS, 2016b).

Rural: This refers to those areas which are not part of any urban area in the Australian Statistical Geography Standard (ASGS) (Australian Bureau of Statistics (ABS), 2016a). It often refers to areas of open land used for agricultural purposes, with low density housing and population (National Geographic, 2024).

Table of Contents

Acknowle	dgements	i
Statemen	t of the Contribution of Others	ii
Statemen	t of the use of Generative Al	iii
Statemen	t of Originality	iv
Abstract.		v
Glossary	of Key Terms	vii
Table of C	Contents	xi
Table of F	igures	XV
Table of T	ables	xvii
Chapter 1	: Introduction and thesis overview	1
1.1	Aim of the study	1
1.2	Research questions	2
1.3	Background	3
1.4	Significance of the study	8
1.5	Structure of the thesis	9
Chapter 2	2: Background	12
2.1	Introduction	12
2.2	Palliative care	13
2.2.1	Definition and history of palliative care	13
2.2.2	Palliative approach to care	14
2.2.3	Person-centred approach	15
2.2.4	Models of care	20
2.3	Known barriers to accessing palliative care	23
2.4	Population shift	24
2.4.1	Urbanisation	24
2.4.2	Counter-urbanisation	25
2.5	Peri-urban	26
2.5.1	Definition of peri-urban	26
2.5.2	Residential development	27
2.6	Impact on service delivery in peri-urban areas	28
2.7	Urban-rural divide	29
2.7.1	Health divide: The result of the urban–rural divide	30
2.8	Resource allocation	31
2.8.1	Australian healthcare system overview	31
2.9	Palliative care funding	33
2.10	Geographical statistical data	34
2.10.	1 Classification of location	38

2.11	Summary	40
Chapter	r 3: Social impact	41
3.1	Introduction	41
3.2	Clinical challenges	41
3.3	Psychosocial support	45
3.4	Ageing population	46
3.5	Mortality rates	49
3.6	Impact and challenges of COVID-19	52
3.6.	1 Background	52
3.6.	2 Government response	52
3.6.	3 Impact of counter-urbanisation	53
3.6.	4 Modes of service delivery	56
3.6.	5 COVID-19 deaths	58
3.7	Summary	60
Chapter	4: Methodology	62
4.1	Introduction	62
4.2	Exploratory, inductive, and interpretive approaches	62
4.3	Conceptualising the problem space	63
4.4	Literature review	64
4.4.	1 Results	66
4.4.	2 Data analysis	68
4.4.	3 Discussion	68
4.5	Theoretical framework	69
4.6	Study design	72
4.6.	1 Study inclusion criteria	74
4.6.	2 Sample descriptors	75
4.6.	3 Recruitment	76
4.6.	4 Data collection	78
4.7	Data analysis	78
4.7.	1 Qualitative analysis	78
4.7.	2 Thematic analysis	81
4.7.	3 Quantitative analysis – Descriptive Statistics	81
4.8	Summary	83
4.9	Limitations	83
Chapter	r 5: Pilot study	85
5.1	Introduction	85
5.2	An overview of palliative care provision in peri-urban areas	85
5.3	Study design	86
5.4	Participants	87

5.5	Data collection	88
5.6	Data analysis	88
5.7	Results	88
5.8	Discussion	91
5.9	Summary	93
Chapter	6. Practitioners' perspective: barriers to access	94
6.1	Introduction	94
6.2	Analysis of survey data	94
6.3	Profile of practitioner participants	95
6.4	Findings	96
6.4.	1 Home location	96
6.4.2	2 Increased risk to healthcare professionals	105
6.5	Summary	108
Chapter	7. Family caregivers' perspective: barriers to access	110
7.1	Introduction	110
7.2	Analysis of survey data	110
7.3	Profile of family caregiver participants	110
7.4	Findings	111
7.4.	1 Home location	112
7.4.2	2 Impact from unmet needs	125
7.5	Summary	128
Chapter	8. Patients' perspective: barriers to access	131
8.1	Introduction	131
8.2	Analysis of survey data	131
8.3	Profile of patient participants	132
8.4	Findings	133
8.4.	1 Patient perspective	133
8.5	Summary	141
Chapter	9: Conclusion	143
9.1	Introduction	143
9.2	Report to Chief Executive Dr Robyn Lawrence	145
Bibliogra	aphy	154
Appendi	ces	207
Appendi	x A: Geographical Classification	207
Appendi	x B: Literature Review summary papers	233
Appendi	x C: Ethics Approval	235
Appendi	x D: Survey questions	236
Appendi	x E: PCA Consent	256

Appendix F: PCA e-newsletter link	257
Appendix G: Participant Information Sheet	258
Appendix H: Grief Supports	
Appendix I: Quantitative Results	

Table of Figures

Figure 1 Placing the person at the centre of care (Department of Health and Aged
Care (DoHAC), 2019)16
Figure 2 Domains of palliative care (Department of Health Western Australia
(WADOH), 2008)17
Figure 3 Indication of service provision based on needs (PCA, 2018b)18
Figure 4 Community-based palliative care transitions (Kamal et al., 2013)19
Figure 5 Common barriers to quality palliative care (Australian Healthcare Associates
(AHA), 2019)23
Figure 6 Map of Australia showing where people live (MapPorn, 2019)
Figure 7 Organisation of the health care system in Australia (Glover, 2020)
Figure 8 Number of suicide deaths, by psychosocial risk factors in 2022 (AIHW,
2023c)
Figure 9 Comparison of age distribution, 1982–83, 2022–23 and 2062–63 (AIHW,
2023b)
Figure 10 Typical illness trajectories towards end-of-life (RACGP, 2023)49
Figure 11 All deaths, Australia, 3 January 2022–27 August 2023 vs baseline
benchmarks (ABS, 2023f)51
Figure 12 COVID-19 confirmed cases and deaths in Australia until November 2023
(WHO, 2023)
Figure 13 Confirmed COVID-19 deaths by age group in Australia November 2023
(ABS, 2023c)
Figure 14 PRISMA diagram adapted from 'The PRISMA 2020 statement: an updated
guideline for reporting systematic reviews' (Page et al., 2021)67
Figure 15 A definition of access to health care (Levesque et al., 2013b)70
Figure 16 Levesque conceptual framework of healthcare access (Levesque et al.,
2013b)72
Figure 17 Research Design73
Figure 18 Gioia thematic analysis survey results80
Figure 19 Aggregated theme relationship81
Figure 20 Greater Adelaide Development Area Map (Attorney-General's Department,
2021)
Figure 21 ASGC Structural Chart (ABS, 2009b)209
Figure 22 Hierarchical Levels ASGC Structure (ABS, 2006a)
Figure 23 Five categories of remoteness (AIHW, 2004)214
Figure 24 ARIA+ Service Centres by population (ABS, 2001b)

Figure 25 Map of 2016 Remoteness Areas of Australia (ABS, 2017, 2018a)	217
Figure 26 Comparison main structures by spatial units (ABS, 2012)	218
Figure 27 Interrelatedness ABS and Non-ABS Structures (ABS, 2022a)	219
Figure 28 ASGS ABS Structures (ABS, 2016c)	220
Figure 29 ASGS Non-ABS Structures (ABS, 2016a)	224
Figure 30 SOS Classification (ABS, 2017)	226
Figure 31 SOSR Classification (ABS, 2017)	227
Figure 32 RRMA Structure (Department of Primary Industries and Energy and	
Department of Human Services and Health, 1994)	228

Table of Tables

Table 1 Methods of statistical classification used in health	36
Table 2 Health care professional demographics	95
Table 3 Scenarios related to access barriers	97
Table 4 Descriptive statistics survey Question 6.19 How much do you agree	with the
following statements?	102
Table 5 Family caregiver demographics	111
Table 6 Descriptive Statistics Survey Q2.7 How much do you agree with the	following
statements?	114
Table 7 Scenarios related to access barriers	116
Table 8 Patient demographics	132
Table 9 Descriptive Statistics Survey Q5.6 How strongly do you feel about th	e
following statements?	138
Table 10 Descriptive statistics survey Q5.20 How much do you agree with th	е
following statements?	140
Table 11 Summary Survey Access	262
Table 12 Participants by group	262
Table 13 Question 2.1	262
Table 14 Question 2.2	263
Table 15 Question 2.3	263
Table 16 Question 2.4	263
Table 17 Question 2.5	263
Table 18 Question 2.7	264
Table 19 Question 2.8	264
Table 20 Question 2.9	265
Table 21 Question 2.10	265
Table 22 Question 2.10	265
Table 23 Question 2.11	265
Table 24 Question 2.12	
Table 25 Question 2.13	266
Table 26 Question 2.14	267
Table 27 Question 2.15	267
Table 28 Question 2.16	268
Table 29 Question 2.17	
Table 30 Question 2.18	269

Table 31 Question 2.18 (Descriptive Statistics)	271
Table 32 Question 2.19	273
Table 33 Question 2.20	274
Table 34 Question 2.21	274
Table 35 Question 2.21 (Descriptive Statistics)	275
Table 36 Question 2.22	276
Table 37 Question 6.1	276
Table 38 Question 6.2	276
Table 39 Question 6.3	277
Table 40 Question 6.4	277
Table 41 Question 6.6	277
Table 42 Question 6.7	277
Table 43 Question 6.8	278
Table 44 Question 6.9	278
Table 45 Question 6.9	278
Table 46 Question 6.10	279
Table 47 Question 6.11	280
Table 48 Question 6.12	280
Table 49 Question 6.13	281
Table 50 Question 6.14	281
Table 51 Question 6.15	282
Table 52 Questions 6.15 (Descriptive Statistics)	285
Table 53 Question 6.16	286
Table 54 Question 6.17	286
Table 55 Question 6.18	287
Table 56 Question 6.19	287
Table 57 Question 6.20	
Table 58 Question 5.1	
Table 59 Question 5.2	288
Table 60 Question 5.3	289
Table 61 Question 5.4	289
Table 62 Question 5.6	289
Table 63 Question 5.7	290
Table 64 Question 5.8	290
Table 65 Question 5.9	290
Table 66 Question 5.10	290
Table 67 Question 5.11	291

291
292
293
293
293
294
297
298
298
298
299

Chapter 1: Introduction and thesis overview

1.1 Aim of the study

The aim of the research is to understand if choice of home location in a peri-urban area influences access to community palliative care services. The inequities of access related to rural and remote Australia are well known, and often termed the urban–rural divide (Rainsford, 2019). What is less understood, however, is how rapid acceleration in population growth in areas on the fringes of urban and regional centres influences access to services. Often termed peri-urban areas, these locations are a popular choice for relocation, offering affordable housing in a rural environment but still within close proximity to urban centres (McFarland, 2015). Residential development planning includes the infrastructure and services needed to meet the needs of new communities; however, infrastructure and service growth in peri-urban locations has consistently lagged behind the housing construction (Kent et al., 2019; Newton et al., 2017).

Peri-urban health services, including local hospitals, are generally smaller facilities with less available infrastructure, a limited workforce, poor access to specialist care, and a reliance on generalist primary healthcare providers (Bradford et al., 2016; Wenham et al., 2020). Residential development changes the land use and occurs as land becomes available, subsequently blurring the lines between urban and rural when defining peri-urban locations (Meeus & Gulinck. 2008). Health service planning assumes that the close proximity to metropolitan centres is an advantage for peri-urban residents; it plans modes of service delivery with the expectation that patients will travel to receive these services. This expectation follows the need for planners to provide a fair distribution of limited resources to locations and services considered to be in most need; in Australia, that means, urban centres (Garrard, 1996). Peri-urban locations, however, remain partially rural and the inconsistencies with funding and allocation of resources to these locations and to palliative care invoke questions of fairness (Simon, 2008). Feedback sought by health services from clinicians and consumers regarding service delivery aids program development; however, it also has the potential for bias and does not necessarily reflect actual needs (Weng et al., 2022).

This research seeks to explore the perceptions and views of health professionals, family caregivers and patients on accessing effective, person-centred, community palliative care services. The complexities involved in palliative and end-of-life care provision invoke potentially differing priorities, needs and perspectives among all individuals and services involved. Health professionals are in the unique position of understanding access barriers from a systemic, workplace and patient viewpoint. Family caregivers are central to the care needs

of a palliative patient throughout the journey and, to optimise quality of life, will need to interact with the healthcare system to access required resources (Appelin et al., 2005). However, what constitutes required resources differs for every patient, influenced by the presence or absence of personal support, knowledge of palliative care, needs, preferences and culture (Maxwell et al., 2022).

Regardless of the differences, every viewpoint is important for gaining an understanding of the barriers to accessing quality community palliative care in peri-urban locations. Understanding barriers may guide future peri-urban health policies which influence patient and family caregiver accessibility to in-home palliative services, choice of end-of-life care, place of death, quality of life and bereavement services (Saurman et al., 2022). This chapter will introduce the study and provide an overview of the following chapters.

1.2 Research questions

The objective of this research is to create recommendations that address any identified discrepancies in policies and practices to ensure that best practice in community palliative care services is maintained in peri-urban locations. Key research questions have been formulated to identify if a relationship exists between peri-urban locations and barriers to provision of person-centred in-home palliative care.

To reach the aim of the study, the following overall research question was developed:

How does counter-urbanisation influence community palliative care provision in periurban Australia?

To help address this overall research question, a set of additional research questions were formulated:

- How are peri-urban areas defined?
 - o How is a peri-urban location defined in community health policies?
- What are the key influences of counter-urbanisation on community palliative care accessibility in peri-urban Australia?
- What role does digital technology play in peri-urban community palliative care delivery?
- Do patients' and family members' expectations of peri-urban community in-home palliative care differ to the care that health professionals provide?
- How does location influence accessibility to in-home palliative care services?

1.3 Background

In Australia in the mid-1990s, a trend of internal migration emerged, with people moving away from densely populated urban areas (Amirinejad et al., 2018; Nygaard & Parkinson, 2021; Ruiz-Martínez & Esparcia, 2020). This trend, termed 'counter-urbanisation', has seen people enticed to relocate for reduced housing prices and a lifestyle change (Davies, 2021; Guaralda et al., 2020). In recent times, this trend has been linked with COVID-19, as people fled the major cities following the implementation of infection control measures, including lockdowns, social isolation requirements and orders to work from home (Amirinejad et al., 2018; Attorney-General's Department, 2021; Davies, 2021; McManus, 2022). Financial year data from 2020–21 reflects the influence of COVID-19 in accelerating Australian internal migration, although it is unclear if the response will remain a temporary or a permanent transition as control measures have been gradually removed (ABS, 2022c; Davies, 2021; Guaralda et al., 2020; McManus, 2022).

Regardless of the reasons behind the trend, counter-urbanisation has resulted in accelerated growth outside of Australian major cities and a blurring of the lines between the previously distinct categories of urban and rural (Guaralda et al., 2020; Hugo, 2002; McManus, 2022). The resulting potential for uncontrolled urban sprawl has been restrained to some extent, by State and Territory government development plans and policies (Australian Digital Health Agency, 2021). New residential developments in peri-urban areas are co-located with major and local road network expansion to improve accessibility to urban centres (Brousselle et al., 2020; Department of the Treasury, 2021). The term 'peri-urban' is not clearly defined but it generally refers to those areas at the broad interface between metropolitan and rural surrounds, which encompass diverse locations, a range of land uses and a variety of communities (Liu & Robinson, 2016).

Peri-urbanisation generally refers to the transitional process whereby rural areas on the fringe of urban centres, change in physical, economic and social terms to become more urban in character (Simon, 2020; Webster & Muller, 2009). The prefix peri is derived from the Greek *peri*, meaning around, enclosing or near, so the term translates to mean urbanisation processes on the urban fringe (Follmann, 2022). These areas often remain categorised as rural due to the challenges associated with their distance from major cities, despite being deemed within the urban sprawl and accessible to services using improved road networks (Attorney-General's Department, 2021; Liu & Robinson, 2016).

Healthcare and resource allocation in Australia have historically followed the approach of service delivery for defined populations based on geographic statistical data provided by the

3

ABS (PCA, 2005; Senate Standing Committee on Community Affairs, 2007). This data classifies locations as urban and non-urban based on spatial units aggregated by population density and remoteness using a calculation of indexed accessibility to urban service centres (Senate Standing Committee on Community Affairs, 2007). The Australian Institute of Health and Welfare (AIHW) releases data at a range of geographical levels from States and Territories to smaller areas which improve understanding for clinicians and policymakers at a local level (AIHW, 2022b). Data used at a local level includes ABS statistical data, incorporating Estimated Resident Population (ERP), and localised data including Primary Health Network (PHN) areas, Local Health Network (LHN) areas and Local Government Areas (LGAs) (AIHW, 2022b). Peri-urban areas are not currently viewed as separate locations in geographic statistical data calculations; rather, they are considered a continuation of their surrounding urban or rural areas.

Areas of increased population growth and residential development can be identified using ABS data, which provides evidence of increased population growth and dwelling density (ABS, 2022a). Using current statistical data generated from current geographic classifications as a standalone measure to determine the character of an area and the needs of individual communities, however, can be misleading and result in difficulties for policymakers (Wood et al., 2023). Although statistical data may highlight an increased population in a rural area, the relative proximity of a location to metropolitan services can influence the model of health service delivery adopted by providers. Health providers may recognise the increased urban style growth in a rural area but continue to provide a rural mode of healthcare delivery under the belief that specialised healthcare is readily accessible via proximity to urban centres.

A failure to recognise the diversity of peri-urban locations ignores the complexities involved in service delivery for palliative clients and their families and is unlikely to result in effective, person-centred care. Accessibility, however, is not simply related to geographic distance – it includes availability and suitability of services to need, affordability and acceptability (Taylor et al., 2021b). A lack of accessibility to health services is one of the reasons why people living in rural areas experience poorer health status in comparison with their metropolitan counterparts (AIHW, 2019; Thomas et al., 2015). The term 'urban–rural health divide' is used to explain this inequity in healthcare provision. It includes a lack of local specialist expertise, limited allied health and in-home support, and a reliance on local General Practitioners (GPs) (RACGP, Rainsford, 2019; 2023). Inequities are also due in part to an older population with increased health needs and chronic medical conditions.

The COVID-19 pandemic had an unintentional consequence of removing some of the barriers of access by transforming the way Australians access healthcare services (Caffery et

4

al., 2022). Research during the COVID-19 pandemic found that Australian medical clinics developed their digital technology to enable healthcare services to continue (Jackson, 2022; Petrie et al., 2021). Digital technologies has removed the barriers of time and location to accessing healthcare services and has been of particular benefit to people living outside of metropolitan centres (Cherry et al., 2018). Digital health promotes advocacy for people to participate in needs assessments, enables contact with specialists in tertiary centres and enhances the number of patient–physician encounters in the patient's own home, improving wellbeing and quality of life (Erdiaw-Kwasie & Alam, 2016; Lally et al., 2021; Milch et al., 2021).

Whilst the change from a traditional face-to-face model of service delivery to digital health improved accessibility in the short term, it is important to refrain from viewing digital health as the panacea to removing the inequities of heath service access in all situations in the long term. Palliative care practitioners have voiced concerns about the difficulties in conveying empathy and communicating sensitive information, responding to body language and facial cues during telephone consults, and a lack of physical contact when touch is often used to support someone in distress (Collier et al., 2016; Milch et al., 2021). The uptake of digital health has been limited to the willingness of healthcare services to change their healthcare delivery and the quality and accessibility of existing digital infrastructure (Ruiz-Martínez & Esparcia, 2020). The urban–rural digital divide extends to internet use and accessibility, despite the pandemic resulting in an acceleration of a national program of development and implementation of digital technology to enhance the Australian health system (Davies, 2021; Rainsford, 2019). The known challenges associated with the urban–rural health divide are factors that have the potential to influence the accessibility and quality of healthcare service delivery for peri-urban residents (Schulte et al., 2022).

Care for a person diagnosed with a life-limiting illness, including end-of-life care, is termed palliative care. This type of care responds to the needs, values, and preferences of people receiving care and their families (Saurman et al., 2022). In-home or community palliative care and a home death have been associated with a higher quality of life and are often used as markers of a 'good death' (Boamah et al., 2021; Gerber et al., 2019). Patients diagnosed with a life-limiting illness commence a palliative approach to care to provide physical, psychological, social, and spiritual comfort (Callisto et al., 2021). Palliative care is provided in many locations by many health professionals and multidisciplinary teams (MDTs) and includes end-of-life care and the choice of a home death (Agllias, 2018). Specialist palliative care is available to those persons with complex care needs to support the care given by the primary care provider in home settings, but the extent of the support provided depends upon available resources (Pask et al., 2018).

Palliative care differs to other health services because it responds to individual needs at any time in the palliative journey, which can span weeks to years, depending on the disease trajectory. Healthcare services required by this group include tertiary centres for both inpatient and outpatient care and local specialist and allied health care providers, both in-clinic and inhome. Consumers in peri-urban areas may well have access to local hospitals, but these facilities often lack access to the specialised services required by this group (Pesut et al., 2014). Ensuring care is responsive to patient preferences may be difficult in a community context due to environmental and personal or behavioural barriers (Kwame & Petrucka, 2021).

There is a known benefit for persons who live outside of metropolitan areas to remain in their own homes for as long as possible. Findings from a Royal Commission into Aged Care found that many Australians verbalise a preference for end-of-life care in their home, regardless of location (Commonwealth of Australia, 2021; Gerber et al., 2019; Kenny et al., 2021; KPMG, 2020). Forced relocation for treatment, rehabilitation, or aged care services can result in disconnection from community, friends and family (Taylor et al., 2021b). Unfortunately, for older Australians and those with a life-limiting illness, there are significant waiting lists for supports provided through a federally-funded Home Care Package which aims to assist people to remain in their own home (Commonwealth of Australia, 2021). Rationale for the study

Peri-urban areas are heterogeneous, being rural areas transitioning to urban land uses, unique in diversity of location, structure, culture, community and geographical area (Buxton & Butt, 2020). Development has not been constrained to planned greenfield development sites, with ad-hoc residential development occurring as land becomes available (Newton, 2010). Residential growth has often outpaced the ability of governments and private entities to plan and develop appropriate infrastructure and provide the services required to meet the needs of local communities (Infrastructure Australia, 2018).

Ensuring that every person has access to the health care and medication they need, when, and where they need it, is recognised as a basic human right. The rights of patients receiving healthcare in Australia can be found in the Australian Charter of Healthcare Rights, legislation and common law (ACSQHC, 2020; Attorney-General's Department, n.d.). Health care services include palliative care, which is a person-centred approach, aimed at improving the quality of life for people diagnosed with a life-limiting illness, with little to no prospect of a cure (Palliative Care Victoria, 2017). It is widely accepted that provision of good quality palliative care in the patient's choice of location, improves overall wellbeing for patients and their family members, enables choice of place for end-of-life care and death, and facilitates the grieving process for family and friends. The WHO recognises palliative care as an ethical responsibility of health systems globally (WHO, 2021). As an ethical responsibility, the

6

provision of quality, person-centred palliative care in the location of choice becomes critical in maintaining quality of life of individuals and communities.

Barriers to access result in unmet needs for patients and family caregivers, increasing potential for relocation, distress, suffering, and complex grief (Driessen et al., 2023). Healthcare professionals face feelings of frustration, stress and burnout due to an inability to provide the holistic community palliative care they believe is appropriate (Baqeas et al., 2021). These are core reasons for understanding the influence of a peri-urban location to accessing community palliative care service delivery. The ethical responsibility for Australia to ensure equitable access to person-centred palliative care services in the location of patient choice further supports the need to conduct this research.

A mix of specialist and generalist providers from public and private sectors provide palliative care, depending on the needs and wishes of the individual and the availability of services (AIHW, 2022d; Wenham et al., 2020). Palliative care services are delivered in multiple settings to provide care throughout the palliative journey; however, these services lack uniformity as each differs in team structure, location of care, and available services (Hui & Bruera, 2020). Community palliative care service delivery is influenced by funding arrangements, variations in organisational structures and patterns of service delivery, with geographic location a particular influence on service availability and models of care outside of urban areas (Eagar et al., 2006; Groeneveld et al., 2017).

Current structure, availability, and allocation of resources for service delivery in fringe areas experiencing urban-style growth, is based on a rural mode of service delivery with an expectation that proximity to urban centres ensures equitable access to services (Rural Health Standing Committee, 2018). However, this view appears to fail to consider that community palliative care service delivery is person-centred, tailored to the individuals' needs and preferences for services and support, which differs for every person. The structure, availability and allocation of services and resources in each peri-urban location need to be explored and evaluated for efficacy so that improvements can be made to existing services to meet the needs and expectations of an increasing number of peri-urban residents. The surge in popularity of peri-urban living necessitates an understanding of the basis for health service policy development in these areas which have a history of known health service challenges. If community health service delivery is based on a mode of delivery that assumes peri-urban areas are homogenous with their surrounding rural or urban areas, then it is unlikely to result in effective, person-centred care. Research demonstrates that Australians living in rural areas experience poorer access to, and use of health services, in comparison with metropolitan counterparts, with access issues increasing with remoteness (AIHW, 2019; End of Life Directions for Aged Care (ELDAC), 2024; Thomas et al., 2015). One area of research which seems to have been overlooked, however, is the influence of peri-urban locations on community palliative care provision. Population growth in these outer fringe areas has potential to not only increase the number of people impacted by the urban-rural inequities but also the demands on existing primary services. In 2019, 14.8% of all registered deaths in Australia, or 24,970 deaths, were registered at a person's home or place of residence (not including aged care facilities) with an unknown number of family members and caregivers impacted (AIHW, 2021a).

1.4 Significance of the study

Death and dying are inevitable parts of life that no one can avoid. The subject remains taboo for many, but the way we care for dying people can be viewed as a measure of our individual values and of our society. The quality and accessibility of palliative and end-of-life care will affect all of us in some manner and must be a priority to providers, policy makers and governments. In the words of Dame Cicely Saunders, founder of the modern hospice movement;

how people die remains in the memory of those who live on so we should strive to make a good death the expectation rather than the exception in all settings. (cited in Deloitte, 2014, p. 1)

All people have the right to expect equitable access to person-centred, quality healthcare at any point in their life journey, including at the end-of-life (ACSQHC, 2023a; Wenham et al., 2020). Evidence has shown that this basic right is challenged by inequities of access due to geographical location (Bakitas et al., 2015; Matsumoto et al., 2013). The choice to relocate from an urban centre should not influence access to quality, person-centred palliative care services. Barriers to access influence preferred end-of-life care and place of death, impact patient quality of life, remove patient choice, increase likelihood of emergency and acute presentations and impede the grieving process of family and friends (KPMG, 2020; Saurman et al., 2022).

The experience of caring for a loved one at home will linger in memories long after the person has died, with potential for long-term psychosocial and physical impacts on family caregivers. Healthcare providers have an ethical and moral right to ensure the wellbeing of the community they serve. Any perception that the care provided has failed to meet the needs of patients and families can also be frustrating and distressing for all clinicians involved,

8

increasing the likelihood of increased compassion fatigue, stress and burnout (Chan et al., 2021; Terry et al., 2015).

The physical, social, financial, and emotional implications for the increasing number of Australians choosing to relocate to peri-urban locations is why it is important to understand the challenges to palliative care service delivery in these fringe areas, now and in the future. We need to understand the specific needs of peri-urban locations to allocate resources appropriate to the diverse communities involved. Current policies and procedures rely on generic urban or rural modes of service delivery as a one-size-fits-all approach.

1.5 Structure of the thesis

The thesis comprises 9 chapters:

- Chapter 1 has provided the context for the thesis and presented the aims, research questions and thesis structure.
- Chapter 2 provides the background of the problem space with an overview of palliative care and difficulties associated with community service delivery related to peri-urban locations. Known barriers to care in rural and remote Australia are outlined, as are the methods of geographical classification of location used in definitions of urban, rural and remote. The chapter summarises changes imposed on healthcare provision that influence community palliative care in peri-urban areas.
- Chapter 3 presents the impact on the physical and psychosocial health and wellbeing of those affected by death and dying and the influence of inequities of access. Perspectives of patients, family caregivers, clinicians and the health system will be discussed, along with the influence of the COVID-19 pandemic.
- Chapter 4 outlines the research process and the corresponding methods applied to each stage. The research process employed an exploratory, inductive and interpretative approach which aimed to respond to the problem space. The chapter will outline the research design and discuss the use of a literature review and pilot study to help identify the problem space and develop the research questions and aim. This chapter will outline how a survey to collect data and validate results was developed, considering the risk of emotional distress to participants due to the research subject. The next stage outlines the qualitative and quantitative data analyses tools, in particular the use of a thematic analysis.

- Chapter 5 provides a summary of the pilot study which adopted an ethnographic approach, conducted to ascertain the feasibility of the research method being developed. Observation and informal discussions were undertaken with a small number of health professionals providing community palliative care in a peri-urban region of South Australia prior to survey development. Identified key issues will be discussed, including home location and barriers to accessing resources, and the risks experienced by both practitioners and family givers.
- Chapter 6 explores the views of health practitioners by thematic analysis of free-text survey responses and a descriptive statistical analysis of quantitative data. The relationship between barriers to accessing required resources, including anticipatory and end-of-life medication, trained staff, and after-hours support, due to home location distance is discussed. An emphasis was placed on a lack of system-led policies and organisational characteristics, including poor support from leadership by research participants. Practitioners highlighted the emotional toll associated with a perceived inability to provide appropriate care to palliative patients and family caregivers due to a home location being situated in a peri-urban area.
- Chapter 7 explores the views and perceptions provided by family caregivers related to barriers to accessing appropriate care in their home locations. Family caregivers highlight the difficulties accessing services in their peri-urban home location and the relationship between person-centred care access and distance. Descriptive responses highlight the consequences of unmet needs and expectations associated with access barriers, including the need to relocate; or transfer for end-of-life care, and the disappointment at the inability to reassure their loved ones that their wishes would be met. The research participants' preference for face-to-face encounters was highlighted, rather than a reliance on digital health service delivery.
- Chapter 8 explores the perceptions of patients. Patients are those who have been diagnosed with a life-limiting illness and who are accessing care in their home location. The similarities and differences between the responses of patients and family caregivers is discussed; the level of appreciation of available services voiced by patients was noticeably higher. The loss of choice of treatment, care and place of care was discussed, with time and distance to home location influencing access.

• Chapter 9 provides a summary of the issues relating to community palliative care provision in peri-urban areas which have become evident in this research. The focus will be on implications and future recommendations for service providers in the form of a paper directed to the Chief Executive of the Department of Health and Wellbeing, SA Health, Dr Robyn Lawrence.

Chapter 2: Background

2.1 Introduction

In chapter 2, the exploration of the problem space will consider the complexities involved in planning and delivering effective palliative care, with an emphasis on the significance of the context of community service delivery in peri-urban locations. The chapter begins by presenting a comprehensive background on the fundamental elements that have shaped the landscape of palliative care, and how it has evolved into a holistic approach. The focus extends to the challenges involved in delivering palliative care services in peri-urban areas, to establish a foundational understanding of the influence of counter-urbanisation on community palliative care provision in peri-urban locations.

By addressing the difficulties associated with community service delivery, the chapter aims to provide understanding of the unique circumstances prevalent in these semi-rural regions. This includes an examination of the barriers that often impede the seamless provision of palliative care, offering insight into the distinctive challenges faced in peri-urban locales. The narrative proceeds to illuminate the divide between urban and rural and remote Australia, outlining the limitations and barriers related to geographical location.

To distinguish between urban, rural, and remote locations, the chapter provides a comprehensive summary of the known methods employed for geographical classification. This detailed exploration provides background to the research questions by helping us to understand the diverse landscapes in which palliative care services are delivered and highlights the lack of clarity on defining peri-urban locations. The chapter also provides a thorough summary of the changes that have been imposed on healthcare provision that directly impact the delivery of community palliative care in peri-urban regions.

By examining these shifts, the chapter aims to contextualise the evolving nature of healthcare practices and their implications for palliative care services. Through this analysis, the chapter sets the stage for understanding the dynamic interplay between healthcare policies, geographic classifications, and the challenges encountered in delivering effective palliative care within the unique context of peri-urban areas.

2.2 Palliative care

2.2.1 Definition and history of palliative care

Palliative care originated in the modern hospice movement in England in the late 1950s and early 1960s with a focus on care for the dying (Loscalzo, 2008). Dame Cicely Saunders opened St Christopher's Hospice in London in 1967 and challenged the view at that time, that lack of a cure constituted a failure on the part of the physician, and introduced a multidisciplinary approach to end-of-life care (Alcalde & Zimmermann, 2022; Saunders, 2001). This approach has been adopted throughout this research. Inspired by this philosophy, Dr Balfour Mount, founded a hospice-style inpatient unit in Montreal, Canada in 1973 (Duffin, 2014). To remove the negative stigma associated with the word 'hospice', he introduced the term 'palliative care', based on the Lain *palliare* meaning 'to cloak' (Duffin, 2014).

Australia was relatively slow to transition from provision of terminal care in an acute setting to programs providing palliative care services in the home (Malden et al., 1984). Over the past 40 years, the impetus for the transition from hospice to palliative care services has come from oncology services that recognised that patients' needs were being unmet (Currow & Phillips, 2013). The report 'Palliative care for cancer patients' commissioned in 1983, is viewed as a defining moment in the early evolution of palliative care in Australia, rejecting the concept at that time of admission to acute care for terminally ill patients (PCA, 2023; Webster, 1985). Since that time, palliative care services in Australia have transitioned to a strong network of providers focused on delivering quality care in many settings, including in-home and community care. The definition of palliative care used in this research is as follows:

...the active holistic care of individuals across all ages with serious healthrelated suffering due to severe illness and especially of those near the end of life. It aims to improve the quality of life of patients, their families and their caregivers. (IAHPC, 2024)

Community or community-based palliative care is defined as palliative care delivered outside of hospital and outpatient clinics, with the aim of providing home-based individualised multidisciplinary support to patients and family members (Chan et al., 2021; Olvera et al., 2023). There are many benefits for people who choose to remain living in their own homes for as long as possible, including improved overall wellbeing, the ability to maintain healthy lifestyles and maintain relationships with family and community (Taylor et al., 2021b; Thornton et al., 2022). Provision of palliative care services in the community recognises that the term

'own home' could include personal residences, retirement villages, communal living arrangements, residential aged care facilities or mobile homes. Community palliative care services often extend into Residential Aged Care Facilities (RACF); however, this research will not include this care service because these facilities are already well-equipped to provide prompt and effective palliative care (Sandsdalen et al., 2016). RACFs generally have access to a GP and staff who are adequately skilled in palliative care on site over a 24 hour period, which can reduce the need for transfer to an acute setting and provide necessary support for family members.

2.2.2 Palliative approach to care

Palliative care is an approach which aims to optimise quality of life and the dying experience for people diagnosed with life-limiting illnesses, including cancer, dementia, and non-malignant degenerative diseases (AIHW, 2022d; PCA, 2018b). Unfortunately, palliative care patients generally have multiple and complex psychological, social and symptom concerns (Pask et al., 2018), which makes the care process complicated and dynamic. Patients and family members go through changing and escalating physical, spiritual and emotional care that influences their quality of life and wellbeing during the palliative journey (Chan et al., 2021). Palliative care involves treating each person as an individual with support tailored to meet the person's unique needs and situation (PCA, 2018a).

A palliative approach to care recognises that death is inevitable and transfers the focus from a cure to holistic care to support the needs and values of the patient, caregivers, and family members (Agllias, 2018; Saurman et al., 2022). The goal of a palliative approach is to improve comfort and function by addressing physical, psychological, social, emotional, cultural and spiritual needs (PalliAged, 2022). The following provides a taxonomy of the palliative care objective(s):

- provides relief from pain and other distressing symptoms
- affirms life and regards dying as a normal process
- intends neither to hasten nor postpone death
- integrates the psychological and spiritual aspects of patient care
- offers a support system to help patients live as actively as possible until death
- offers a support system to help the family cope during the patient's illness and in their own bereavement
- uses a team approach to address the needs of patients and their families, including bereavement counselling, if required
- will enhance quality of life, and may also positively influence the course of illness

 is applicable early in the course of illness, in conjunction with other therapies that are intended to prolong life, such as chemotherapy or radiation therapy, and includes those investigations needed to better understand and manage distressing clinical complications. (PCA, 2015)

The World Health Assembly recognises that:

...palliative care, when indicated, is fundamental to improving the quality of life, well-being, comfort and human dignity for individuals, being an effective personcentred health service that values patients' need to receive adequate, personally and culturally sensitive information on their health status, and their central role in making decisions about the treatment received. (WHO, 2013)

2.2.3 Person-centred approach

Patient or person-centred care is defined as, 'providing care that is respectful of and responsive to individual preferences, needs and values and ensuring that patient values guide all clinical decisions' (ACSQHC, 2024).

A patient or person-centred approach to healthcare means recognising that every person is a human being and not simply a medical condition or disease to be treated (Coulter & Oldham, 2016). Person-centred care involves effective communication, dignity, trust and respectful relationships, to seek and understand the needs, wishes, preferences of patients, family members and caregivers (ACSQHC, 2023b). A person-centred approach is fundamental to effective and quality palliative care, to ensure access is available to appropriate information, support, and care, when and where it is needed by the person(s) involved. Eight dimensions have been identified as important for providing person-centred care:

- respect for patients' values, preferences and needs
- provision of information and education
- access to care
- emotional support
- involvement of family, friends and caregivers
- continuity and seamless transition between health care settings
- physical comfort
- coordination of care. (Galekop et al., 2019)
The importance of safety and quality healthcare when using a person-centred approach is embedded throughout all Australian Healthcare Standards developed by the Australian Commission on Safety and Quality in Health Care (ACSQHC) (2023b). This includes the National Safety and Quality Health Service Standards and the Primary and Community Healthcare Standards (ACSQHC, 2021b, 2021c). The approach involves the patient, their families and support people at the centre of all decision-making about care, as shown in Figure 1. The primary focus is on symptom management and quality of life rather than life-prolonging treatments.



Figure 1 Placing the person at the centre of care (Department of Health and Aged Care (DoHAC), 2019)

Person-centred care is a core value in palliative care, ensuring dignity and respect for individuals and family needs, cultural and religious preferences. A person-centred approach is central to holistic care which supports the needs and values of the patient, caregivers, and family members in service delivery (Agllias, 2018; Saurman et al., 2022). The holistic scope of palliative care is highlighted in Figure 2.



Figure 2 Domains of palliative care (Department of Health Western Australia (WADOH), 2008)

Palliative care differs from other health services in that patients must access care based on their individual needs at any time during the palliative journey, from diagnosis to end-of-life. This can span weeks to years, depending on the disease trajectory (RACGP, 2023). Figure 3 provides a visual representation of individual needs and service involvement across the palliative journey. Healthcare services required by this group include tertiary centres for both inpatient and outpatient care and local specialist and allied healthcare providers, both in-clinic and in-home. Integration of palliative care early in the disease process is effective at improving quality of life and reducing unnecessary hospitalisations and the associated costs (Namasivayam et al., 2022; Seow et al., 2021).



Figure 3 Indication of service provision based on needs (PCA, 2018b)

Palliative care is provided in many settings with person-centred care supporting choice of location for care, including end-of-life care and the choice of a home death (Agllias, 2018). It is widely accepted that provision of good quality palliative care in the patient's choice of location improves overall wellbeing for patients and their family members and facilitates the grieving process for family and friends. The palliative journey differs for each individual and is not linear in nature, with the needs and choices of the person at the centre of care constantly changing in an effort to optimise their health (Gorod et al., 2021).

Several factors influence the decisions or choices made by every individual. These factors, include past experience and memories, social and cultural rules and expectations, cognitive biases, age and individual differences, belief in personal relevance, and a perception of value (Dietrich, 2010; Latif, 2020). Individuals have the right to appoint a medical decision-maker if they are too unwell to voice their choice of medical treatment, care provider or location of care, including hospital, hospice or home (Zolkefli, 2017). Supporting patient choice for community-based palliative care not only aids the patient and family, but it also fosters seamless transitions between the settings required and reduces use of hospital resources (Kamal et al., 2013), as demonstrated in Figure 4.



Figure 4 Community-based palliative care transitions (Kamal et al., 2013)

Many Australians have verbalised a preference to remain in their own home for as long as possible, with 60–70% expressing a wish for end-of-life care and a home death (Commonwealth of Australia, 2021; Gerber et al., 2019; Kenny et al., 2021; KPMG, 2020). This view is supported by the findings of the Royal Commission into Aged Care which found that many Australians verbalise a preference for end-of-life care in their home, regardless of location (Commonwealth of Australia, 2021; Gerber et al., 2019; Kenny et al., 2021; KPMG, 2020). In-home or community-based specialist palliative care and a home death have been associated with reduced hospital costs and a higher quality of life and are often indicators of a good death (Boamah et al., 2021; Gerber et al., 2019; Spilsbury & Rosenwax, 2017).

There are, however, no common measures of a good death, with preferred location only one of many core themes identified in the literature. Other themes include; pain-free status, life completion, dignity and quality of life (Meier et al., 2016). In 2019, however, only 14.8% of all registered deaths in Australia, or 24,970 deaths, were registered at a person's home or place of residence (not including aged care facilities) (AIHW, 2021a; Swerissen & Duckett, 2015). The proportion of institutionalised deaths has increased, with 51% or 86,276 of deaths in 2019 occurring in hospitals or medical centres and 29.5% or 49,896 in residential care facilities (AIHW, 2021a).

Statistics, however, fail to acknowledge all factors associated with the specific setting or place of end-of-life care and death (Gu & Wang, 2020). A home death is more likely to occur when the person involved has expressed this wish, and not every person expresses a home death as their preferred option. A preference for hospital care may be verbalised due to concerns regarding quality of care available in the home, the appropriateness of family

members attending to personal care, ambivalence towards professional carers in the home or not wanting to be a burden on family (Munday et al., 2007). Despite the common wish for endof-life care at home, many people in the terminal stage are transferred to acute settings because their deteriorating condition and unstable symptoms can no longer be adequately managed in the home. Other reasons for being transferred to another location include a decision by the patient or family, lack of caregiver ability to manage 24-hour care or a lack of services to support the family (Champion et al., 2015).

2.2.4 Models of care

Evidence suggests that the complex and multidimensional needs of palliative patients are best met through an MDT approach, particularly due to the importance of continuity of care (Fernando & Hughes, 2019; Hudson et al., 2019). A team approach enables expert clinicians, including doctors, psychologists, pharmacists and nurses to share the responsibility, and to work together to communicate and coordinate the delivery of care (Bowen, 2014). Promoting one person who has the accumulated knowledge of the team as the point of contact encourages a respectful and trusting relationship, which can enhance care provision, reduce hospital admissions and support choice of end-of-life care (Hudson et al., 2019). Relational continuity of care goes beyond information and care provision. It seeks to promote feelings of satisfaction from patients and family members, and helps build and maintain an emotional bond and mutual understanding which ultimately optimises quality of life and care provision (Aghaei et al., 2020). Access to the MDT services and support required by people during the palliative journey, including at end-of-life is influenced by multiple factors which may not reflect the changing needs of communities. The health system structure, a combination of public and private providers, funding arrangements, and models of service delivery all have a direct effect on the quality of care and the eventual outcomes (Davidson et al., 2006).

A model of care is a multifaceted concept which broadly outlines the way in which health services are organised to deliver best practice patient outcomes (Luckett et al., 2014). Models provide a framework for implementing and evaluating care and services through core components which describe who delivers what interventions, when, where, how and to whom (PalliAged, 2017; Siouta et al., 2016). Regardless of the setting in which palliative care is delivered, models provide benefits to patients and carers (Brereton et al., 2017). There is no preferred organisational model of care, with various models available to meet the specific needs of particular groups that are associated with their illness or condition, setting, specialities, or services available (Davies et al., 2011; Rumbold & Aoun, 2021).

The curative model of care does not fit for medical conditions that may eventually lead to death and is not appropriate for palliative care service delivery (Neuberger, 2003).

20

Communities in rural Australia have typically relied on a traditional primary care model, with an emphasis on diagnosis and management by an individual practitioner and referrals where necessary (Woods, 2001). Integrated models of palliative care focus on the potential role of primary care and PHNs as hubs to facilitate integration of specialist palliative care and community services, with GPs an integral part of care provision (Rumbold & Aoun, 2021). Research has demonstrated an improvement in quality of life, symptom burden and psychological symptoms and an increased likelihood for a home death from an integrated model of care (Agency for Clinical Innovation (ACI), 2022a). GP-led clinical integrated models of care may benefit patients who have an established and trusting relationship with their GP, thereby promoting a relational continuity of care with their provider (Hudson et al., 2019). Challenges arise, however, if the GP lacks specialist palliative skills or is unable to prioritise the perspectives of the patient or family (Brown et al., 2018; Gilbert, 2020).

Subacute models of palliative care aim to optimise quality of life for the patient with an active and advanced life-limiting illness with care delivery under the management of a clinician with specialist palliative care expertise and evidenced by an individualised multidisciplinary management plan (Independent Health and Aged Care Pricing Authority (IHACPA), 2024). Subacute care needs and treatments differ from those of acute care as the goal is not driven by the diagnosis; rather, the goal is to maximise the functional status and quality of life of the patient (Pearse et al., 2011). Limitations occur with this type of model due to a lack of clarity on the term 'specialised expertise', resulting in varying responses to perceived goals of care, service delivery and care implementation (ACI, 2017). What is considered an appropriate composition of medical, nursing and allied health clinicians in local MDTs has the potential to limit the patient's ability to meet their goals (ACI, 2017). Limitations to subacute models of care are exacerbated by the barriers to accessing specialist clinicians and expertise in peri-urban, rural and remote areas (Pearse et al., 2011)

The team approach is indispensable within models of palliative care service delivery and necessary to address the complex needs of the patient from diagnosis to the point of bereavement for loved ones (WHO, 2020a). Communication, information, and practical issues can be as overwhelming as physical symptoms such as fatigue, pain, reduced mobility, and loss of appetite (Bone et al., 2016). Allied health professionals can support medical practitioners in managing physical symptoms, advising on nutritional status or muscle strength, and offering support with finance, legalities, or psychosocial issues (Chan et al., 2021). The team structure and interactions between members varies depending on the model of teamwork adopted; however, evidence strongly suggests that palliative care is best delivered though an MDT-based model of care (Fernando & Hughes, 2019; Gilbert, 2020).

21

Commonly used terminology to describe palliative care teams include multidisciplinary, interdisciplinary, and transdisciplinary. A purely MDT has clearly defined responsibilities for each individual within the team of professionals from multiple disciplines, who function independently with goals and treatments that are discipline-specific (Martin et al., 2022). In an interdisciplinary team, members work interdependently to augment patient care and a transdisciplinary approach involves overlapping responsibilities between members (Fernando & Hughes, 2019).

Home and community-based models aim to provide appropriate and timely palliative care in-home, with community-based models of care promoting seamless transitions between palliative care settings (ACI, 2022a; Kamal et al., 2013). Studies have shown the benefits of palliative care outside of the inpatient care setting, with or without input from a specialist palliative care team, in terms of patient satisfaction, quality of life, reduced number of hospitalisations, length of hospital stays and symptom severity (Brereton et al., 2017; Cohn et al., 2017; Mathew et al., 2020). Person-centred models include 'Respecting Choices', an advance care planning model which aims to assist individuals to consider, choose and communicate their preferences for care and treatment (Agency for Clinical Innovation (ACI), 2022b). Another person-centred model is the 6S model which has Self-image as the central concept, with the patient and their own experience of the situation as the starting point. The other concepts – Self-determination, Symptom relief, Social relationships, Synthesis and Strategies – are all interrelated to Self-image (Österlind & Henoch, 2021).

Despite the significance of an effective model of care to support patient needs, a universally agreed model of care for palliative care does not exist (Siouta et al., 2016). It is essential for quality care provision that models developed and used by palliative care providers are dynamic and responsive to community and population changes and are underpinned by the diseconomies of scale resulting from distance and small dispersed populations (Wakerman et al., 2008). Organisations develop and adopt models of care tailored to local needs, resources, strategic plans and initiatives, and the costing, funding and revenue opportunities provided (ACI, 2013). The model adopted by an organisation is reflected in the type and structure of management and the organisation's characteristics and philosophy which influences priorities of care and allocation of resources (Parreira et al., 2021). Providers need to consider feedback related to patient preferences, experiences, and needs, rather than rely on benchmarks and formal indicators of ideal location of death or perceived quality of life (Siouta et al., 2016). Barriers to accessing community palliative care in rural locations within peri-urban areas will be inherent in models of care that emphasise revenue opportunities or that fail to prioritise and adapt methods that cater to individual needs and preferences.

2.3 Known barriers to accessing palliative care

Despite evidence that palliative care services can improve quality of life and be provided alongside treatment options, multiple barriers remain to accessing care. The challenges faced by someone who is diagnosed with a life-limiting illness are already substantial and potentially overwhelming for the person and the family involved, without the added complexity of accessing services. Availability of palliative care services is influenced by geographical location, provider funding arrangements, variations in organisational structures and patterns of service delivery (Eagar et al., 2006; Wenham et al., 2020). Common barriers are provided in Figure 5.





In addition to common barriers, the following 9 populations of interest have been identified as having co-existing vulnerabilities and/or complex needs in relation to accessing quality palliative care and community services (AHA, 2019):

• Aboriginal and Torres Strait Islander peoples

- care leavers and people affected by forced adoption
- people from culturally and linguistically diverse (CALD) backgrounds
- · people with disabilities
- people experiencing homelessness
- · people who are incarcerated

• people who identify as lesbian, gay, bisexual, transgender, intersex, queer/questioning, asexual (LGBTIQA+)

- refugees
- veterans (AHA, 2019)

Understanding barriers to accessing palliative care in Australia is multifaceted, however concerns related to accessibility are primarily related to the vast geographical area of the country (Shukla et al., 2020). Geographic locations are a known barrier to accessing quality and person-centred palliative care for Australians living outside of high-density urban centres, influencing available services, choice and location of care (Schulte et al., 2022; Taylor et al., 2021b; Thomas et al., 2015). This is why it is important to explore the shifting trends in the distribution of the Australian population.

2.4 Population shift

2.4.1 Urbanisation

History has shown that the first European settlers gravitated to the colonial settlements on the Australian coastline, with these settlements later becoming the current major cities (Dodson, 2016). This trend of Australians settling in urban areas continued throughout the nineteenth century, as people sought to live close to employment opportunities (Amirinejad et al., 2018; Nygaard & Parkinson, 2021; Ruiz-Martínez & Esparcia, 2020). Despite the popularity of living in major cities, issues such as overcrowding, a lack of consistent infrastructure, housing shortages and high housing costs emerged as the population increased (Dodson, 2016; Slavko et al., 2020). The amount of land available for urban planning, however, was finite, with population growth in cities accommodated by either an increase in density or expansion into surrounding rural areas (Gurran et al., 2021; McFarland, 2015). Increased housing pressures resulted in an expansion of city boundaries, with a pattern of suburban

development intensifying from the late 1940s to the early 1960s (Berry, 1999; Dodson, 2016; Macintyre, 2018).

Uncontrolled urban sprawl was initially made possible by access to rail transport, which allowed residents to commute from new residential developments to the city centre (Berry, 1999). Suburban developments expanded with more affordable housing than metropolitan centres, and the dominant aspiration of many people was to live in an owner-occupied, single family, fully detached home (Berry, 1999; Macintyre, 2018). The Australian suburban model was intensified by the removal of wartime petrol rationing in 1951 which then enabled the expansion of motor vehicle manufacturing, development of freeway schemes and local road access to augment existing rail transport (Davison & Yelland, 2004; Dodson, 2016).

The Commonwealth established a national housing program in 1945 to ensure that land was available to maintain the ever-increasing suburban development. This was a coordinated approach that also included State and Local Government Areas (LGAs) (Dodson, 2016). The government recognised the importance of home ownership to Australians – their desire for the 'great Australian dream' of a suburban quarter acre block and the growing trend of car ownership (Macintyre, 2018; Walters, 2021). Census data demonstrates that in 2016 over 70% of the Australian population lived in major cities, with the remainder living in regional and remote areas (Thomson et al., 2019). Subsequently, LGAs were encouraged to rezone greenfield areas to allow for increased suburban development and infrastructure including transport, sewerage, mains water, electricity and telephone lines (Macintyre, 2018). The term 'greenfield' refers to those areas on the urban fringe that are zoned for future urban development. It identifies that infrastructure, policies and actions will need to be resolved prior to planned residential development in these areas (Attorney-General's Department, 2021).

2.4.2 Counter-urbanisation

The urban trend of living in the densely populated cities and suburban areas of Australia began to change in the mid-1990s, with internal migration emerging (Amirinejad et al., 2018; Nygaard & Parkinson, 2021; Ruiz-Martínez & Esparcia, 2020). Relocating from major metropolitan areas and settling in outer metropolitan, regional or rural areas has been termed counter-urbanisation (Guaralda et al., 2020; McManus, 2022). Resettlement was considered a viable alternative to city living, influenced by the desire of some for an improved quality of life and better environmental conditions and escaping densely populated and polluted areas (Jończy et al., 2021). The reasons for internal migration, however, are multi-faceted and not necessarily due to a failure in urban design or an inability of urban areas to accommodate increased populations (Han et al., 2017; McManus, 2022).

There are links between household mobility and the housing pathway of downsizing (Jończy et al., 2021). Reasons for downsizing include wanting to carry out less maintenance and decrease floorspace due to physical inability, or a desire for a lifestyle change, often due to a life event such as retirement, divorce, illness, death of a partner or becoming an emptynester (Han et al., 2017). Housing mobility has been linked to a financial strategy to release home equity, or the inability or unwillingness of residents to bear the cost of urban living expenses, including house prices and mortgage stress (Davies, 2021; Guaralda et al., 2020). Evidence supports the significance of relocating within proximity of family members, for financial, emotional and physical care and support, both young families and retirees (Argent & Plummer, 2022).

Downsizing and lifestyle changes, often termed tree or sea changes, are no longer deemed the domain of retirees (Ghin & Ainsworth, 2022). This is due in part to State Government planning and policy, locating new residential developments in conjunction with major and local road network expansion (Australian Government, 2021; Department of Planning Transport and Infrastructure (DPTI), 2017). Improved commute times have contributed to accelerated population growth to areas that were sought after for natural amenities but not previously considered viable for relocation due to their distance from metropolitan cities (Amirinejad et al., 2018). Improved accessibility, housing affordability, an increasing population and economic restructuring has resulted in accelerated growth in peri-urban locations on the urban fringes and a blurring of the lines between cities and rural regions.

2.5 Peri-urban

2.5.1 Definition of peri-urban

There is no universally accepted definition of urban or rural areas, nor criteria for differentiating the terms peri-urban or peri-urbanisation (Follmann, 2022; McBride & Moucheraud, 2022). A level of ambiguity exists in defining the space between urban and rural surrounds due to the number of terms in use that, often depend on the classification system being used. Terms are used interchangeably and include urban fringe, semi-urban, rural living zone, country living zone, exurban, peri-urban, metropolitan fringe or rural, which add to the lack of clarity in research (Buxton & Butt, 2020). The term peri-urban is increasingly used to describe those areas on the urban periphery, with an informal boundary defined by a significant economic relationship with the nearby urban centre due to commuting trends and often related to population density (Buxton & Butt, 2020; Gottero et al., 2021; Simone, 2022). For the purpose of this research, the term peri-urban will be used to describe the outer metropolitan

fringe space between metropolitan and rural surrounds, which has seen increased residential development.

2.5.2 Residential development

Peri-urban areas at the broad interface between metropolitan areas and their rural surrounds encompass diverse locations and a variety of communities and land uses, and are generally within 150 km of urban centres (Liu & Robinson, 2016; Webster & Muller, 2009). The encroaching urban sprawl into peri-urban areas is helping to meet Australians' need for affordable housing; however, development has outpaced projections and this places pressure on water catchments, native vegetation, agricultural land and infrastructure (Buxton & Butt, 2020; Harman et al., 2015; Infrastructure Australia, 2018). Planning for sustainable growth and residential development in peri-urban areas to ensure future liveability and community wellbeing has been identified as a challenge for State and LGAs (Buxton & Butt, 2020; Gurran et al., 2021). Population increases, changes in the forms of land use, environmental concerns, and misconceptions about how peri-urban areas are identified all contribute to challenges in developing policies and planning and available infrastructure (Buxton & Butt, 2020). The lack of distinction between rural and urban classifications also increases difficulties for real estate developers seeking space for urban development and agricultural industries trying to retain land (Taylor et al., 2017).

Developing large tracts of available land tends to result in a development market dominated by large corporations which are able to consult with planning authorities (Walters, 2021). These corporations often consolidate the land into small, more affordable allotments, rather than the existing larger, lifestyle allotments, which alters the characteristics of the area and transforms it into an urban environment in a rural area (Walters, 2021). Thus, because expansion in peri-urban areas occurs as land becomes available, they are often irregular and do not fit within the Greater Capital City Statistical Areas (GCCSAs) or greenfield boundaries (Low Choy et al., 2008; McFarland, 2015).

The type and level of growth and changes in these outer fringe areas is influenced by two contrasting perspectives on how the spatial relationships between urban centres and the surrounding rural areas are viewed (Buxton & Butt, 2020). An urban perspective views the land surrounding metropolitan centres as a means of providing required land and resources for residential development and allowing urban centres to constantly expand (Buxton & Butt, 2020). However, population growth and the increasing length of commute times raise issues about the expansion of urban areas in relation to social, economic and environmental sustainability (Beer, 2014; Kent et al., 2019). A rural perspective recognises that although

27

residential development may generate challenges, this internal migration may stimulate rural areas through new business ventures, new income streams, and skills and knowledge sharing (Buxton & Butt, 2020).

2.6 Impact on service delivery in peri-urban areas

Peri-urban areas have historically been viewed as a viable option for sustainable urban growth, due to their proximity to urban amenities (Argent & Plummer, 2022; Attorney-General's Department, 2021; Infrastructure Victoria, 2021; Wolff et al., 2021; Wynne et al., 2020). Land uses include a changing mix of established township, low density residential development, large rural living or lifestyle allotments that often involve some level of rural activity, and agricultural land (Attorney-General's Department, 2021; State Planning Commission, 2020).

New residential developments in greenfield and urban fringe areas are intentionally colocated with major and local road network expansions (Brousselle et al., 2020; Department of the Treasury, 2021). Improved infrastructure and new roadways have improved accessibility to services in urban centres, although improving commute times does not negate accessibility issues for those residents who are frail, elderly, unwell or rely on public transport (Schulte et al., 2022). A lack of public transport and a reliance on private transport have been identified as sustainability issues for future development in peri-urban areas due to time, financial costs and loss of productivity associated with increased travel (Newton et al., 2017; North Western Melbourne Primary Health Network (NWMPHN), 2021; Shao et al., 2019).

Outer fringe areas often receive a rural model of service delivery due to geographical location, meaning that it is more difficult to access the same goods, services and social interactions as metropolitan centres (Attorney-General's Department, 2021; Liu & Robinson, 2016). Populations in rural areas have historically relied heavily on the services delivered by primary health providers, including GPs, community-based health services and specialist community clinics (Halcomb et al., 2021). Primary health care services vary but include acute presentations, health screening and prevention and management of chronic conditions (Halcomb et al., 2021).

Modelling suggests that there is a problem with health service access in peri-urban areas, and this problem is comparable to the already acknowledged issues present in rural and remote areas of Australia (Olson et al., 2019). The concern is that modelling has indicated that these peri-urban areas share similarities with rural and remote regions, regarding inequities in health service access. This implies that, despite proximity and assumed access to urban centres and services, peri-urban areas face challenges in ensuring equitable access to health services. The problem is multifaceted, involving issues such as inadequate

healthcare infrastructure, limited availability of healthcare professionals, and socioeconomic factors (Mortel et al., 2017; Schulte et al., 2022). Regardless of the location, allotment size or commute distance, the increased population growth in peri-urban locations corresponding to housing development has increased the number of people potentially impacted by the urban-rural inequity (Schulte et al., 2022). Understanding and addressing these challenges are crucial for ensuring that residents in peri-urban areas have fair and timely access to essential health services.

2.7 Urban–rural divide

The term 'urban-rural divide' refers to the unique challenges that people who live in rural and remote locations face due to geographical location. In Australia, around 7 million people, or approximately 28% of the population, live in rural and remote areas (AIHW, 2022f). The term, however, is often used in the context of multifaceted problems associated with location, including poorer access to goods and services (such as health and education) and employment opportunities, and poor-quality housing and digital infrastructure (Smith et al., 2008). People living outside of major cities in 2022 were less likely to have completed Year 12 or a non-school qualification, less likely to be employed than their counterparts in metropolitan areas, and in 2019–2020 had 15% less household income per week and paid higher prices for goods and services (ABS, 2022b; 2023d). A visual representation of the distribution of population is shown in Figure 6.



Figure 6 Map of Australia showing where people live (MapPorn, 2019)

2.7.1 Health divide: The result of the urban–rural divide

People living in rural areas experience poorer health in comparison with their metropolitan counterparts (Bradford et al., 2016; Smith et al., 2008). Research shows that people living outside of major cities have higher rates of hospitalisations, deaths and injury, have poorer access to primary healthcare services and are more likely to engage in risky behaviours such as smoking, than their metropolitan counterparts (AIHW, 2022f). The move towards digital health has the potential to remove the barrier of distance, however, rural residents are more likely to face reduced access to digital infrastructure and have limited capacity to use available technology (Warr et al., 2021). Areas outside of the metropolitan areas have historically had an older population who are frail and have increased health needs, chronic medical conditions, decreased mobility and lower socioeconomic status (AIHW, 2021d; Smith et al., 2008; Weng et al., 2021).

These issues are exacerbated by inequitable access due to a lack of public transport and geographic accessibility to the primary health services they require to maintain quality and timely healthcare (Taylor et al., 2021b). The result is a reduction in quality of life, increased likelihood of unnecessary and preventable hospitalisations and presentation to emergency departments, and unmet healthcare needs (Khanassov et al., 2016; Kurpas et al., 2014). Rural residents may be forced to relocate to urban centres due to difficulties associated with

accessing treatment, rehabilitation, care and support services or residential aged care (Taylor et al., 2021b).

People living outside of major cities face difficulties in both access and capacity to use the modern and up-to-date digital technologies that urban residents generally access, use and expect in order to participate in society (Department of Infrastructure, 2020; Thomas et al., 2020). Despite ongoing infrastructure policies, these inequities, also referred to as the digital divide or digital exclusion, remain in rural areas of Australia (Davies, 2021; Erdiaw-Kwasie & Alam, 2016; Park, 2017). Digital divides can be divided into two orders. First order divides are associated with inequities of access to required infrastructure and second order divides contribute to circumstances which limit the capacity to use and benefit from available technology (Warr et al., 2021). The Australian Digital Inclusion Index (ADII) measures : key dimensions of digital inclusion: access, affordability and digital ability (Thomas et al., 2020). Although the overall digital inclusion score increased in 2020, the gaps between the digitally included and excluded widened, with the latter group having lower levels of education, employment and income (Thomas et al., 2020).

People living in rural and remote areas are more likely to be overlooked for telecommunication upgrades, resulting in limited or no access to high-speed internet infrastructure or reliable mobile phone coverage which increases disconnection to specialist health services (Guenther et al., 2020; Marshall et al., 2021). While digital exclusion often refers to limited internet access and infrastructure, it is a multilayered concept which recognises the way in which people engage with Information and Communications Technology (ICT) due to motivation, skills and opportunities to benefit (Park, 2017).

Ensuring that care is accessible to all Australians is one of the 6 principles viewed as fundamental to good palliative care, reflected with initiatives to remove accessibility barriers, throughout the National Strategy (DoHAC, 2019). The known challenges associated with the urban–rural health divide are factors that have the potential to influence the accessibility and quality of healthcare service delivery for peri-urban residents (Schulte et al., 2022).

2.8 Resource allocation

2.8.1 Australian healthcare system overview

Australia is a geographically large and sparsely populated country with an excellent health system and highly trained medical workforce. To achieve equal access to health services for all and maintain the balance between supply and demand, it is essential that finite resources, including an appropriate mix of medical specialities, are well distributed

geographically (Khashoggi & Murad, 2020). The Australian healthcare system is complex, governed by a number of agreements developed by the Council of Australian Governments (COAG) (Senate Standing Committee on Community Affairs, 2012). The system operates under a mixed model of private and public health and hospital services, which is shown in Figure 7. The Federal, State and Territory, and local governments broadly share responsibility for funding, operating, managing, and regulating the health system (AIHW, 2022c). The private for-profit and not-for-profit sectors and voluntary organisations also play a role in operating and funding some health services (AIHW, 2022c). Approximately 55%, or 14.42 million Australians, now have private health cover, which is encouraged by a tax rebate on premiums (Private Healthcare Australia, 2023).

The 3 levels of government are collectively responsible for providing universal health care. The main responsibilities of each are as follows:

- Federal Government: Develops national health policy, administers the Medicare Benefits Schedule (MBS), provides funds to States and Territories for public hospital services, oversees PHNs, funds medicines through the Pharmaceutical Benefits Scheme (PBS), regulates private health insurance, funds community-controlled Indigenous primary healthcare, organises health services for veterans and funds health and medical research.
- State and Territory Governments: fund and manage public hospitals, regulate and license private hospitals, oversee LHNs, deliver public community-based and primary health services, and deliver preventive, ambulance and health complaints services.
- Local Governments: provide environmental health-related services such as waste disposal and water fluoridation, provide community and home-based health and support services, and deliver health promotion activities.



Figure 7 Organisation of the health care system in Australia (Glover, 2020)

2.9 Palliative care funding

As noted above, responsibility for providing and funding healthcare services is shared between the Federal and State and Territory Governments. The Federal Government does not directly fund palliative and hospice care services, but financial assistance is provided to State and Territory Governments to operate palliative services within health and community service provision responsibilities (Senate Standing Committee on Community Affairs, 2012). The Federal Government provides funding for national palliative care projects to assist universities, health services and palliative care organisations with research, education, and training and skills. State and Territory Governments provide grants to non-government organisations to promote access to these services.

The Federal Government pays approved aged care providers a subsidy for each person receiving government subsidised aged care (DoHAC, 2023a). Additional funding is given to aged care providers to look after those who have been assessed as needing palliative care, using the Aged Care Funding Instrument (ACFI) (AIHW, 2022d). In 2020–21 there were 90,700 admissions to hospital where palliative care was provided during all or part of the

episode of care, with the average length of stay almost double the average for all overnight hospitalisations (AIHW, 2022d). Of these, 50,000 were for palliative care and 40,700 for other end-of-life care. One in two hospitalisations (53%) ended with the patient dying in hospital.

These figures demonstrate a 23% increase in the number of palliative care-related hospitalisations between 2015–16 and 2020–21 (AIHW, 2022d). In 2020–21, 14,500 people received 69,100 MBS-subsidised palliative medicine attendance and case conference services provided by palliative medicine physicians/specialists (AIHW, 2022d). Data does not identify avoidable hospitalisations; however the term 'potentially preventable hospitalisation' is an umbrella term used by researchers to describe hospital stays which could have been avoided with access to appropriate primary healthcare and which result in an increased burden on the health system (Glenister & Archbold, 2022).

Public and private hospitals both receive funding from the Federal, State and Territory Governments, private health insurance funds and individual out-of-pocket payments by individuals. However, the relative contributions made by these sources of funds vary across the sectors, reflecting the types of patients they treat, the services they provide, and the administrative arrangements in which they operate (AIHW, 2021c).

2.10 Geographical statistical data

The ABS collects and releases survey data which can then be used to support important decision-making (ABS, 2023g). This data is used by health service providers when developing, designing and planning policies for health service delivery within Australia (ABS, 2023g). The Census of Population and Housing (referred to as the Census) captures a large amount of the statistical data every 5 years which reflects the characteristics of Australian people. The ABS has developed a geographical classification system, which is hierarchical in structure, to provide accurate data on specific locations in Australia (ABS, 2021). The structure enables small areas to be grouped together to provide data about larger areas and enable comparison between old and new areas, reflecting any changes. Spatial categorisation systems include State and Territory boundaries, suburbs, LGAs, electoral divisions and postcodes. Each system has a specific purpose. Limiting data for planning and service delivery based on pre-defined area units, fails to consider the complexities of access, the fit between supply and demand, and the contextual variables associated with the location (McGrail et al., 2017; Wang, 2020).

Table 1 lists the main methods of classifying the locations used in health data analysis, research, resource allocation and service provision in Australia. The list is not exhaustive, and

data obtained from the ABS is limited to Census data. Table 1 also provides a brief overview of the geographic or spatial classification used in healthcare, the method of classification and the measurements used to assist with comparison.

Table 1 Methods of statistical classification used in health

Model	General Use	Classifications	Measurement
The ASGS which replaced the Australian Standard Geographical Classification (ASGC) in use 1984–2011	Commonly used by AIHW in reports on smaller areas. Health services use ABS data in analysis and research Data may be used standalone or in conjunction with other statistical and/or social data	Classifies locations in hierarchical structure. Main structure has 7 levels from State/Territory (S/T), Statistical Areas Level 4 (SA4s), Statistical Areas Level 3 (SA3s), Statistical Areas Level 2 (SA2s), Statistical Areas Level 1 (SA1s), Mesh Blocks (MBs) The GCCSAs and Urban Centres and Localities (UCLs) and Section of State (SOS) are separate structures	Classifies areas as urban or rural based on aggregates of units of population size, dwelling density and urban character Data provided by ABS
Australian Statistical Geography Standard- Remoteness Area (ASGS- RA) developed from Australian Standard Geography Classification – Remoteness Area (ASGC- RA)	Purpose of releasing and analysing statistics to tailor services to meet the needs of regional and remote Australia	Classifies Australia into 5 classes of remoteness by accessibility to services Remoteness Areas (Ras) are aggregates of SA1s that are grouped together based on their average ARIA+ score	Classifies areas as remote using measurement of physical distance to an urban centre and the ARIA+ index of accessibility calculation to goods, services and opportunities for social interaction.
Indigenous Regions (IREG)	Health data directed at Indigenous health, particularly primary health care	Part of the ABS Indigenous structure enabling statistical analysis of data related to Indigenous communities and regions	Based on Indigenous population size in a hierarchical structure
Accessibility/ Remoteness Index of Australia Plus (ARIA+) developed in 2003 from Accessibility/Remoteness Index of Australia (ARIA)	Enable services to be tailored to meet the needs of regional and remote Australia	Continuous index of remoteness classifying locations into 5 areas of remoteness and 5 classes of accessibility	Defines remoteness in an index which calculates accessibility by levels per road distance to service centres Australia is divided into 1 km grids for ARIA and ARIA+ indexes which differs to SA1s used in ASGS-RA and ASGC-RA
Rural Remote and Metropolitan Area (RRMA), developed in 1994 and replaced with the ARIA	Used for general purpose policy and procedures related to rural and remote areas and workforce allocation	Classifies Australia into 3 zones (metropolitan, rural and remote areas) and 7 classes within the zones	Measures location by population size using Census data to classify locations
Modified Monash Model (MMM) introduced in 2015, developed from the Monash Model (MM)	Used in targeted programs to attract health professionals to more remote communities and to determine access to healthcare	Classifies locations as cities, rural, remote or very remote on an MM scale with 7 categories from major city to very remote.	Measures remoteness and population size of the area. Remoteness is determined by ASGS-RA.
PHN replaced Medicare Locals in 2015	Commonly used by AIHW in reports of smaller areas to meet local area needs	31 PHN areas which are classed as metropolitan or country Provides care for population in local area	Views locations as metropolitan or country, based on subsets of populations using data from ASGS- RA

LHN or Local Health District (LHD)	Provides public hospital services and may contain one or more public hospitals	Every public hospital is included in an LHN with boundaries defined by State/Territory governments.	LHNs are classified by S/T governments as metropolitan or country including at least one public hospital
Population Health Area (PHA)	Research and policy development requiring geographical statistical data for analysis and comparison in health datasets	Comprises a number of whole SA2s and aggregates of smaller SA2s	Classified per the SA2s which comprise the PHA to enable comparison of regions within the whole of Australia
Postcode and Postal Areas (POAs)	Postcodes assist with processing/delivery of mail POAs are an ABS postcode approximation POAs are used for a range of statistical data e.g. dashboards	Postcodes are defined geographic areas which are subject to change POAs are an ABS approximation of postcodes using one or more SA1s	Measures data in small spatial units of postcode areas or SA1s and aggregates
Local Government Area (LGA)	Datasets – e.g. life expectancy, mortality rates, rural and remote health outcomes	Comprises administrative areas of S/T which local governing bodies are responsible for	LGAs comprise a diverse range of communities with boundaries defined by S/T governments
ERP	Used by policymakers in planning	Links people to a usual place of residence and provides estimate by S/T and lower hierarchical levels	ABS official estimate of population by updating Census base by components of population change
Cities, Towns & Villages (CTV)	Used to analyse services	Extends statistical geography to consistently include joint population and service centres of any population size	Spatial distribution based on 2 characteristics – population centre and service centre –containing at least one identified service
Small towns	Used to gain insights into the makeup of small towns and implications for future policy and service delivery within regions	Locations with urban features grouped by population size	No explicit definition of towns, but urban centres and localities grouped by populations over 200

Table 1 demonstrates the complexities involved in the process of geographical classifications and measurements of accessibility and remoteness, which add to the service provision issues evident in peri-urban areas.

2.10.1 Classification of location

ABS data determines major classifications used to define areas in Australia as urban, rural, regional, or remote. The most commonly used classifications are as follows:

- RRMA classification (Senate Standing Committee on Community Affairs, 2007)
- ASGC 1984–2011 defines areas as urban/rural based on population density (ABS, 2006a, 2016b, 2022a)
- ASGC-RA determined 7 levels of remoteness between 1984–2011, based on distance by road to service centres (AIHW, 2004)
- ASGS 2011 to current (Edition 3) defines areas as urban/rural based on population density (ABS, 2006a, 2016b, 2022a)
- ASGS-RA which replaced the ASGC-RA in 2011, determines 5 levels of remoteness based on distance by road to service centres
- RAs and the ASGC-RA, based on ARIA or ARIA+ index values, classify remote areas
- ARIA or ARIA+, based on index values, classify remote areas based on the level of accessibility to goods, services and opportunity for social interaction (ABS, 2001b)

Ensuring that care is accessible to all Australians is one of the 6 principles viewed as fundamental to good palliative care, reflected in initiatives to remove accessibility barriers throughout the National Palliative Care Strategy (DoHAC, 2019). Health service policy decisions and the allocation of limited resources for service delivery are based on the degree of 'urbanness or ruralness' as defined by the ABS (AIHW, 2022b; 2009).

The ASGS collects and disseminates ABS geographical statistical data to classify locations in Australia based on hierarchical aggregates by population size, dwelling density and urban character to define areas as urban or non-urban (rural) (ABS, 2020). The smallest units of information provide the most accurate data and enable identification of smaller townships and communities which assists with effective resource allocation (ABS, 2022a). The boundaries range from small geographical areas to larger areas, but small areas are designed to ensure a size large enough to provide privacy and confidentiality to residents (ABS, 2020). Practitioners are already familiar with the challenges of maintaining privacy and confidentiality in small communities, including small rural pockets in peri-urban locations (Roufeil & Battye, 2008).

The importance of maintaining privacy has the unintended result of providing a larger spatial unit which provides a generalisation of the area as urban or non-urban. However, periurban locations are not homogenous; they have differing characteristics and often no discernible growth pattern (Buxton & Butt, 2020; McFarland, 2015). The smallest spatial unit used by health services is an SA2 with a population of 3,000 to 25,000 people. Small towns using Sections of State Ranges (SOSR) to compare small, medium and large towns with major cities (ABS, 2018a). Challenges become obvious in peri-urban areas due to the diversity of population and housing density that may be evident within each SA2. Definitions of Other Urban and Bounded Locality as urban or rural are too simplistic for use in peri-urban areas and fail to capture the needs of the community without additional local input.

The ASGC-RA classifies areas into 5 classes of remoteness via an index of accessibility, which has limitations when used in peri-urban areas due to a failure to consider consumer expectations of choice, quality and cost of available services (The Senate Community Affairs Committee Secretariat, 2012). ARIA and ARIA+ define remoteness in terms of relative access to a particular service. This definition uses the average distance of a location from a service centre and assumes that the greater the distance, the more remote the area and the less opportunities for social interaction and supply of goods and services.

Defining remoteness based on a simplistic measure of access alone does not consider the complexities involved, including location of the individual, distance to the service, the particular service involved and its affordability and appropriateness in the situation. Accessibility also becomes questionable when the requirement for a service centre does not include a requirement for specialist health services. Defining accessibility more broadly would lead to considering the dimensions of affordability, availability, accommodation, acceptability, and appropriateness.

The MMM is used by health providers to define a location as metropolitan, rural, remote or very remote, based on the ASGS-RA framework of geographical remoteness and town size (DoHAC, 2021). An MM category from 1 to 7 is used to determine eligibility for health workforce programs, including rural bulk billing and attracting health professionals to work in more remote and smaller communities; it is not relevant for peri-urban areas due to their proximity to major urban centres.

The ABS is not the sole provider of information used in health policy and research (Rural Health Standing Committee, 2018). Data provided by LGAs provide a smaller spatial unit than State or Territory, although the area has the potential to cover a diverse range of communities. It can be difficult to define peri-urban areas within LGA boundaries using LGA

geography due to the variety of land uses that may exist. Classifying LGAs as peri-urban or metropolitan fringe due to their general proximity to urban centres has the potential to oversimplify challenges to accessing community health services that may exist for some residents within the LGA (ABS, 2020).

CTV geography was developed to assist with analysing the spatial distribution of services and access in regional areas of Australia. The Bureau of Infrastructure, Transport and Regional Economics (BITRE) (2019) uses ABS data to provide information on the spatial distribution of the Australian population by clustering people into locations where they both live and where services are located and can be accessed. Defining locations as CTVs extends statistical geography by identifying only those population centres which are also service centres and includes locations with populations of fewer than 200 persons (BITRE, 2019).

2.11 Summary

This chapter has addressed the research questions relating to how peri-urban areas are defined and how peri-urban locations are defined in community health policies. Rural locations challenge palliative care providers because they rely on primary healthcare providers and there are increased travel times for both patients and health workers which impact effective use and availability of resources and provision of quality care (Cinnamon et al., 2008). Healthcare resource allocation is influenced by funding arrangements, organisational structures and policies, and are based on geographical statistical data which uses population density and distance from services to ascertain accessibility. Distance is calculated using road distance between home location and the boundary of the closest urban centre. This simplistic calculation disregards the local geography, road surface, transport options and weather conditions, which all influence the actual time it takes to travel to an urban centre and the cost and accessibility of doing so. Healthcare providers calculate the distance to the closest hospital to ensure equitable access but that hospital may or may not have the capability to meet the complex healthcare needs of the patient (Barbieri & Jorm, 2019).

The surge in popularity of peri-urban living indicates the need for a better understanding of the basis for health service policy development in these areas which have a history of known health service challenges. It would appear unlikely that community health service delivery based on a model of delivery that assumes peri-urban populations are homogenous with their surrounding rural or urban areas will result in effective, person-centred care. Geographical classifications are provided in more detail in Appendix A.

40

Chapter 3: Social impact

3.1 Introduction

The term palliative care can have negative connotations for many people, because it can conjure visions of hospices and impending death (Collins et al., 2020). However, evidence suggests that early palliative care reduces physical and emotional distress and optimises quality of life for individuals, their caregivers, family members and friends (Aoun et al., 2020; Hui & Bruera, 2020; Seow et al., 2020). As the journey continues and death approaches, palliative support intensifies to relieve the patient's physical, spiritual and psychosocial suffering by controlling pain and other symptoms and avoiding prolonging the dying process (Mitchell, 2021; Rome et al., 2011). The focus of palliative care then shifts to bereavement and supporting the family following the patient's death.

Evidence has shown, however, that inequities of access to health and community services exist for those living outside metropolitan areas (Bakitas et al., 2015). Australians living in rural areas are known to experience poorer health and are far less likely to receive the palliative and end-of-life care needed, in comparison with their metropolitan counterparts (AIHW, 2022f; Bradford et al., 2016; Smith et al., 2008; Thomas et al., 2015). It is the similarities to rural locations which suggest that the urban–rural health divide will influence access to community palliative care service provision in peri-urban areas, thereby increasing the potential for physical and emotional distress, decreasing quality of life, and removing personal choice.

This chapter will address the overall research question by exploring some of the challenges of the urban–rural health divide in peri-urban locations and highlighting the implications of barriers to access for the patient, family caregiver and healthcare system. Attention will be given to the clinical challenges of care provision, the importance of psychosocial support, models of service delivery and the impact of the COVID-19 pandemic.

3.2 Clinical challenges

A diagnosis of a life-limiting illness presents substantial and overwhelming challenges for the person involved. Challenges include depression, feelings of hopelessness, loss of independence, fear of pain and other symptoms, loss of dignity and control, and fear of dying (Woo et al., 2006). The key components of community palliative care support include symptom assessment and management, support and coordination of services subject to needs and psychosocial care (Chan et al., 2021). Palliative care is holistic, person- and family-centred care; therefore, it is important to remember that the patient is not the only person impacted by an impending death. Families and loved ones are significantly affected by the diagnosis. There

is an expectation that family members will manage the physical care and practical difficulties associated with daily living activities, cope with changed familial roles and responsibilities, and manage the associated financial and legal implications while handling their own home and work roles, emotional distress and concerns about future needs (Kristjanson & Aoun, 2004). Barriers to accessing in-home services can result in unplanned hospitalisation admission, presentation to emergency departments, reduced quality of life, strained and broken relationships and increased financial costs (Namasivayam et al., 2022; Pesut et al., 2014; Virdun et al., 2020).

Buxton and Butt (2020) noted that there are significant similarities and differences between areas classified as rural and those viewed as peri-urban. Peri-urban health services, including local hospitals, are generally smaller facilities with less available infrastructure, a limited workforce, and poor access to specialist care. They also rely on generalist primary healthcare providers to provide specialist palliative care (Bradford et al., 2016; Wenham et al., 2020). Although rural GPs play an essential role in providing palliative care, they may have limited knowledge and skills in managing end-of-life symptoms and inadequate training in facilitating difficult conversations, instead emphasising the pathophysiology of the disease (Brown et al., 2018; Ding et al., 2019; Erdiaw-Kwasie & Alam, 2016; Glare & Virik, 2001; Johnson et al., 2020). Any perception that the care provided has failed to meet the needs of patients and family members can be frustrating and distressing for all clinicians involved, increasing the likelihood of increased compassion fatigue, stress and burnout (Chan et al., 2021; Terry et al., 2015).

Challenges to accessing quality care can result in poor symptom management, increased anxiety, and poor coping strategies for patients. Pain and the fear of pain often drive patient behaviour and are common at the end-of-life, particularly cancer-related pain (Heinrich et al., 2022; Rome et al., 2011). Unrelieved pain can exacerbate other symptoms such as fatigue and anxiety, and increase psychological distress for patients and families, influencing decisions regarding the ability to cope and care location (Heinrich et al., 2022). Psychological distress for people facing an uncertain and impending death can result in depression, evidenced by feelings of hopelessness, helplessness, guilt, worthlessness, loss of perspectives, self-harm and suicidal ideation (Block, 2000).

Research has found that patients with cancer, especially those with a diagnosed psychiatric disorder, have a higher risk of dying by suicide (Cheung et al., 2017). There is an increased risk of suicide in the first year following diagnosis, especially in the first 3 months, which may arise through a desire for self-autonomy and control, with or without an associated diagnosis of depression (Cheung et al., 2017). The relatively higher risk of suicide has also

42

been found to depend on the cancer type, generalised to those with a poor prognosis and high symptom burden in the first two years following diagnosis, and cancer types associated with long-term quality of life impairments (Hu et al., 2023). Figure 8 demonstrates the most frequent psychosocial risk factors for death by suicide in 2022. The factor 'limitation of activities due to disability', includes all types of disabilities and health conditions which affect an individual's abilities and/or a perceived limitation of ability due to a newly diagnosed illness.



Figure 8 Number of suicide deaths, by psychosocial risk factors in 2022 (AIHW, 2023c)

Although suicidal thoughts may be fleeting in many people facing end-of-life, 3,249 deaths by suicide were recorded in Australia in 2022 – an average of 9 deaths per day (AIHW, 2023a). Data reveals that 1 in 5, or 20%, of suicides in New South Wales in 2019 were people with a terminal condition or those who had experienced a significant decline in physical health and functional ability (Go Gentle Australia, 2022). The perception of loss of independence, quality of life, value, dignity, pleasure in life, and sense of usefulness and purpose is associated with increased risk of suicidal ideation (Cheung et al., 2017).

Some people with a terminal illness will resort to tragic methods to end their lives, believing that they have no other option to end their suffering. Palliative care aims to optimise quality of life and remove distress through symptom management. For example, palliative sedation helps relieve the refractory distress of a dying patient but does not hasten death (Mroz et al., 2017). This is not physician assisted suicide (PAS), euthanasia or voluntary assisted dying (VAD). Euthanasia refers to a health practitioner intentionally ending the life of a patient through medical means following the explicit request of the patient, and PAS involves the health practitioner providing or prescribing drugs for a patient to end their own life (Mroz et al., 2017). VAD laws and legislation in relevant States and Territories provide an option for

terminally ill people to die peacefully at a time and place of their choosing, either by selfadministration or by a physician stepping in to assist (White et al., 2021). Further discussion about VAD, euthanasia, or PAS are outside the scope of this research.

It is important to remember that every number in the total of lives lost to suicide is a person, and a life which was valued and will be missed. The impact from the loss of this person on family, friends and the community cannot be underestimated, with an unnatural death such as suicide likely to result in traumatic grief (Swarte et al., 2003). Tragic deaths also impact the personal and professional lives of the health practitioners and support staff involved, resulting in consequences for staff wellbeing, recruitment, retention and patient care (Croft et al., 2023). Although palliative staff are frequently exposed to death, the emotional effects following a death by suicide have been described as strong feelings of guilt, sadness, grief, responsibility and shock, in addition to fear of post-death investigation and concern about blame from employers or professional bodies (Croft et al., 2023; Fairman et al., 2014). Long-term repercussions from a terminal patient taking their own life may result in loss of confidence, hypervigilance regarding suicide risk in patients, aversion to caring for terminal patients, career change or early retirement (Croft et al., 2023; Fairman et al., 2014).

The place where people with life-limiting illness receive care and die, and the option of how death occurs, depends heavily on accessibility to quality healthcare services. The choice may be a consequence of limited resources and a lack of available alternatives, rather than preference (Gomes & Higginson, 2006). Peri-urban locations offer limited options for treatment and care and limited choice of location for end-of-life care, often resulting in relocation or foregoing preferences for care, treatment and place of death (KPMG, 2020; Saurman et al., 2022). Patient preferences for a specific dying process, in-home or community palliative care, or a home death have been associated with a higher quality of life and are often used as a marker of a good death (Boamah et al., 2021; Gerber et al., 2019).

Factors which influence patient accessibility to a choice of in-home services and preferred place for end-of-life care and death, not only influences quality of life and wellbeing but also impedes the grieving process for family and friends (KPMG, 2020; Saurman et al., 2022). Families in rural environments feel unsupported if they're unable to fulfil the final wishes of a loved one to die at home due to a lack of support from service providers (Spelten et al., 2019). Generalist primary health care services and specialist services delivering community palliative care in rural locations have limited capacity and resources to support palliative patients, families, and communities, resulting in challenges to meet both demand and expectations for quality care (Namasivayam et al., 2022; Weng et al., 2022).

44

3.3 Psychosocial support

Grief is the normal emotional reaction to loss. Providing grief and bereavement support for people dealing with impending death and their family members coping with loss are core components of palliative care that have a significant, positive impact on the people involved (Hudson et al., 2012; PCA, 2020). In many instances, however, this support may be suboptimal or even non-existent in areas outside of urban centres, with emotionally-burdened family members hindering patients' care and adjustment to their illness (Hall et al., 2012). Following a death, most individuals will utilise their own strategies and support networks to deal with the grief reactions, but the consequences of bereavement vary for each individual (Hall et al., 2012; Hudson et al., 2018; Thompson, 2017).

Grief is a broad term which incorporates the losses that an individual may experience over a lifetime and is evidenced by a series of psychological, physical, behavioural and emotional responses (Murray, 2023). Anticipatory or preparatory grief refers to the distress a person may feel prior to their own death or the death of a loved one. Distress, in this context, includes the concept of social death, which refers to the ways in which someone is treated as if they were dead or non-existent although they remain physiologically alive (Borgstrom, 2017). Social death involves a loss of personhood, loss of social identity, a loss of social connectedness, and losses associated with physical deterioration (Králová, 2015). Social relevance refers to the manner in which clinicians interact and perceive a person who is dying which subsequently diminishes their self-identify (Borgstrom, 2017).

Bereaved family caregivers may experience a different form of social death, resulting from isolation caused by caring for a dying person, and a loss of identity and role (Schneider, 2006). The term bereavement pertains specifically to the state of having lost a loved one to death and the time period after, although the terms grief and bereavement are often used interchangeably (Johns, 2015). The emotional and physical pain experienced following the death of a loved one or by someone experiencing anticipatory grief or social death, is different for everyone (Johns, 2015; PCA, 2020). Factors such as personality, available support symptoms, relationships and disease progression all contribute to coping ability (Grief Australia, 2022).

Known as complicated or prolonged grief disorder, family caregivers experiencing intense feelings of grief for extended periods can face severe mental and physical health problems (Boelen & Prigerson, 2007; Prigerson et al., 1996). A lack of preparedness for death, unmet needs or an unnatural death have been associated with more intense and persistent symptoms of distress in an estimated 10% to 20% of family caregivers coping with bereavement (Hall et al., 2012). All individuals experience grief in the context of their whole

45

being; however, traumatic grief can manifest as insomnia, substance misuse, depression, depressed immune function, hypertension, cardiac problems, cancer, and work and social impairment (Hall et al., 2012; Thompson, 2017). Prolonged or complicated grief increases the likelihood of frequent hospitalisation and higher use of medical services (Hall et al., 2012). When bereavement is the result of suicide, those left behind are at increased risk of ongoing mental health concerns and are two to five times more likely to die by suicide themselves (National Suicide Prevention Project Reference Group, 2020).

There is an expectation that all palliative care providers have the capacity to assess and provide referrals for family members and caregivers needing bereavement support (PCA, 2018c). Access to grief and bereavement support for individuals living outside of metropolitan areas, however, is often poor due to a lack of professionals experienced and educated in grief and loss, despite evidence suggesting the effectiveness of intervention (Currier et al., 2008; Johns, 2015). Palliative teams in peri-urban locations are often insufficiently resourced and may have limited access to generalist services let alone specialist bereavement support (PCA, 2020; Thompson, 2017). Furthermore, there has been a shortage of evidence-based strategies or accepted bereavement standards to help health professionals provide psychosocial and bereavement support (Hudson et al., 2018).

Bereavement support then falls to GPs or nursing staff with an interest in this area or generalist social workers who may be unwilling to accept a referral due to a lack of experience and skills in grief and loss (Bryce et al., 2021). Skilled psychiatry support via telehealth is available to support people in the community; however, even telehealth availability is limited by a lack of skilled clinicians (Nelson et al., 2023; Scheil-Adlung, 2015). A lack of bereavement support has the potential to significantly impact the wellbeing, physical and mental health and financial security of people unable to cope with the death of a loved one (Johns, 2015; PCA, 2020). The standard of quality care and access to skilled and competent staff should not depend on geographical location.

3.4 Ageing population

Another critical challenge is the current and projected increased rate of our ageing population. Australians are expected to live longer than ever, with the population projected to reach 40.5 million in 2062–63 (Department of the Treasury, 2023). Australia's population was 26,473,055 on 31 March 2023, indicating an annual growth of 2.2% or 563,200 people (ABS, 2023e). Australia's population will continue to age over the next 40 years, with the number of people aged 65 and over more than doubling, and those aged 85 and over expected to more than triple; the number of centenarians is expected to increase six-fold (Department of the Treasury, 2023). Figure 9 shows the distribution of age structures and the change between

1982–83, 2022–3 and projected age distribution in 2062-63. The reference to 'older Australians' used throughout this chapter, will refer to people aged 65 years or older.



Figure 9 Comparison of age distribution, 1982–83, 2022–23 and 2062–63 (AIHW, 2023b)

Many older Australians report a desire to age in place, but the relationship between this desire and the reality of housing and financial circumstances will ultimately influence decision-making (AIHW, 2013; Urban Development Institute of Australia (UDIA), 2023). Research has estimated that 440,000 older, low-income households in Australia will be unable to find or afford suitable housing by 2031, due to home ownership rates falling, reduced access to public housing and more people carrying a mortgage debt into retirement (Faulkner et al., 2023). The need for homeowners and those seeking rental accommodation to find affordable housing is predicted to significantly increase population growth in peri-urban, rural and regional locations (Faulkner et al., 2023). In addition, lifestyle and downsizing are important factors for older Australians when considering whether to relocate, with peri-urban population growth driven by

proximity to major cities, lower cost of living and retirement amenities (Centre for Population, 2019).

Future trends in population growth are difficult to project, but in the 12 months to June 2021, the combined population of Australian capital cities declined, while regional areas gained 49,000 people – an increase from 30,000 in 2019–20 (Centre for Population, 2022b). One of the challenges of predicting future growth in peri-urban locations is the lack of uniform distribution of growth, as development occurs at differing rates (Foster et al., 2013). Past trends, lifestyle factors, proximity to amenities and financial factors all indicate population growth of older Australians in peri-urban locations which will correspond with increased demand for community palliative care services (Faulkner et al., 2023; PCA, 2022).

Older Australians generally live healthier lives than previous generations, however, many people live with chronic health conditions. The ABS estimated that in 2017–18, 1.1 million older Australians lived with one chronic condition, 831,000 had two and 1.0 million lived with three or more (AIHW, 2021d). Corresponding with an ageing and growing population living with chronic conditions is the expectation that more people will be diagnosed with chronic illnesses and substantially increase the demand for palliative and end-of-life care (AIHW, 2023b; Nevin et al., 2019). The estimated demand for palliative care in Australia is expected to increase by 50% between now and 2035, and double by 2050 (KPMG, 2020). The trend for older Australians to relocate to peri-urban areas highlights the need to revise current healthcare services and methods of service delivery and reinforces the significance of this research.

One of the difficulties for healthcare and palliative care services providing care and treatment for people with chronic conditions in locations outside of urban centres, is the low level of exposure to the conditions by some primary care providers and the differing care needs of the illnesses. Most people die from chronic diseases that progress along one of 3 typical illness trajectories – chronic disease management, to treatment of advanced illness, to a palliative approach, end-of-life care and terminal phase (Reymond et al., 2018; Royal Australian College of General Practitioners (RACGP), 2023). The trajectories influence all facets of service provision, care needs and treatment provided, with pre-emptive planning for end-of-life care difficult for clinicians who lack specialist expertise. The trajectories are shown in Figure 10 and are as follows:

Trajectory 1: Cancer (short decline). This typically progresses steadily with slight decline in physical health over months to years and periods of positive or negative effects of cancer treatment. This is followed by a short period of evident decline with a clear end-of-life

phase of increasing symptoms and rapid decline in weight and functional status in the last weeks or months of life (RACGP, 2023).

Trajectory 2: Organ failure (intermediate with acute episodes). Organ failure typically occurs in non-malignant, life-limiting illness (e.g. advanced heart disease, lung disease). There is an intermediate rate of decline in function over years with long-term limitations and acute, often life-threatening exacerbations and hospital treatment, followed by further deterioration. Death often seems 'sudden' and may occur at any time along the trajectory, with symptoms of end-stage organ failure (RACGP, 2023).

Trajectory 3: Frailty or dementia (gradual dwindling). Typically, this trajectory is a gradual dwindling decline of physical and/or cognitive function. People with dementia have a long, variable disease course up to 6 to 8 years prior, early impairment of memory and reduced capacity for decision-making and communication. The last year of life is characterised by a steady slow decline in overall function, rather than a sudden decline in any one domain. Frailty and dementia together predict a more rapid decline (RACGP, 2023).



Figure 10 Typical illness trajectories towards end-of-life (RACGP, 2023)

3.5 Mortality rates

The concept of a good death or 'dying well' has been described as a multifaceted and unique experience for each person, shaped by individual attitudes, spirituality and cultural background (Meier et al., 2016). A good death gives people dignity, choice and support to

address their physical, psychological, social and spiritual needs (Swerissen & Duckett, 2015). A good death involves being able to choose and control the location of death, who is present, and what support, care and treatment is provided (Robinson et al., 2016). Dying well also requires good end-of-life care which involves palliative care involvement with support and services to optimise quality of life and reduce pain and suffering (Swerissen & Duckett, 2015).

Australians consistently voice a preference for a home death, surrounded by loved ones, yet over the past 100 years home deaths have declined and institutionalised deaths have increased (Swerissen & Duckett, 2015). There are multiple reasons why a person changes their preference for place of death; however, it is important to understand if barriers to accessing in-home palliative services are a significant factor. It is therefore important to explore mortality rates and places of death to provide a more holistic picture of peri-urban community requirements and to encourage more investment in community-based support to carry out people's wishes to die at home.

Figure 11 outlines all deaths across Australia from January 2022 to August 2023. There were 190,939 deaths in Australia in 2022, which is 20,000 higher than 2021 and is not considered to be a typical year for mortality in Australia; COVID-19 was a significant contributor to the mortality rate during this period (ABS, 2023b). In 2023, there were 122,112 deaths that occurred by 31 August, which is 11,974 deaths (10.9%) more than the baseline average (ABS, 2023f). In August 2023 there were 15,736 deaths, 1.5% more than the baseline average, and 13,856 of these deaths were doctor certified with the remainder coroner certified (ABS, 2023f). COVID-19 had a significant impact on life expectancies in countries other than Australia; in the US life expectancy fell by 2 years for males, and in the UK it fell by 1.5 years (Centre for Population, 2022a). The medium- and long-term effect of future COVID-19 waves and variants on mortality in Australia is unclear.



Figure 11 All deaths, Australia, 3 January 2022–27 August 2023 vs baseline benchmarks (ABS, 2023f)

The Australian Hospital Statistics provide numbers of deaths in hospitals and the National Coronial Information Systems for coroner-referred deaths (ABS, 2023f). Place of death can also be collected through the Civil Registration system on either a Death Registration Form or the Medical Certificate of Cause of Death for provision to the ABS by the Registries of Births, Deaths and Marriages (RBDMs) (ABS, 2023f).

It is important to consider that indicators related to place of death do not acknowledge patient preference, and emergency department presentations or number of hospitalisations may not reflect accessibility issues (Barbera et al., 2015). The ability of a service to maintain overall patient volume may represent effective maintenance of services, however, it may not represent quality person-centred care provision (Barbera et al., 2015; Lally et al., 2021). Statistics for place of death do not capture any important factors which contribute to the end-of-life experience. These include if the place of death was the person's preferred place of death, pain or comfort levels, psychological care, spiritual aspects or family support. Statistics do not include reasons for transfer, or who made the decision to transfer if place of death was not the patient's preference. Regardless of this lack of additional factors, the place of death results for 2019 were as follows:

- The majority (51.0%) of deaths in Australia occurred in a hospital/medical service area.
- Residential aged care facilities were the second most common place of death, accounting for 29.5% of all deaths.
- Other specified locations were the least common place of death (1.4%).
- Of the 169,301 records in scope, 5,792 records (3.4%) did not contain sufficient free text place of death information to be assigned to a specified place of death category in the Civil Registration system.
- Despite 60–70% of Australians expressing a wish for end-of-life care and a home death, only 14.8% of all registered deaths in Australia, or 24,970 deaths, were registered at a person's home or place of residence (not including aged care facilities) (AIHW, 2021a).

3.6 Impact and challenges of COVID-19

3.6.1 Background

The global scale of the economic and social disruption and devastating impact on life from the SARS-CoV-2, the virus that causes COVID-19, was unprecedented. More than 3 years into the pandemic, COVID-19 continues to have direct and indirect impacts on health and dying, particularly in areas outside of major Australian cities. COVID-19 was declared a Public Health Emergency of International Concern on 30 January 2020 with an official death toll of 171 (WHO, 2020b). As of 2 November 2023, there were 6,977,023 deaths reported to WHO globally – a number that is likely to be a gross underestimation (WHO, 2023). The first case of COVID-19 in Australia was confirmed by the Victorian Health Authorities on 25 January 2020 and was linked to an international flight from Wuhan to Melbourne (DoHAC, 2020). The first wave began in early 2020 and ended in April, the second wave emerged from 11 June 2020 and variants of the virus have continued to evolve.

3.6.2 Government response

Australian State and Federal Governments responded quickly to the impending health crisis with a whole of government response, aiming to minimise the number of people becoming infected, manage the demand on the health system and make COVID-19 vaccines available to everyone (DoHAC, 2022b). The result was sudden social, economic and technological changes across Australia, which significantly changed life for all Australians (Brousselle et al., 2020; Infrastructure Victoria, 2021). Government responses included developing guidelines and action plans to guide the response to COVID-19 from organisations and individuals. International travel restrictions and border control were a strong focus of the Federal Government's response, with a ban on travel from China implemented on 1 February 2020 (Costantino et al., 2020). Regular reviews resulted in travel advice and border restrictions rapidly adapting as situations evolved, culminating in a complete border closure to all non-citizen/non-resident international arrivals on 20 March 2020 (Costantino et al., 2020).

At the national level, economic support packages were released to secure jobs, financial support packages were made available to upgrade healthcare services, COVID-19 testing centres were established, traditional models of health service delivery were changed with telehealth services established, adequate medical supplies were made available and up-to-date information and education were provided to the public (DoHAC, 2023b). At State and Territory levels, responses included testing all suspected viral infections to ensure early detection, contact tracing and non-pharmaceutical intervention measures were implemented, such as quarantine, mandatory lockdown, social distancing and mandatory mask (Errett et al., 2020). Interstate and intrastate travel restrictions were initiated to protect vulnerable communities (Nguyen et al., 2022) and State and Territory Governments implemented policies such as the South Australian Acute Care Response Strategy, to support decanting patients to peri-urban hospitals in an effort to retain capacity at metropolitan acute sites for COVID-19-related care at peak infection times (SA Health, 2021).

3.6.3 Impact of counter-urbanisation

A relationship between the start of the COVID-19 pandemic and an acceleration of counter-urbanisation has been established (Beck & Hensher, 2021; McManus, 2022). Increased population growth in these peri-urban areas has had a direct influence on the quality and accessibility of palliative and end-of-life care due to the corresponding increase in the number of people potentially impacted by urban–rural inequity.

Media used the term 'urban refugees' to describe the unprecedented population growth in rural and regional areas due to people fleeing to the assumed safety of rural life (Argent & Plummer, 2022; Regional Australia Institute, 2021). People relocated away from densely populated urban areas solely to escape the risk of infection (Argent & Plummer, 2022; Denham, 2021; McManus, 2022; Nagel, 2020). Population statistics reflect the accelerated trend of counter-urbanisation in those metropolitan fringes which were previously viewed as rural (Guaralda et al., 2020). Reduced housing prices in peri-urban areas in comparison to metropolitan centres, increased the viability of relocation which was reflected in increased demand (Nygaard & Parkinson, 2021).

Improved road networks increased viability for relocating home for those employed in Fly-in Fly-out (FIFO) and Drive-in Drive-Out (DIDO) arrangements, thereby changing the traditional relationship between place of work and home (Charles-Edwards, 2021). The pandemic also influenced choice of permanent home location for those with dual residences such as a holiday house, and for couples living apart together (LAT) (Charles-Edwards, 2021).

Relocation was also a viable option as more people were able to work remotely from home; the routine of a daily commute was removed by integrating digital technologies into work routines (Amirinejad et al., 2018; Attorney-General's Department, 2021; Davies, 2021). Prior to the pandemic, approximately only 8% of employees worked from home, however, government requirements resulted in many finding the experience beneficial and effective (Productivity Commission, 2021).

The flexibility of working remotely provided the opportunity to choose a residential location based on lifestyle and affordability rather than commute distance and time (Beck & Hensher, 2021; Centre for Population, 2020). Remote working options offered the potential for increased autonomy, job satisfaction and significant financial and time benefits for employees, which could be invested into activities that improve quality of life for individuals and families (Beck & Hensher, 2021; Nagel, 2020). Major employers have indicated that flexible remote working arrangements are likely to remain in place, forcing employees that otherwise may have been reluctant to work from home, to adapt quickly to new technologies (Centre for Population, 2020; Nagel, 2020).

Working from home includes flexible working arrangements, which require employees to be present in the office for a portion of a working week. This practice, also known as telecommuting or hybrid, needs to consider whether the existing digital infrastructure can meet requirements. It also limits the choice of location for those seeking a rural lifestyle and working from home options to areas within commuting distance to urban centres (Denham, 2018). Periurban locations subsequently have become a long-term location of choice for many because it can generally support remote working or study options (Ruiz-Martínez & Esparcia, 2020).

COVID-19 unexpectedly accelerated the expansion of Australian major cities with growth in peri-urban locations increasing the already blurred lines between urban and rural (Argent & Plummer, 2022; Attorney-General's Department, 2021; Infrastructure Victoria, 2021; Wynne et al., 2020). Large high-density residential developments had already begun as land became available in areas previously viewed as rural (Walters, 2021). Development planning included required infrastructure to meet the needs of growing residential developments; however, infrastructure and service growth consistently lagged behind housing construction prior to the unanticipated growth from COVID-19 refugees (Newton et al., 2017).

The need to accelerate development of services and infrastructure in outer fringe locations should be an investment priority, given the scale of likely growth (McDougall & Maharaj, 2011). However, development creates a challenge for planning due to the need for a coordinated approach between the many local and state authorities involved (Reinhard &

Buckland, 2022; The Committee for Sydney, 2017). Financial and investment constraints further complicate decisions about whether to initiate a program of investment to retrofit ageing existing urban infrastructure or to develop peri-urban areas to meet community needs (Newton et al., 2017). People often have preconceived views of rurality – often a psychological construct of a lifestyle interacting with nature, a false belief of understanding life in areas previously designated for agricultural purposes (Gregory, 2009). There are benefits to rural living, but these locations lack uniformity in characteristics, demography and available services due to varying distances to urban centres (Hart et al., 2005; Taylor et al., 2017). This often results in deep chasms between expectations based on beliefs and goals, and the actual lived experience (Proulx et al., 2012; Ragusa, 2010). These tree- or sea-changers may not consider the possibility of poor mobile reception or unreliable internet or the time required to commute to friends and family, or actual proximity to local amenities, utilities and services, including pharmacies, medical clinics, tertiary hospitals or hospice facilities (Ragusa, 2010).

There is a view that the pandemic did not cause but merely accelerated an already existing trend of counter-urbanisation, especially when the daily commute was not an issue due to upgraded road networks (Amirinejad et al., 2018; Attorney-General's Department, 2021; Davies, 2021). In the 2020–21 financial year, internal migration resulted in 49,200 people leaving Australian capital cities, which contributed to the 70,900 people that moved to regional Australia (ABS, 2022d). The pandemic prompted people to change lifestyle. Many individuals realised that location was no longer a factor when they had to stay at home, social distance, and work, and study remotely (Guaralda et al., 2020; McManus, 2022).

A survey of internal migration patterns in Victoria post COVID-19 found that people aged 18–24 years were more likely to be influenced in choice of location by employment opportunities and those older than 45 years seeking lifestyle changes (Ghin & Ainsworth, 2022). Eighty percent of the surveyed population relocated to within 125 km of Greater Melbourne, which is considered peri-urban and within the distance considered as commuting distance and accessible to goods and services (Ghin & Ainsworth, 2022).

Financial year data from 2020–21, also suggests that COVID-19 impacted decisions to relocate, although it is unclear if the response will remain a temporary or a permanent transition as control measures change over time (ABS, 2022d; Davies, 2021; Guaralda et al., 2020; McManus, 2022). The ABS is also projecting that populations in major Australian cities will double between 2018 and 2066, with strong links between peri-urban areas and adjacent cities (ABS, 2018b; Thornton et al., 2022).

3.6.4 Modes of service delivery

The beginning of the COVID-19 pandemic posed significant and unexpected challenges for all healthcare services but particularly palliative care clinicians planning the delivery, availability, and quality of palliative and end-of-life care. At the commencement of the pandemic, the projections of hospitalisations and deaths due to COVID-19 posed enormous challenges to provision of person-centred care. The pandemic caused traditional primary healthcare services to dramatically change in a short timeframe (Stockdill et al., 2021; Taylor et al., 2021a). Primary care providers had access to very little evidence to inform, guide and support the required changes, resulting in negative reports from providers who felt that they were unable to provide appropriate care (Halcomb et al., 2021) The result was a rapid transition to telehealth for palliative clients at particularly high risk of poor outcomes through exposure to the virus (Lally et al., 2021).

The transformation of healthcare delivery to digital health had the potential to remove the barriers of time and location, to provide after-hours support and education and to remove feelings of isolation for palliative patients (Guenther et al., 2020; Lally et al., 2021; Namasivayam et al., 2022; Schulte et al., 2022). Telehealth provided the opportunity to enhance the number and quality of patient–physician encounters in the patient's own home and enabled family members to participate in clinical encounters, regardless of their location, with reduced risk of infection (Lally et al., 2021). Palliative care community providers were able to maintain contact with specialists in tertiary centres and participate in education via video conference. Multidisciplinary meetings were also able to continue in a virtual and safe environment (Milch et al., 2021; Ray et al., 2014). Other potential financial benefits to providers included reduced travel, shorter interactions and economies of scale (Snoswell et al., 2020).

The ability to maintain effective communication and collaboration between patients, families and palliative care service providers is crucial for shared and informed decision-making, which is essential when providing person-centred care (Saurman et al., 2022; Schulte et al., 2022). It is important to provide advocacy for people to participate in needs assessment and development of local initiatives to improve wellbeing and quality of life interventions in palliative care (Erdiaw-Kwasie & Alam, 2016). Transitioning to phone and video consults was necessary for palliative care services to maintain communication and complex care needs in a convenient and safe environment for patients and practitioners (Dolan et al., 2021).

The sudden change from a predominantly face-to-face method of community palliative care service delivery to one that incorporated telehealth was generally well accepted by clinicians; however, some voiced concerns which added stress to an already fragile health service (Bate et al., 2021). Practitioners and consumers were concerned about privacy,

security and a lack of accessibility and capability to use digital technologies (Dolan et al., 2021; Lally et al., 2021; Milch et al., 2021; Organisation for Economic Co-operation and Development (OECD), 2020). Other concerns were about the difficulty in conveying empathy and communicating sensitive information, particularly due to a lack of physical contact when touch is often used to support someone in distress (Collier et al., 2016; Milch et al., 2021). Practitioners faced difficulties responding to body language and facial cues due to an inability to view patients during telephone consults when video consultations were not possible (Eastman et al., 2021; Lally et al., 2021).

Cultural, psychological and disease-related factors influence the communication styles adopted by practitioners to convey necessary information to patients who may have misunderstandings about their prognosis and hold unrealistic hopes of a cure (Ghandourh, 2016). A review of people receiving community end-of-life care during COVID-19 identified feelings of concern about finances and infection but it was the loss of direct connection with their regular healthcare providers at that time that triggered increased distress and anxiety (Mitchell et al., 2021). Feedback has been sought by health services to understand practitioners' experiences and expertise with using digital technologies in palliative care provision so that programs can be developed and improved but feedback may be biased and not necessarily reflect the patient's or family's unmet needs (Dolan et al., 2021; Weng et al., 2022).

Whilst the change from a traditional face-to-face method of service delivery to digital health improved accessibility in the short term, it is important to refrain from viewing digital health as the panacea to removing the inequities of heath service access in all situations over the long term. Policies regarding service delivery need to consider the benefits of maintaining face-to-face contact, regardless of the geographic location of care, and instead view digital health as a complementary service method (Collier et al., 2016; Mills et al., 2021). Services such as palliative care are founded on respectful relationships developed through face-to-face delivery and generally include interdisciplinary face-to-face encounters (Lally et al., 2021; Milch et al., 2021; Namasivayam et al., 2022).

The Australian Government had already recognised the potential of telehealth in meeting the needs of Australians experiencing access barriers to healthcare services prior to the pandemic when they introduced a policy on telehealth funding in 2011 (Bradford et al., 2016). During the years that followed, Australian healthcare providers increasingly adopted digital health technologies in an effort to meet stakeholder needs and respond to issues in the health system (Hii et al., 2022). The COVID-19 pandemic, however, had an unintentional consequence of accelerating the national program of digital technology development as health

services sought methods to maintain service delivery (OECD, Beck & Hensher, 2021; 2020). The use of digital technologies removed the barriers of time, location and accessibility to all health care services whilst providing the benefit of reduced risk of exposure to the virus (Cherry et al., 2018; Hambleton & Aloizos AM, 2019; Schulte et al., 2022; St Clair & Murtagh, 2019).

All Australian healthcare providers were encouraged to move away from traditional face-to-face health care delivery due to risk of infection and develop their available digital technology to enable delivery of healthcare services during the pandemic (Petrie et al., 2021). Research found that rural Australian medical clinics developed their use of digital technology as the need arose and per the available technology (Petrie et al., 2021). Globally, the potential impact of digital transformation on future health reforms is tremendous (Nwosu et al., 2022; Taylor et al., 2021a). However, COVID-19 highlighted the challenges facing Australians living outside of major cities that may not be able to access reliable internet connections or have the digital literacies and skills required, particularly those aged over 65 years (Marshall et al., 2021).

A study in 2021 found that older people often fail to employ the full functionality of electronic devices such as the software tools required to enable video consultations, preferring the familiarity of telephone interactions (James et al., 2021). From a provider viewpoint, the uptake of digital health was limited to a trained workforce, the willingness of healthcare services to change healthcare delivery from face-to-face to digital, and the quality and accessibility of existing high-speed digital infrastructure in rural areas (Ruiz-Martínez & Esparcia, 2020).

3.6.5 COVID-19 deaths

The growing demand for palliative care from Australia's ageing population was already leading to an increase in demand for quality care prior to the COVID-19 pandemic. In 2019–20, around 43,400 or 50% of all palliative care hospitalisations were cancer-related (AIHW, 2022a). Of the cancer-related hospitalisations involving palliative care, 52% ended in death, 13% were transferred to another facility, and 30% were discharged to the person's home (AIHW, 2022a). In the initial stages of the pandemic, cancer screenings, including breast ultrasounds, mammography breast magnetic resonance imaging and colonoscopy, were suspended in an effort to reduce the risk of infection to patients and healthcare workers (AIHW, 2020; 2022). Delays in screening not only increased stress in patients but also had the negative effect of potentially delaying diagnosis and treatment and consequently allowing the disease to advance to a higher stage, which increases risk of death (Luo et al., 2022). Delaying cancer treatment by even one month may increase the risk of death by 6 to 13%, with the risk increasing as the delay progress (Hanna et al., 2020). A 4-week delay in accessing cancer

treatment is associated with increased mortality and an increase in demand for palliative care (Hanna et al., 2020).

In Australia, from January 2020 to 2 November 2023 (as shown in Figure 12), there were 11,629,977 confirmed cases of COVID-19 with 23,289 deaths reported to the WHO (2023). The following figure presents the distribution of daily new cases and numbers of deaths, from the beginning of the Australian pandemic experience to November 2023.



Figure 12 COVID-19 confirmed cases and deaths in Australia until November 2023 (WHO, 2023)

Key Statistics

- 20,170 of the 627,645 death registrations received by the ABS between March 2020 and September 2023 were people who died from or with COVID-19.
- The underlying cause of death for 15,920 (78.9%) of these people was COVID-19.
- There were a further 4,250 people who died of other causes (e.g. cancer) but COVID-19 contributed to their death.
- Chronic cardiac conditions were the most common pre-existing chronic condition for those who had COVID-19 certified as the underlying cause of death.
- 369 Aboriginal and Torres Strait Islander people died from or with COVID-19 since August 2021 (ABS, 2023c).
- Australia experienced low mortality during the early years of the COVID-19 pandemic. However, as infection rates increased significantly from the beginning of 2022, deaths

from both COVID-19 and other causes increased. ABS provisional mortality data shows that in 2022, the total number of deaths (191,000) was 10.9% higher than in 2021 and 15.5% higher than the historical average.

• The number of monthly deaths in Australia peaked in July 2022 at 18,000 deaths and then slowly declined before rising again slightly in December 2022 (ABS, 2023c).

Evidence suggests that the risk of severe disease, hospitalisation and death increases with age: 1.04% for people aged 20–29 years, increasing to 18.40% for those aged 80 years and older (Holt et al., 2020). Australians aged over 65 years and those with comorbidities remain disproportionately vulnerable to hospitalisation and death due to COVID-19 (as shown in Figure 13) (Holt et al., 2020).

AGE	DEATHS	%
0-19	23	0.1%
20-29	30	0.2%
30-39	72	0.5%
40-49	151	0.9%
50-59	441	2.8%
60-69	1,128	7.1%
70-79	3,033	19.1%
80-89	6,005	37.7%
90+	5,037	31.6%

Figure 13 Confirmed COVID-19 deaths by age group in Australia November 2023 (ABS, 2023c)

3.7 Summary

Most people will be touched by death in their lifetime, either personally or through family and friends. The benefits of accessing palliative care services in the location of choice, early in the palliative journey are well known (Hui & Bruera, 2020). However, delays in diagnosis and treatment are likely to result in difficulties supporting patients and families in the transition to end-of-life (Pesut et al., 2017). The full impact of the potential delays in cancer diagnosis due to COVID-19 and the impact on excess mortality in patients with cancer will not be known for several years and will require long-term monitoring (Lai et al., 2020). What is known is that the risk of being diagnosed with cancer increases with age. With more Australians living to an older age, the number of cancer cases diagnosed each year is estimated to increase by 22% by 2031 (AIHW, 2022a).

The social impact of barriers to accessing quality, person-centred care on health and dying are substantial and significant. The risk for physical and emotional distress is real for those individuals seeking palliative care but facing accessibility barriers. The removal of personal choice and preferences in care, treatment and locations for care and death has the potential to impede quality of life, wellbeing and the grieving process. The demand and need for community palliative care services is increasing and geographic location should not be a barrier to access. After all, the way we treat our vulnerable, our sick and those that are dying is a measure not only of ourselves, but of the society in which we all live. As Dame Cicely Saunders said, "How people die remains in the memory of those who live on" (cited in Berry, et al., 2017, p. 1).

Current research largely focuses on the barriers to accessing healthcare for vulnerable populations and the geographical location barriers in rural and remote areas of Australia. Few studies have focused on access barriers in outer fringe or peri-urban locations of Australia. The aim of this study is to explore the challenges to community palliative care provision in these peri-urban locations. In doing so, the research aims to understand how the classification of peri-urban locations influences health service policy and the impact of methods of service delivery, including digital technology.

Chapter 4: Methodology

4.1 Introduction

Chapter 4 outlines the research process and explains the corresponding methods applied to each stage. The research process employed an exploratory, inductive, and interpretative approach to respond to the research questions and meet the aim of the research. The chapter will outline the research design and discuss why the researcher chose to first conduct a literature review to provide evidence to confirm the problem space that was initially identified by the researcher's lived experience and to conduct a pilot study to confirm feasibility of the larger study. An ethnographic approach to the pilot study was adopted to understand the experience of clinicians providing community palliative care. To fulfil the aims of this study however, it was important to adopt a methodology that enabled the researcher to understand the views and experiences of people involved within the community palliative care setting.

It was deemed appropriate to develop a survey to collect data for the larger study and validate results from the perspective of participants from both service providers and recipients of care, including family caregivers and patients. The survey design, method of distribution, recruitment of participants and data collection will be discussed. The next stage outlines the qualitative and quantitative data analysis tools, in particular the use of a thematic analysis to allow the overriding challenges to emerge from qualitative data from the lived experience of participants. Descriptive statistics data analysis was used as a tool to compare and contrast emerging qualitative themes.

4.2 Exploratory, inductive, and interpretive approaches

The objective of a qualitative inquiry is to explore and explain the phenomena and the complex reality for those involved (Renjith et al., 2021). Qualitative research methods are often used in healthcare research to help understand health-related behaviours and to describe lived experiences by examining the 'how' and 'why' through the narratives provided (Creswell & Creswell, 2018; Miller et al., 2023). The use of free text and open-ended questions in a survey, allows participants to explain their intended meaning and offers an important opportunity to reveal issues that might otherwise go uncaptured (Rich et al., 2013; Riiskjær et al., 2012). One of the chief reasons for conducting a qualitative study is that it is exploratory, allowing the researcher to explore the research topic which has had minimal coverage within the literature (Renjith et al., 2021). In this study, the flexibility of this approach guided the research priorities and helped the researcher to develop definitions of location – a key component of the research (Casula et al., 2021).

Stebbins (2001, p. 3) defined exploratory research as 'a broad-ranging, purposive, systematic, prearranged undertaking of an area designed to maximize the discovery of generalizations leading to descriptions and understanding of an area of social or psychological life'. A qualitative exploratory approach illuminates how a phenomenon is manifested and allows the researcher to listen to the voices of participants to uncover the full nature of the phenomenon and contribute to the development of new knowledge in the area (Polit & Beck, 2017). To conceptualise the problem, the exploratory approach was broken down into stages (Miller et al., 2023). The first stage involved identifying and clarifying the problem space through a literature review, observation of clinicians in the field of research and input from experts in their respective relevant fields. This guided the next stage, which was formulating the research aim and questions. The final stage involved developing data collection methods which were based on the initial stages of research, and subsequently data analysis.

An inductive research approach was also undertaken. The primary purpose of an inductive approach is to allow findings to emerge from observation and the data provided, rather than rely on predefined expectations (Brantnell et al., 2019). An inductive approach helps the researcher to develop views and an understanding of the problem space and is used to guide exploratory research. Understanding is gained by observation and gathering data, and analysis searches for patterns and emerging themes to generate findings. The researcher did not compare or analyse emerging themes in the data until the intended data collection process was completed. The inductive approach provides additional rigour to the exploratory approach used to guide the research direction. Qualitative data analysis was conducted via organising data into themes by categories which allowed the researcher to discover emerging connections and similarities and compare aspects of the data (Gioia et al., 2013).

The interpretative approach was then applied to support the exploratory and inductive approaches to promote a better understanding of the identified problem space. An interpretative approach focuses on the tentative reality gained from the meaningful account of an experience provided by an individual that is intended to yield applicable insight into the collective experience (Hunt, 2009).

4.3 Conceptualising the problem space

The problem space was identified by the researcher's lived experience as a healthcare professional employed in community palliative care in peri-urban locations. The researcher drew on this experience to explore the perceptions of others involved in the clinical or operational aspects of care provision in peri-urban locations. The potential for bias or prejudice towards a position based on a personal viewpoint rather than a truth was recognised by the

researcher. Subsequently, methodology was selected to ensure appropriate fit for the research questions developed and to reduce researcher bias throughout the process (Florczak, 2021).

A pilot study was conducted to assist with developing methodology for the larger study and to ascertain the feasibility of the research (Hassan et al., 2006; Polit & Beck, 2017). Due to the exploratory nature of the pilot study, an ethnographic approach was used to examine the experiences, views and perceptions of health professionals involved in the clinical and operational aspects of providing community palliative care. Focused ethnography is the study of social interactions, perceptions and behaviours that occur within the particular culture being observed, including shared attributes relating to gender, customs, experiences and beliefs (Goodson & Vassar, 2011; Reeves et al., 2008). A focused ethnography is useful within a healthcare setting as it provides the social context and explores the social conditions in which participants live and work, their perceptions and how they are influenced by a diagnosis, treatment, or service (Holloway & Galvin, 2017; Wall, 2015). In this study, the cultural groups are involved in community palliative care; the philosophy of palliative care differs from other approaches to care (Friedrichsen et al., 2021).

Data was collected over the period January 2021 and July 2023 through observation and informal conversations, with overall concepts and relevant verbatim data documented in a diary. Multidisciplinary palliative team members were observed individually and in team meetings, both face-to-face and via on-line video communication platforms. Indirect conversations were conducted randomly and often in a reflective context during the debrief of a complex situation. The exploratory approach was supported by a thematic content analysis of diary notes. This identified patterns and connecting themes that supplemented the lived experience of the researcher (Mishra & Dey, 2022; Pope & Mays, 2020).

Themes emerged during the process of the researcher reading and re-reading data. This helped to identify the interrelatedness between and overlapping nature of the themes and their relationships to the problem space and guided the research design (Pope & Mays, 2020). However, experience, interactions, observation and discussions could lead to unintentional bias when developing research questions and analysing data, influencing the outcome (Simundić, 2013). The researcher first conducted a literature review that was used as an evidence base for the research (Elliott & Timulak, 2005).

4.4 Literature review

To guide the search strategy and ensure that a broad range of literature was identified, the following research questions were developed:

- How are peri-urban areas defined?

- How is a peri-urban location defined in community health policies?"

A systematic review of the peer-reviewed and grey literature was carried out with the aim of identifying and collating empirical evidence within the pre-specified eligibility criteria to address the research questions (Liberati et al., 2009). A review protocol was developed and followed, based on the Preferred Reporting Items for Systematic Reviews and Meta-Analysis (PRISMA) statement to increase clarity, transparency, quality and value of the review (Liberati et al., 2009; Page et al., 2021). PRISMA is a standardised collection of essential elements for documenting findings in systematic reviews and meta-analyses, based on solid evidence (Page et al., 2021). The review was conducted in five phases: – planning, searching, screening, appraisal and synthesis – to minimise bias, provide reliable findings and allow conclusions to be made (Page et al., 2021).

The online databases PubMed, CINAHL, Web Of Science and Scopus were selected for the literature review to provide comprehensive results. An initial literature search revealed that the term 'peri-urban' is not used consistently; instead a number of terms are used interchangeably, often depending on the classification system being used. Alternate terminology included urban fringe, semi-urban, exurban, peri-urban, metropolitan fringe, country living and rural, which add to the lack of clarity (Buxton & Butt, 2020). Therefore, key search terms used were 'peri-urban', 'rural', 'outer urban', 'outer metropolitan', 'outer urban', 'fringe', and 'definition', (policy or policies or strategy or planning).

A search of Australian Government websites determined that health service delivery is based on geographical statistical data provided by the ABS. This data is used to define locations as urban, rural, or remote, purely by population density and physical distance from urban centres. Non-metropolitan areas are viewed as those with less than 100,000 inhabitants and are further differentiated by a calculated level of remoteness. Peri-urban areas are not specifically classified in geographical statistical data.

Studies were eligible for inclusion if they were published in English, published after 1985, and set in locations deemed relevant by the researcher. 1985 was selected as a start date as it coincided with the commencement of the ASGC (ABS, 2009a). The initial search criteria were not confined to Australia and included global classifications of location searching for a common theme of rurality/peri-urban because there are common global issues related to peri-urban locations.

The researcher undertook all data extraction and all article analyses. The researcher read and reviewed articles for content and relevance to peri-urban location definition and relationship to health service delivery. It became evident very early in the research that excluding the rural element would limit the effectiveness of the review. Information pertinent to

the PRISMA review process was captured in a Microsoft Excel spreadsheet (Microsoft Corporation, 2018; Page, 2021).

4.4.1 Results

Figure 14 outlines the PRISMA process used for the literature review. 1,245 articles were identified using the search terms. Following removal of ineligible records due to language and duplication, abstracts were screened for relevance to the research questions, resulting in 59 full-text articles screened for eligibility by the researcher. An additional 45 articles were removed in this process, with 15 articles remaining. One article was removed because it was deemed irrelevant to the research and 4 articles added following a review of cited references. This resulted in 18 papers deemed eligible for the study (refer to Appendix B).



Figure 14 PRISMA diagram adapted from 'The PRISMA 2020 statement: an updated guideline for reporting systematic reviews' (Page et al., 2021)

Of the 18 articles used for this review, five were identified which discussed classification of peri-urban areas and the lack of consensus on a clear definition. The remaining 13 articles partially fulfilled the criteria by discussing the current methods used to classify rural areas or suggested improved methods of classification to meet the specific needs of rural locations for service delivery.

4.4.2 Data analysis

Using an inductive approach, extracted data relating to the research questions were grouped thematically to identify recurring elements in relation to peri-urban location classification and definitions. Thematic analysis required the researcher to read and re-read the full-text articles, code using complete thoughts and sentences derived from texts, and identify themes from the codes (Saunders et al., 2023). Three general themes emerged from the overall evidence by comparing and contrasting the data:

- There is no clear definition/classification of peri-urban areas for policy and planning,
- Classifications are geographical and in terms of accessibility are limited to population size, density and distance to urban centres, and
- Measures need to be developed to enable a clear definition of rurality so that meaningful policy and planning can occur. Measures would assist with workforce planning and GP clinic placement.

Development of themes identified themes-within-themes, or sub-themes, which demonstrate the hierarchy of complex meaning within the data (Braun & Clarke, 2006). A consistent sub-theme throughout all articles was the diversity in non-urban areas, regardless of the classification of peri-urban or rural. Two articles discussed the urban–rural health divide and the corresponding difficulties with resource allocation with use of geospatial classification derived from census data. Neither article suggested changes to current methods of measuring rurality based on population size and density, and distance from metropolitan centres.

The inadequacies of using SA2s as the smallest measure of an area as urban or nonurban was a common theme, with many authors voicing suggestions for alternative methods to be used in addition to current classification methods. Suggestions included thematic mapping, urban–rural gradients, spatial analysis or inclusion of other local factors that are deemed as relevant to the specific policy being developed. Additional factors to be considered included infrastructure characteristics, economic diversification, rate of urbanisation, and local employment. Five of the 18 articles discussed the need to recognise the uniqueness of periurban locations and the importance of classifying the area in a way that doesn't rely on the urban–rural perspective.

4.4.3 Discussion

The literature review supported the identification of a problem space, in terms of understanding the barriers to accessing healthcare services in peri-urban locations. What becomes clear in the literature is the importance of a definition of peri-urban areas that

considers the unique characteristics of that area. Current population density indicators are not reliable guides of populations living in a mix of high density, planned residential developments, rural living allotments and agricultural dwellings (Buxton & Butt, 2020). Defining peri-urban areas as rural infers low-density living, lower population sizes and a reliance on primary health care. The historical view that rural populations constitute an older and lower socio-economic demographic and high-risk, agricultural employment are no longer relevant in rural areas within high-density peri-urban locations. COVID-19, the ability to work from home, and housing prices in outer city locations have changed the demographic of the population moving from urban centres to these outer fringe areas, which lack appropriate access to infrastructure and services (Argent & Plummer, 2022).

Current classification methods rely on census data and may not encourage healthcare providers to make allowances for the culture, mindset or expectations from the populations moving from high-density urban centres to a rural living or country living area. There is an expectation that healthcare services would be accessible at a level available in urban centres. Measuring physical road distance between where people live and the urban centre locations where people need to travel in a straight line and viewing that as a reasonable commute distance appear to lack consideration of the multitude of challenges to service provision (ABS, 2001b). Physical road distance itself is not a clear and reliable indicator of the time taken to travel, the physical capability of the driver or passengers, the road conditions, the influence of climate conditions, geographic characteristics or socioeconomic factors, which ideally need to be factored into the measurement (ABS, 2001b).

Health service delivery based on a whole of population view of peri-urban areas as rural would appear to be doing a disservice to the communities because it does not focus on providing person-centred care for the individuals concerned. The development of PHAs, PHNs and LHNs appear to be a step in the right direction, but the areas involved continue to rely on SA2s which remain a significantly large area for which to provide adequate in-home services (ABS, 2020). There ABS also recognizes the need to provide access to relevant data about peri-urban areas to decision-makers and they are planning to release experimental boundaries in future ASGS data releases.

4.5 Theoretical framework

Access to palliative care services can be broadly defined as the capacity of people to obtain appropriate services in response to recognised care needs at any given time along the palliative journey (Corscadden et al., 2017). Subsequently, access represents a fit between

supply and demand, the interaction between patient needs and the services that meet those needs (Corscadden et al., 2017). There are many concepts and methods of measuring access to healthcare, demonstrating the importance of accessing health service performance worldwide. Some methods conceptualise access by utilisation of services due to characteristics of the provider, whereas others consider the roles of the consumers and providers in accessing services (Tanner et al., 2020). Penchansky and Thomas (1981, p. 128) define access as 'a concept representing the degree of "fit" between the clients and the system'.

Access to healthcare is vital, the Australian health system's key role is to provide safe, effective, appropriate and accessible treatment and other services to all Australians (AIHW, 2022d). In this context, access is defined as an attribute of the health services providing the opportunity to reach and obtain appropriate services in proportion to individual needs for care. Others remove the emphasis of access from the health provider and view access as the result of the interface between the characteristics of individuals and their environments and the characteristics of the providers and health systems involved (Penchansky & Thomas, 1981).



Figure 15 shows the Levesque model, highlighting how Levesque and colleagues define access as the opportunity to identify healthcare needs, seek, obtain or use services, reach resources and be offered appropriate health services to have the need for services fulfilled (Levesque et al., 2013b).



Figure 15 A definition of access to health care (Levesque et al., 2013b)

Levesque's framework (2013b) expands on the view of access as opportunity to reach and obtain services and explores the notion of accessibility. This suggests healthcare access is a complex concept of utilisation of services to realised access (Levesque et al., 2013b). The framework describes dimensions and determinants that integrate variations in access from demand and supply factors (Levesque et al., 2013b). Five dimensions of accessibility are proposed, which are inter-related, continuous and fluid (Levesque et al., 2013b). The 5 dimensions as represented on the upper part of Figure 16 are as follows (Levesque et al., 2013b):

- Approachability: services within reach, how the provider ensures that services meet the constraints and preferences of the client, including hours of operation.
- Acceptability: social and cultural acceptance of services and providers.
- Availability: timely delivery of services, the extent to which the provider has the resources to meet client needs.
- Affordability: financial capacity and willingness to use services.
- Appropriateness: fit between services needed and obtained.

The Levesque framework (2013b) proposes that five corresponding dimensions of abilities of populations interact with the dimensions of accessibility to generate healthcare access. They are represented on the lower part of Figure 16 and as follows:

• Ability to perceive: level of health literacy, knowledge and beliefs about healthcare and ill health.

- Ability to seek: personal autonomy, the capacity to choose to seek care.
- Ability to reach: personal mobility, availability of transportation and occupational flexibility.
- Ability to pay: economic resources to pay for health services.
- Ability to engage: participation and involvement in decision-making and treatment options.

The five dimensions of abilities are inherently person-centred, attempting to mirror the actual process of seeking the care needed by a patient on their palliative journey (Haj-Younes et al., 2022). The dimensions of accessibility and abilities are not independent constructs, often influencing each other and acting at different times during the episode of care (Levesque et al., 2013b). The conceptual framework guided the development of the research questions.



Figure 16 Levesque conceptual framework of healthcare access (Levesque et al., 2013b)

4.6 Study design

To reach the research goal, it was important to explore the expectations, feelings, and opinions of the people within the community palliative care setting. Therefore, a qualitative approach was deemed appropriate (Taherdoost, 2022). Qualitative data collection enables researchers to study the nature of phenomena from the lived experience of individuals (Busetto

et al., 2020; Gehman et al., 2018). Quantitative data, on the other hand, allows the researcher to determine cause-and-effect relationships, test hypotheses and generalise the opinions, attitudes and experiences of a large group of people (Verhoef & Casebeer, 1997).

The subject of death and dying is difficult and for some individuals and groups it is inappropriate to discuss the topic for personal, cultural, or religious reasons. The methodology adopted in this research was developed to reduce the risk of emotional distress to participants due to the research subject. The survey was voluntary, online and accessed via a URL link provided. Participants were able to leave the survey at any time, data collected was not linked to participants, and contact details for grief and bereavement support services were provided (Braun et al., 2021; Koenig et al., 2003). The requirement for a mandatory response was reduced to the questions which determined eligibility to participate in the survey, as per the National Statement on Ethical Conduct in Human Research guidelines (National Health and Medical Research Council, 2007). The research project ID H8936 received approval from the HREC at James Cook University (JCU). Refer Appendix C.

Figure 17 outlines the research design employed in this research. An exploratory design was used to develop an online survey as the data collection tool for the study. The researcher's lived experience, observation and informal discussions with community palliative care clinicians and a literature review guided the feasibility of the research and the development of the survey questions. Reliability and validity of the study were addressed by pretesting the questionnaire (n = 10) with multidisciplinary clinicians working in a community palliative care context, including clinical nurses, a social worker and a psychologist experienced in grief support (Hu, 2014; Kistin & Silverstein, 2015). The pretesting determined whether questions were clear, unambiguous and met the aim of the study (Shakir & ur Rahman, 2022). After feedback, minor revisions were made to the number of questions in the survey to reduce the length of time needed to complete. Other changes were made to improve clarity and the wording of individual questions (Choi & Pak, 2005). The survey was developed using the web-based software Qualtrics (2020). A copy of the survey can be found in Appendix D.



Figure 17 Research Design

4.6.1 Study inclusion criteria

To gain an understanding of the way people view their experiences accessing community palliative care, this study targeted those involved in both providing and receiving care, as follows:

- Health professionals employed in any role within a multidisciplinary health team, both paid and unpaid, full-time, part-time and casual, who provide in-home (community) palliative care services to patients and/or family members in Australia.
- Family members or care-givers of people who were currently receiving or had received, in-home (community) palliative care (this included direct patient care and bereavement care) in Australia. This group of people may have received community palliative care for bereavement support for an indefinite period following the death of the person

diagnosed with a life-limiting illness and would have had all direct patient services stopped at the time of death.

 People diagnosed with a life-limiting illness and receiving in-home (community) palliative care, including those receiving active treatment and those who had ceased treatment.

Exclusion criteria for all groups:

- Children and young people under the age of 18 years were excluded from the study due to the increased risk of psychosocial problems associated with this age group and grief and loss.
- Patients/persons requiring extensive medical interventions for effective symptom management due to either their current treatment regime or advanced illness stage were not excluded but were not actively recruited.
- Participants in any group, who lacked capacity to provide informed consent due to cognitive, intellectual or mental impairment.
- Persons who deemed themselves as vulnerable to emotional upset due to the nature or subject of the study and questions being asked for any reason, or who were experiencing diagnosed depression or PTSD, receiving ongoing psychological support to manage uncontrolled grief and loss, or cultural or religious reasons. Those recently bereaved, or at risk of chronic grief, were requested to exclude themselves.

Specific disease diagnosis, culture, gender, age (other than ensuring that the participant was over 18 years), specific health conditions or stage of disease were not sought in survey data to remove the risk of participant identification. Access to all methods of care, including face-to-face and digital health, were included and the type of care was not differentiated between symptom management, psychological support, or terminal, pre-terminal, comfort or bereavement care when identifying groups as patients, family members/caregivers or support persons/care givers who had received, or were receiving, community palliative care. Patients and family members were not excluded if active medical treatment was provided at the same time as community palliative care.

4.6.2 Sample descriptors

At the commencement of the survey, participants were asked to identify their role within one of 3 groups: healthcare professional, family caregiver or patient. Questions were allocated to participants depending on the group selected. A free text box was available prior to survey submission for every participant to include additional details if they desired to do so. Participants were asked to identify their location by State/Territory, or LGA and location type as either rural, remote, peri-urban, metropolitan, unsure or other. The researcher allocated participants to a location using the participants' identified LGA if location was identified as 'unsure'. Allocation was made using a manual search of the identified LGA in an online search of individual LGA determinations of location and peri-urban groups or networks of rural council lists (Peri-Urban Councils Victoria, 2022; Wollondilly Shire Council, 2022). Questionnaires collected information on participant characteristics, their perceptions of barriers to services, models of service delivery including digital health and impacts to wellbeing, choice of location and care received in the participant's location of care. The use of the term quality was not used in reference to care provided in the questionnaire, despite quality and palliative care being intrinsically linked. Individual perceptions of what constitutes high quality care are likely to be fluid and subject to change, therefore perceptions of care were related to individual expectations, values and needs, and subsequent satisfaction, to understand barriers to access (Elwyn et al., 2007).

The questionnaire directed at health professionals consisted of a combination of 20 Likert and multiple-choice questions. Quantitative data was sought to support the descriptive data. Open-ended comments were invited at 12 points in the survey and responses to all questions were voluntary. Two Likert questions with strongly agree–strongly disagree statements were included so that participants could rate their personal view of how services met the needs of the local community and how easy it was to interact with digital health services.

The family caregiver questionnaire consisted of 22 Likert and multiple-choice questions with open-ended comments invited at 10 points in the survey.

The patient questionnaire consisted of 20 Likert and multiple-choice questions with open-ended comments invited at 10 points in the survey.

4.6.3 Recruitment

To reach the research goals, it was critical to gather perceptions of access to services during the palliative care journey directly from everyone involved in the provision and receipt of care. It was not possible to identify, or contact every potential participant in the targeted population, due to the emotional subject involved in this research. There was potential for

participants to find the subject and the questions or responses provided by participation in the survey, to be a grief trigger (Wilson et al., 2021). This may have resulted in an unexpected rekindling of emotions of joy and gratitude of happier times or feelings of sadness, loneliness, regret and longing (Labott et al., 2013, 2016).

People cope with grief in different ways and some individuals, including those with diagnosed depression or someone who has only recently lost a loved one, will be more vulnerable to emotional distress from participation in this research. The researcher is an experienced palliative care clinician and was aware that the feelings of grief stirred up by the survey was likely to be of a short duration for participants who were not deemed vulnerable to emotional distress and they were unlikely to require ongoing support. The researcher chose to send the survey link to connections via an online community who was already interested in the research subject in order to minimise the risk of upsetting participants.

Palliative Care Australia is the national peak body for palliative care and represents all those who work towards high quality palliative care for all Australians who need it. Consent was obtained from the PCA to include the survey link in one e-newsletter (see Appendix E). Potential participants would only be able to access the link if they were accessing PCA online, had voluntarily subscribed to the e-newsletter or received the e-newsletter in a professional capacity (see Appendix F). The researcher acknowledged the potential for participants to post the link on social media sites or promote access in some other method so, a statement requesting that this did not occur was included in the Participant Information Sheet (Appendix G). Those with pre-existing emotional vulnerabilities, PTSD, depression or believed that they were in immediate danger were asked to not participate and to seek assistance, with contact details provided, including those for immediate assistance. Questions in the survey asked about the length of time services had been provided with responses linked to a message requesting participants reconsider participation if deemed at increased risk of emotional upset due to early or chronic grief. (Refer to the grief support information page included in the survey in Appendix H.

The risk to potential participants was reduced by including reminders in the survey link, the survey questions and the Participant Information Sheet, that participation was voluntary and to consider not participating or withdrawing at any time prior to submission. A link to the PCA resource on grief and loss was included in the Participant Information Sheet and grief support page, with participants able to download and/or print information. Further contacts for support were provided in the Participant Information Sheet with links that respected diverse cultures and genders and one link specifically for First Nations people. Risk was also minimised by making participants aware of anonymity in the survey and the choice to not answer questions.

The potential for any participant to voluntarily refer the survey link to anyone in their online network, outside of the targeted population pool, was outside the control of the researcher, and would be considered snowball sampling. The researcher did not actively request participants to recruit other participants, as it was possible that would generate bias (Gile et al., 2018). This form of recruitment does have the potential to increase the participation numbers but was not actively sought by the researcher (Gile et al., 2018). The researcher is unable to clarify the contribution of snowball sampling to survey responses.

PCA has over 22,000 subscribers and social media followers and participants were recruited from this pool. There was no way of estimating how many followers and subscribers are health professionals, patients, family caregivers or interested community members. Participants were given 10 weeks to access the e-newsletter via the PCA website to ensure adequate exposure within Australian States and Territories. The survey link was available between 9 February and 20 April 2023. The data collection method, convenience sampling, relied on the willingness and motivation of participants to provide information so it can be associated with poor participation rates. However, a benefit of this type of data collection is anonymity which potentially reduces social desirability bias and encourages participants to give honest and detailed responses to sensitive experiences (Luckett et al., 2021).

The study aimed for a sample size of 90 health professionals and 90 patient/family caregivers. This number of health professionals was chosen because the palliative care workforce is relatively small and approximately 52% of palliative nurses work in hospital settings (AIHW, 2022e). National 2019 workforce data showed that 9 in 10 (87%) palliative medicine physicians and 3 in 4 (72%) palliative care nurses worked in major cities, which may have swayed participation rates for a survey looking at peri-urban locations (AIHW, 2022e). Due to the emotional subject of this research, it was difficult to estimate the number of responses that would be received. Ultimately 94% of people who commenced the survey, submitted a completed questionnaire with 2 withdrawing at the initial research description, 4 declining consent, 4 failing to submit and 1 excluded as ineligible.

4.6.4 Data collection

All questions relating to access to services and models of service delivery were voluntary to reduce risk of possible grief triggers, to rule out any false claims and to increase the reliability and validity of the results (Wilson et al., 2021). Qualitative and quantitative data was captured concurrently, in the one survey, using Qualtrics (2020). The online survey was closed on the

advertised closure date to ensure that no additional responses were received outside of the targeted date range. Any details provided by participants in the free text boxes that could potentially lead to identification was removed by the researcher during reporting (Rich et al., 2013; Riiskjær et al., 2012). Reports were developed for completed surveys only, using Qualtrics and qualitative and quantitative data transferred to Excel spreadsheets for analysis by the researcher (Microsoft Corporation, 2018; Qualtrics, 2020). A total of 182 participants commenced the survey; 2 withdrew at the initial research description, 4 declined to give consent, 4 failed to continue to completion and 1 was excluded as location was identified as nealth professionals, 91 as family caregivers and 30 as patients (people diagnosed with a life-limiting illness). A total of 21 participants accessed the Participant Information Sheet and 32 accessed the grief support information page.

4.7 Data analysis

4.7.1 Qualitative analysis

The exploratory and descriptive approach was supported by Gioia's method for qualitative rigour which enables the researcher to find structure in unstructured qualitative forms (Gioia et al., 2013). The Gioia method requires the researcher to categorise the data into 3 different phases (first, second and third order or aggregate dimensions) (Gioia, 2021). First order concepts are identified from the data, using the voice of the participant to allow terms, phrases, and meanings to emerge as building blocks that form patterns (Gioia et al., 2013; Waeraas, 2022). The second order analysis looks for similarities and differences in the terminology, words and phrases used by participants, seeking commonalities that lead to patterns and themes in meaning and experiences provided. Finally, the third order or aggregate dimensions, reveal the common underlying patterns which comprise all of the second and first order elements (Gioia, 2021).

The researcher analysed the data using Qualtrics and Excel spreadsheets to manually search for common words and phrases (Microsoft Corporation, 2018; Qualtrics, 2020). Using Latent Semantic Analysis (LSA), the third order dimensions were identified by word frequency, based on the hypothesis that words and phrases occurring within similar text have similar meaning (Landauer et al., 1998). Phenomenological themes and sub-themes emerged during the process of reading and re-reading data, and the researcher was able to derive the range of factors that contributed to the whole lived experience of participants and identify the interrelatedness and overlapping nature of the themes. The researcher then interpreted the

findings, seeking to find meaning and respond to the research question whilst also taking into account the available literature to describe the phenomena involved (Gioia et al., 2013).

The Gioia methodology was developed with the purpose of creating an approach that enables a systematic and coherent presentation of both a first order analysis (using terms and codes based on participants' perspectives) and a second order analysis (using concepts, themes, and dimensions based on the researcher's perspective). The process of theme development is flexible and evolves through familiarity with the data (2006). It is possible for the researcher to provide a comprehensive and accurate classification of the connections between the data and the development of new ideas and themes (Gioia et al., 2013). Using the Gioia methodology, a data structure detailing the first order concepts, second order themes and third order dimensions and their relationships, is provided in Figure 18.



Figure 18 Gioia thematic analysis survey results

4.7.2 Thematic analysis

The results of the thematic analysis demonstrated the relationship between peri-urban locations and barriers to access. The data demonstrates some key areas that need to be addressed by the healthcare organisations involved, related to the influences of organisational characteristics and healthcare system-led support, culture, funding, policies and procedures, community infrastructure and individual needs and expectations. These factors will be discussed in the following chapters using direct participant quotes to directly connect data to theory and reaffirm the existence of relationships. Figure 19 demonstrates the relationships between themes and aggregated dimensions and their level of influence.



Figure 19 Aggregated theme relationship

4.7.3 Quantitative analysis – Descriptive Statistics

An objective and systematic approach was used to analyse responses to closed-ended questions in the online survey. Closed questions require participants to choose from a predefined set of responses, provided by the researcher (Creswell & Creswell, 2018). To help the researcher understand the participants' views and attitudes, this research utilised dichotomous, multiple-choice and Likert-scale questions to collect data. Dichotomous questions typically invoke a quick answer to a choice of two responses; in this study, this included 'yes/no' and 'agree/disagree' (Schaeffer & Dykema, 2011). Multiple-choice, including

radio-choice, questions were used. For some questions, participants could only choose one answer among a set of predefined options and checkboxes, and for other questions participants could choose multiple options from a given set of options (Strunk & Mwavita, 2021). Validity of the responses from multiple-choice questions was provided by pretesting questionnaire with (n = 10) multidisciplinary palliative care clinicians, to reduce potential for researcher bias (Considine et al., 2005).

Responses were filtered within each of the three groups of participants by Qualtrics, to locations identified by participants. These consisted of peri-urban, rural, regional and metropolitan. The number and percentage of responses were calculated using actual response numbers to each question provided by Qualtrics (2020) reporting and percentage calculated following export of data to Excel (2018), using the following formula:

 $P = \frac{F}{N} \times 100\%$

- P: Number of percentage
- F: The frequency of answer
- N: Number of responses

Likert-scale questions were also used in the survey, to measure the extent to which participants agreed or disagreed with a given statement regarding a particular topic (Strunk & Mwavita, 2021). Participants were asked to indicate their level of agreement by way of an ordinal scale with 5 options as alternative answers. They were: Strongly agree, Somewhat agree, Neither agree nor disagree, Somewhat disagree, and Strongly disagree.

Descriptive statistics attempt to aggregate data into groups to measure typical values and the spread of values within each dataset (Guetterman, 2019). In this study, measures of central tendency were used to analyse Likert-scale responses as interval data (Creswell & Creswell, 2018). A core element of descriptive statistics is the measure of central tendency which attempts to measure an entire sample to determine the collective attitude or opinion of the participant group with a single number (Strunk & Mwavita, 2021). The three most common measures of central tendency are the mean, median and mode, with the mean or average being the most frequently used. Results of analysis of quantitative data will be displayed in tables reporting mean and standard deviation for Likert scale responses, which were calculated using Qualtrics and Excel (Microsoft Corporation, 2018; Qualtrics, 2020). The spread of values is represented through measures of variability including standard deviation, which calculates the extent to which the values differ from the average (Guetterman, 2019). The researcher used convergent parallel design to capture both quantitative and qualitative data within the one survey (Creswell & Creswell, 2018; Fetters et al., 2013). Data was collected simultaneously and remained separate during analysis, which was conducted using Qualtrics reporting tools, Excel spreadsheets and manually by the researcher (Microsoft Corporation, 2018; Qualtrics, 2020). The intent was to merge, compare and contrast results from data analysis to add a greater depth of understanding to participant meaning and experiences reported in emerging themes from the qualitative analysis (Creswell & Creswell, 2018) – a process called triangulation. In this research design, validity of the research arises from capturing three different perspectives from three distinct groups of participants (Jick, 1979; Schoonenboom & Johnson, 2017).

4.8 Summary

This research applied an exploratory descriptive approach to provide an in-depth analysis of the problem space. The problem space was identified and conceptualised through qualitative analysis of the researcher's lived experience and observation and informal discussion with palliative care clinicians who work in peri-urban locations. To confirm the research gap and problem space, an extensive literature review of secondary sources was undertaken. This supported the development of the overarching research design, which is comprised of a qualitative descriptive approach, applying the qualitative interpretive framework. This informed the design of the survey, which was pre-tested with a group of 10 community palliative care clinicians and experts in their field, to ensure validity and reliability prior to distribution of the final questionnaire (Shakir & ur Rahman, 2022).

The broad data collection was undertaken through an online survey, which was distributed via an e-newsletter to provide anonymity of participants and to encourage honest and open feedback. Qualitative data analysis focused on thematic analysis applying the Gioia Methodology for Qualitative Rigor to ensure that the identified core themes and aggregate dimensions were reflective of the words of participants. The analysis of quantitative data identified the core statistical data, used to provide insight and understanding into responses provided by participants. The analysis and reporting were used to develop a number of recommendations with the aim of informing future policy and practice.

4.9 Limitations

Several limitations must be considered when interpreting our results. The finding that peri-urban areas are a factor contributing to barriers of access to community palliative care services must be interpreted within the limits of the study methodology and relatively small number of participants (n = 171). As this was an online survey, the researcher is unable to

calculate response rates. It is possible that the views of people who do not access PCA enewsletters and non-participants differ from those who did respond, with respect to accessibility of services and the outcome. Voluntary participation could represent participant bias, potentially contaminating results by being motivated to voice grievances and negative opinions. This research did not differentiate between specific disease diagnosis, a patient's care needs, prognosis, patient age or existing family relationships and extent of available support network, with potential for responses to differ depending on circumstances.

Chapter 5: Pilot study

5.1 Introduction

As discussed earlier, palliative care aims to minimise suffering and optimise quality of life for people diagnosed with a life-limiting illness and their family unit. Any barrier to access can result in anguish, reduced quality of life, and unresolved grief, and it removes choice of and preference for place of care. This chapter will discuss findings and focus on the challenges identified in peri-urban community palliative care from a pilot study conducted prior to developing the larger research study (Hassan et al., 2006). The study involved a small group of professionals providing community palliative care in a peri-urban region of South Australia which included in-home palliative and bereavement care. Evidence is presented by clustering and presenting the views of this group of community palliative care practitioners from South Australia to support the feasibility of a larger research study. A brief summary extracted from chapter 2 will be provided of the main features of the peri-urban location in order to present underlying accessibility limitations for peri-urban residents and palliative care service delivery.

As a reminder, it is worth noting at this point that health services typically base their policies on classifications of location, but the data is general and includes only two classifications – urban or non-urban – based on population size, density, and an index of assumed accessibility to goods and services (AIHW, 2022b; McGrail & Humphreys, 2009). The implications of this reliance on location classification methods, to accessing person-centred palliative care services in peri-urban areas will be discussed as findings are presented. Practitioners believed that challenges existed in care provision that could be directly attributed to geographical location and allocation of resources.

5.2 An overview of palliative care provision in peri-urban areas

A diagnosis of a life-limiting illness is distressing, and patients and family members often need health professional support in referrals for in-home services. In-home palliative care may be provided for days, weeks, or months and on some occasions for years (Tieman et al., 2016). However, not every person chooses to receive community palliative care and available care depends on the needs, available services, and location.

It is important to note that any factor which influences patient accessibility to in-home services may not only impact preferred end-of-life care and place of death, but also impact patient quality of life, remove patient choice, and impede the grieving process of family and friends (KPMG, 2020; Saurman et al., 2022). All people have a right to dignity, comfort and privacy, to direct their own care or have their wishes acknowledged, refuse medical treatment,

and expect equitable access to respectful and compassionate care at any point in their life journey, including at the end-of-life (ACSQHC, 2023a; Wenham et al., 2020). Evidence has shown that access to palliative care improves patient outcomes, including quality of life, and caregiver outcomes, including reduced stress and financial savings (Hawley, 2017). This basic right, however, is challenged by access inequities due to geographical location for those choosing to live outside metropolitan areas, as noted in the literature review.

The ability to physically access palliative care and/or the ability to travel is important for ensuring quality of life and wellbeing (Gerber et al., 2019; Miller et al., 2021). Geographical statistical data defines accessibility to healthcare services using the traditional Euclidean distance to model accessibility (Khashoggi & Murad, 2020). This approach is simply a measure of road distance in a straight line between home and the location of care (ABS, 2022a). This definition, however, appears to be too simplistic. Instead, access can be a complex concept, involving issues that include distance and time.

This research adopts Levesque et al.'s (2013a) concept of access to healthcare. Access is defined as the opportunity to identify healthcare needs, to seek, reach, obtain or use services and to have healthcare needs fulfilled. The Levesque framework broadens this definition by viewing access to healthcare services as the result of interaction between individuals and healthcare services. The framework proposes dimensions of accessibility and abilities that are interrelated and fluid (Levesque et al., 2013b). The 5 dimensions of accessibility are as follows; affordability, availability, accommodation, acceptability, and appropriateness. The corresponding dimensions of abilities of the population to utilise healthcare services are; the ability to perceive, seek, reach, pay and engage with the health services (Levesque et al., 2013a). The view of healthcare access described by Levesque and colleagues, suggests factors related to peri-urban location, structural constraints and characteristics of both individuals and services need to be considered to remove access barriers (Levesque et al., 2013b).

5.3 Study design

Due to the exploratory nature of the research, an ethnographic approach to the pilot study was selected to examine the experiences of health professionals providing community palliative care in one peri-urban region within South Australia. The researcher drew on her own lived experience as a community palliative care clinical nurse to explore perceptions of others involved in the clinical or operational aspects of care provision to support the feasibility of a larger study. Ethnography is the study of social interactions, perceptions and behaviours that
occur within communities, teams and organisations (Reeves et al., 2008). A focused ethnographic approach is useful within a healthcare setting, such as community palliative care, as it provides the social context and explores the social conditions in which participants live and work, their familiarity with the setting, their perceptions and how they are influenced (Holloway & Galvin, 2017; Kovan & de Vries, 2010).

5.4 Participants

The research was conducted in a peri-urban region in South Australia. The region extends into seven LGAs within commuting distance of metropolitan Adelaide and has experienced increased population growth. The region is characterised by picturesque countryside, high-density residential developments, small townships, and agricultural areas serviced by a road network of major highways, sealed and unsealed roads. The participants involved worked in three teams and were experienced in providing community palliative care in roles including registered nurses, social worker, occupational therapist, nurse consultants and home support workers/paramedical aids from two providers – one public and one private. **Error! Reference source not found.** provides a visual representation of the areas designated as peri-urban in South Australia and their proximity to metropolitan Adelaide.



Figure 20 Greater Adelaide Development Area Map (Attorney-General's Department, 2021)

5.5 Data collection

Data was collected from January 2021 to May 2023 through observation and informal conversations. Overall concepts and relevant verbatim data were documented in a diary. Participants were observed individually and in team meetings, both face-to-face and via Zoom and Teams (Barbu, 2014; Microsoft Corporation, 2019). Indirect conversations were conducted randomly and often in a reflective context during a debrief after a complex or traumatic situation. Demographic data with the potential to identify participants was not collected.

5.6 Data analysis

A thematic analysis was conducted to supplement the lived experience of the researcher. The process involved becoming familiarised with the data, developing codes and identifying themes as they emerged. Analysis involved the researcher reading and re-reading the hand written diary notes to identify recurring words, sentences and their meanings from the perspective of participants (Pope & Mays, 2020). Interrelated primary themes were identified including (a) risks to related to location, (b) challenges related to distance, and (c) lack of resources.

5.7 Results

The perceptions of community palliative care practitioners working in this region, supported the evidence from the literature that peri-urban locations influence access to care, services and support. Frequent comments were made about the level of care being akin to a lottery, where the level of care was predetermined by geographic location. The literature also describes this phenomenon as a 'postcode lottery' of locational disadvantage (Javanparast et al., 2022). This is despite palliative care standards which dictate that provision of person-centred care should not be challenged by limitations, including location (PCA, 2018a). Practitioners voiced their frustration at their perceived lack of support from policymakers, the difficulties of providing care in peri-urban locations and the increased risks to both practitioners and patients.

Risks to health professionals exist in all community care settings, due to complex care needs and family situations. However, practitioners clearly voiced the view that potential risks are exacerbated in the rural and isolated locations within peri-urban areas (AHA, 2019). Palliative care professionals often work autonomously, in unfamiliar environments, with a likelihood of a delayed response by emergency services or support services due to location and poor mobile phone coverage. Practitioners discussed the difficulties and lack of confidence they often faced due to working in isolation and travelling long distances to enter unfamiliar

environments, containing any number of extended family and friends, who are often highly emotional, overwhelmed, and unpredictable. There is great potential for any home to include domestic violence situations, weapons, domestic and working animals, people with mental health conditions, drug and alcohol addictions and generally unsafe working and living conditions (Terry et al., 2015).

Nurse D during a team meeting advised:

... need to reject this referral due to all the safety issues detailed ... The client also lives at ... a remote location with minimal mobile phone reception.

This view was supported by an informal discussion:

...Seriously I am not comfortable working with this gentleman...All history highlights the risk to workers, especially health workers. Also mention of sexual assault and physical assault history ... access to weapons ... but they are friends so how is this ok? (Social Worker A)

Evidence supports the benefits of relational continuity of care for optimal patient outcomes (Hudson et al., 2019). Small communities in rural locations promote an environment where overlapping relationships are inevitable and valued but potentially problematic when maintaining safe and ethical care (Campbell & Gordon, 2003; Szumer & Arnold, 2023). Urban healthcare professionals are generally able to separate their personal and professional lives, but rural workers tend to live, shop, and work amongst their patients, peers, and possibly extended family members, resulting in the need to navigate overlapping relationships (Campbell & Gordon, 2003; Cook & Hoas, 2019). The lines between maintaining professional, ethical behaviour to protect patients from harm and remaining objective by ensuring a clear separation between clinical and personal worlds in a small community can become problematic (Endacott et al., 2006). Many voiced a view that leadership either 'downplayed' or dismissed any potential risks voiced by staff, with many comments linking a perceived lack of support to overlapping professional and social relationships.

Home support worker X stated in an informal meeting:

... I was scared at the house but I didn't have a choice ... we were told to go.

Community palliative care professionals in peri-urban areas are more likely to experience physical hazards such as smoke, chemical and mechanical exposure related to agricultural activities, and physical injury related to manual handling (Terry et al., 2015).

Providers are required to complete a home risk assessment prior to a home visit, to mitigate risk and promote worker safety. However, risk assessments rely on access to an accurate history and an honest response from potential clients. Providers have a legal obligation to ensure that workers have a safe workplace and that risks are mitigated; however, practitioners recognised that the workplace in community palliative care is also the patient's home.

It is important to allocate resources so that quality, person-centred palliative care can be provided safely to those diagnosed with a life-limiting illness and their families. Current practices involve practitioners working in isolation and travelling long distances, with potential for impaired communication with specialists or emergency services. Difficulties exist in providing access to the services, support and equipment required to assist family members to provide quality in-home care:

...would like to spend more time ... instead of rushing through a shower And running out the door. It makes me sad. (Interview Paramedical Aide P)

Ideally, any person making the decision to receive care at home has considered the increased risk to themselves from receiving care at home rather than in a hospital or facility (Coombs et al., 2017). Community palliative care services can support individuals to make an informed choice of location for care and reduce this risk; however, this ability depends on accessibility:

I am over capacity and don't have time to follow up care coordination needs. (Nurse G, informal discussion)

Policies that influence all aspects of community palliative care are developed and implemented using classifications of location which generalise areas as urban or rural (National Rural Health Alliance (NRHA), 2022). Health policies on clinical procedures, medication management and digital health often fail to consider the implications of policy in areas outside of high-density urban locations. These implications include the accessibility of 24-hour care, in-home services, hospice care, and specialist input; the availability, storage, and transport of medications; and grief and bereavement support and choice of provider. In a team meeting, Nurse J outlined the policy requirements for nursing staff and the inferred implications for family members from clinicians following this medication policy:

Never, ever collect medication for a client ... You leave and tell family to ring when meds in house. They can go to hospital. Not our problem.

A lack of access to expected services in peri-urban areas results in distress and relocation for patients and families, especially when services to support in-home care is inadequate, insufficient, or unavailable:

This location adds increased difficulty to service provision, with difficulty delivering equipment needed ... finding staff ... telehealth doesn't work ... clients are sometimes moved to hospital because there is no other option. (Nurse L, informal discussion)

Health professionals discussed their frustration and stress when unable to provide the right care to patients and families, both face-to-face and via telehealth, including grief counselling and bereavement support that might otherwise be available in another location:

We have decided that palliative care involvement with a family ceases the moment a client dies and no further visits needed. (Nurse Consultant J, team meeting)

This was critiqued by many, for example:

Desperately need a ... grief/bereavement counsellor ...we are not meeting needs. I am frustrated that I cannot do what I am trained to do. (Nurse E, informal discussion)

The perspectives of health professionals demonstrate that we need to have a deeper understanding of challenges in peri-urban environments and promote the importance of future research into policy development.

5.8 Discussion

These findings provide insight into the views of community palliative care practitioners in a peri-urban region and support the feasibility of a larger study. Increased population growth in this peri-urban region, correspondingly increases the number of people potentially impacted by the urban–rural inequity, with the demand and expectations for healthcare challenging existing services. The literature demonstrates the known challenges associated with geographically rural locations that influence the accessibility and quality of healthcare provision for peri-urban residents (Schulte et al., 2022). Practitioner views highlighted the difficulties for populations in the rural and peri-urban locations involved, which have historically relied heavily on health services from primary health providers, including General Practices and community-based health services (Halcomb et al., 2021).

Current geographical classifications of locations as urban or non-urban are calculated using data such as population size, density, and an index of accessibility. Classifications are the result of an aggregation of spatial units that must also ensure residents' privacy and confidentiality. The large size of these spatial units results in broad generalisations of population density which influences the classification and ultimately affects health service provision. Practitioners discussed the implications of failing to consider access to community palliative care in peri-urban areas from a complex perspective.

Services such as community palliative care are based on person-centred care which these clinicians believed could not be provided in a safe, timely, reliable, or appropriate manner unless barriers to access are removed. The level of frustration with current policies that promoted a 'one size fits all' approach in service delivery was evident. Themes demonstrated the need for policy which considered the complexity of the area, the dynamic mix of high- and low-density urban living, the rural locations, distances involved, the available infrastructure and resources, and appropriate models of delivery for that specific location (Collis et al., 2013; Hart et al., 2005).

Findings from observations and informal discussions with clinicians highlighted the importance of developing strategies to improve access in peri-urban locations. Planning needs to be responsive to local community needs and include the experiences of local GPs, community members and the health workers providing the physical care. Development and planning for methods of palliative care service delivery need to consider feedback from community members to ensure that funding is available to provide appropriate resources in the form of trained staff, equipment, and infrastructure. Consideration is required to understand community needs regarding grief counselling and choices of location for end-of-life care. Furthermore, the risks identified by practitioners to both healthcare staff and patients need to be addressed as a matter of urgency, to maintain safety and to retain trained staff.

Digital health access played an important role in ensuring equitable access; however, digital health needs to be viewed as an important supplementary tool and not a panacea to access. Multiple factors influence the use of and access to digital technology, including motivation and willingness to embrace digital technologies and access to resources such as a reliable, trained, and qualified workforce and access to reliable internet (Bradford et al., 2016). There will always be a need for people to travel for specialist medical treatment that is only available in major metropolitan centres, but many voiced a preference for services, including palliative care, to be provided face-to-face. Given the expected growth in peri-urban areas and the challenges to health services to provide more services, it is important to work proactively to ensure vital community services such as palliative care are delivered effectively.

5.9 Summary

This study has explored the view of clinicians who provide community palliative care, obtained through an ethnographic pilot study. Views were obtained from observation and informal discussions conducted with a small group of clinicians, working in one peri-urban region of South Australia, to support feasibility of a larger study. This pilot study demonstrated the existence of a problem space, with the views of practitioners closely aligned with evidence from the literature. The voices of healthcare professionals promoted a view that the one-size-fits-all approach in service delivery methods in the peri-urban region under review was problematic. They also provided a depth of emotion related to the issues in service delivery that was not anticipated by the researcher. Themes highlighted the need for further research to consider the complexities involved with service delivery from a broader range of peri-urban locations and participants from both provider and consumer groups. Analysis of the data collected in diary notes from this small study not only reinforced the existence of the problem space but guided the development of the data collection method, participant recruitment and survey questions to be used in the larger study.

This study will continue through the following chapters which adopted an exploratory and descriptive approach to discuss the key issues from the larger study, derived from data collected from three clusters of participants: healthcare practitioners, family caregivers and patients. Chapter 6 will discuss the views of practitioners on accessing community palliative care in peri-urban locations, which aligned closely with the results from this pilot study. Key issues identified include the home location influencing accessibility to resources, including medication and staff, and the resulting impact on the healthcare professionals themselves. The results identified the common issues to care provision in many peri-urban locations, regardless of the State or Territory or organisation involved.

Chapter 7 will discuss barriers to access services from the perspectives of family caregivers, including the key issues associated with a move from face-to-face service delivery to telehealth. Chapter 8 concludes the survey results with the perspectives of patients on access to care and their expectations of service provision. Given the expected growth in periurban areas and the challenges to health services to provide more services, it is important to work proactively and listen to the voices of all involved, to ensure that vital community services such as palliative care are delivered effectively.

Chapter 6. Practitioners' perspective: barriers to access

6.1 Introduction

The complexities involved in palliative and end-of-life care provision invoke potential for differing priorities, needs and perspectives among patients, family members and the various healthcare professionals regarding treatment and services. Health professionals are in the unique position of understanding the access barriers from a systemic, workplace and patient viewpoint. Regardless of the differences, every viewpoint is important in gaining an understanding of barriers to accessing quality community palliative care in peri-urban locations. Therefore, the following three chapters aim to expand on the findings from the pilot study, discussed in chapter 5. The views of survey participants supported those identified in chapter 4, which identified the existence of barriers to accessing resources in peri-urban locations. However, participants in the survey expanded on the simple discussion of barriers to access and detailed their perceptions of associated risks and emotional toll for care providers in peri-urban locations. Many viewed the source of risk as a perceived inability to provide appropriate care.

6.2 Analysis of survey data

A qualitative approach was utilised to collect and capture data collected via an online survey. Participants were divided into three distinct groups, depending on their self-identified role. An exploratory and descriptive approach to analysing qualitative data supplied in the form of responses to open-ended questions and free-text fields was supported by a thematic content analysis to present the patterns and connecting themes of the details provided by participants (Mishra & Dey, 2022; Pope & Mays, 2020). Themes and sub-themes emerged during the process reading and re-reading of data, identifying the interrelatedness and overlapping nature of the themes and their relationships to the experiences of healthcare professionals (Pope & Mays, 2020). The findings are interpreted to detail the meaning within the criteria of the scope of the research question and positioned within the available literature to describe the phenomena involved.

Quantitative data was analysed using Qualtrics (2020) reporting tools, Microsoft Excel spreadsheets (2018) and manually by the researcher. The use of voluntary responses to questions to reduce risk of grief triggers resulted in varying response numbers to each question, which is reflected in the statistical results. A descriptive approach enables the voices of the individuals to be understood within the context of their lived reality and experiences and then compared with factual data to establish cause and effect (Busetto et al., 2020; Verhoef &

Casebeer, 1997). The next stage outlines the participants and provides the results of analysis, in particular the overriding challenges to emerge from the lived experiences of participants.

6.3 **Profile of practitioner participants**

Table 3 outlines the demographics of the participants who identified as health professionals. Of the practitioner participants (n = 52) who commenced the survey, one was withdrawn automatically because they worked outside of Australia and one voluntarily withdrew prior to completing the demographic questions. Of the remaining (n = 50) number of health practitioners who submitted a completed questionnaire, 70% (n = 35) self-identified as working in a peri-urban area, 16% (n = 8) in other non-urban (rural and regional) and 14% (n = 7) in metropolitan centres. Data that could reveal participant identity, such as gender and organisation name, location by postcode and specific age, were not requested to aid anonymity. The term 'non-urban' will be used to refer to participants who identified location as rural or regional, due to the small number involved. The median survey completion time for the group identifying as health care professionals was 662.5 seconds or 11 minutes 04 seconds. Table 2 presents the demographic details of the participants who identified as health care professionals.

Characteristic	Total family caregiver		Peri-Urban		Other non- urban (Rural & Regional)		Metropolitan	
	(<i>n</i> = 50)		(<i>n</i> = 35) (70.0)		(n = 8) (16.0)		(<i>n</i> = 7) (14.0)	
	n	%	n	%	n	%	n	%
Age								
18–34 years	5	10.0	2	5.7	1	12.5	2	28.6
35–49 years	14	28.0	9	25.7	3	37.5	2	28.6
50–64 years	15	30.0	11	31.4	2	25.0	2	28.6
65+ years	0	0	0	0	0	0	0	0
Prefer not to answer	16	32.0	13	37.2	2	25.0	1	14.2
Years working palliative care								
Less than 1 year	4	8.0	2	5.7	1	12.5	1	14.3
1–5 years	17	34.0	12	34.3	1	12.5	4	57.1
>5 years	29	58.0	21	60.0	6	75.0	2	28.6
Role								
Clinical:								
General Practitioner	1	2.0	1	2.8	0	0	0	0
Nurse Practitioner	2	4.0	1	2.8	0	0	1	14.3
Nurse Navigator	4	8.0	2	5.7	2	25.0	0	0
Clinical Nurse/ANUM	9	18.0	7	20.0	2	25.0	0	0
Registered Nurse	14	28.0	8	23.0	1	12.5	5	71.4
AIN/Carer/Home support	6	12.0	5	14.3	1	12.5	0	0
Allied Health-OT, physio	3	6.0	2	5.7	0	0	1	14.3
Psychologist/Social Worker	4	8.0	2	5.7	2	25.0	0	0
Clinical support:								

Table 2 Health care professional demographics

Volunteer	3	6.0	3	8.6	0	0	0	0
Admin/referral	2	4.0	2	5.7	0	0	0	0
Equipment/transport	2	4.0	2	5.7	0	0	0	0
Location by State/Territory								
Queensland	6	11.6	2	5.7	3	37.5	1	2.0
South Australia	42	80.8	33	94.3	5	62.5	4	8.0
New South Wales	2	3.8	0	0	0	0	2	4.0
Type of Organisation								
employed								
Specialist Palliative Care Team	16	32.0	11	31.4	4	50.0	1	14.3
Community Health	10	20.0	9	25.7	1	12.5	0	0
Public hospital	7	14.0	5	11.4	2	25.0	1	14.3
Private hospital	1	2.0	0	0	0	0	1	14.3
GP/Medical clinic	1	2.0	1	2.8	0	0	0	0
Home Care Provider	10	20.0	7	20.0	0	0	3	42.8
RACF Community team	3	6.0	1	2.9	1	12.5	1	14.3
Mental health service	1	2.0	1	2.9	0	0	0	0
Charity	1	2.0	1	2.9	0	0	0	0

6.4 Findings

Two initial questions were mandatory (informed consent and identification by participant group) in all three groups of participants, with the remaining questions all voluntary and differing in number and wording, pertinent to the participant role. All participants, regardless of role, were requested prior to consenting to participation, during the survey, and on submission, to read the information in the Participant Information Sheet (see Appendix G) and grief support page (see Appendix H) and withdraw from the study and seek assistance if required. The voluntary nature of the questions means that response numbers vary by question, therefore all responses provided were used and all percentages calculated from responses provided. There were 20 voluntary questions allocated to healthcare professionals with quantitative results provided in Appendix I.

Analysis of the data provided by healthcare professionals resulted in identifying two main themes that characterised the barriers to access experienced by people working to provide community palliative care – namely, the role of the home location in influencing resource access and the resulting risk to health care professionals. Sub-themes emerged related to the resources involved – namely, difficulties accessing staff, equipment, medication and training. These factors all increased access barriers for patients and families. Participants also noted a lack of support and recognition of staff needs by organisations involved in peri-urban care provision.

6.4.1 Home location

The factors associated with the location of the patient's home emerged as an important and complex issue. Of all palliative care professionals, 63.3% overwhelmingly highlighted that

the need to travel long distances influenced patient and family decision-making about treatment options, indicated with the low variance (m=4.7, sd=.7, v=.5, ci=.1) (refer Appendix I: Table 50) which aligns with the response from those specifically working in peri-urban locations: (n = 30) or 85.7%. Concerningly, 75.5% of all professional respondents indicated that their patients have been forced to relocate to avoid travelling long distances for healthcare services, demonstrated by the moderate variance in responses (m=3.8, sd=1, v=1, ci=.14) (refer Appendix I: Table 50); the responses from those in peri-urban locations was even higher, with even more believing that travel distance resulted in relocation: (n = 33) or 94.3%.

Participants were asked if they were aware of any of the following scenarios with the high number of responses from peri-urban professionals representative of the existence of access barriers in peri-urban locations. Table 3 outlines their responses.

Table 3 Scenarios related to access barriers

Scenario	Peri-urban respondents (<i>N</i> = 50)		
	Ν	%	
Hospital admissions extended due to in-home services unable/unavailable to support discharge	30	85.7	
Clients needing to attend medical clinics or outpatient clinics as no in-home services to provide medication administration	23	65.7	
Clients needing to attend medical clinics or outpatient clinics as no/insufficient in- home services available to assist with wound/catheter/stoma care	25	71.4	
Ambulance being called for assistance with pain/symptom control as community services unable to be contacted	29	82.9	
Respite in aged care facility as no/insufficient services available in-home to support hospital discharge	28	80.0	
End-of-life care provided in hospital as appropriate services unable to be provided in the home	31	88.6	
People relocating due to services being unable to provide the level of support needed in the home	32	91.4	

Provision of quality palliative care services requires access to the resources required by health care professionals when needed, regardless of setting or geographical location (ACSQHC, 2021a). Research has identified five categories of resources: (a) organisational characteristics and healthcare system-led resources, (b) social resources, (c) job resources, (d) home resources, and (e) personal resources (Lee et al., 2020). The categories are interrelated, with characteristics of organisations determining leadership support and culture, and system-led policies, legislation, funding incentives and clinical practice guidelines directly influencing task-related resources required by clinicians to meet patient needs (Lee et al., 2020; Scholl et al., 2018). Task-related resources can be defined as all the materials, equipment, medication, personnel, facilities, funds, training, educational resources and anything else that can be used to provide healthcare services (Ransom & Olsson, 2017). Equipment used for in-home palliative care includes pressure relieving

devices, oxygen concentrators, mobility aids, beds and syringe drivers, which are all vital for quality end-of-life care (ACSQHC, 2021a). Access to training, education, networking and support improves knowledge and self-efficacy which contributes to delivery of safe and quality care, and is crucial to those clinicians who do not routinely provide end-of-life care (Dehghani et al., 2020; Latter et al., 2020; Moorhouse et al., 2000; Prem et al., 2012).

Locations outside of urban centres challenge palliative care providers because of the increased travel times which impacts effective use and availability of resources and provision of quality care (Cinnamon et al., 2008). Health care resource allocation is influenced by funding arrangements, organisational structure and policies and are based on geographical statistical data which uses population density and distance from services as a guide to ascertain accessibility. Accessibility to healthcare is calculated on a straight-line measure of the road distance between home location and the boundary of the closest urban centre (Cerni et al., 2023). Equitable access to a hospital in Australia is calculated from home location to the closest hospital, which may or may not be able to meet the complex healthcare needs of the patient (Barbieri & Jorm, 2019; Cerni et al., 2023). What is not considered in this measure of access is the time and the financial cost involved and the subsequent influence on available in-home services and location of care.

In response to the statement 'Home location does not influence access to services', the response was low, with only (n = 2) or 5.7% of peri-urban participants in agreement. According to healthcare professionals, the distance and time involved in travelling to/from patient homes in rural areas was not factored into their pre-prepared schedule or roster of home visits for the day, subsequently limiting the time spent providing one-to-one care and the number of home visits possible in one day. When asked 'What barriers to accessing services do you believe apply in your location', one Assistant in Nursing (AIN) responded:

Rural location of some homes severely impacts ability to provide services in the home and how frequently. Peri-urban locations are viewed as urban with associated amenities and infrastructure, but in reality there are often areas of unsealed and poorly maintained roads that are also impacted by extreme weather conditions, roaming livestock and wildlife.

When asked 'What barriers to accessing services do you believe apply in your location', one Volunteer responded:

The roads and I don't drive in bad weather.

Other AINs explained how travel time influenced care provision:

Unable to visit when clients want me to, time is given to me to attend without consideration of distance needed to travel or ridiculous routes expected to take which waste time and petrol and annoy clients.

Provision of equipment and support services are aimed at lessening the burden on families; however, not all families appreciate the home becoming a hospital environment and the 'swinging door' of staff that may be in the home at any given time (Morris et al., 2015). A person-centred approach ensures that families have access to all services, supports and equipment that practitioners believe will assist during the carer journey. These caregivers are enabled to assert their right to choose which resources are provided and when, depending on their individual needs and wishes.

Regardless of preference, effective management of symptoms, including pain, breathlessness and fatigue, is an essential aspect of palliative care in any location, but particularly in the home environment (Baillie et al., 2018).

Any failure to address symptoms due to poor communication or planning will have tragic results, namely unnecessary pain, discomfort and suffering for the patient, distress for family members, and potential for relocation to hospital. The goal of palliative care is symptom control which is provided by a combination of non-pharmacological and pharmacological measures, with drug therapy an important component (Masman et al., 2015). Every patient has individual needs, however, many acute events during the palliative journey can be predicted by the experienced practitioners providing care, and anticipatory measures can be put in place. These include ensuring a supply of medications are available in the patient's home, the patient and family caregivers are educated in the use and administration of the medications, and appropriate supports are available (Ding et al., 2019).

Timely access to anticipatory medication is vital in assisting people with emergent symptoms, and access to these essential medications are vital to those who wish to die at home (Latter et al., 2020). A reasonable number, (n = 8) or 22.8%, of peri-urban participants reported difficulty accessing medication required for palliative and end-of-life care. There are multiple barriers to accessing medication in rural locations which has a detrimental effect on palliative care provision and symptom management. Access to prescriptions becomes problematic in areas outside of metropolitan locations, particularly those with GPs who lack experience in caring for palliative patients and may lack the confidence to prescribe opioids (Latter et al., 2020; Le et al., 2017). A multidisciplinary approach to developing organisational policies and procedures to promote best practice in community medication management is advisable to meet needs (Caring@home, 2020).

Palliative care is a fundamental component of general practice with the GP assisting with the transition from chronic disease management to end-of-life care, including symptom management, a terminal care plan, care after death and bereavement support for patients, families and colleagues (RACGP, 2023). Research has shown, however, that despite close relationships with patients and a willingness to provide care, many rural GPs lack the training and skills required for optimal palliative and end-of-life care (Abbey et al., 2020; Ding et al., 2019). GPs must deal with the time factor challenges of visiting patients in their own homes because they're not mobile; travel distances impact their time spent with the patient (Ding et al., 2019; Näppä et al., 2023). Access to specialist palliative care physicians can be difficult, with the benefits of digital health options depending on the willingness of medical staff seeking support and the availability of mobile phone and internet in the location involved (Ding et al., 2019).

Local community hospitals are generally smaller and often lack access to a hospitalbased pharmacist or pharmacy stock (Tan et al., 2012). This results in a reliance on smaller, community pharmacies that do provide strong community support but may delay access to medication via reduced opening hours, and a reduced range of palliative medication or available stock (Kuruvilla et al., 2018; Masman et al., 2015; Ogi et al., 2021; Tan et al., 2012). Essential medications for symptom management in palliative patients are generally well understood by clinicians; however, the choice of medication, dose and method of administration are tailored to meet the needs of individual patients at any given time (Masman et al., 2015). The medication, dose, timing and model of delivery are likely to change frequently, depending on need, and it is difficult for smaller pharmacies to provide timely access to a core standardised list of medications for the terminal phase (Masman et al., 2015; Tait et al., 2020).

Medication safety standards, principles, legislation and organisational policies determine administration, storage, dispensing, documentation and transport related to safe medication use (DoHAC, 2022a). All authorised activities involving controlled medications must be within 'the lawful practice of their profession', as determined by the standards and guidelines from all registration boards and professional practice peak bodies (Caring@home, 2020). Individual clinicians must also consider their personal scope of practice related to lawful practice (Department of Health (DoH), 2020). Restrictions on use of controlled substances are necessary to prevent illicit use, but these need to be carefully considered, as excessive restrictions have been found to hinder access to palliative care (Gomez-Batiste, 2017).

System-led standards and policies influence access to medication outside the metropolitan centres, but there are factors that can effect how well these are maintained in

peri-urban areas. There is a lack of trained staff to administer drugs, difficulties related to safe storage due to challenging family dynamics and prescribing, dispensing and transport issues related to distance and time (Scholl et al., 2018). The home location potentially increases stress on family members faced with the dilemma of leaving patients on their own at home or driving to the pharmacy if home delivery options are unavailable. Access to imprest or emergency medication for home visits would alleviate issues for clients impacted by location barriers, but these are subject to organisational, State, Territory and national policies, guidelines and legislation (Caring@home, 2020). Despite the common wish for end-of-life care at home, many people in the terminal stage are transferred to acute settings because of distressing symptoms that are unable to be adequately managed in the home (Champion et al., 2015).

Barriers to access which negatively influenced the ability of health professionals to provide quality person-centred care were evident in the responses to several statements. A high number of participants, (n = 25) or 71.4%, wanted the opportunity to provide a choice of day/time of service to patients and only 14, or 40.0%, of peri-urban professionals agreed that the choice for end-of-life in the home was supported, regardless of location. The importance of a person-centred approach to care is fundamental in end-of-life care, and the needs and wishes of the individual in all decision-making regarding treatment and care and the location of care, including a home death, should be the highest priorities (Slater, 2006).

Person-centred care is a core value in palliative care. It ensures individuals' autonomy, dignity, respect and choice and supports family needs, cultural and religious preferences (Agllias, 2018; Wilson et al., 2014). Provision of person-centred care supports a choice of options for care, with only 14, or 40.0%, of peri-urban professionals believing that choices for end-of-life care, including hospice, home death and voluntary assisted dying, were supported in their peri-urban location.

Participants pointed out that distance and time to attend a face-to-face visit influenced the model of service delivery, with one Assistant Nurse Unit Manager (ANUM) responding to the question 'What barriers to accessing services do you believe apply in your location' thus:

Specialists only providing care digitally.

A digital model of service delivery was a contentious issue with many participants voicing strong opinions about the importance of providing a face-to-face service. Over half the participants, or 57.1%, agreed that digital health is an important tool for ensuring regular communication with patients, and 67.4% of all professionals believe that digital health supports contact with patients, demonstrated by the low variance (m=3.8, sd=.9, v=.7, ci=.1) (refer to Table 4).

Table 4 Descriptive statistics survey Question 6.19 How much do you agree with the following statements?

						Variance	Confidence
Statement	n	min	max	mean	SD		Interval
6.19(1) (Professional) Digital health is an							
important tool in ensuring regular							
communication with patients	46	2	5	3.8	0.9	0.7	0.1
6.19(2) (Professional) I have the							
resources and confidence to use digital							
health options effectively	47	2	5	4.0	1.0	1.0	0.1
6.19(3) (Professional) Telehealth							
consultations are an acceptable							
alternative to face-to-face consultations							
with medical professionals	46	2	5	3.4	0.9	0.8	0.1
6.19(4) (Professional) Telehealth means							
that patients can access medical							
support that would not be readily							
accessible due to distance	47	3	5	4.0	0.7	0.5	0.1
6.19(5) (Professional) I feel that patient's							
choice of place of care is supported							
with digital health options	47	1	5	3.8	0.8	0.8	0.1
6.19(6) (Professional) Patients and							
family members are comfortable to							
remain at home knowing that they can							
access support at any time	47	1	5	3.4	0.9	0.9	0.1

These views supported the literature that found that, whilst the change from a traditional face-to-face model of service delivery improved accessibility in the short term, it was important to view digital health as a means of complementing current service methods (Collier et al., 2016). Services such as palliative care are founded on relationships developed through face-to-face delivery to convey empathy, communicate sensitive information and enable an appropriate response to body language and facial cues (Lally et al., 2021; Milch et al., 2021; Namasivayam et al., 2022).

Participants were asked to select problems they had experienced with digital health options. Over a third, (n = 13) or 37.1%, reported poor mobile coverage and 11, or 31.4% reported unreliable or non-existent internet issues. Participants also reported digital health service delivery issues relating to patients with hearing and language difficulties and people not answering their phones. Digital health problems reported by participants support the view that peri-urban locations retain issues known to exist in rural locations. The literature found that people living in rural areas have limited or no access to high-speed internet infrastructure or reliable mobile phone coverage, despite programs to fund improved connectivity and coverage (Department of Infrastructure, Transport, Regional Development, Communications and the Arts, (DITRDCA), 2022; Guenther et al., 2020).

The National Palliative Care Strategy framework for palliative care service development recommends 24-hour access to palliative care services (DoHAC, 2019). The transformation from traditional healthcare delivery to digital health has the potential to remove some of the barriers of time and location, help provide after-hours support and education, and lessen feelings of isolation for palliative patients (Guenther et al., 2020; Lally et al., 2021; Namasivayam et al., 2022; Schulte et al., 2022).

In a worrying response to the statement 'Access is available for support 24/7', only 4, or 11.17%, of peri-urban participants agreed. Ongoing 24/7 support is viewed as the 'gold standard' of quality palliative care service delivery because it to ensures access is available to information, decision-making and support for real or perceived crises in the home (Phillips et al., 2008). Unfortunately, this level of support is often only available in larger specialist palliative care units in metropolitan centres (Phillips et al., 2008). The desire for changes in peri-urban palliative policy so that 24-hour access could be implemented was evident in responses, with 27, or 77.1%, wanting to be able to provide 24/7 contact for support and 29, or 82.8%, wanting 24/7 in-home care options.

In response to the question 'What barriers to accessing services do you believe apply in your location?', 33, or 94.3%, cited a lack of resources/capacity from providers and 30, or 85.7%, said that care needs had increased beyond capacity of existing services. A lack of resources and capacity could explain the high number of responses citing that it took too long for services to commence: (n = 28) or 80.0%. Although the commencement of services can be crucial for effective symptom management and quality care, the referral time can be outside the control of palliative care staff due to administrative wait times. One ANUM commented that;

Time taken for assessments with MAC [My Aged Care] for in-home and ACAT [Aged Care Assessment Team] is ridiculous.

When asked to respond to the question 'Thinking about the organisation in which you provide community palliative care, are there any changes you would like to see?', the overwhelming response from participants in all roles was for more resources to support clients. They specifically cited the need for more staff, funding and equipment. Others wanted increased frequency of services (n = 30, or 85.7%), more available equipment (n = 22, or 62.9%) and higher staff numbers to support service available (100%). As one psychologist responded:

More staff, more funding then we could increase the services and increase the distance we could cover.

Most participants did not believe that appropriate bereavement care was available, despite this being a core component of palliative care (PCA, 2020). Considering the value of bereavement support, it was concerning to find only 6, or 17.1%, of the participants believed that appropriate care was available in their location. The need for more palliative care personnel was a recurrent theme:

More resources available to support clients, quick access to allied health. (Clinical nurse)

More staff would mean more people could attend appointments (Transport Care Coordinator)

Responses were compatible with the literature; bereavement support is often suboptimal or non-existent and there is a lack of service capacity for sustainable care for those experiencing complex grief in areas outside of urban centres (Hall et al., 2012; Reid et al., 2023).

Participants expressed frustration at patients' and family members' expectations of the level, quantity and type of services that were available in their location, with only 13, or 37.1%, of the participants believing that existing health services were meeting the needs and expectations of people in the peri-urban location where they worked.

When asked 'What barriers to accessing services do you believe apply in your location?', one AIN responded:

Clients tell me that they want me to come to their house for longer, at different times, and on different days.

The response from one Administrative Support participant was:

Expectation from clients for services that are beyond our capacity.

Health expectations have been defined as 'beliefs about future outcomes, formed through cognitive processes and influenced by previous knowledge and past experiences' (Janzen et al., 2006). Current health models encourage consumers to seek health information, to be empowered in the healthcare process, with the person-centred approach encouraging individual preferences and values to be acknowledged. Information is abundant and easily available through social media, meaning that patient expectations are high and they are likely to complain if there are inconsistencies between expectations and experiences (Coulter &

Oldham, 2016; Proulx et al., 2012). Peri-urban services provided in a rural model of service delivery are not likely to meet the preconceived views and expectations of people relocating to peri-urban and expecting the same level of service provided in metropolitan centres.

6.4.2 Increased risk to healthcare professionals

The literature highlights the challenges and risks that exist in all community care settings, due to complex care needs and family situations and the fact that potential risks are exacerbated in rural and isolated locations (AHA, 2019). Palliative care professionals often work autonomously in unfamiliar environments, likely in locations with poor mobile coverage which delays any potential response by emergency or support services. Providers are required to complete a home risk assessment prior to a home visit, to mitigate potential physical risk of injury to staff and to promote worker safety.

Regardless of their role, all palliative care professionals face many challenging situations in addition to everyday stressors, including the need to break bad news and absorb negative emotions, recurrent exposure to death, situations which challenge personal and religious beliefs, and secondary trauma (Breen et al., 2014; Cross, 2019; Wilkes & Beale, 2001). Working in palliative care involves constant exposure to deep human suffering that encompasses physical, spiritual and psychosocial dimensions on a daily basis (WHO, 2020a). It is reasonable to assume that this constant exposure to suffering and death would predispose clinicians to emotional exhaustion, stress, burnout or compassion fatigue (Clayton & Marczak, 2023; Cross, 2019). Indeed, palliative and end-of-life care is an area of healthcare that evokes strong emotions and opinions amongst health professionals and communities alike.

While many palliative care health professionals agree that the role is challenging, many also find the experience emotionally rewarding, known as compassion satisfaction (Aparicio et al., 2022; Baqeas et al., 2021; Sinclair, 2011). This is due in part to the relationships that develop between the patient, family and health professionals throughout the palliative journey and the sense of satisfaction or achievement at making a meaningful difference in peoples' lives (Aparicio et al., 2022; Slocum-Gori et al., 2013; Webster & Kristjanson, 2002). The majority of professionals agreed that staff maintain good relationships with patients/family caregivers receiving care, demonstrated by a small variance in responses (m=4.3, SD =0.6) (refer Appendix I: Table 50).

Spontaneous expressions of gratitude from families are valued by healthcare professionals and add to the rewards and incentives of providing palliative care (Aparicio et al., 2022). Coping strategies aimed at finding a deep and inner meaning in the work they do are key to health professionals managing the emotionally challenging palliative care

environment and not succumbing to burnout and compassion fatigue (DiTullio & MacDonald, 1999; Sapeta et al., 2022).

Peri-urban participants, regardless of role or location, recognised that the service provided was fundamentally beneficial to the community, using terms such as 'we do our best' and 'we provide good service'. However, in response to the question, 'Overall, are you satisfied with the access to palliative care in your location', peri-urban responses were poor with only 8, or 22.8%, of the participants feeling satisfied. A positive view of the service provided to the community through terms such as 'outstanding' and 'accessible to anyone' were provided only by the group that self-identified as working in metropolitan areas.

The overwhelming theme from peri-urban participants throughout the survey responses was frustration due to an inability to provide a level of care that they believed clients deserved because of a lack of resources, including equipment and staff. Research has shown that the perception of a poor outcome or 'bad death' is often viewed by clinicians as attributable to circumstances outside their control, including an inability to meet patient needs or patient characteristics (Semino et al., 2014). A perceived lack of support from leadership or an organisation was viewed by participants as contributing to poor outcomes. Research has demonstrated that a supportive leadership recognises the importance of their staff as not just a resource but as an asset to be developed, supported and retained (Sypniewska et al., 2023).

Organisations which equip nursing and other healthcare staff with the necessary resources, including equipment and appropriate staff levels, increase staff satisfaction and decrease the likelihood of adverse medical events, poor patient care and unmet patient needs (Duffield et al., 2011; Perry et al., 2018). Words such as 'sad', 'overworked', 'stressed' and 'exhausted' were used when talking about barriers to access. Sharing thoughts and feelings is a necessary coping strategy, but negative emotions indicate that there are barriers to accessing quality care and risk factors involved in palliative care delivery (DiTullio & MacDonald, 1999). As one peri-urban Occupational Therapist with 1 to 5 years' experience commented:

Staff are overworked and stressed. Client needs are not being met properly...palliative care is not recognised as important by [organisation named].

This comment was supported by two Personal Care Attendants, each with over 5 years' experience:

Would like to spend more time with them instead of rushing through a shower...and running out the door. It makes me sad that we can't do more.

We do our best but it could be much better, more staff, more time, more client focused.

A clinical nurse with over 5 years' experience summarised:

Worked for years to provide best possible care but it is disheartening not to be supported, no resources, no equipment, no staff, not supported financially for the work we do, not recognised, clients moving to hospital to die because family cannot cope and we cannot do anymore to help them, clients moving to aged care because family cannot cope and we cannot do anymore to help them, clients dying in pain because we just don't have time to get to them or given them the time they deserve, zero bereavement and grief support is pathetic and not good enough.

Caring for dying patients and the family dynamics involved can be difficult for any health professional. However, environmental stressors are magnified in peri-urban and urban locations because smaller communities are more likely to be closely tied together and it is more likely that there are pre-existing relationships between health professionals and their patients and family members (Campbell & Gordon, 2003; Szumer & Arnold, 2023). Overlapping relationships makes it more difficult to ensure a clear separation and setting of boundaries between clinical and personal relationships for staff; they are often unable to escape work-related situations and conversations in everyday social activities (Endacott et al., 2006).

Peri-urban locations retain the systemic difficulties associated with rural areas, including difficulties attracting and retaining experienced staff, heavy workloads, and peer isolation (Reid et al., 2023). The travel distances involved in rural locations further challenge staff in their ability to meet daily workload demands without compromising care or the flexibility to attend to the individual needs of dying patients (Sørstrøm et al., 2023; Wilkes & Beale, 2001). Working autonomously further increases stressors for staff, particularly those with limited experience and training, working to fulfil multiple roles, as clinician, friend and counsellor, with minimal or nil support for attending to the needs of patients and families.

Support for health professionals in these environments is vital to ensure that appropriate symptom control and personal care is provided to patients and families (Khalil et

al., 2022). A supportive workplace culture is vital for promoting informal debriefing with peers, MDT meetings or structured and formal support (Wilkes & Beale, 2001). One administrative support worker did not feel able to express an opinion in the workplace with this response:

I would personally like to see more staff, more equipment, more time and more care given to people in the community but I would not say that at work.

The literature demonstrates that personal factors such as experience, age, gender and character play a role in influencing attitudes and negative emotions which affect stress, frustration and burnout in the palliative care workforce (Baqeas et al., 2021; Frey et al., 2018). It is possible that providers are concentrating on physical risk mitigation and ensuring worker safety and promoting personal wellbeing and mindfulness, minimising or negating the role of the organisation.

Responses demonstrate the interrelatedness and overlapping nature of the themes. Many of the participants expressed negative emotions and thoughts about some of the themes. The influence of a scarcity of resources, combined with overworked staff and an emotionally challenging environment, cannot be overlooked when considering the prevalence of risks to palliative care clinicians (Gómez-Urquiza et al., 2020). The literature highlights the key issues required to meet the emotional needs of palliative care staff, including open communication with peers, organisational support with training and education, and access to resources (Yu et al., 2023).

6.5 Summary

This chapter has summarised the findings from healthcare professionals, focusing on the following key research questions:

- How does counter-urbanisation influence community palliative care provision in periurban Australia?
- What role does digital technology play in peri-urban community palliative care delivery?
- How does location influence accessibility to in-home palliative care services?

Those working in a palliative care provider role reported that distance in rural areas of peri-urban locations was a significant barrier to their ability to enable equitable access for patients and family members. Participants recognised that the difficulties accessing the equipment, medication, staff and support required to remain in the home environment ultimately have the potential to result in caregiver burnout, stress and relocation of patients. The level of negative emotion expressed by clinicians, regarding how difficult they found it to

provide resources in these rural areas, was an unexpected finding. A lack of satisfaction and increased frustration due to an inability to provide appropriate palliative care have the potential to lead to burnout, stress and an inability to retain the experienced and knowledgeable staff required in the palliative environment. Resources, including experienced health professionals, are vital for providing optimal, quality community palliative care that ensures quality of life, relief from distressing symptoms, and the choice of dying at home.

Whilst the literature identifies that barriers to access exist in all locations, the responses from health professionals in peri-urban locations support the view that peri-urban areas suffer the same inequities of access as in rural locations. Levesque et al.'s conceptualisation of access to healthcare was used to support this research, recognising access as a complex framework with integrated barriers identified by five dimensions of accessibility and corresponding abilities of populations (Levesque et al., 2013b). Responses highlighted the existence of all five dimensions of accessibility on the provider or supply side, which are: approachability, acceptability, availability and accommodation, affordability, and appropriateness (Levesque et al., 2013b). The five dimensions of abilities attempt to mirror the process of seeking care needed by a patient on their palliative journey, which is inherently overwhelming and emotionally challenging (Haj-Younes et al., 2022). Notwithstanding the role of system-led influences, the organisational role as a resource in itself, is an important factor due to the implications of culture and policies on equipment, staff, training, and time allocated in this sector of healthcare.

Rural locations and the tyranny of distance are factors that providers need to seriously consider when developing appropriate funding and policies. The role of digital technology in community palliative care delivery was an issue for many, who perceived the role of digital health as secondary to face-to-face consultations. The lack of available resources that health care professionals can provide to patients and the resulting negative influence on the clinicians themselves becomes a negative cycle, with a poor patient experience being the outcome.

Chapter 7. Family caregivers' perspective: barriers to access

7.1 Introduction

Effective palliative care in the home environment is provided by an MDT which includes both clinicians and informal caregivers, who may or may not include family members (Hulme et al., 2016). Informal family caregivers are the foundation of the palliative workforce, providing the majority of the care throughout the palliative journey, including end-of-life care and facilitating a home death (C. Gardiner et al., 2020; Hulme et al., 2016). Family and caregivers are central to the care needs of a palliative patient throughout the journey, providing practical support, transport, financial assistance and emotional support (O'Sullivan et al., 2021). However, there are demands associated with caring for a loved one at home, and family caregivers are often unprepared for the physical care needs, are unfamiliar with medical terminology, lack the knowledge and skills required to deliver safe care, and are unaware of the type or amount of care needed by the patient (Appelin et al., 2005).

To ensure that quality of life for the patient is optimised, family caregivers must interact with and navigate the healthcare system to access required resources (Appelin et al., 2005). However, what constitutes required resources differs for every patient, influenced by the presence or absence of personal support, knowledge, needs and preferences. Therefore, while presenting a second set of data as evidence, this chapter aims to describe the perspective of family and caregivers receiving community palliative care to help evaluate and identify required resources and the barriers that exist in peri-urban locations.

7.2 Analysis of survey data

Data analysis for the group who identified as family caregivers, followed the same approach outlined in chapter 6. Questions in the survey were modified for each group to enable participants to respond to experiences relevant to their role, with similar questions enabling the researcher to compare responses between groups, if required.

7.3 Profile of family caregiver participants

Of the 92 participants who identified as family caregivers at survey commencement, one voluntarily withdrew prior to completing the demographic questions. Of the remaining (n = 91) who submitted a completed questionnaire, 49.4% (n = 45) identified the home location as within a peri-urban area, 33% (n = 30) in other non-urban (rural and regional) and 17.6% (n = 16) in metropolitan centres. Data that could reveal participant identity, such as gender, location by postcode and specific age, were not requested, to aid anonymity. The median survey completion time for the group identifying as family caregivers was 769.5 seconds or 12 minutes

08 seconds. Table 5 presents the demographic details of the participants who identified as family caregivers.

Table 5 Family caregiver demographics

Characteristic	Tota cai part	al family regiver icipants n = 91)	Peri-Urban (n = 45) (49.4)		Other urbar Regio	r non- 1 (Rural & nal) 30) (33.0)	Metropolitan $(n = 16) (17.6)$	
	n	%	n	%	'n	%	n	%
Age								
18–34 years	9	9.9	5	11.1	2	6.6	2	12.5
35–49 years	21	23.1	10	22.2	7	23.3	4	25.0
50–64 years	29	31.9	13	29.0	8	26.7	8	50.0
65+ years	13	14.3	6	13.3	5	16.7	2	12.5
Unsure/PNA	19	20.8	11	24.4	8	26.7	0	0
Time receiving palliative care								
Less than 1 month	15	16.5	11	24.4	3	10.0	1	6.2
1–6 months	44	48.3	21	46.7	15	50.0	8	50.0
6 months to 1 year	18	19.8	8	17.8	6	20.0	4	25.0
>1 year	11	12.1	3	6.7	5	16.7	3	18.8
Unsure	3	3.3	2	4.4	1	3.3	0	0
Relationship to patient	07	40.7			45	50.0		05.0
Spouse/partner	37	40.7	18	40.0	15	50.0	4	25.0
	38	41.7	20	44.4	8	20.7	10	62.5
family/friend	16	17.6	1	15.6	1	23.3	2	12.5
Location by State/Territory								
Queensland	6	6.6	4	8.9	1	3.3	1	6.2
South Australia	59	64.8	30	66.7	23	76.7	6	37.5
New South Wales	11	12.1	4	8.9	3	10.0	4	25.0
Victoria	12	13.2	5	11.1	3	10.0	4	25.0
ACT	2	2.2	2	4.4	0	0	0	0
Tasmania	1	1.1	0	0	0	0	1	6.3

7.4 Findings

Family caregivers were asked to respond to 22 voluntary questions, following provision of informed consent. Analysis of data provided by participants in this group resulted in identifying two main themes that characterised the barriers to accessing services, as perceived by family caregivers – namely, (1) the influence of the home location and (2) the impact on family caregivers from unmet needs. The perceived need for specific resources was influenced by factors pertaining to the patient and family involved; the home location was identified as the underlying factor in all barriers to accessing the services and support required by peri-urban family caregivers. The impact on family caregivers emerged through identification of negative emotions and the need to relocate loved ones at end-of-life, due to barriers to access. Sub-

themes emerged which related to the home location – namely, the influence of time/distance and the reliance on local medical clinicians. Ultimately, barriers to accessing the services and support perceived by family caregivers as essential to the caring experience left a lasting effect on the family unit.

7.4.1 Home location

Although providing care for a loved one in the home environment can be rewarding, it is not without challenges. In the context of in-home palliative care, the role of family caregivers far exceeds a simple caring role and often involves the responsibility and skills required to effectively recognise and manage symptoms and administer medication, including opioids (Hudson, 2003; WA Country Health Service (WACHS), 2021). Many family caregivers lack the skills, education, experience or knowledge to cope with the demands of caring for a person's physical and emotional needs (Becqué et al., 2021). There comes a time when family members come to the realisation that there is no hope for a cure, their loved one is no longer 'living with an illness' but 'dying from an illness' and that death is inevitable (Steele & Davies, 2015). The challenges of coping with physical care needs are then accentuated by simultaneously coping with the overwhelming emotions associated with the impending loss (Steele & Davies, 2015).

Family caregivers are forced into a situation whereby they are 'learning as they go', needing support and services from health professionals to provide information, education, services and assistance to maintain the family unit and effectively participate in the decision-making process (Becqué et al., 2021). An optimal outcome for the family unit requires access to services to assist family caregivers to be successful in their role (Miller & Porter, 2021; Steele & Davies, 2015). In response to the question, 'Overall, are you satisfied with the access to palliative care in your location?', family caregivers living in peri-urban locations expressed the highest level of dissatisfaction: (n = 21) or 63.6%. Non-urban (rural and regional) and metropolitan responses were unexpectedly similar, with 9, or 50.0%, of the non-urban family caregivers and 5, or 62.5%, of metropolitan family caregivers expressing dissatisfaction.

Additional descriptive responses provided by peri-urban participants demonstrated an underlying level of appreciation of palliative care input, regardless of the amount, quality or model of delivery. Words such as 'wonderful', 'happy' and 'amazing' were used, particularly when the support eventuated with a home death. However, the predominant theme that emerged was the influence and impact of limited access to services on the caring experience for family caregivers. Negative emotions, including feelings of dissatisfaction, disappointment and anger, were expressed, due to perceptions of poor, limited or a lack of access to community palliative care services that family caregivers believed were required in the home to provide appropriate support. As one spouse expressed:

...appreciate palliative care nurses visiting and providing access to doctors... visits need to be more often and I need more help since he needs 2 people to wash him, clean up after using his bowels and to transfer him from bed to chair. I need someone 24/7 but palliative care provide someone 3 times a week for an hour...Can't get a hospital bed as he is not sick enough or about to die. I cannot cope.

Another spouse expressed the same view that access to services was limited in their periurban location:

Palliative care were helpful but was nowhere enough support. I could not do it alone and I kept telling the nurse but I only received one person to assist with washing my husband who could not get out of bed. There was an expectation that I would help which I did not appreciate.

An extended family member responded:

Too little, too much expectation for family to provide everything, took a long time to find anyone who would come to the farm and then they really didn't know how to look after someone in a wheelchair.

A child in the family caregiver role expressed:

Didn't like services just coming when they felt like it and sometimes just not turning up at all...

The quality of the care and the amount of support that is provided during the palliative journey lives on in the memories of the people left behind, long after death has occurred. So, it is imperative that palliative care service delivery meets the needs of the people involved. Considering the access to healthcare and medication needed by every person, when, and where they need it, is recognised as a basic human right, then family caregivers have every right to expect their loved one has access to the services needed (WHO, 2021). This expectation in the general community was reflected in survey responses.

All, (n = 45) or 100%, of peri-urban family caregivers agreed that 'There was an expectation that health services were accessible', highlighted by the low variance (m=4.8, sd=0.4, v=.16, ci=0.05) (refer to Table 6) which indicates that people believe that healthcare should be accessible across all locations. This was consistent across all locations, with 94.6%

of all respondents agreeing that health services should be accessible regardless of location, with a low variance (m=4.7, sd=0.5, v=.2, ci=.11) (refer to Table 6).

Table 6 Descriptive Statistics Survey Q2.7 How much do you agree with the following statements?

	n	min	max	mean	SD	Variance	Confidence
2.7(1) (Family) Existing health services were considered when selecting the home location							
Peri-Urban	45	1	5	3.2	1.6	2.60	0.24
Rural	23	1	5	2.1	1.2	1.40	0.24
Regional	7	1	5	3.0	1.5	2.30	0.57
Metro	16	1	5	2.8	1.4	1.80	0.36
Total	91	1	5	2.8	1.4	2.00	0.35
2.7(2) (Family) The location	was sel	ected as	health s	ervices v	were in t	he planning o	levelopment
stage	45		-	0.5	4 7	0.00	0.00
Peri-Urban	45	1	5	2.5	1./	2.80	0.22
Rural	23	1	4	1.7	1	0.90	0.20
Regional	7	1	4	2.0	1.1	1.10	0.40
Metro	16	1	3	1.7	0.7	0.50	0.26
Total	91	1	5	2.0	1.1	1.30	0.27
2.7(3) (Family) Consideration (eg hospice, aged care facility)	on was gi lity) whe	iven to a n selecti	ccessing ng the h	g service ome loca	ers for po ation	ssible future	health needs
Peri-Urban	45	1	5	2.1	1.5	2.10	0.21
Rural	23	1	4	1.8	0.9	0.85	0.17
Regional	7	1	5	2.6	1.6	2.50	0.60
Metro	16	1	5	2.2	1.5	2.40	0.40
Total	91	1	5	2.2	1.4	2.00	0.30
2.7(4) (Family) There was a	n expect	ation tha	t health	services	were ac	cessible	
Peri-Urban	45	4	5	4.8	0.4	0.16	0.05
Rural	23	3	5	4.4	0.6	0.32	0.11
Regional	7	4	5	4.7	0.5	0.20	0.20
Metro	16	4	5	4.8	0.3	0.10	0.08
Total	91	3	5	4.7	0.5	0.20	0.11

Despite their expectations, a large number, (n = 32) or 71.1%, of peri-urban family caregivers reported difficulty accessing services when required, and only 11, or 24.4%, found that services met their needs and expectations. Of the remaining responses to needs and expectations being met, 16, or 35.5%, said they were partially met and 18, or 40.0%, said they were not met.

Almost half, (n = 19) or 42.2%, of the participants in peri-urban locations complained that they were unable to access the service they wished to use, at a time or day of the week they wanted, and (n = 18) or 40.0% were unable to access the service as frequently as wanted. A further 10, or 22.2%, did not believe the local providers had the resources to provide what they wanted. Concerningly, 12, or 26.7%, were unsure why access was unavailable, and 16, or 35.5%, believed that the service they wanted to access was not available in their location. As the child of a person diagnosed with a life-limiting illness expressed: Is it policy or lack of funding??? why is there not enough hours of help?? mum wants to stay home. she does not want to die in hospital or at my house with her grandchildren there. Tokenistic at best.

An optimal outcome can be difficult to achieve in any location, due to unrealistic expectations from patients and families. Information is abundant and easily available through social media, meaning that expectations are high and people are likely to complain if care and services fall short of expectations (Coulter & Oldham, 2016). This scenario often results when people relocate from urban centres to peri-urban areas with expectations that the same services will be accessible (Ragusa, 2010; Taylor et al., 2017). Often this group of people have been swayed by the 'hype' of 'country living within commuting distance' and have failed to consider the possibilities of poor access to required services and the distances involved in travel (Ragusa, 2010). As 2 spouses living in peri-urban locations said:

Moved from NSW and expected better health services in the area we moved to...

we probably lived a long way from everything but that was our choice... I struggled to give him all the meds he needed to ensure he passed peacefully and quickly.

A large portion of peri-urban family caregivers, (n = 25) or 55.5%, agreed that 'existing health services were considered when selecting the home location'. In addition, 75.6% (m=2.8) of peri-urban respondents in this group did not select home location based health services that were still in the planning or development stage, but there was a relatively large variance (v=2.8, ci=0.22) (refer to Table 6). The decision to relocate to a peri-urban location, based on anticipated completion of required infrastructure to meet the needs of growing residential developments, can result in frustration due to unmet expectations (Newton et al., 2017).

Only 11, or 24.4%, of the peri-urban participants agreed that 'consideration was given to accessing services for possible future health needs (e.g. hospice, aged care facility) when selecting the home location'. This type of response follows the view that people are more likely to plan for death through completing advance care directives, making funeral plans and writing wills, than they are to plan for any future illness or decline in functional ability (Carrese et al., 2002). Refusing to consider future needs by leaving the future in 'God's hands' or having the 'take it as it comes' mentality, fails to consider the potential distress resulting from a lack of access to local services in peri-urban areas. Table 7 lists the responses to scenarios which illustrate that barriers to access exist in peri-urban locations, with the potential to cause distress and suffering to patients and family caregivers.

Table 7 Scenarios related to access barriers

Scenario	Peri-urban respondents (n = 70)		
	n	%	
One or more hospital stays have been extended as services unavailable in-home to	30	85.7	
support discharge			
Needed to attend clinic or hospital outpatients for medication administration as in-home	6	17.1	
services unavailable			
Needed to attend clinic or hospital outpatients for assistance with wound/catheter/ stoma	6	17.1	
care as in-home services unavailable			
Needed to call an ambulance for assistance with pain/symptom control as unable to contact	18	51.4	
community services			
Respite in aged care facility was required as in-home services unavailable to support	7	20.0	
hospital discharge			
Will need to transfer to hospital for end-of-life care to remain in local area	35	18.0	

Participants were asked 'What barriers to accessing services do you believe applies in their situation?'. A third of the peri-urban participants, (n = 15) or 33.3%, believed 'home location and the distance involved for service providers' was a barrier to access. Responses from non-urban (rural and regional) family caregivers also cited distance as a barrier (n = 23) or 76.7%. However, geographic locations are a known barrier to accessing quality and person-centred palliative care for Australians living in non-urban or rural, regional and remote areas (Schulte et al., 2022). Home location was not cited as a barrier by any metropolitan participants.

Corresponding to the previous question, 13, or 28.9%, of the peri-urban family caregivers cited 'distance and time involved in travelling to some health services not available locally' as a barrier to access in their situation. Again, non-urban responses were high, (n = 19) or 63.3%, and distance to health services was not considered a barrier to access for any metropolitan participants.

There is no current agreed definition of what constitutes reasonable access to healthcare services in Australia, regarding the time or distance a person needs to travel (F. Gardiner et al., 2020). There are multiple types of spatial measures of accessibility used in healthcare policymaking, with methods of measuring access including area-based or distance-based factors, as discussed in detail in chapter 4 (Wood et al., 2023). Only 5, or 11.4%, of the peri-urban participants travelled less than 50 kms to healthcare services, with the majority, (n = 35) or 79.5%, of participants, travelling 50–150km. The responses correspond within the general definition of peri-urban areas being situated at the broad interface between

metropolitan areas and their rural surrounds, within a 150 km distance from urban centres (Liu & Robinson, 2016). The remainder, (n = 4) or 9.1%, travelled over 150km.

Around a third, (n = 14) or 31.1%, of peri-urban participants believed 'time/distance influenced home location of the person requiring care', with 17, or 37.8%, believing that 'relocation closer to services that provide a higher level of care will be needed'. The need to relocate due to the peri-urban home location was expressed by several peri-urban family caregivers:

...whole family relocated from rural South Australia to be closer to Adelaide for treatment. we moved as close as we could to the city but still a long drive which mum couldn't manage when she got sicker.

We moved mum to a unit closer to...so she get more services as she got sicker.

We relocated to a house in the township to be easier and closer.

The time and distance involved in travel to healthcare appointments, treatments and tests recommended for palliative patients also has the potential to influence the decisionmaking of patients and their families. Geographic distance increases the time required to travel, with poor road surfaces and weather conditions creating difficulties for patients in poor health and who already face discomfort when travelling in any form (Cerni et al., 2023; Lalani, 2022). Responses supported this barrier, with 34, or 75.5%, admitting that 'travel was difficult due to pain, discomfort and exhaustion'.

Just over half, (n = 24) or 55.8%, of peri-urban family caregivers agreed that 'distance influenced decisions', which was similar to the non-urban responses: (n = 21) or 70.0%. For peri-urban family caregivers, decision-making about travel was further influenced by a lack of public transport options, (n = 37) or 82.2%, and managing the cost of petrol and parking, lost wages, and organising private transport options, (n = 35) or 77.8%. Family caregivers reported:

Mum kept missing her FMC appts because they were too much for her.

Mum has blood transfusions at hospital every 2 weeks and has other appointments. Travel is expensive 420kms per week.

Need a taxi for appointments costs \$1000000000.

Although only 6, or 13.3%, of the peri-urban participants reported financial barriers to accessing services, descriptive responses highlighted the number of family caregivers that believed that the only way to receive services needed to support their needs was to pursue private services, despite the financial outlay. Comments include:

Can't get anyone to help me overnight unless I pay for it and it is too expensive.

Couldn't afford to get help privately.

We needed more help at home but couldn't afford to pay for it so dad had to go to hospital.

Palliative care were helpful but was nowhere enough support. I could not do it alone and I kept telling the nurse but I only received one person to assist with washing my husband who could not get out of bed. There was an expectation that I would help which I did not appreciate. When I got nowhere asking for more help I told the nurse I would pay for what I wanted and suddenly there was enough staff to cover 24 hours a day for a week. I could sleep well finally.

The responses highlight the view that resources were unavailable to meet the individual identified needs and choices of palliative care consumers in peri-urban locations. Some, (n = 6) or 13.3%, peri-urban family caregivers believed that 'care needs were beyond the scope of existing services', which corresponds with the small number, (n = 7) or 15.5%, believing that 'physical symptoms were managed in the home'.

Research has shown that GPs and nurses working outside of metropolitan centres often lack knowledge and have limited confidence in managing end-of-life medication, including managing breakthrough medication for adequate symptom relief (Khalil et al., 2022). Palliative and end-of-life care in peri-urban and rural locations is often provided by generalist community nurses who are not routinely involved in palliative care (Khalil et al., 2022). As one peri-urban participant providing care to her mother responded:

Mum's GP didn't really know what to do when mum was dying.

A digital model of service delivery has the ability to bridge the gap in available services and provide after-hours support to family caregivers in peri-urban locations and decrease the need to attend emergency departments or transfer to hospital for end-of-life symptom management (Guenther et al., 2020; Lally et al., 2021; Namasivayam et al., 2022; Schulte et al., 2022). To meet the needs of people living outside of urban centres, a national program of development and implementation of digital technology to enhance the Australian healthcare system is underway (Davies, 2021; Rainsford, 2019). Regardless of the plan to develop digital technology to support healthcare, only 3, or 8.1%, of the participants believed that 'digital health enabled access to medical support that would not be available otherwise due to distance' or that 'their choice of place of care was supported with digital health', The majority of peri-urban family caregivers, (n = 37) or 88.1%, reported receiving a face-to-face method of community palliative care service delivery. Many in this group, (n = 26) or 61.9%, also reported receiving phone consultations, video consultations (n = 8) or 19.0% and text messages (n = 5) or 11.9%. Despite the progress in digital transformation in healthcare, only 4, or 10.8%, of the peri-urban participants believed that 'digital health consultations were an acceptable alternative to face-to-face delivery'.

Regardless of location, 84% of family members felt confident in their ability to use digital health effectively (m=4.3, sd=.7, v=.5, ci=.2), demonstrated by a low variance (refer Appendix I: Table 35). All 45 peri-urban participants responded to the question regarding problems with digital health options, and multiple responses were allowed. Only 19, or 42.2%, did 'not perceive any problems using digital health', although problems were identified with the remaining participants. A small number, (n = 5) or 11.1%, reported 'a lack of computer access/skills', 13, or 37.1%, reported 'poor mobile coverage', and 11, or 31.4%, reported 'unreliable or non-existent internet access issues in home locations'. Problems identified by peri-urban family caregivers support the literature on digital access in rural locations (Guenther et al., 2020). Few participants, (n = 4) or 8.8%, reported 'not using digital options' themselves or having 'unreliable/non-existent internet access' in their location. The remaining participants, (n = 10) or 22.2%, were 'not offered digital options by their local providers'. As one participant responded:

Phone and internet are dreadful. Need to use landline.

Only 7, or 18.9%, of the peri-urban participants agreed that 'digital health is an important tool in ensuring regular communication with the palliative care team'. This is contrary to the 100% of metropolitan participants who agreed. These views support the literature that has found that whilst digital service delivery improves accessibility in the short term, it is important to view it as a way of complementing current service methods (Collier et al., 2016).

Palliative care aims to control the significant symptom burdens, including dysphoea, pain, nausea and fatigue that result from disease progression, the treatment provided, or a concurrent mixture of both, and which results in distress, suffering and decreased quality of

life (Howie & Peppercorn, 2013). The literature has identified that the symptom burden also impacts the family caregivers, who often report anxiety and a need for more psychological and practical support to guide the patient and the family unit through the disease progression (Philip et al., 2021). Identifying and addressing the holistic needs of patients and their family caregivers, attending to advance care planning and committing to honest goals-of-care discussions should not be confined to the terminal phase (Pereira & Chasen, 2016). In 2002, the WHO recognised the need to implement palliative care earlier in the illness trajectory by modifying the definition of palliative care, replacing the term 'terminal illness' with 'life-threatening illness' (Pereira & Chasen, 2016).

Considering access to palliative care using the dimensions of ability involves ensuring that users, in this instance family caregivers, can identify that services exist, can be reached and will have the appropriate impact on the person involved (Haj-Younes et al., 2022; Levesque et al., 2013b). It is vital that family caregivers are provided with the information they require to perceive the need for palliative care services and support, which is determined by individual factors such as, health literacy, health knowledge and personal, cultural and religious beliefs about health (Levesque et al., 2013b). Many family members will be overwhelmed by a diagnosis of a life-limiting illness and have difficulty accepting that death will occur. Discussions with appropriate healthcare professionals can ensure understanding of the palliative phases and accessing services in a timely manner (Levesque et al., 2013b).

More than half, (n = 27) or 60.0%, of the peri-urban family caregivers reported 'difficulty finding information about service/s and identifying what services were needed.' A mother, aged 65 years and over, of a person with a life-limiting illness, described her experience as follows:

GP or oncologist/unit should have completed referral earlier. Should not have been left for me to do when everyone in the house was struggling.

The majority of peri-urban family caregivers, (n = 37) or 82.2%, reported receiving information about available services from a health professional such as, GP, specialist, or clinic staff. A referral to palliative care or a coordinated approach for a GP-led palliative approach to care is required before services can commence, which involves perceiving the notion that care is required and understanding what services are appropriate and when. The literature demonstrates that Australian GPs in non-metropolitan locations are more likely to report having the skills to manage palliative patients without support, and less likely to be positive about a good outcome from specialist palliative care teams (Johnson et al., 2011). Subsequently, many exhausted and overwhelmed family caregivers in non-urban locations totally rely on their GP to provide information.

Relegating a palliative approach to only the end-of-life period, removes the patient's right to be informed about the illness and its likely trajectory, thereby influencing effective planning and decision-making (Philip et al., 2021). Delaying commencement of care increases the likelihood of unnecessary hospitalisations, costly and futile treatments, visits to emergency departments, increased suffering and the associated financial costs to patients and the health system (Mason et al., 2021; Namasivayam et al., 2022; Newton et al., 2020; Radbruch et al., 2020). Despite the evidence of the benefits, palliative care is often underutilised or initiated late in the palliative journey (Sarradon-Eck et al., 2019). A scarcity of palliative care resources may result in referrals being deferred to an advanced stage of the illness (Hui et al., 2022). Differing disease trajectories and subtle signals in a patient's dependency may influence recognition of the need for referral to palliative care, resulting in delayed support (Hui et al., 2022). Regardless of the reasons for timing of referrals, research demonstrates an increased risk of hospitalisation for those unable to access or receiving specialist palliative care services late in the disease trajectory (Earp, 2021).

Cancer trajectories are often characterised by a clear differentiation between curative and palliative stages, with a short period of decline and a foreseeable death (RACGP, 2023). This differs from diseases, such as cardiac failure which is characterised by a gradual decline in function and unpredictable periods of exacerbations and remissions, resulting in a sudden death (RACGP, 2023). Patients may rarely see their GP during treatment and the need for palliative care becomes evident when the curative phase ceases and the patient is returned to their primary caregiver for comfort care (Claessen et al., 2013). Signals that needs are changing may only be evident to the GP following reports from family members and home support workers and every patient has differing environments, needs and preferences that need to be considered (Claessen et al., 2013).

One participant, a spouse in the 18–34-year age group, voiced her frustration:

...don't know where to look. Husband is 30 and doctor said everything is for old people.

Other family caregivers voiced similar responses:

...want to access...tried to access...nothing to access. Friends, family, neighbours all help.

Don't know how to access, what to access, where to start. Nobody ever said we could get help from palliative care. Told me package was more than enough but it isn't. I am nearly 80.

If we are not using it then we either do not need or want it or know it is available if and when we need it.

We didn't know help was available and so busy caring for mum we didn't know where to look.

Although initially the focus of palliative care was about relieving suffering at end-of-life, the practice of implementing care earlier in the illness trajectory is now considered best practice (WHO, 2013). Early focus on care is no longer relegated to cancer patients and is now integrated into symptom management for all life-limiting health conditions, ranging from advanced heart, liver, renal and neurological diseases (Pereira & Chasen, 2016; Radbruch et al., 2020). Care aimed at improving quality of life early in the disease progression has been shown to reduce depression and anxiety and improve patient satisfaction and quality of life (Howie & Peppercorn, 2013). Less than half, (n = 21) or 46.7%, of peri-urban family carers reported an improved quality of life due to accessing community services. As one spouse reported:

We were so busy living that we didn't think about needing any help...only had 2 visits from palliative care nurse before sending my wife to hospital where she died.

Implementing palliative care early in the journey is more likely to result in personcentred care, consistent with patient preferences and a home death (Howie & Peppercorn, 2013). Studies have shown that early palliative care may improve survival rates in a comparison of 2 groups of cancer patients – one receiving palliative care and one receiving traditional care (Mason et al., 2021).

The desire for earlier intervention was expressed by 2 spouses:

Wife relied on her spiritual healer and alternative methods so palliative care was very late in the journey. Might have helped more earlier but can't change the past.

Palliative nurse was lovely but referral was too late. My wife was 63 years old we thought we had more time. Oncology nurse referred us to palliative care and 3 days later she died.
Waited weeks for palliative care to contact then it was another week waiting for a visit to house. Then nothing happened. Husband went to hospital a week later and died there. I had to phone palliative care a week after he died to tell them he was dead.

The home location has the potential to significantly influence the timing of coordination and commencement of palliative care services required to meet the diverse needs of patients. Peri-urban areas retain the challenges known to exist in rural locations and research has shown that many rural GPs lack the training and skills required for optimal palliative and endof-life care (Ding et al., 2019). The distance involved in travelling to the home location influences time constraints and the financial impact of being away from the clinic, which are barriers to accessing the in-home palliative care visits needed to assess patients and support family members (RACGP, 2016). Furthermore, the time/distance that patients must travel for specialist palliative care needs may influence GP decisions regarding referrals and commencement of services (Currow et al., 2012).

GPs in peri-urban clinics are often generalists, and require access to, and availability of, upskilling opportunities in palliative care. Access to GP-led coordination of palliative care services in these communities is limited by service deficits, including smaller clinics, less access to specialist input, and smaller, less equipped local hospitals and less exposure to palliative care patients and medications (Latter et al., 2020). The influence on outcomes, the lasting effect on families and the compromises made due to limited access to services required by palliative patients is evident in the following response from a peri-urban family caregiver:

Wanted overnight care sometimes to have a break, the help we got in the house was great but more visits a day to help with incontinence would have been much better, it was wonderful having her home and we love that we could but we still sent her to local hospital at the end when it got too hard and she died there. It wasn't hospice but it wasn't a big ward in town so it was a compromise.

The inability to access services when wanted or needed in the home, was reported by several family caregivers as contributing to not achieving a wish for a home death or visits to emergency departments due to the inability to manage symptoms in the home. It is often viewed as a privilege to fulfil a person's wish to remain at home, surrounded by their loved ones, which would not be possible without the dedication and commitment of family members providing 24-hour care to ensure comfort, safety and dignity (Breen et al., 2018). A supportive family environment has been found to be one of the determinants of a good death (Meier et

124

al., 2016; Saarinen et al., 2023). Family caregivers may not understand what resources are needed, what is available and how to access help, with a lack of resources resulting in a preference for relocation due to concerns regarding increased risk to the patient from poor care in the home (Coombs et al., 2017; Munday et al., 2007).

Despite the common wish for end-of-life care at home, many people in the terminal stage are transferred to acute settings because symptoms cannot be managed in the home 24/7 or because there are a lack of services to support the family (Champion et al., 2015). Whilst it is an option that provides quality, person-centred care, a hospital death may not be the patient's preference, and it is more intrusive than a home death and increases the burden on the health system (Mortel et al., 2017). Ensuring that family caregivers have access to appropriate palliative care means that they can avoid unnecessary hospitalisations and presentations to emergency departments (Khanassov et al., 2016; Taylor et al., 2021b). Based on 2020–21 National Hospital Cost Data, the total palliative care expenditure in public hospitals was \$581.4 million (1.1% of total public hospital expenditure) (AIHW, 2023b).

Considering the importance of supporting family caregivers, the low responses from caregivers about access to help was concerning – only 5, or 11.1%, of the family caregivers living in a peri-urban area reported that they 'had access to support 24 hours a day, 7 days a week'. Furthermore, 60.8% of family members did not agree that digital health supported their choice of home care, highlighted by the low variance (m=3.3, sd=0.9, v=.9, ci=.3) (refer to Appendix I: Table 35).

Family caregivers said the following:

Room to improve and provide more after hours instead of calling an ambulance or the local hospital.

Wanted services in the house to provide 24-hour care. Didn't happen.

I cannot cope. It is too hard. If his family weren't staying here 24/7 I would have just given up by now. He shouldn't have to go to a nursing home just because he can't walk, his mind is ok and he still enjoys life.

Very unhappy being told that services not available to support my husband at home while I am at work in the city. He cannot remain home alone, particularly when I often stay overnight in the city due to distance involved with commuting. He now has to come to the city with me and stay at our daughter's home where someone can watch him. I thought palliative care could support us more than an occasional 1 hour. Not good enough. We now have to move from our forever home in a peaceful valley where we intended to stay and move into the city. 24hour care should be available anywhere. We are less than 100 kms from the city so winery area that is rural but close to city for commuting.

7.4.2 Impact from unmet needs

Caring for a person with a life-limiting illness and the death of that person are stressful events and overwhelmingly emotional times for family caregivers (Aoun et al., 2018). Although palliative care professionals provide community support, family caregivers provide a significant portion of the practical, emotional and existential care in the home (Norinder et al., 2021). The literature, policies and standards, recognise the vital role that informal caregivers play as members of the multidisciplinary palliative care team (Bowen, 2014). However, it is unrealistic to expect that family caregivers are prepared to meet the physical care needs of the patient or have the knowledge and skills to deliver safe care. Most will not be emotionally equipped to manage the stress involved in a journey with their loved one which will ultimately end in an untimely death. The child of one person receiving care expressed:

We didn't know anything. We were overwhelmed. No doctor told us about palliative care. We were struggling. Mum asked me to phone community health at the hospital and they asked if I wanted palliative care. I didn't understand what palliative care meant. I cried when they asked if mum was dying and she was listening and said yes. It was awful and I will never forget.

Every patient, every family unit will have differing knowledge levels, experience, available personal and community supports, financial capability, history, needs and preferences. It is essential that family caregivers interact with healthcare professionals to seek the information and knowledge they need and ensure that the quality of life for the patient and the family unit is optimised (Norinder et al., 2021). Research has shown that family caregivers benefit from information about prognosis and the challenges of providing end-of-life care, however, are unlikely to initiate conversations with clinicians and unlikely to understand or retain complex medical terminology (Mangan et al., 2003). Reducing the burden of caregiving by meeting physical, financial and emotional care needs and providing the information and knowledge to allow preparation for an impending death, results in reduced depression, anxiety and complicated grief in family caregivers (Aoun et al., 2018; Miller & Porter, 2021).

As 2 spouses expressed:

It is too hard. If his family weren't staying here 24/7 I would have just given up by now...Not fair and not good enough.

Nobody came to help even when I asked and I did everything they told me to. They said 3 months for services. He died 4 weeks after I asked. I am really disappointed nobody helped me.

A preference for end-of-life care and a home death is common, either by a desire to follow patient wishes or feelings of obligation (Woodman et al., 2016). Regardless of the reason to remain at home, many people are transferred to acute settings because symptoms cannot be managed in the home, or there is a perceived lack of services to support family (Champion et al., 2015). The preference and support provided by family caregivers is a known environmental factor consistently associated with a home death (Callisto et al., 2021). However, it is unrealistic to expect family caregivers to effectively manage symptoms in the home, particularly administration of breakthrough medication at end-of-life, without information, education and support (Miller & Porter, 2021; Munday et al., 2007; WACHS, 2021). Transfer to emergency departments or acute settings for end-of-life care, can increase distress, suffering and caregiver burden, with feelings of unmet needs being associated with complicated and traumatic grief (Hall et al., 2012).

The literature has identified the highest unmet needs reported by family caregivers, namely psychological issues and patient care and support, and unmet needs are associated with anxiety and decreased quality of life (Driessen et al., 2023; Hart et al., 2022; Robinson et al., 2018).

The reality for family caregivers providing end-of-life care in the home, is that the experience of caregiving and bereavement will be closely intertwined. Death may bring a measure of relief to exhausted family caregivers, seen as an ending of their loved one's suffering and to the strain and burden of the caregiving role, and many will find their own coping strategies to manage the symptoms of depression and grief (Boerner & Schulz, 2009). However, there will be a minority who will suffer long-term, with stress, anxiety, depression and complicated grief (Hall et al., 2012).

Family caregivers experiencing complicated grief will require assistance from skilled clinicians to develop coping strategies and validate their feelings (Johns, 2015). However, many participants reported difficulties accessing services, including bereavement and grief support, with only 6, or 13.3%, of the peri-urban family givers responding positively to the statement 'support has been provided with advance care planning, completing will and grief

127

counselling'. One family caregiver, a mother in the 18–34 year age group of a young child reported:

We don't qualify for any support at home. No NDIS. We need a home care nurse and the social worker is at the hospital. Financially we are in tatters even with carers payment which is so little. I am trying to get grief counselling through Carer Gateway before he dies so I can cope with his death.

A very small number, (n = 2) or 4.4%, of the peri-urban family caregivers reported receiving support from a psychologist or psychiatrist for anticipatory grief or bereavement care, and 6, or 13.3%, received bereavement care. The literature demonstrates the high value that family caregivers place on the close relationship developed with the palliative care team prior to the patient's death and the established 'lifeline' with staff that was expected to continue (Näppä & Björkman-Randström, 2020). However, it becomes 'luck of the draw' with some peri-urban or rural palliative care teams if a skilled healthcare professional is available for provision of psychosocial care. If a social worker is not available, or lacks the skills, competency or personal interest in providing bereavement support, then a nurse will assume the role (Johns et al., 2019). A skilled clinician in the MDT, regardless of their role, must be able to support family members and identify potential complicated grief (Sealey et al., 2015). The number of responses about the lack of support following a death supports this idea:

Only 1 visit then a phone call after John died.

I don't know why mum couldn't get more help. I don't know why we weren't offered help when she died.

Palliative care did nothing. Nobody did anything to help us when mum died. The hospital referred mum to palliative care with advanced pancreatic cancer. She was going home for us to care for her at home until she died. Palliative care nurse came out before Christmas then never again. When she phoned we kept telling that mum wasn't doing well and she did nothing until she sent a dietician out because we said mum was losing weight and not eating. That was just perfect. next day we called an ambulance. She died next day. Didn't even get the hospital bed mum was promised. No help. No nothing. When she phoned days later I told her mum was dead and what I thought of her. Never heard another thing.

Palliative care should be person-centred care, and tailored to individual needs and preferences, with required information, services and support differing for every family caregiver and the person that care is provided for (Coulter & Oldham, 2016). The importance and necessity of any community palliative care support will differ, influenced by the presence or absence of personal support, knowledge, needs and preferences (Dionne-Odom et al., 2019).

Family caregivers were asked to respond about barriers to accessing services in their location. It is reasonable to assume that a response citing a barrier to access would be concerning a service or resource that the family caregiver perceived as important or necessary. The large number of participants that reported barriers to access in peri-urban locations highlights the number of family caregivers with the potential for long-term physical ill-health, anxiety, depression or complicated grief as а result of unmet needs.

7.5 Summary

This chapter summarised the findings from healthcare professionals, seeking to address the following key research questions:

- How does counter-urbanisation influence community palliative care provision in periurban Australia?
- What role does digital technology play in peri-urban community palliative care delivery?
- How does location influence accessibility to in-home palliative care services?

In addition, this chapter sought to respond to the research question:

• Do patients' and family members' expectations of peri-urban community in-home palliative care differ to the care that health professionals provide?

Participants reported the distress, emotional toll and eventual outcome on the patient and the family unit of barriers to accessing community palliative care in their location. A home location, the distance and time involved in travel, was identified as a significant barrier to equitable access for those living in close proximity to urban centres. The expectation that services would be available to meet their needs was evident, with many reporting difficulties accessing services when, as frequently, and for the length of time that family caregivers wanted and expected. Responses did not differentiate palliative care services between teams or

providers, with no recognition of specialist palliative care teams as separate entities in service provision.

Unlike health professionals, family caregivers did not view a lack of equipment or medication as significant barriers to quality care provision, nor was blame attributed to the healthcare system or organisations involved; rather, the overriding theme that emerged related to supporting the carer experience. Participants reported a lack of resources related to generalised services and support, with 'palliative care' used as a term to refer to overall inhome support to improve the quality of the palliative care experience. Emotions were generally voiced as feelings of disappointment at assumed accessibility to services not meeting expectations and subsequently influencing person-centred care, in particular, choice of location of death for the patient.

Whilst the literature identifies that barriers to access exist in all locations, the responses from peri-urban family givers in peri-urban locations, support the view provided by health professionals that peri-urban areas retain the inequities of access known to exist in rural locations. Barriers to all 5 dimensions of abilities on the user or demand side were also evident in family caregiver responses, albeit in varying degrees, depending on the participant (Levesque et al., 2013b). The dimensions to abilities for users were the ability to perceive, ability to seek, ability to reach, ability to pay, and ability to engage (Haj-Younes et al., 2022; Levesque et al., 2013b). The influence of barriers is significant considering that the dimensions are interrelated and inherently person-centred, subsequently influencing quality of life and wellbeing.

Distance involved in travelling to/from a home location influenced decision-making about treatment and place of care, including end-of-life and local providers lacked the resources and the capacity to support community palliative care. Rural GPs' and generalist community nurses' limited experience in end-of-life care and use of palliative care medications influenced commencement of community palliative care services and symptom management in the home.

The significance of commencing community palliative care services early in the journey and quality of life has been discussed in the literature, and the home location was identified as a contributing barrier to access. There was a general consensus that digital technology played a minor role in service delivery and that face-to-face support remained the preference. The extent of the role of digital technology differed to that voiced by healthcare professionals but the preference to retain the face-to-face component was comparable. The barriers of geographic location and distance and the resulting long-term impact from unmet needs on family caregivers are factors which need to be considered by community palliative care providers. The experience of caring for a loved one at home will linger long after the person has died, and healthcare providers have an ethical and moral right to ensure the wellbeing of the community they serve.

Chapter 8. Patients' perspective: barriers to access

8.1 Introduction

The term palliative care can conjure visions of imminent death, however death does not always come suddenly (Collins et al., 2020). For most people, it is the inevitable result of one or more diseases that are managed by multiple clinicians over weeks, months or years. Evidence suggests that access to early person-centred palliative care reduces physical and emotional distress and optimises quality of life for the patient, family members and friends (Aoun et al., 2020; Hui & Bruera, 2020; Seow et al., 2020). Quality of life is viewed as an important overall measure of wellbeing and happiness in relation to an individual's perceived value and social context in which they live (Phyo et al., 2020). The role of the family and caregivers is recognised as vital to the care and psychosocial needs of a palliative patient throughout the palliative journey, but the patient is the focus of care (O'Sullivan et al., 2021). Commencing community palliative care involves consultations focused on patient needs, and building a relationship with health professionals to encourage open and honest conversations about illness and implications, physical and psychosocial care and coordinating the care needs and resources per need and patient preference (Gomes et al., 2013). Interactions with family caregivers may be purely incidental, non-existent or fully integrated in all face-to-face and digital health interactions.

As the journey continues and death approaches, the focus of palliative support shifts and intensifies to relieve the patient's physical, spiritual and psychosocial suffering by controlling pain and other symptoms and avoiding prolonging the dying process (Mitchell, 2021; Rome et al., 2011). Elements of a good death are individual, unique and different, although dignity and autonomy are common elements (Krikorian et al., 2020; Semino et al., 2014). Elements are shaped by the patient's culture, life experience, environment, age, financial situation and disease progression, with clinicians and family members all exerting an influence with their own notions of a good death (Meier et al., 2016). The importance, is therefore, to ensure provision of person-centred care which recognises and supports the views of individual patients and their preferences and needs. Although integrated with family caregiver requirements, the patient's preferences remain integral to optimising wellbeing and quality of life. Therefore, this chapter aims to describe the perspective of patients receiving community palliative care.

8.2 Analysis of survey data

Analysis of data for the third group of participants, identified as patients, followed the exploratory and descriptive approach used in chapters 6 and 7. The relationship between

themes identified by the patients' experiences and that voiced in chapter 7 by family caregivers is discussed, to identify similarities or differences.

The following section outlines patient demographics and provides the results of quantitative and qualitative data analyses, allowing challenges perceived by the patients themselves, to accessing services, to emerge.

8.3 **Profile of patient participants**

People diagnosed with a life-limiting illness and receiving community palliative care services were invited to participate in the survey, Table 8 provides the demographics for the participants of this study. Nil participants who commenced the survey voluntarily withdrew, with 30 submitting a completed survey. Of these, 14, or 46.7%, identified the home location as within a peri-urban area, 13, or 43.3% in other non-urban (rural or regional) and 3, or 10.0%, in metropolitan centres. Data with potential to identify participants, such as gender, location by postcode, specific age, diagnosis or disease stage, were not requested to aid anonymity. The median survey completion time for the group identifying as patients was 658.0 seconds or 10 minutes 58 seconds. Table 8 identifies the demographic details of the participants who identified as palliative care clients (patients).

Characteristic	Total patient participants (n = 30) n %		Per (n = [,] n	i-Urban 14) (46.7) %	Other non- urban (Rural & Regional) (n = 13) (43.3) n %		Metropolitan (n = 3) (10.0) n %	
Age								
18–34 years	3	10.0	1	7.11	1	7.7	1	33.3
35–49 years	4	13.3	3	21.40	1	7.7	0	0
50–64 years	11	36.7	5	35.80	4	30.8	2	66.7
65+ years	9	30.0	3	21.40	6	46.1	0	0
Prefer not to answer	3	10.0	2	14.30	1	7.7	0	0
Time receiving palliative care								
Less than 1 month	5	16.7	4	28.60	0	0	1	33.3
1–6 months	14	46.7	5	35.70	8	61.5	1	33.3
6 months to 1 year	2	6.7	1	7.20	1	7.7	0	0
>1 year	7	23.3	3	21.40	3	23.1	1	33.4
Unsure	2	6.6	1	7.10	1	7.7	0	0
Location by State/Territory								
Queensland	4	13.3	2	14.30	2	15.4	0	0
South Australia	20	66.7	9	64.30	8	61.5	3	100.0
New South Wales	2	6.7	1	7.10	1	7.7	0	0
Victoria	3	10.0	2	14.30	1	7.7	0	0
Tasmania	0	3.3	2	14.30	1	7.7	0	0

Table 8 Patient demographics

8.4 Findings

Participants who identified as patients were asked to respond to 21 voluntary questions following provision of informed consent. The data analysis process resulted in identifying 2 main themes that characterised the perspective of palliative care patients. These were 1) the appreciation of the support of palliative care professionals in maintaining a sense of normality, managing symptoms and providing information – being the 'ear' when needing to talk, and 2) the disappointment when services were not accessible or not able to meet individual needs, resulting in relocation. Sub-themes emerged which related to the home location – namely, the influence of time/distance on decision-making regarding treatment and place of care. Ultimately, however, the existence of barriers to accessing the services and support expected by people diagnosed with a life-limiting illness was perceived by patients as not ideal but not with the level of emotion that was expressed by family caregivers.

8.4.1 Patient perspective

A variety of complex emotional responses occur when a person receives bad news, including a diagnosis of an incurable disease, being informed of disease progression, or being told that their life is ending prematurely (Derry et al., 2019). Emotional reactions include anxiety, sadness, shock, denial and anger and people will develop strategies to manage their emotions in an attempt to achieve a sense of normality in their lives (Fringer et al., 2018). People receiving palliative care experience multiple changes or transitions during their illness trajectory. Change can be confronting and often occurs suddenly and unexpectedly, resulting in new situations, accompanied by distress and anxiety (Derry et al., 2019). Common misconceptions, knowledge gaps and fear of engaging in taboo conversations may deter people from accessing palliative care until late in the illness journey (McIlfatrick et al., 2021)

In an endeavour to preserve a balance in their lives, people focus on making the most of limited time, which often leads to defining their priorities and preferences (Groot et al., 2007). Early and open discussions about end-of-life issues with patients allows time for questions, planning and reflection, and can alleviate anxiety and fear (Cheluvappa & Selvendran, 2022). Person-centred palliative care then becomes very important to people receiving care because it promotes individual goals, including continuing to live a meaningful life with dignity and respect and as autonomously as possible (Nysæter et al., 2022). For many, a meaningful life means remaining in their own home, amongst memories and loved ones (Nysæter et al., 2022). Research has shown that the vast majority of people voice a preference to remain in their own home for end-of-life care and a home death (Swerissen & Duckett, 2015). Implementing

134

palliative care early in the journey is more likely to result in provision of person-centred care, enabling choices in care and treatment consistent with patient preferences and choice for a home death (Howie & Peppercorn, 2013).

Supporting this view, 9, or 64.3%, of the peri-urban patients reported 'my quality of life has been improved by remaining in my own home'. The same number agreed with the statement 'I have been supported in my choice to remain in my own home'. This view was reiterated by the majority of all participants, regardless of location.

In response to the question, 'Overall, are you satisfied with the access to palliative care in your location', patients expressed a higher level of satisfaction than any other group of participants; 14, or 60.9% of all patients, regardless of location were satisfied with access to services in their location. Additional descriptive responses provided to support survey responses demonstrated the underlying level of appreciation and satisfaction with palliative care service delivery. Words such as 'happy', 'wonderful', 'perfect' and 'grateful' were used, particularly in relation to palliative care support for a home death. As one patient in the 35–49-year age group, receiving care for 1–6 months in a peri-urban location, expressed:

My nurse comes out daily my aid comes out 3 times a week and will come more often if I ask her to. Social worker has made sure everything is ready for my daughter, a memory box, my funeral is arranged and paid for and my house will be signed over to my daughter next month when she turns 18. I have a massage therapist and my doctor visits. I love my comfort box he provides. I'm gonna die happy and peaceful at home. Palliative care is not a death sentence but a new beautiful beginning.

Another patient in the 50–64-year age group expressed the same sentiment:

Aware that I am not eligible for aged services as I am under 65 however, I only need social worker support which palliative care are providing and she is wonderful. I know the nurses are there if I need them but I am hoping to remain well for as long as possible. I have moved closer to Adelaide to make travel to LMH easier.

A patient aged over 65 expressed their appreciation for the support received:

Palliative care have been wonderful for over 18 mths. Have stayed in the background for support while I have remained independent including driving myself everywhere. Recently needed help with pain when treatment ceased and

palliative care arranged for a specialist to see me. Couldn't be happier with their support.

Although most patients were satisfied with access to palliative care services, some recognised that relocation would be required when their disease progressed and care needs increased. An over 65 years patient, receiving care for 1–6 months said:

Very happy with support from palliative care. My husband and I talk with the nurse regularly and are able to vent our frustrations and just talk openly which helps us greatly. I am managing well at home and have a good quality of life with support from family. I know that this will change in the future and I will need more help and will probably have to move into a nursing home or hospital as my husband will not be able to care for me at home, but for now am very grateful that we have someone to contact when we need anything.

Some negative emotions emerged in comments, within the group of peri-urban patients who were not satisfied with access to palliative care services. In general, comments were about their dissatisfaction with needing to relocate due to a lack of expected service access. A patient in the 18–34 age group provided the following succinct response:

I have to move to get the help I should be able to get locally....is supposed to be modern and with all facilities.

A peri-urban patient under 65 years of age and ineligible for Commonwealth Home Support packages expressed their view:

I have tonsil cancer but not sick enough for palliative care. Live alone and need help but can't get any. Can't afford petrol to drive anywhere and not well enough to drive. Have to sell my house and move but house at...not worth a lot so can't buy anything and can't afford to rent.

The importance of person-centred care to enable a meaningful life for the time remaining for people receiving palliative care highlights the importance of removing barriers to accessing services in a timely manner. The expectations of palliative patients were reflected in survey responses.

Similar to the family member and health professional responses, 86.6% of patients indicated that they had an expectation that healthcare services were accessible, regardless of location, with a low variance (m=4.5, sd=0.8, v=.6, ci=0.1) (refer to Table 9). The level of

expectation of accessibility to health services was so high overall, that (n = 28) or 93.3% of all responses, regardless of location, had expectations of access to services. Of the remaining number of responses, (n = 2) or 6.7%, one in a peri-urban and the other in a non-urban (rural), location, responded with 'somewhat disagree'. These responses were comparable with responses received from family caregivers, with 100% of participants having an expectation of access to health services.

Despite their expectations, however, only 7, or 50.0%, found that services met their needs and expectations. Of the remaining responses to needs and expectations being met, 16, or 35.5%, of the participants said their needs were partially met, and 18, or 40.0%, said they were not met. It is worth noting that palliative patients were generally happier with services received than family caregivers, with only 24.4% agreeing that needs and expectations had been met. Studies have shown that caregiver expectations associated with stress and caregiver burden can influence perceptions of the caregiving experience, which may be reflected in the differing responses between the 2 groups (Burns et al., 2015). Comments from peri-urban patients who said that services did not meet their needs and expectations revealed their frustration. Two 50 –64 year olds said the following:

Palliative care accepted me but then did nothing as they said I wasn't about to die from cancer so they left me with nothing even community nursing refused to come out to my house cos too far away.

Need gardening, housework, help with transport to appointments and treatment, respite in my home so my husband can leave the house, meals and shopping and help with showers sometimes but cannot access as NDIS refuses and too young for My Aged Care.

Only 5, or 35.7%, of the peri-urban patients reported difficulty accessing services when required, with the same number also finding it difficult to access identified services when required. Although small in number, one very emotional response was received from a peri-urban patient:

No social worker my body and mind isn't coping I'm drowning internally and externally don't know how I will get thru I can't afford MRI. It cost \$625 – they offered me a payment plan.

The importance of understanding patient perceptions of unmet needs is evident in the close relationship between patient and family caregiver responses to barriers to access; 85.2% of patients agreed that they have a good relationship with the professionals providing their care, highlighted by the low variance (m=4.3, sd=1) (refer Appendix I: Table 74).

What is perceived as an unmet need by palliative care patients may differ to family caregivers, due to personal preferences, culture and life experience. The effect of unmet needs includes increased or unrelieved pain and physical symptoms such as nausea, fatigue and dyspnoea, and emotional distress (Wang et al., 2018). The family caregiver role is to support the needs of the patient, therefore any perception of unmet patient care needs will increase caregiver distress and burden, which then influences the distress of the patient, impacting quality of life and negatively influencing the physical health of both patient and caregiver (Wang et al., 2018). Person-centred care involves the needs and preferences of the family unit as a whole, therefore understanding the needs and preferences of everyone involved and removing barriers to access will increase the wellbeing of the family unit, central to the aim of palliative care services (WHO, 2020a).

Only 6, or 42.9%, of the peri-urban patients reported 'difficulty finding information about service/s and identifying what services were needed.' However, no peri-urban patients agreed to being involved with seeking services or information, with the majority, (n = 11) or 78.6%, agreeing that they were referred to palliative care, with 64.3% of referrals coming from the health professional providing treatment. As discussed in chapter 7, the role of family caregivers is to support loved ones with physical, spiritual, psychosocial and financial needs and this cannot be understated (Norinder et al., 2021). Family caregivers require support to continue in their caregiving role and often make the decisions about services and supports for in-home palliative care in order to improve quality of life and enable the option of a home death (Steele & Davies, 2015).

Patients may not have a role in initiating access. This is often left to family caregivers and primary healthcare clinicians and GPs. Almost half, 42.2%, of family caregivers in periurban locations complained that they were unable to access the service they wished to use, at a time or day of the week they wanted, and 40.0% were unable to access the service as frequently as they wanted. It is unknown if this level of dissatisfaction was the view of the family caregiver or a report on behalf of the patient.

Studies have demonstrated that palliative care patients themselves often create barriers to in-home services due to personal preferences, reluctance to accept strangers in their home, an expectation of family support, misconceptions and the stigma associated with palliative care, and self-perceived views of their health and functional condition (Cardenas et al., 2022; Collins et al., 2020; Flieger et al., 2020). Denial or an unrealistic self-perception of their needs and abilities was evident in responses to the question, 'If you are not using all available community services, why not?' Half of all palliative care patients reported 'I don't need the service at this time'; 7, or 50.0%, of these respondents were peri-urban patients. As one patient, who did not provide any personal details responded:

Not old enough for home help but can't shower myself or go down steps to get to garden. Nurse said I have to pay for ramps myself but too expensive. Palliative nurse organised someone to come out and shower me but I don't always want a shower when they get here so I tell them to go away. OT that came out was rude. Husband and I fight all the time. He cooks me food but that's all. I have a car that takes me...for chemo and it costs me \$50 each time. Nurse keeps asking me about moving to...[urban centre]...with my son but I don't want to. His house is small and my grandchildren I love but they are noisy. I am too young for nursing home and don't want to go there. Will die here on the farm. Nurse phones me sometimes but doesn't visit. Doctor phones me and he doesn't visit. Friends don't visit.

Peri-urban patients were asked 'if location was chosen as health services needed were close' and only 5, or 35.7%, agreed. Even less, (n = 4) or 28.6%, agreed that consideration was given to services in the planning or development stage. High variance across responses highlights that, although people denied selecting their home location based on healthcare access (m=2.3, sd=1.4, v=1.9, ci=.25), the majority disagreed with this statement, which links to the assumption that healthcare will be easy to access, regardless of location (refer Table 9).

Table 9 Descriptive Statistics Survey Q5.6 How strongly do you feel about the following statements?

Statement	n	min	max	mean	SD	Variance	Confidence Interval
5.6(1) (Patient) I chose this location as							
health services I need are close by	30	1	5	2.3	1.4	1.9	0.25
5.6(2) (Patient) I chose this location as							
services I need/may need, are in the							
planning or development stage	30	1	5	1.9	1.3	1.7	0.24
5.6(3) (Patient) I chose this location as							
health services that provide for increased							
care needs are close by (eg hospice, aged							
care facility, specialist care)	30	1	4	1.7	1.1	1.3	0.20
5.6(4) (Patient) There was an expectation							
that health services were accessible	30	2	5	4.5	0.8	0.6	0.10

However, a theme relating to home location posing challenges to access emerged when answers relating to barriers to access were expressed. Although only 3, or 21.4%, of the patients perceived difficulties in accessing services in their peri-urban location, 8, or 57.1%, admitted that they 'have had one or more hospital stays extended as services were unavailable to support discharge home'.

A reasonable number, (n = 4) or 28.6%, of the peri-urban patients believed that 'home location and the distance involved for service providers' was a barrier to access. The majority, (n = 13) or 92.9%, were more concerned with the barrier from 'home location and distance involved in travel to healthcare services not available locally'. However, the response from non-urban or rural, regional and remote patients was as expected because geographic locations are a known barrier to accessing quality and person-centred palliative care for Australians living in these locations (Schulte et al., 2022). Home location was not cited as a barrier for service providers by any metropolitan participants.

Only one, or 7.1%, of peri-urban patients travelled less than 50 kms for healthcare services, with the majority, (n = 13) or 92.9%, of the participants, travelling 50–150km. No patients reported travelling over 150 km.

Geographic distance increases the travel time and associated difficulties for all patients already in poor health and facing pain and discomfort when travelling (Lalani, 2022). Responses supported this notion, with 6, or 42.8%, of peri-urban patients agreeing that 'time/distance influenced home location of the person requiring care' and 'the distance required to travel to some health services has influenced where I live'. Eight participants, or 57.1%, believed that 'relocation closer to services that provide a higher level of care will be needed'. The need to relocate due to the peri-urban home location and barriers to access was expressed by patients aged under 65 years:

I can't access any other support. I have tried cancer council, Ferros care, NDIS and My Aged Care. Cancer Council helped with transport and a lawyer. When I need more help I go to hospital.

And another:

Aware that I am not eligible for aged services as I am under 65 however, I only need social worker support which palliative care are providing and she is wonderful. I know the nurses are there if I need them but I am hoping to remain well for as long as possible. I have moved closer to Adelaide to make travel to LMH easier.

140

The time and distance involved in travel for health purposes has the potential to influence patient decision-making, particularly if they believe that their care needs are becoming a burden on family caregivers (Munday et al., 2007). Research has demonstrated the significant role of rurality as a predictor of low rates of outpatient chemotherapy and radiotherapy treatment (Cerni et al., 2023). Six, or 42.8%, of the peri-urban patients agreed that 'distance influenced decisions' which was similar to the non-urban responses: (n = 6) or 46.1%. For peri-urban patients, decision-making regarding travel was further influenced by a lack of public transport options reported by the majority, (n = 12) or 85.7%. Travel was also deemed complicated, which also affected decision-making, with the same number of participants saying that it was difficult for family caregivers to manage the cost of petrol and parking, lost wages and time with private transport options. An expectation that a community car would be available in peri-urban locations in proximity to urban centres was not met, with 9, or 64.3%, of the participants reporting a lack of availability.

The responses highlighted a lack of person-centred care, demonstrated by a lack of available resources to meet the individual identified needs and choices of palliative care consumers in peri-urban locations. Six, or 13.3%, of peri-urban family caregivers believed that 'care needs were beyond the scope of existing services,' which corresponds with the small number, (n = 7) or 15.5%, who stated that 'physical symptoms were managed in the home'.

In stark contrast to the negative opinions of family caregivers, palliative patients were supportive of the role of digital technology in community palliative care. Ten, or 71.4%, of the patients believed that 'digital health enabled access to medical support that would not be available otherwise due to distance' and 8, or 57.1%, said that 'they were comfortable to remain at home knowing that I can access support at any time.' Of all patients, 80.7% indicated that digital health provided access to supports that they would otherwise not have access to due to distance, highlighted by the low variance (m=3.9, sd=.8, v=.6, ci=.2) (refer to Table 10).

Table 10 Descriptive statistics survey Q5.20 How much do you agree with the following statements?

						Variance	Confidence
Statement	n	min	max	mean	SD		Interval
5.20(1) (Patient) Digital health is an							
important tool in ensuring I have							
regular communication with my							
palliative care team	27	1	5	3.6	0.9	0.8	0.2
5.20(2) (Patient) I have the resources							
and confidence to use digital health							
effectively	27	2	5	3.9	1.1	1.2	0.2
5.20(3) (Patient) Digital health							
consultations are an acceptable	26	2	5	3.5	0.8		

alternative to face-to-face consultations with medical professionals						0.7	0.2
5.20(4) (Patient) Digital health means that I can access medical support that I would not be able to access due to distance	26	2	5	3.9	0.8	0.6	0.2
5.20(5) (Patient) I am comfortable to remain at home knowing that I can access support at any time	26	1	5	3.6	1.1	1.2	0.2

Most peri-urban patients, (n = 10) or 71.4%, reported receiving a face-to-face method of community palliative care service delivery. This same number also reported receiving phone consultations, although other digital health technologies were not preferred. Despite the progress of the digital transformation in healthcare, only 6, or 42.9%, of peri-urban patients believed that 'digital health is an important tool in ensuring I have regular communication'. It is unclear how much decreased function and increased need influences the method of service delivery to community palliative care patients who require face-to-face consultations to support end-of-life and terminal care. It is also worth considering that these views support the literature that posits that digital health is preferable complementary to current service methods (Collier et al., 2016).

8.5 Summary

This chapter has summarised the findings from the perception of a small number of community palliative care patients – all patients on a palliative journey with an inevitable ending. The goal was to address the overall aim of this research by focusing on the following key research questions:

- How does counter-urbanisation influence community palliative care provision in periurban Australia?
- What role does digital technology play in peri-urban community palliative care delivery?
- How does location influence accessibility to in-home palliative care services?

In addition, this chapter sought to address the following research question:

• Do patients' and family members' expectations of peri-urban community in-home palliative care differ to the care that health professionals provide?

Participants reported their disappointment at unmet needs and the need to relocate for additional services, to reduce travel time to treatment and appointments and for end-of-life

care. The factors associated with travelling over 50 kms for healthcare were identified as a barrier to equitable access to services for those living in a community viewed as having appropriate healthcare access due to proximity to services. There was an expectation by patients that services would be available, to the point that many did not report factoring in a need to source healthcare access prior to their changing needs. Consideration was not given to available healthcare services in the local community when relocating or choosing to remain in their local area.

Patients reported satisfaction with the method of service delivery received and believed that digital health was an important tool in service delivery. These results are difficult to use for comparison purposes between groups, as that stage of the palliative journey may have influenced patient perspectives. However, comparable to family caregivers, there were no specific resources identified as barriers to access, merely a generalised view that access to palliative care services in their home were not available as expected or needed. No blame was attributed to the healthcare system or organisational characteristics involved in care provision. Emotions inferred a level of disappointment, rather than anger or frustration at an inability to source services as desired. Disease progression, increased symptoms including pain and fatigue, functional and cognitive decline could play a part in this virtual acceptance of the situation, as could the desire to not burden or upset family caregivers. Whilst the literature has identified that barriers to access exist in all locations, the responses from peri-urban patients does add support to the views provided by family caregivers and health professionals, that peri-urban areas retain the same inequities of access as rural locations.

Similar to family caregiver responses, patients reported varying levels of barriers to access in all 5 dimensions of access and abilities, from the user perspective (Levesque et al., 2013b). The dimensions are interrelated, continuous, and fluid throughout the palliative journey (Haj-Younes et al., 2022; Levesque et al., 2013b). Any barriers to access, at any point in the journey, will have an immediate effect on patient wellbeing, which ultimately influences the physical and emotional wellbeing of health professionals involved and the family caregivers.

Chapter 9: Conclusion

9.1 Introduction

Chapter 9 begins with a summary of the research methodology, which employed an exploratory, inductive, and interpretative approach, with the aim of responding to the research questions. The significance of the findings will be discussed, including the implications for public healthcare providers. This chapter will also propose recommendations for evaluation by providers in the form of a report, outline key implications of the research and suggest areas for potential future research.

The research sought to understand if home location in peri-urban areas in Australia influences access to community palliative care service delivery. The inequities of access related to geographical location in rural and remote Australia are well known. What is less understood, is the state of access in the rapidly growing outer urban and regional fringes (Rainsford, 2019). Often termed peri-urban areas, these locations are a popular choice for relocation, however, infrastructure and service growth in peri-urban locations has consistently lagged behind housing construction (McFarland, 2015; Newton et al., 2017).

This study was designed to describe, through an ethnological framework, the perceptions and experiences of barriers to access of those providing or receiving, palliative care in the community setting. An online survey was used for data collection, with the majority of participants citing peri-urban areas within South Australia as the location for provision or receipt of care. Qualitative data was collected using free text and open-ended questions to allow participants to explain themselves and reveal issues that might otherwise go uncaptured (Rich et al., 2013; Riiskjær et al., 2012). Providing the option of free-text feedback, as opposed to just closed questions, also helped overcome the limitations to potential data (Bowyer et al., 2019). Qualitative data was organised into themes by categories to allow connections, similarities and comparisons to emerge (Gioia et al., 2013). Quantitative data in the form of Likert and multiple-choice questions was collected to support the qualitative data. The intent was to merge, compare and contrast results from data analyses to add a greater depth of understanding to participant meaning and experiences reported in emerging themes from the qualitative analysis (Creswell & Creswell, 2018).

The data analysis identified themes that characterised the barriers to access experienced by people working to provide community palliative care. The prominent barrier was the role of the home location in accessing resources, including the staff, equipment and medication required to support patients and families. Many healthcare professionals also cited

the physical and emotional risks involved in providing care in rural and isolated areas within peri-urban locations, with many stating that they were unable to provide appropriate care to patients and family caregivers. Healthcare professionals also noted that the difficulties accessing required resources and a perceived lack of support from their organisations, with many voicing feelings of stress, exhaustion and frustration.

Emerging themes from family caregiver and patient responses mirrored those from providers, with the home location cited as the underlying factor in all barriers to accessing the services needed or expected, with distance required to travel to services influencing decisions related to care, treatment and relocation. Many spoke about their negative experiences of palliative care resulting from barriers to accessing the services perceived as essential to the caring experience, citing the emotional toll on the family unit.

The key research findings in relation to the research aims and questions, have been summarised in a report, a copy of which is included in this chapter. The report has been directed to Dr Robyn Lawrence, the Chief Executive of the Department for Health and Wellbeing, responsible to the Minister for Health and Wellbeing for the governance, leadership and management of the South Australian health system. The research seeks to create impact in the public health sector, guided by the WHO (1987) public health approach to palliative care, and engage communities in palliative care. The concept of a public health approach to palliative care aims to effectively integrate palliative care into society through appropriate choices, medication availability, education, and implementation of palliative care services at all levels (Kellehear, 1999; Stjernswärd, 2007). These initiatives have the potential to lead to public policies to support dying, death, loss and grief, and reduce vulnerable populations' access barriers to community and hospital-based palliative services.

9.2 Report to Chief Executive Dr Robyn Lawrence

This report summarises the key findings of the research.



1 April 2024

Dr Robyn Lawrence Chief Executive, Department for Health and Wellbeing, PO Box 287, Rundle Mall Adelaide SA 5000

Dear Dr Lawrence,

RE: Research titled: 'Access to Australian community palliative care services: The influence of peri-urban location'

This research was undertaken to fulfil requirements for the degree of Doctor of Philosophy at James Cook University. The aim of this research was to explore the challenges of community palliative care provision in peri-urban locations. The researcher welcomes the opportunity to engage with the Department of Health and Wellbeing to provide recommendations to guide future planning and policies in service delivery.

Background

Current research largely focuses on the barriers to accessing healthcare for vulnerable populations and to the geographical location barriers in rural and remote areas of Australia. Few studies have focused on access barriers in outer fringe or peri-urban locations. Peri-urban areas in South Australia are those in close proximity to Adelaide – including Mt Barker, Victor Harbor and Gawler – that are subject to urban sprawl into designated country areas. These

areas have a rural model of service delivery, with services jointly planned across both metropolitan and country.

The researcher's lived experience as a specialist palliative care clinical nurse in periurban locations guided the initial research. The problem space of existing barriers to access was identified and conceptualised through qualitative analysis of observation and informal discussions with palliative care clinicians who work in South Australian peri-urban locations. To confirm the research gap, an extensive literature review was undertaken to understand the methods of classification and the definition of peri-urban locations to understand the barriers to accessing healthcare services. An online survey was then developed to collect data from clinicians, patients and family caregivers. Data analysis resulted in the development of recommendations for future change to service delivery to enable equitable access to services.

Literature review

A review of the literature provided evidence of the benefits of palliative care provision for people diagnosed with a life-limiting illness and their family caregivers, particularly early in the palliative journey. Person-centred palliative care aims to optimise quality of life by prioritising the needs and values of individuals in every interaction with healthcare services. Peri-urban areas, offer a rural lifestyle in proximity to urban centres and have witnessed a surge in popularity and unprecedented population growth, impacting the provision of healthcare services.

What became clear in the literature is the importance of developing an approach in defining peri-urban areas considering the unique characteristics of that area. The literature highlighted the ambiguity in terminology used to define peri-urban areas, with no clear definition currently being used. Current geographical classification methods to determine locations as urban or non-urban generally use indicators of population density and/or indexes of accessibility based on distance in a straight line. Evidence suggests that the current methods of location classification used to allocate resources and methods of service delivery do not reliably indicate the needs and expectations of the increasing number of the population living in peri-urban locations.

Survey results

A survey was developed to collect open and honest perspectives of experiences related to community palliative care access, from both providers and users. Data collected was not linked to participants and the researcher provided confidentiality by removing any identifiable data in reporting. The survey consisted of a combination of closed-ended questions, including Likert and multiple-choice, open-ended comments and free text boxes.

Quantitative data was sought to support the descriptive data. Participants were divided into three groups within the one survey, following participant identification as either a health professional providing community palliative care in any role, and a person receiving care as either a family caregiver or a person diagnosed with a life-limiting illness. The survey link was distributed by Palliative Care Australia in February 2023 via an e-newsletter, with snowball sampling extending distribution via social media – a method viewed as promoting less risk to participants due to the emotional research subject.

A significant number responded to the survey which closed in April 2023 – 182 clinicians, family caregivers and patients. The majority of completed surveys came from family caregivers (91 or 53%), followed by community palliative healthcare professionals (50 or 29%) and patients (30 or 18%).

The majority (33, or 73%) of participants identifying as health professionals were providing care in peri-urban areas of South Australia. Many voiced concerns about the difficulties accessing equipment, trained staff and medication due to the time and distance associated with location. Travel was viewed as a challenge, with staff voicing an inability to meet daily workload demands without compromising person-centred care or the flexibility to attend to the individual needs of dying patients, thereby compromising place of death. The majority wanted to provide 24/7 contact for support and in-home care options, with a method of delivery that promoted relational continuity of care. A reliance on a digital model of service delivery was a contentious issue; many participants felt strongly about the importance of providing a face-to-face service, due to the emotional environment of palliative care.

Of most concern, was the overwhelming level of emotional distress expressed by clinicians who believed they were unable to provide appropriate care. Words were used such as 'frustrated', 'sad', 'overworked', 'stressed' and 'exhausted'. Clinicians perceived that barriers to access derived from a lack of support from leadership or the organisation for which they worked, which they believed demonstrated a lack of recognition of staff as an asset to be developed, supported and retained. Organisations that equip nursing and other healthcare staff with the necessary resources, including equipment and appropriate staff levels, increase staff satisfaction and decrease the likelihood of adverse medical events, poor patient care and unmet patient needs.

Of the group who identified as family caregivers, 45, or 49%, received care in periurban locations, with 30 living in South Australia. Data analysis highlighted a large proportion of responses expressing negative emotions including feelings of dissatisfaction, disappointment and anger, due to perceptions of poor, limited or lack of access to services that caregivers believed were required in the home. Blame was not directly attributed to specific

148

providers; instead, participants provided a generalised view of palliative care services in the health system.

Barriers to access resulting in unmet needs was a common theme, namely removal of choice which influenced decisions regarding treatment, care and place of care. Many reported the need to relocate, particularly at end-of-life, as a direct result of barriers to care in peri-urban locations. The highest unmet needs reported by caregivers were psychological issues, direct patient care and support and these were associated with anxiety, decreased quality of life and complicated grief. Continuity of care with trusted and skilled clinicians is required to develop coping strategies and validate caregiver responses following the death of a loved one; however, many reported difficulties accessing services, including bereavement and grief support in their location.

A small number of patients responded to the survey, with 9, or 30%, receiving care in South Australian peri-urban locations. Their responses were similar to family caregivers', although responses demonstrated that patients were generally more appreciative of the care received. Coping strategies used by those dealing with a life-limiting illness may have influenced their responses however, because this group tends to focus on wellbeing rather than fixing issues (Garg et al., 2018). Distance and time involved in travelling to receive care, attend appointments and receive treatment was voiced by many as likely to influence future decisions regarding treatment, care, place of death and the need to relocate despite their wishes to remain at home.

Key findings

This research suggests that current classifications of location used to determine palliative care service policy, methods of delivery and allocation of resources, fail to consider the heterogeneity and complexities of peri-urban locations. A lack of clear definition of periurban areas became evident early in the research. Locations currently viewed as peri-urban are classified as urban or non-urban (rural) depending on the classification used but are considered to have an advantage of over purely rural areas of being relatively close to metropolitan services.

The data analysis refuted this view of proximity providing access and offering an option of sustainable urban growth. Many participants receiving care cited distance, time, financial outlay and discomfort involved in the need to travel for complex symptom management. They also noted a scarcity of local resources and infrastructure which was a significant barrier to access, choice and quality of life. The disappointment at the need to relocate for care, treatment or place of death due to unmet needs and expectations was an emotion reiterated by many participants.

Palliative care clinicians expressed an understanding of the challenges and risks that exist in all community care settings, due to complex care needs and family situations. The potential exists for any home to include domestic violence situations, weapons, domestic and working animals, people with mental health conditions, drug and alcohol addictions and generally unsafe working and living conditions. However, the fact that potential risks are exacerbated in rural and isolated locations within peri-urban areas was clearly voiced. The likelihood of a delayed response by emergency or support services due to location and poor mobile coverage, difficulty accessing resources including staff, medication and equipment was reported by many who cited a lack of confidence, frustration and disappointment in a perceived lack of support from their organisation.

Analysis of responses demonstrated the interrelated and overlapping relationships between peri-urban locations, system-led and organisational policies, procedures, guidelines, funding, leadership, models of care and culture, and the needs and expectations of communities. Overall, the access barriers known to exist in rural areas were reported in South Australian peri-urban locations. The voices of clinicians, family caregivers and patients all focused attention on the need for the health system and organisations providing palliative care, to understand and explore access barriers in peri-urban locations.

Recommendations

The research findings have led to a number of key recommendations concerning policies, practices, guidelines, and effective approaches which need to be evaluated by SA Health to improve access to person-centred palliative care in peri-urban areas. Key recommendations are as follows:

- R.1 Build flexibility into allocation of resources and methods of service delivery. Explore the method of location classification used to allocate resources and methods of service delivery to meet the needs of the individual communities within peri-urban locations to reduce access barriers and promote person-centred care. A rural model of service delivery and a view that Adelaide is within proximity for access does not optimise quality of life for all.
- **R.2 Provide access to equipment.** Access to the equipment and appropriate home modifications are vital to helping people to remain at home for as long as possible, and to optimise quality of life. Access includes the equipment itself, and the resources for

providers to appropriately assess, modify, deliver and set up equipment within an appropriate timeframe.

- **R.3 Provide 24/7 after-hours telephone support for ongoing education and information.** Develop an after-hours support service for all community palliative care clients that meets community needs and enhances existing relational continuity of care to optimise care and quality of life. Continuity of care helps family caregivers address any concerns or problems, reduces the likelihood of ambulance attendance and transfers to hospital, and supports choice of location for care and death.
- R.4 Support access to essential medication for end-of-life care. Home-based palliative care at end-of-life requires optimal pharmacological management of symptoms related to the disease process. Detailed protocols and procedures tailored to individual services need to be developed for effective medication management to support end-of-life care in the community in locations where barriers to medication access exist. This includes options for timely delivery of medication to the home for clients unable to collect, clinician access to imprest or emergency medication stock, in addition to 24/7 pharmacist access.

This recommendation excludes any reference to medication for patients who have commenced the voluntary assisted dying process.

- **R.5** Increase funding to support choice of location for end-of-life care. Limited options for place of death are provided for people who prefer to remain within their peri-urban community. Support for home deaths is dependent on the individual situation and is influenced by complex needs and available family supports in the home location, limiting the option for home environment for many. Local hospitals provide private rooms and trained staff as available but is often not a preferred option. There are limited options available within residential aged care facilities and individuals who choose a hospice or who have complex needs are transferred to Adelaide, which can provide challenges for family members at a difficult time.
- **R.6** Target strategies to improve physical access to specialist services in peri-urban areas. The time, distance, financial implications, and physical discomfort related to travelling to specialist services in metropolitan hospitals continues to provide a barrier to access for peri-urban residents and influences home location and decisions about care and treatment. Participants confirmed the usefulness of digital models of service delivery, but they prefer face-to-face visits due to the emotional context of palliative care and poor and unreliable internet and mobile access.

- **R.7** Increase workforce capacity. Increased resources for direct and indirect care workers and clinicians would help people to remain in their own home for longer, enhance continuity of care, provide choice of provider and person-centred care, support face-to-face visits, optimise quality of life for patients and family caregivers, and support choice of place of death.
- **R.8** Support access to bereavement and grief support. Difficulties accessing counselling and grief support in non-urban areas, both anticipatory and following the death of a loved one, was a common complaint. The potential long-term detrimental physical, emotional and financial outcomes for those experiencing complicated grief should be a concern for healthcare providers.
- **R.9 Promote a supportive working environment.** Almost all the clinicians cited an inability to voice concerns or provide feedback in a supportive environment with a lack of organisational and leadership support leading to reports of stress, exhaustion, and burnout. The emotional and physical challenges generated by a lack of resources and the distance required to travel, combined with an emotionally challenging environment, cannot be overlooked when considering the prevalence of risks and the need to retain trained, qualified and highly skilled palliative clinicians to optimise quality of life and wellbeing for communities.

The social impacts on health and dying from barriers to accessing quality, personcentred care are substantial and significant. The risk for physical and emotional distress is real for those individuals seeking palliative care but facing accessibility barriers. The removal of personal choice and preferences in care, treatment and locations for care and death has the potential to impede quality of life, wellbeing and the grieving process. The demand and need for community palliative care services is increasing and geographic location should not be a barrier to access. This research is complementary to the long-term investment by the Government in policies to improve access to palliative care. After all, the way we treat our vulnerable, our sick and those who are dying is a measure not only of ourselves, but of the society in which we all live.

Implications from the research

This research has highlighted implications for people living and working in peri-urban areas and their healthcare providers. It is widely accepted that provision of good quality, holistic palliative care in the location of patient choice improves overall wellbeing for patients and their family members, enables choice of place for end-of-life care and death, and facilitates the grieving process for family and friends. The palliative care being provided in peri-urban

locations cannot be labelled holistic if barriers to access are ignored, placing increased pressure on the traditional support mechanisms of extended family, friends and neighbours.

Ethically, individuals and communities are entitled to quality, person-centred palliative care in the location of choice to maintain a good quality of life. This research, however, strongly suggests that current models of healthcare service delivery and allocation of the resources required for community palliative care in peri-urban areas requires urgent evaluation. The reliance on proximity to urban centres and an expectation that individuals will be able to travel an acceptable commute distance, have implications for palliative care patients and family caregivers. The findings suggest that the issues related to distance have negative outcomes for quality of life, wellbeing, and decision-making and influence the relocation decisions of families and communities.

A lack of resources results in clinicians having to prioritise and ration resources to meet the needs of the dying. Skilled palliative care staff have developed systems of balancing demands and available resources to the detriment of their own wellbeing, which is unsustainable in the long term. The urgency in palliative care services holds implicit ethical content which requires immediate consideration. Government and healthcare providers are well placed to navigate the challenges of resource allocation through evaluating and developing appropriate policy guidelines and procedures and models of service delivery and care tailored to the contextual needs of peri-urban locations. Future funding is needed to enhance local infrastructure and resources to improve access to community palliative care services in the location of choice and simultaneously support the clinicians providing the care. Evaluation of organisational and leadership culture is also required to retain and expand the experienced palliative care clinicians essential to care provision in peri-urban locations.

Direct contribution to knowledge

The findings from this research contribute to previous research undertaken to understand the barriers to accessing palliative care in rural locations in Australia, but it also confirms that more research needs to take place that focuses specifically on peri-urban locations. The findings indicate that government and healthcare providers rely on proximity to urban centres as an assumed predictor of accessibility to goods and services required. There is no clear or defined appropriate distance to commute for the purpose of work, nor a recognised acceptable distance to expect palliative patients to travel for complex healthcare needs.

Proximity or distance as standalone factors appear to fail to recognise frailty, disease progression, pain and symptoms exacerbated by travel, lack of functional ability, financial status, transport options or support networks available to those individuals requiring palliative treatment or care in urban centres. Peri-urban areas are neither urban nor rural, with policies dictating service delivery and resource allocation falling short for those living in fringe areas. This research demonstrates that peri-urban locations do not represent one issue or challenge for government in Australia – rather, it raises a number of interrelated questions.

Future research

Current research related to healthcare and palliative care needs in the peri-urban environment is limited. Much of the research is directed at rural communities, with the boundaries of peri-urban locations blurred, fitting neatly into neither urban nor rural locations. Future research is needed to provide clear and uniform definitions and classifications of periurban. Resource allocation is guided by outdated methods of classifying locations by variations of 'urbanness'. This research indicates that a rural model of service delivery is no longer applicable to those relocating to peri-urban areas who expect to be able to access the same quality, local services as found in metropolitan centres. Future research into methods of service delivery that currently focus on strengthening the use of digital health to support access need to consider the perspective of peri-urban residents. This research indicates that digital health in these locations is a contentious issue from the view of both clinicians and patients. It is hoped this research provides an opportunity for discussion, debate, and evaluation to ensure that service delivery progresses in the future.

Should you require any further information in relation to the recommendations raised in this report, please contact the researcher.

Shirley Papavasiliou

JCU PhD Candidate

Email: Shirley.papavasiliou@my.jcu.edu.au

Bibliography

- Abbey, E., Craig, C., & Mayland, C. R. (2020). General practitioners' perceptions of compassionate communities: a qualitative study. *BMC palliative care*, *19*(1), 97-97. https://doi.org/10.1186/s12904-020-00597-y
- Agency for Clinical Innovation (ACI). (2013). *Understanding the process to develop a Model* of Care: An ACI Framework. Government of NSW. Retrieved 27 January 2024 from <u>https://aci.health.nsw.gov.au/__data/assets/pdf_file/0009/181935/HS13-</u> 034 Framework-DevelopMoC_D7.pdf
- Agency for Clinical Innovation (ACI). (2017). *Sub-Acute Care Type Policy Guidance*. Aged Health, Palliative Care and Rehabilitation Networks. Retrieved 26 January 2024 from <u>https://aci.health.nsw.gov.au/___data/assets/pdf_file/0003/251553/ACI-Sub-acute-care-type-policy-guidance.pdf</u>
- Agency for Clinical Innovation (ACI). (2022a). *Palliative care models of care: Evidence check*. New South Wales Ministry of Health issuing body. Retrieved 26 January 2024 from <u>https://aci.health.nsw.gov.au/ data/assets/pdf file/0003/729822/ACI-Palliativecare-models-of-care-evidence-check.pdf</u>
- Agency for Clinical Innovation (ACI). (2022b). *What is the Patient Centred Medical Home Model?* NSW Government. Retrieved 24 April 2023 from <u>https://aci.health.nsw.gov.au/nhn/patient-centred-medical-home-model/what-is-the-patient-centred-medical-home-model</u>
- Aghaei, M. H., Vanaki, Z., & Mohammadi, E. (2020). Emotional Bond: The Nature of Relationship in Palliative Care for Cancer Patients. *Indian J Palliat Care*, *26*(1), 86-94. <u>https://doi.org/10.4103/ijpc_ljpc_181_19</u>
- Agllias, K. (2018). A snapshot of Australian social workers in palliative care and their work with estranged clients. *Social work in health care*, *57*(8), 620-636. https://doi.org/10.1080/00981389.2018.1474163
- Alcalde, J., & Zimmermann, C. (2022). Stigma about palliative care: origins and solutions. ecancer, 16 1377. <u>https://ecancer.org/en/journal/article/1377-stigma-about-palliative-care-origins-and-solutions</u>
- Amirinejad, G., Donehue, P., & Baker, D. (2018). Ambiguity at the peri-urban interface in Australia. Land use policy, 78, 472-480. <u>https://doi.org/10.1016/j.landusepol.2018.07.022</u>
- Aoun, S. M., Abel, J., Rumbold, B., Cross, K., Moore, J., Skeers, P., & Deliens, L. (2020). The Compassionate Communities Connectors model for end-of-life care: a community and health service partnership in Western Australia. *Palliative care and social*

practice, *14*, 2632352420935130-2632352420935130. https://doi.org/10.1177/2632352420935130

Aoun, S. M., Ewing, G., Grande, G., Toye, C., & Bear, N. (2018). The Impact of Supporting Family Caregivers Before Bereavement on Outcomes After Bereavement: Adequacy of End-of-Life Support and Achievement of Preferred Place of Death. *Journal of pain and symptom management*, 55(2), 368-378.

https://doi.org/https://doi.org/10.1016/j.jpainsymman.2017.09.023

- Aparicio, M., Centeno, C., Robinson, C. A., & Arantzamendi, M. (2022). Palliative
 Professionals' Experiences of Receiving Gratitude: A Transformative and Protective
 Resource. *Qual Health Res*, 32(7), 1126-1138.
 https://doi.org/10.1177/10497323221097247
- Appelin, G., Brobäck, G., & Berterö, C. (2005). A comprehensive picture of palliative care at home from the people involved. *European Journal of Oncology Nursing*, 9(4), 315-324. <u>https://doi.org/https://doi.org/10.1016/j.ejon.2004.11.001</u>
- Argent, N., & Plummer, P. (2022). Counter-urbanisation in pre-pandemic times: disentangling the influences of amenity and disamenity. *Australian Geographer*. <u>https://doi.org/10.1080/00049182.2022.2043807</u>
- Attorney-General's Department. (2021). Land Supply Report for Greater Adelaide: Background & Context. Retrieved 2 August 2022, from <u>https://plan.sa.gov.au/ data/assets/pdf file/0003/830982/Land Supply Report for</u> <u>Greater Adelaide - Background and Context.pdf</u>
- Attorney-General's Department. (n.d.). *Right to Health: public sector guidance sheet.* Australian Government. Retrieved 23 February 2024 from <u>https://www.ag.gov.au/rights-and-protections/human-rights-and-anti-</u> discrimination/human-rights-scrutiny/public-sector-guidance-sheets/right-health
- Australian Bureau of Statistics (ABS). (2001a). *1216.0 Australian Standard Geographical Classification (ASGC)*. Australian Government. Retrieved 6 February 2023 from https://www.abs.gov.au/AUSSTATS/abs@.nsf/66f306f503e529a5ca25697e0017661f/4D48C5A6832F9EE3CA256AD4007F680A?opendocument
- Australian Bureau of Statistics (ABS). (2001b). *1244.0 Information Papers: ABS Views on Remoteness*. Australian Government. Retrieved 05 November 2022 from <u>https://www.ausstats.abs.gov.au/ausstats/free.nsf/0/FCC8158C85424727CA256C0F0</u> <u>0003575/\$File/12440_2001.pdf</u>
- Australian Bureau of Statistics (ABS). (2006a). 1216.0 Australian Standard Geographical Classification (ASGC) - Electronic Publication, 2005 - Summary Commentary.
 Australian Government. Retrieved 28 January 2023 from

https://www.abs.gov.au/Ausstats/abs@.nsf/0/61F7610179905110CA256F190012653 B?opendocument

Australian Bureau of Statistics (ABS). (2006b). 2901.0 - Census Dictionary, 2006 (Reissue). Australian Government. Retrieved 29 January 2023 from

https://www.abs.gov.au/AUSSTATS/abs@.nsf/2f762f95845417aeca25706c00834efa/ 413876f3bae9cc70ca25720a000c428b!OpenDocument#

- Australian Bureau of Statistics (ABS). (2009a). *1216.0 Australian Standard Geographical Classification (ASGC)*. Australian Government. Retrieved 25 January 2023 from https://www.ausstats.abs.gov.au/ausstats/subscriber.nsf/0/27232652B2BE98ADCA25 https://www.ausstats.abs.gov.au/ausstats/subscriber.nsf/0/27232652B2BE98ADCA25 https://www.ausstats.abs.gov.au/ausstats/subscriber.nsf/0/27232652B2BE98ADCA25 https://www.ausstats.abs.gov.au/ausstats/subscriber.nsf/0/27232652B2BE98ADCA25 https://www.ausstats.abs.gov.au/ausstats/subscriber.nsf/0/27232652B2BE98ADCA25 https://www.ausstats.abs.gov.au/ausstats/subscriber.nsf/0/27232652B2BE98ADCA25
- Australian Bureau of Statistics (ABS). (2009b). *1216.0 Australian Standard Geographical Classification (ASGC) ASGC Structural Chart*. Australian Government. Retrieved 28 January 2023 from

https://www.abs.gov.au/ausstats/abs@.nsf/Latestproducts/7F46393B062B19ADCA25 76320019809B?opendocument#

Australian Bureau of Statistics (ABS). (2011a). *1216.0 - Australian Standard Geographical Classification (ASGC) - Introduction*. Australian Government. Retrieved 25 January 2023 from

https://www.abs.gov.au/ausstats/abs@.nsf/Latestproducts/E41F233CFD636824CA25 791F000EEB82?opendocument

- Australian Bureau of Statistics (ABS). (2011b). *1216.0 Australian Standard Geographical Classification (ASGC) Summary Main Features*. Australian Government. Retrieved 28 January 2023 from https://www.abs.gov.au/ausstats/abs@.nsf/mf/1216.0#
- Australian Bureau of Statistics (ABS). (2012). 2011.0.55.001 Information Paper: Census of Population and Housing -- Products and Services 2011. Australian Government. Retrieved 6 February 2023 from

https://www.abs.gov.au/ausstats/abs@.nsf/Previousproducts/2011.0.55.001Main%20 Features702011?opendocument&tabname=Summary&prodno=2011.0.55.001&issue =2011&num=&view=#

- Australian Bureau of Statistics (ABS). (2016a). *1270.0.55.001 Australian Statistical Geography Standard (ASGS): Volume 1 -Main Structure and Greater Capital City Statistical Areas, July 2016*. Australian Government. Retrieved 29 January 2023 from <u>https://www.abs.gov.au/ausstats/abs@.nsf/Lookup/by%20Subject/1270.0.55.001~Jul</u> <u>y%202016~Main%20Features~Mesh%20Blocks%20(MB)~10012#</u>
- Australian Bureau of Statistics (ABS). (2016b). 1270.0.55.004 Australian Statistical Geography Standard (ASGS): Volume 4 - Significant Urban Areas, Urban Centres and Localities, Section of State, July 2016. Australian Government. Retrieved 16 July

2022 from

https://www.abs.gov.au/ausstats/abs@.nsf/Lookup/by%20Subject/1270.0.55.004~Jul y%202016~Main%20Features~Different%20Definitions%20of%20Urban~6#

Australian Bureau of Statistics (ABS). (2016c). *Fact Sheet - ABS Structures*. Australian Government. Retrieved 6 February 2023 from

https://www.abs.gov.au/websitedbs/censushome.nsf/home/factsheetsas?opendocum ent&navpos=450

- Australian Bureau of Statistics (ABS). (2016d). *Fact Sheet Changes to geographic areas between Censuses*. Australian Government. Retrieved 6 February 2023 from <u>https://www.abs.gov.au/websitedbs/censushome.nsf/4a256353001af3ed4b2562bb00</u> 121564/e45142948304157cca257a4b000a6126!OpenDocument
- Australian Bureau of Statistics (ABS). (2016e). *Fact Sheet Non-ABS Structures*. Australian Government. Retrieved 6 February 2023 from

https://www.abs.gov.au/websitedbs/censushome.nsf/home/factsheetsnas?opendocu ment&navpos=450

Australian Bureau of Statistics (ABS). (2016f). *Fact Sheet - Postcodes & Postal Areas*. Australian Government. Retrieved 8 April 2023 from

https://www.abs.gov.au/websitedbs/censushome.nsf/home/factsheetspoa?opendocu ment&navpos=450

Australian Bureau of Statistics (ABS). (2017). *1270.0.55.004 - Australian Statistical Geography Standard (ASGS): Volume 4 - Significant Urban Areas, Urban Centres and Localities, Section of State, July 2016*. Australian Government. Retrieved 06 February 2023 from

https://www.abs.gov.au/ausstats/abs@.nsf/Lookup/by%20Subject/1270.0.55.004~Jul y%202016~Main%20Features~Design%20of%20SOS%20and%20SOSR~12

Australian Bureau of Statistics (ABS). (2018a). 2071.0 - Census of Population and Housing: Reflecting Australia - Stories from the Census, 2016. Australian Government. Retrieved 28 January 2023 from

https://www.abs.gov.au/ausstats/abs@.nsf/Lookup/2071.0main+features1132016

Australian Bureau of Statistics (ABS). (2018b). *Population Projections Australia*. Australian Government. Retrieved 2 August 2022 from

https://www.abs.gov.au/statistics/people/population/population-projectionsaustralia/latest-release

Australian Bureau of Statistics (ABS). (2020). *1270.0.55.003 - Australian Statistical Geography Standard (ASGS)*. Australian Government. Retrieved 17 July 2022 from https://www.abs.gov.au/ausstats/abs@.nsf/mf/1270.0.55.003 Australian Bureau of Statistics (ABS). (2021). *ABS geography products*. Australian Government. Retrieved 28 January 2023 from

https://www.abs.gov.au/statistics/statistical-geography/abs-geography-products

- Australian Bureau of Statistics (ABS). (2022a). *Australian Statistical Geography Standard* (ASGS) Edition 3. Australian Government. Retrieved 23 January 2023 from <u>https://www.abs.gov.au/statistics/standards/australian-statistical-geography-standard-asgs-edition-3/latest-release#cite-window1</u>
- Australian Bureau of Statistics (ABS). (2022b). *Household Income and Wealth, Australia*. Australian Government. Retrieved 30 November 2023 from <u>https://www.abs.gov.au/statistics/economy/finance/household-income-and-wealth-australia/latest-release#data-downloads</u>
- Australian Bureau of Statistics (ABS). (2022c). *More growth in the regions during the pandemic* <u>https://www.abs.gov.au/media-centre/media-releases/more-growth-regions-</u> <u>during-pandemic</u>
- Australian Bureau of Statistics (ABS). (2022d). Regional population Statistics about the population and components of change (births, deaths, migration) for Australia's capital cities and regions. Australian Government. Retrieved 2 August 2022 from https://www.abs.gov.au/statistics/people/population/regional-population/latest-release
- Australian Bureau of Statistics (ABS). (2023a). *Australian Statistical Geography Standard* (ASGS) - Statistical Geography. Australian Government. Retrieved 28 January 2023 from <u>https://www.abs.gov.au/statistics/statistical-geography/australian-statistical-geography-standard-asgs</u>
- Australian Bureau of Statistics (ABS). (2023b). *Causes of Death, Australia*. Australian Government. Retrieved 23 March 2024 from

https://www.abs.gov.au/statistics/health/causes-death/causes-death-australia/latestrelease#:~:text=Media%20releases-

,Key%20statistics,the%20leading%20cause%20of%20death.

- Australian Bureau of Statistics (ABS). (2023c). COVID-19 Mortality in Australia: Deaths registered until 30 September 2023. Australian Government. Retrieved 5 November 2023 from <u>https://www.abs.gov.au/articles/covid-19-mortality-australia-deaths-</u> registered-until-30-september-2023
- Australian Bureau of Statistics (ABS). (2023d). *Education and Work, Australia.* Australian Government. Retrieved 30 November 2023 from <u>https://www.abs.gov.au/statistics/people/education/education-and-work-</u> australia/latest-release
- Australian Bureau of Statistics (ABS). (2023e). *National, state and territory population.* Australian Government. Retrieved 2 December 2023 from
https://www.abs.gov.au/statistics/people/population/national-state-and-territorypopulation/latest-release

Australian Bureau of Statistics (ABS). (2023f). *Provisional Mortality Statistics*. Australian Government. Retrieved 2 December 2023 from

https://www.abs.gov.au/statistics/health/causes-death/provisional-mortalitystatistics/latest-release

Australian Bureau of Statistics (ABS). (2023g). *Statistical geography explained*. Australian Government. Retrieved 6 February 2023 from

https://www.abs.gov.au/statistics/statistical-geography/statistical-geography-explained

Australian Commission on Safety and Quality in Health Care (ACSQHC). (2020). Australian

Charter of Healthcare Rights. Retrieved 23 February 2023 from <u>https://www.safetyandquality.gov.au/publications-and-resources/resource-</u> library/australian-charter-healthcare-rights-second-edition-a4-accessible

- Australian Commission on Safety and Quality in Health Care (ACSQHC). (2021a). *Delivering and Supporting Comprehensive End-of-Life Care: a user guide*. ASQHC. Retrieved 21 January 2024 from <u>https://www.safetyandquality.gov.au/sites/default/files/2021-</u> 05/website_eolc_user_guide.pdf
- Australian Commission on Safety and Quality in Health Care (ACSQHC). (2021b). *National Safety and Quality Health Service (NSQHS) Standards*. Australian Government. Retrieved 15 December 2023 from

https://www.safetyandquality.gov.au/standards/nsqhs-standards

- Australian Commission on Safety and Quality in Health Care (ACSQHC). (2021c). *National Safety and Quality Primary and Community Healthcare Standards.* Australian Government. Retrieved 25 November 2023 from
- Australian Commission on Safety and Quality in Health Care (ACSQHC). (2023a). *National Consensus Statement: essential elements for safe and high-quality end-of-life care.* ACSQHC. Retrieved 30 November 2023 from

https://www.safetyandquality.gov.au/our-work/end-life-care/national-consensusstatement-essential-elements-safe-and-high-quality-end-life-care

- Australian Commission on Safety and Quality in Health Care (ACSQHC). (2023b). *Personcentred care*. Australian Government. Retrieved 26 November 2023 from <u>https://www.safetyandquality.gov.au/our-work/partnering-consumers/person-centred-</u> care
- Australian Commission on Safety and Quality in Health Care (ACSQHC). (2024). *Principles of care*. Australian Government. Retrieved 24 February 2024 from <u>https://www.safetyandquality.gov.au/standards/clinical-care-standards/about-clinical-care-standards/principles-care</u>

Australian Digital Health Agency. (2021). *Australian Digital Health Agency: Corporate Plan* 2021-2022. Australian Digital Health Agency.

https://www.digitalhealth.gov.au/sites/default/files/documents/adha-corporate-plan-2021-2022.pdf

Australian Government. (2021). 2021 Intergenerational Report: Australia over the next 40 years. Department of the Treasury. Retrieved 25 February 2022 from https://treasury.gov.au/publication/2021-intergenerational-report

Australian Healthcare Associates (AHA). (2019). *Exploratory Analysis of Barriers to Palliative Care: Summary Policy Paper*. Australian Government Department of Health. Retrieved 13 October 2022 from <u>https://www.health.gov.au/resources/publications/exploratory-analysis-of-barriers-to-</u>

palliative-care-summary-policy-paper?language=en

- Australian Institute of Health and Welfare (AIHW). (2004). Rural, regional and remote health; A guide to remoteness classifications. *4*(PHE53). Retrieved 1 February 2023, from <u>https://www.aihw.gov.au/getmedia/9c84bb1c-3ccb-4144-a6dd-13d00ad0fa2b/rrrh-gtrc.pdf.aspx?inline=true#:~:text=ASGC%20Remoteness%20categorises%20areas%20as,'%20and%20'very%20remote'.</u>
- Australian Institute of Health and Welfare (AIHW). (2013). *The desire to age in place among older Australians: Bulletin 114*. Australian Government. Retrieved 26 January 2024 from <u>https://www.aihw.gov.au/getmedia/69a6b0b9-6f86-411c-b15d-</u> 943144296250/15141.pdf

Australian Institute of Health and Welfare (AIHW). (2016). *Medicare Locals*. Australian Government. Retrieved 8 April 2023 from <u>https://meteor.aihw.gov.au/content/517967</u>

Australian Institute of Health and Welfare (AIHW). (2020). *How COVID-19 changed the way Australians used health services in 2019-20*. Australian Government. Retrieved 3 December 2023 from <u>https://www.aihw.gov.au/news-media/media-</u> releases/2020/december/how-covid-19-changed-the-way-australians-used-heal

Australian Institute of Health and Welfare (AIHW). (2021a). *Classifying Place of Death in Australian Mortality Statistics*. Australian Government. Retrieved 16 April 2022 from <u>https://www.abs.gov.au/statistics/research/classifying-place-death-australian-</u> mortality-statistics#data-download

Australian Institute of Health and Welfare (AIHW). (2021b). *Geographical variation in disease: diabetes, cardiovascular and chronic kidney disease*. Australian Government. Retrieved 8 April 2023 from <u>https://www.aihw.gov.au/reports/chronic-disease/geographical-variation-in-disease/contents/technical-notes/geographical-structure</u>

Australian Institute of Health and Welfare (AIHW). (2021c). *Interfaces between the aged care and health systems in Australia - where do older Australians die?* AIHW.

https://www.aihw.gov.au/reports/aged-care/where-do-older-australians-die/summary

- Australian Institute of Health and Welfare (AIHW). (2021d). *Older people*. Australian Government. Retrieved 2 August 2022 from <u>https://www.aihw.gov.au/reports-</u> data/population-groups/older-people/overview
- Australian Institute of Health and Welfare (AIHW). (2022a). *Cancer*. Australian Government. Retrieved 4 December 2023 from <u>https://www.aihw.gov.au/reports/australias-health/cancer</u>
- Australian Institute of Health and Welfare (AIHW). (2022b). *Geography*. Australian Government. Retrieved 8 April 2023 from <u>https://www.aihw.gov.au/reports/primary-health-care/medicare-subsidised-health-local-areas-2019/contents/technical-note/geography</u>
- Australian Institute of Health and Welfare (AIHW). (2022c). *Health System Overview*. Australian Government. Retrieved 29 October 2023 from

https://www.aihw.gov.au/reports/australias-health/health-system-overview

- Australian Institute of Health and Welfare (AIHW). (2022d). *Palliative Care Services in Australia*. Australian Government. Retrieved 21 June 2023 from <u>https://www.aihw.gov.au/reports/palliative-care-services/palliative-care-services-in-australia/pdf-documents</u>
- Australian Institute of Health and Welfare (AIHW). (2022e). *Palliative care workforce*. Australian Government. Retrieved 4 October 2022 from

https://www.aihw.gov.au/reports/palliative-care-services/palliative-care-services-inaustralia/contents/palliative-care-workforce

Australian Institute of Health and Welfare (AIHW). (2022f). *Rural and remote health*. Australian Government. Retrieved 2 August 2022 from

https://www.aihw.gov.au/reports/rural-remote-australians/rural-and-remote-health

Australian Institute of Health and Welfare (AIHW). (2023a). *Deaths in Australia*. Australian Government. Retrieved 21 November 2023 from <u>https://www.aihw.gov.au/reports/life-expectancy-deaths/deaths-in-australia/contents/age-at-death</u>

Australian Institute of Health and Welfare (AIHW). (2023b). *Palliative care services in Australia*. Australian Government. Retrieved 2 December 2023 from <u>https://www.aihw.gov.au/reports/palliative-care-services/palliative-care-services-in-australia/contents/summary</u>

Australian Institute of Health and Welfare (AIHW). (2023c). *Suicide & self-harm monitoring*. Australian Government. Retrieved 9 December 2023 from

https://www.aihw.gov.au/suicide-self-harm-monitoring/data/suicide-self-harmmonitoring-data

- Australian Institute of Health and Welfare [AIHW]. (2019). *Rural & remote health*. Australian Government. Retrieved 16 January 2022 from <u>https://www.aihw.gov.au/reports/rural-health/rural-remote-health/contents/rural-health</u>
- Baillie, J., Anagnostou, D., Sivell, S., Van Godwin, J., Byrne, A., & Nelson, A. (2018).
 Symptom management, nutrition and hydration at end-of-life: a qualitative exploration of patients', carers' and health professionals' experiences and further research questions. *BMC palliative care*, *17*(1), 60. <u>https://doi.org/10.1186/s12904-018-0314-4</u>
- Bakitas, M. A., Elk, R., Astin, M., Ceronsky, L., Clifford, K. N., Dionne-Odom, J. N., Emanuel, L. L., Fink, R. M., Kvale, E., Levkoff, S., Ritchie, C., & Smith, T. (2015). Systematic Review of Palliative Care in the Rural Setting. *Cancer Control*, 22(4), 450-464. https://doi.org/10.1177/107327481502200411
- Baqeas, M. H., Davis, J., & Copnell, B. (2021). Compassion fatigue and compassion satisfaction among palliative care health providers: a scoping review. *BMC Palliat Care*, 20(1), 88. <u>https://doi.org/10.1186/s12904-021-00784-5</u>
- Barbera, L., Seow, H., Sutradhar, R., Chu, A., Burge, F., Fassbender, K., McGrail, K., Lawson, B., Liu, Y., Pataky, R., & Potapov, A. (2015). Quality Indicators of End-of-Life Care in Patients With Cancer: What Rate Is Right? *Journal of oncology practice*, *11*(3), e279-287. <u>https://doi.org/10.1200/jop.2015.004416</u>
- Barbieri, S., & Jorm, L. (2019). Travel times to hospitals in Australia. *Scientific Data*, *6*(1), 248. <u>https://doi.org/10.1038/s41597-019-0266-4</u>
- Barbu, C. M. (2014). *zoom: A spatial data visualization tool (Version 2.0.6)*. In <u>https://github.com/cbarbu/R-package-zoom</u>.
- Bate, N. J., Xu, S. C., Pacilli, M., Roberts, L. J., Kimber, C., & Nataraja, R. M. (2021). Effect of the COVID-19 induced phase of massive telehealth uptake on end-user satisfaction. *Internal medicine journal*, *51*(2), 206-214. https://doi.org/10.1111/imj.15222
- Beck, M. J., & Hensher, D. A. (2021). Insights into Working from home in Australia in 2020: Positives, Negatives and the Potential for Future Benefits to Transport and Society. *Transp Res Part A Policy Pract*. (2022 Apr;158), 271-284. Retrieved 2 August 2022, from <u>https://ses.library.usyd.edu.au/handle/2123/24765</u>
- Becqué, Y. N., Rietjens, J. A. C., van der Heide, A., & Witkamp, E. (2021). How nurses support family caregivers in the complex context of end-of-life home care: a qualitative study. *BMC palliative care*, 20(1), 162. <u>https://doi.org/10.1186/s12904-021-00854-8</u>

- Beer, A., Morris, A., Paris, C. (2014). Housing and Local Government in Australia in the 21st Century. Australian Centre of Excellence for Local Government. <u>https://opus.lib.uts.edu.au/bitstream/10453/42141/3/Housing-and-LG-in-the-21st-Century.pdf</u>
- Berry, M. (1999). Unravelling the "Australian Housing Solution": The Post-War Years. Housing, Theory, and Society, 16(3), 106-123. https://doi.org/10.1080/14036099950149974
- Berry, M., Brink, E., Harris, J., & Sleeman, K. E. (2017). Supporting relatives and carers at the end of a patient's life. *BMJ (Online)*, 356(8091), j367-j367. <u>https://doi.org/10.1136/bmj.j367</u>
- Block, S. (2000). Assessing and Managing Depression in the Terminally III Patient. *Annals of Internal Medicine*, *132*(3), 209-218. <u>https://doi.org/10.7326/0003-4819-132-3-</u> <u>200002010-00007</u> %m 10651602
- Boamah, S. A., Weldrick, R., Lee, T.-S., & Taylor, N. (2021). Social Isolation Among Older Adults in Long-Term Care: A Scoping Review. *Journal of Aging and Health*, 33(7-8), 618-632. https://doi.org/10.1177/08982643211004174
- Boelen, P. A., & Prigerson, H. G. (2007). The influence of symptoms of prolonged grief disorder, depression, and anxiety on quality of life among bereaved adults: A prospective study. *European archives of psychiatry and clinical neuroscience*, 257(8), 444-452. <u>https://doi.org/10.1007/s00406-007-0744-0</u>
- Boerner, K., & Schulz, R. (2009). Caregiving, bereavement and complicated grief. *Bereave Care*, *28*(3), 10-13. <u>https://doi.org/10.1080/02682620903355382</u>
- Bone, A. E., Morgan, M., Maddocks, M., Sleeman, K. E., Wright, J., Taherzadeh, S., Ellis-Smith, C., Higginson, I. J., & Evans, C. J. (2016). Developing a model of short-term integrated palliative and supportive care for frail older people in community settings: perspectives of older people, carers and other key stakeholders. *Age and Ageing*, 45(6), 863-873. <u>https://doi.org/10.1093/ageing/afw124</u>
- Booth, M., Hill, G., Moore, M., Dalla, D., Moore, M., & Messenger, A. (2016). The new Australian Primary Health Networks: how will they integrate public health and primary care? *Public Health Research & Practice*. <u>https://www.phrp.com.au/issues/january-2016-volume-26-issue-1/the-new-australian-primary-health-networks-how-will-they-integrate-public-health-and-primary-care/</u>
- Borbasi, J. (2017). *Life Before Death: Improving Palliative Care for Older Australians*. The Centre for Independent Studies. <u>http://nla.gov.au/nla.obj-562304837</u>
- Borgstrom, E. (2017). Social Death. *QJM: An International Journal of Medicine*, *110*(1), 5-7. <u>https://doi.org/10.1093/qjmed/hcw183</u>

- Bowen, L. (2014). The multidisciplinary team in palliative care: a case reflection. *Indian J* Palliat Care, 20(2), 142-145. <u>https://doi.org/10.4103/0973-1075.132637</u>
- Bowyer, A. V., Finlay, I., Baillie, J., Byrne, A., McCarthy, J., Sampson, C., Snow, V., & Nelson,
 A. (2019). Gaining an accurate reflection of the reality of palliative care through the use of free-text feedback in questionnaires: the AFTER study. *BMJ supportive & palliative care*, 9(1), e17-e17. <u>https://doi.org/10.1136/bmjspcare-2015-000920</u>
- Bradford, N. K., Caffery, L. J., & Smith, A. C. (2016). Telehealth services in rural and remote Australia: A systematic review of models of care and factors influencing success and sustainability. *Rural and Remote Health*, *16*(4), 3808-3808. https://pubmed.ncbi.nlm.nih.gov/27744708/
- Brantnell, A., Baraldi, E., & van Achterberg, T. (2019). An inductive exploration of the implementation knowledge of research funders. *Health Research Policy and Systems*, *17*(1), 67. <u>https://doi.org/10.1186/s12961-019-0472-8</u>
- Braun, V., & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative Research in Psychology*, *3*(2), 77-101. <u>https://doi.org/10.1191/1478088706qp063oa</u>
- Braun, V., Clarke, V., Boulton, E., Davey, L., & McEvoy, C. (2021). The online survey as a qualitative research tool. *International journal of social research methodology*, 24(6), 641-654. <u>https://doi.org/10.1080/13645579.2020.1805550</u>
- Breen, L. J., Aoun, S. M., O'Connor, M., Howting, D., & Halkett, G. K. B. (2018). Family Caregivers' Preparations for Death: A Qualitative Analysis. *Journal of pain and symptom management*, 55(6), 1473-1479.
 https://doi.org/https://doi.org/10.1016/j.jpainsymman.2018.02.018
- Breen, L. J., O'Connor, M., Hewitt, L. Y., & Lobb, E. A. (2014). The "specter" of cancer: Exploring secondary trauma for health professionals providing cancer support and counseling. *Psychological Services*, *11*(1), 60-67. <u>https://doi.org/https://doi.org/10.1037/a0034451</u>
- Brereton, L., Clark, J., Ingleton, C., Gardiner, C., Preston, L., Ryan, T., & Goyder, E. (2017).
 What do we know about different models of providing palliative care? Findings from a systematic review of reviews. *Palliative Medicine*, *31*(9), 781-797.
 https://doi.org/10.1177/0269216317701890
- Brousselle, A., Brunet-Jailly, E., Kennedy, C., Phillips, S. D., Quigley, K., & Roberts, A. (2020). Beyond COVID-19: Five commentaries on reimagining governance for future crises and resilience. *Canadian Public Administration*, *63*(3), 369-408.
 https://doi.org/10.1111/capa.12388
- Brown, C. R. L., Hsu, A. T., Kendall, C., Marshall, D., Pereira, J., Prentice, M., Rice, J., Seow,H.-Y., Smith, G. A., Ying, I., & Tanuseputro, P. (2018). How are physicians deliveringpalliative care? A population-based retrospective cohort study describing the mix of

generalist and specialist palliative care models in the last year of life. *Palliative Medicine*, *32*(8), 1334-1343. <u>https://doi.org/10.1177/0269216318780223</u>

- Bryce, E., Katz, J., Pema Lama, T., Khatry, S. K., LeClerq, S. C., & Munos, M. (2021). Antenatal care processes in rural Southern Nepal: gaps in and quality of service provision-a cohort study. *BMJ Open*, *11*(12), e056392. https://doi.org/10.1136/bmjopen-2021-056392
- Bureau of Infrastructure and Transport Research Economics (BITRE). (2019). *An Introduction to where Australians live*. Department of Infrastructure, Regional Development and Cities. Retrieved 9 April 2023 from <u>https://www.bitre.gov.au/</u>
- Burns, E. J., Quinn, S. J., Abernethy, A. P., & Currow, D. C. (2015). Caregiver Expectations: Predictors of a Worse Than Expected Caregiving Experience at the End of Life. *Journal of pain and symptom management*, *50*(4), 453-461. <u>https://doi.org/https://doi.org/10.1016/j.jpainsymman.2015.04.017</u>
- Busetto, L., Wick, W., & Gumbinger, C. (2020). How to use and assess qualitative research methods. *Neurological Research and Practice*, *2*(1), 14. <u>https://doi.org/10.1186/s42466-020-00059-z</u>
- Buxton, M., & Butt, A. (2020). *The Future of the Fringe: The Crisis in Peri-Urban Planning*. CSIRO Publishing.
- Caffery, L. A., Muurlink, O. T., & Taylor-Robinson, A. W. (2022). Survival of rural telehealth services post-pandemic in Australia: A call to retain the gains in the 'new normal'. *Aust J Rural Health*. <u>https://doi.org/10.1111/ajr.12877</u>
- Callisto, A., Leong, L. J. P., & Crawford, G. B. (2021). Predictors of community death in an Australian specialist palliative care service. *Progress in palliative care*, 1-8. https://doi.org/10.1080/09699260.2021.1965776
- Campbell, C. D., & Gordon, M. C. (2003). Acknowledging the inevitable: Understanding multiple relationships in rural practice. *Professional Psychology: Research and Practice*, 34(4), 430-434. https://doi.org/https://doi.org/10.1037/0735-7028.34.4.430
- Canadian Hospice Palliative Care Nursing Standards Committee. (2014). Canadian hospice palliative care nursing standards of practice.

http://www.chpca.net/interest_groups/nurses_ig.html

- Cardenas, V., Rahman, A., Zhu, Y., & Enguidanos, S. (2022). Reluctance to Accept Palliative Care and Recommendations for Improvement: Findings From Semi-Structured Interviews With Patients and Caregivers. *Am J Hosp Palliat Care*, *39*(2), 189-195. <u>https://doi.org/10.1177/10499091211012605</u>
- Caring@home. (2020). *Guidelines for the handling of palliative care medicines in community services*. Brisbane South Palliative Care Collaborative and NPS MedicineWise.

https://www.caringathomeproject.com.au/Portals/13/Documents/NPS-Palliative-Care-Guidelines-v25-jg260620-ACC.pdf

- Carrese, J. A., Mullaney, J. L., Faden, R. R., & Finucane, T. E. (2002). Planning for death but not serious future illness: qualitative study of housebound elderly patients. *BMJ*, 325(7356), 125. https://doi.org/10.1136/bmj.325.7356.125
- Casula, M., Rangarajan, N., & Shields, P. (2021). The potential of working hypotheses for deductive exploratory research. *Quality & Quantity*, *55*(5), 1703-1725. <u>https://doi.org/10.1007/s11135-020-01072-9</u>
- Centre for Population. (2019). *Planning for Australia's Future Population*. Australian Government. Retrieved 26 January 2024 from <u>https://population.gov.au/sites/population.gov.au/files/2021-</u> 08/planning future population.pdf
- Centre for Population. (2020). *Migration between cities and regions: A quick guide to COVID-19 impacts*. Commonwealth of Australia. Retrieved 2 August 2022 from <u>https://population.gov.au/research/research-migration-between-cities-and-regions-a-</u> <u>quick-guide-to-covid-19-impacts</u>
- Centre for Population. (2022a). 2022 Population Statement. Commonwealth of Australia. Retrieved 2 January 2023 from <u>https://population.gov.au/sites/population.gov.au/files/2023-</u> 01/population_statement_2022_0.pdf
- Centre for Population. (2022b). *Regional Population, Australia, 2020-21*. Australian Government. Retrieved 4 June 2022 from <u>https://population.gov.au/data-and-forecasts/key-data-releases/regional-population-2020-21</u>
- Cerni, J., Hosseinzadeh, H., Mullan, J., Westley-Wise, V., Chantrill, L., Barclay, G., & RheeHideformat, J. (2023). Does Geography Play a Role in the Receipt of End-of-Life Care for Advanced Cancer Patients? Evidence from an Australian Local Health District Population-Based Study. *Journal of Palliative Medicine*, *26*(11), 1453-1465. <u>https://doi.org/10.1089/jpm.2022.0555</u>
- Champion, N., Medigovich, K., & Morrison, P. P. (2015). Exploring why some terminally ill people die in hospital when home was their preferred choice. *33*, 6-12. <u>https://anmj.org.au/wp-content/uploads/2019/06/Jul-Sep-2019ANMJ.pdf</u>
- Chan, H. Y.-I., Chung, C. K.-M., Tam, S. S.-C., & Chow, R. S.-K. (2021). Community palliative care services on addressing physical and psychosocial needs in people with advanced illness: a prospective cohort study. *BMC palliative care*, 20(1), 1-143. <u>https://doi.org/10.1186/s12904-021-00840-0</u>

- Charles-Edwards, E. (2021). Living in more than one place: capturing dual-local lives in the 2026 Census. *Australian population studies*, *5*(1), 3-8. <u>https://doi.org/10.37970/aps.v5i1.75</u>
- Cheluvappa, R., & Selvendran, S. (2022). Palliative Care Nursing in Australia and the Role of the Registered Nurse in Palliative Care. *Nursing Reports*, *12*(3), 589-596. https://www.mdpi.com/2039-4403/12/3/58
- Cherry, J. J., Rich, W. C., & McLennan, P. L. (2018). Telemedicine in remote Australia: The Royal Flying Doctor Service (RFDS) Medical Chest Program as a marker of remote health. *Rural and Remote Health*, *18*(4), 4502-4502. https://doi.org/10.22605/RRH4502
- Cheung, G., Douwes, G., & Sundram, F. (2017). Late-Life Suicide in Terminal Cancer: A Rational Act or Underdiagnosed Depression? *Journal of pain and symptom management*, 54(6), 835-842. <u>https://doi.org/10.1016/j.jpainsymman.2017.05.004</u>
- Choi, B. C., & Pak, A. W. (2005). A catalog of biases in questionnaires. *Prev Chronic Dis*, 2(1), A13. <u>https://www.ncbi.nlm.nih.gov/pmc/articles/PMC1323316/</u>
- Christ, G., Bonanno, G., Malkinson, R., & Rubin, S. (2003). Bereavement Experiences after the Death of a Child. *When children die: Improving palliative and end-of-life care for children and their families*.
- Cinnamon, J., Schuurman, N., & Crooks, V. A. (2008). A method to determine spatial access to specialized palliative care services using GIS. *BMC health services research*, 8(1), 140. <u>https://doi.org/10.1186/1472-6963-8-140</u>
- Claessen, S. J., Francke, A. L., Engels, Y., & Deliens, L. (2013). How do GPs identify a need for palliative care in their patients? An interview study. *BMC Fam Pract*, *14*, 42. https://doi.org/10.1186/1471-2296-14-42
- Clayton, M., & Marczak, M. (2023). Palliative care nurses' experiences of stress, anxiety, and burnout: A thematic synthesis. *Palliative & supportive care*, *21*(3), 498-514. <u>https://doi.org/10.1017/S147895152200058X</u>
- Cohen-Mansfield, J., Skornick-Bouchbinder, M., & Brill, S. (2017). Trajectories of End of Life: A Systematic Review. *The Journals of Gerontology: Series B*, 73(4), 564-572. <u>https://doi.org/10.1093/geronb/gbx093</u>
- Cohn, J., Corrigan, J., Lynn, J., Meier, D., Miller, J., Shega, J., & Wang, S. (2017). Community-Based Models of Care Delivery for People with Serious Illness. *NAM Perspectives*. <u>https://doi.org/10.31478/201704b</u>
- Collier, A., Morgan, D. D., Swetenham, K., To, T. H. M., Currow, D. C., & Tieman, J. J. (2016). Implementation of a pilot telehealth programme in community palliative care: A qualitative study of clinicians' perspectives. *Palliative Medicine*, *30*(4), 409-417. <u>https://doi.org/10.1177/0269216315600113</u>

- Collins, A., McLachlan, S.-A., & Philip, J. (2020). Community knowledge of and attitudes to palliative care: A descriptive study. *Palliative Medicine*, *34*(2), 245-252. <u>https://doi.org/10.1177/0269216319888371</u>
- Collis, C., Freebody, S., & Flew, T. (2013). Seeing the Outer Suburbs: Addressing the Urban Bias in Creative Place Thinking. *Regional Studies*, *47*(2), 148-160. https://doi.org/10.1080/00343404.2011.630315
- Commonwealth of Australia. (2021). *The Royal Commission into Aged Care Quality and Safety; Final Report,*. Commonwealth of Australia. Retrieved 2 August 2022 from <u>https://agedcare.royalcommission.gov.au/publications/final-report-volume-1</u>
- Considine, J., Botti, M., & Thomas, S. (2005). Design, format, validity and reliability of multiple choice questions for use in nursing research and education. *Collegian (Royal College of Nursing, Australia)*, *12*(1), 19-24. <u>https://doi.org/10.1016/S1322-</u> <u>7696(08)60478-3</u>
- Cook, A. F., & Hoas, H. (2019). Challenges with Healthcare in the Rural and Prairie Environment. *Narrative Inquiry in Bioethics*, *9*(2), 85-88. https://doi.org/https://doi.org/10.1353/nib.2019.0034
- Coombs, M. A., Parker, R., & de Vries, K. (2017). Managing risk during care transitions when approaching end of life: A qualitative study of patients' and health care professionals' decision making. *Palliative Medicine*, *31*(7), 617-624. https://doi.org/10.1177/0269216316673476
- Corscadden, L., Levesque, J.-F., Lewis, V., Breton, M., Sutherland, K., Weenink, J.-W., Haggerty, J., & Russell, G. (2017). Barriers to accessing primary health care: comparing Australian experiences internationally. *Australian journal of primary health*, 23(3), 223-228. <u>https://doi.org/https://doi.org/10.1071/PY16093</u>
- Costantino, V., Heslop, D. J., & MacIntyre, C. R. (2020). The effectiveness of full and partial travel bans against COVID-19 spread in Australia for travellers from China during and after the epidemic peak in China. *Journal of Travel Medicine*, 27(5), 1-7. <u>https://doi.org/10.1093/jtm/taaa081</u>
- Coulter, A., & Oldham, J. (2016). Person-centred care: what is it and how do we get there? *Future Hosp J*, *3*(2), 114-116. <u>https://doi.org/10.7861/futurehosp.3-2-114</u>
- Creswell, J. W., & Creswell, J. D. (2018). *Research design : qualitative, quantitative & mixed methods approaches* (5th edition. International student edition. ed.). SAGE Publications, Inc.

https://spada.uns.ac.id/pluginfile.php/510378/mod_resource/content/1/creswell.pdf

Croft, A., Lascelles, K., Brand, F., Carbonnier, A., Gibbons, R., Wolfart, G., & Hawton, K. (2023). Effects of patient deaths by suicide on clinicians working in mental health: A

survey. *International Journal of Mental Health Nursing*, *32*(1), 245-276. <u>https://doi.org/https://doi.org/10.1111/inm.13080</u>

- Cross, L. A. (2019). Compassion Fatigue in Palliative Care Nursing: A Concept Analysis. *Journal of Hospice and Palliative Nursing*, *21*(1), 21-28. <u>https://doi.org/10.1097/NJH.00000000000477</u>
- Currier, J. M., Neimeyer, R. A., & Berman, J. S. (2008). The Effectiveness of Psychotherapeutic Interventions for Bereaved Persons: A Comprehensive Quantitative Review. *Psychological bulletin*, *134*(5), 648-661. <u>https://doi.org/10.1037/0033-2909.134.5.648</u>
- Currow, D. C., Allingham, S., Bird, S., Yates, P., Lewis, J., Dawber, J., & Eagar, K. (2012).
 Referral patterns and proximity to palliative care inpatient services by level of socioeconomic disadvantage. A national study using spatial analysis. *BMC health services research*, *12*(1), 424. https://doi.org/10.1186/1472-6963-12-424
- Currow, D. C., & Phillips, J. (2013). Evolution of palliative care in Australia 1973-2013. *Cancer Forum*, 37(1), 38-42. <u>https://researchonline.nd.edu.au/nursing_article/90/</u>
- Dahlgren, G., & Whitehead, M. (2006). European strategies for tackling social inequities in health: Levelling up Part 2. In U. o. L. WHO Collaborating Centre for Policy Research on Social Determinants of Health (Ed.). Copenhagen: World Health Organization.
- Davidson, P. M., Kristjanson, L., Jackson, D., Bennett, M. L., Phillips, J. L., & Daly, J. (2006). Enhancing palliative care delivery in a regional community in Australia. *Australian health review*, *30*(3), 370-379. <u>https://doi.org/10.1071/ah060370</u>
- Davies, A. (2021). COVID-19 and ICT-Supported Remote Working: Opportunities for Rural Economies. *World*, 2(1), 139-152. <u>https://doi.org/10.3390/world2010010</u>
- Davies, S. L., Goodman, C., Bunn, F., Victor, C., Dickinson, A., Iliffe, S., Gage, H., Martin, W., & Froggatt, K. (2011). A systematic review of integrated working between care homes and health care services. *BMC health services research*, *11*(1), 320-320.
 https://doi.org/10.1186/1472-6963-11-320
- Davison, G., & Yelland, S. (2004). *Car wars : how the car won our hearts and conquered our cities*. Allen & Unwin.
- Dehghani, F., Barkhordari-Sharifabad, M., Sedaghati-Kasbakhi, M., & Fallahzadeh, H. (2020). Effect of palliative care training on perceived self-efficacy of the nurses. *BMC palliative care*, *19*(1), 63-63. <u>https://doi.org/10.1186/s12904-020-00567-4</u>
- Deloitte. (2014). *Transforming care at the end-of-life: Dying well matters*. Deloitte Centre for Health Solutions. Retrieved 1 December 2023 from <u>https://www2.deloitte.com/content/dam/Deloitte/uk/Documents/life-sciences-health-care/transforming-care-at-the-end-of-life.pdf</u>

- Denham, T. (2018). Metro-bound commuting and regional development: evidence from
 Victoria [Conference paper]. 8th State of Australian Cities National Conference, 28-30 November 2017, Adelaide, Australia. <u>https://doi.org/10.4225/50/5b30316f342ff</u>
- Denham, T. (2021). The limits of telecommuting: Policy challenges of counterurbanisation as a pandemic response. *Geographical Research*, *59*(4), 514-521. https://doi.org/10.1111/1745-5871.12493

Department of Health (DoH). (2020). *Requirements of the Medicines and Poisons Legislation: a summary for public health service facilities*. Government of Western Australia. Retrieved 4 December 2022 from <u>https://www.health.wa.gov.au/~/media/Corp/Policy-Frameworks/Public-</u> <u>Health/Medicines-Handling-Policy/Supporting/Requirements-of-the-Medicines-and-</u> <u>Poisons-Legislation.pdf</u>

Department of Health and Aged Care (DoHAC). (2001, 7 February 2023). *Measuring remoteness : accessibility/remoteness index of Australia (ARIA)*. Dept of Health and Aged Care. Retrieved 7 February 2023 from

http://www.health.gov.au/pubs/hfsocc/ocpanew14a.htm

- Department of Health and Aged Care (DoHAC). (2019). *National Palliative Care Strategy* 2018. Australian Government. Retrieved 6 August 2023 from <u>https://www.health.gov.au/resources/publications/the-national-palliative-care-strategy-</u> 2018?language=en
- Department of Health and Aged Care (DoHAC). (2020). *First confirmed case of novel coronavirus in Australia*. Australian Government. Retrieved 5 November 2023 from <u>https://www.health.gov.au/ministers/the-hon-greg-hunt-mp/media/first-confirmed-</u> <u>case-of-novel-coronavirus-in-australia</u>
- Department of Health and Aged Care (DoHAC). (2021). *Modified Monash Model*. Australian Government. Retrieved 3 April 2023 from <u>https://www.health.gov.au/topics/rural-health-workforce/classifications/mmm</u>
- Department of Health and Aged Care (DoHAC). (2022a). *Guiding Principles for Medication management in the community*. Commonwealth of Australia. Retrieved 30 December 2023 from <u>https://www.health.gov.au/resources/collections/guiding-principles-for-</u> <u>medication-management-in-the-community-collection</u>
- Department of Health and Aged Care (DoHAC). (2022b). *Our response to the pandemic*. Australian Government. Retrieved 2 August 2022 from <u>https://www.health.gov.au/health-alerts/covid-19/government-response</u>
- Department of Health and Aged Care (DoHAC). (2023a). *Aged care subsidies and supplements.* Australian Government. Retrieved 18 November 2023 from <u>https://www.health.gov.au/topics/aged-care/providing-aged-care-services/funding-for-</u>

aged-care-service-providers/aged-care-subsidies-andsupplements#:~:text=Subsidies%20are%20for%20approved%20providers,of%20mee ting%20specific%20care%20needs.

- Department of Health and Aged Care (DoHAC). (2023b). *What we did about COVID-19*. Australian Government. Retrieved 5 November 2023 from https://www.health.gov.au/topics/covid-19/about/what-we-did
- Department of Health Western Australia (WADOH). (2008). *Palliative Care Model of Care*. Department of Health, Western Australian. Retrieved 4 November 2023 from <u>https://www.health.wa.gov.au/~/media/Files/Corporate/general-documents/Health-Networks/WA-Cancer-and-Palliative-Care/Palliative-care/Palliative-Care-Model-of-Care.pdf</u>
- Department of Infrastructure, Transport, Regional Development, Communications and the Arts (DITRDCA). (2020). *Telecommunications in new developments*. Australian Government. Retrieved 18 January 2023 from <u>https://www.infrastructure.gov.au/media-technology-</u>

communications/internet/competition-broadband/telecommunications-newdevelopments

- Department of Infrastructure, Transport, Regional Development, Communications and the Arts, (DITRDCA). (2022). *Peri-Urban Mobile Program*. Australian Government. Retrieved 27th December 2022 from <u>https://www.infrastructure.gov.au/media-</u> <u>technology-communications/phone/mobile-services-coverage/peri-urban-mobileprogram</u>
- Department of Planning Transport and Infrastructure (DPTI). (2017). *The 30-Year Plan for Greater Adelaide; 2017 Update*. Government of South Australia. Retrieved 25 March 2022 from

https://livingadelaide.sa.gov.au/__data/assets/pdf_file/0003/319809/The_30-Year Plan for Greater Adelaide.pdf

- Department of Primary Industries and Energy and Department of Human Services and Health. (1994). *Rural, remote and metropolitan areas classification : 1991 census edition / Department of Primary Industries and Energy, Department of Human Services and Health.* Australian Government Publishing Service. Retrieved 7 February 2023 from <u>https://www.pc.gov.au/inquiries/completed/nursing-home-</u> <u>subsidies/submissions/subdr096/subdr096.pdf</u>
- Department of the Treasury. (2021). 2021 Intergenerational report: Australia over the next 40 years. Government of Australia. <u>https://apo.org.au/node/312932</u>

- Department of the Treasury. (2023). *Intergenerational Report 2023 Australia's future to 2063*. Commonwealth of Australia. Retrieved 7 November 2023 from https://treasury.gov.au/publication/2023-intergenerational-report
- Derry, H. M., Epstein, A. S., Lichtenthal, W. G., & Prigerson, H. G. (2019). Emotions in the room: common emotional reactions to discussions of poor prognosis and tools to address them. *Expert Rev Anticancer Ther*, *19*(8), 689-696. https://doi.org/10.1080/14737140.2019.1651648
- Dietrich, C. (2010). Decision Making: Factors that Influence Decision Making, Heuristics Used, and Decision Outcomes. *Inquiries Journal/Student Pulse*, *2*(02). Retrieved 18 August 2024, from http://www.inquiriesjournal.com/a?id=180
- Ding, J., Saunders, C., Cook, A., & Johnson, C. E. (2019). End-of-life care in rural general practice: How best to support commitment and meet challenges? *BMC palliative care*, *18*(1), 51-51. https://doi.org/10.1186/s12904-019-0435-4
- Dionne-Odom, J. N., Ornstein, K. A., & Kent, E. E. (2019). What do family caregivers know about palliative care? Results from a national survey. *Palliative & supportive care*, 17(6), 643-649. <u>https://doi.org/10.1017/S1478951519000154</u>
- DiTullio, M., & MacDonald, D. (1999). The struggle for the soul of hospice: Stress, coping, and change among hospice workers. *American Journal of Hospice and Palliative Medicine*®, *16*(5), 641-655. <u>https://doi.org/10.1177/104990919901600505</u>
- Dodson, J. (2016). Suburbia in Australian Urban Policy. *Built environment*, *42*(1), 23-36. <u>https://www-jstor-</u>

org.elibrary.jcu.edu.au/stable/44132317?sid=primo#metadata_info_tab_contents

- Dolan, H., Eggett, C., Holliday, L., Delves, S., Parkes, D., & Sutherland, K. (2021). Virtual care in end of life and palliative care: A rapid evidence check. *Journal of Telemedicine and Telecare*, 27(10), 631-637. <u>https://doi.org/10.1177/1357633x211046118</u>
- Driessen, H. P. A., Busschbach, J. J. V., Rijt, C. C. D. v. d., Elfrink, E. J., Raijmakers, N. J. H., Roij, J. v., Rietjens, J., & Kranenburg, L. W. (2023). Unmet care needs of patients with advanced cancer and their relatives: multicentre observational study. *BMJ supportive* & *palliative care*, spcare-2023-004242. https://doi.org/10.1136/spcare-2023-004242
- Duffield, C., Diers, D., O'Brien-Pallas, L., Aisbett, C., Roche, M., King, M., & Aisbett, K. (2011). Nursing staffing, nursing workload, the work environment and patient outcomes. *Applied nursing research*, 24(4), 244-255.
- Duffin, J. (2014). Palliative Care: The Oldest Profession? *Canadian Bulletin of Medical History*, 31(2), 205-228. <u>https://doi.org/10.3138/cbmh.31.2.205</u>
- Eagar, K., Owen, A., Masso, M., & Quinsey, K. (2006). The Griffith Area Palliative Care Service (GAPS): an evaluation of an Australian rural palliative care model. *Progress in palliative care*, *14*(3), 112-119. <u>https://doi.org/10.1179/096992606X112315</u>

- Earp, M., Cai, P., Fong, A., Blacklaws, K., Pham, T., Shack, L., & Sinnarajah, A. (2021).
 Hospital-based acute care in the last 30 days of life among patients with chronic disease that received early, late or no specialist palliative care: a retrospective cohort study of eight chronic disease groups. *BMJ Open*, *11*(3), e044196-e044196.
 https://doi.org/doi.org/10.1136/bmjopen-2020-044196
- Eastman, P., Dowd, A., White, J., Carter, J., & Ely, M. (2021). Telehealth: rapid adoption in community palliative care due to COVID-19: patient and professional evaluation. *BMJ Supportive & amp; Palliative Care*, bmjspcare-2021-002987.
 https://doi.org/10.1136/bmjspcare-2021-002987
- Elliott, R., & Timulak, L. (2005). Descriptive and interpretive approaches to qualitative research. In J. M. P. Gilbert (Ed.), A handbook of research methods for clinical and health psychology (Vol. 1, pp. 147-159). Oxford University Press.
- Elwyn, G., Buetow, S., Hibbard, J., & Wensing, M. (2007). Measuring quality through performance. Respecting the subjective: quality measurement from the patient's perspective. *BMJ*, 335(7628), 1021-1022. https://doi.org/10.1136/bmj.39339.490301.AD
- End of Life Directions for Aged Care (ELDAC). (2024). *Rural and Remote*. ELDAC. Retrieved 5 January 2024 from <u>https://www.eldac.com.au/Resources/Diverse-Population-</u> <u>Groups/Rural-and-Remote</u>
- Endacott, R., Wood, A., Judd, F., Hulbert, C., Thomas, B., & Grigg, M. (2006). Impact and Management of Dual Relationships in Metropolitan, Regional and Rural Mental Health Practice. *Australian & New Zealand Journal of Psychiatry*, *40*(11-12), 987-994. <u>https://doi.org/10.1080/j.1440-1614.2006.01922.x</u>
- Erdiaw-Kwasie, M. O., & Alam, K. (2016). Towards understanding digital divide in rural partnerships and development: A framework and evidence from rural Australia. *Journal of Rural Studies*, *43*, 214-224. <u>https://doi.org/10.1016/j.jrurstud.2015.12.002</u>
- Errett, N., Sauer, L., & Rutkow, L. (2020). An integrative review of the limited evidence on international travel bans as an emerging infectious disease disaster control measure. *Journal of Emergency Management*, *18*(*1*), 7-14.
- Fairman, N., Thomas, L. P., Whitmore, S., Meier, E. A., & Irwin, S. A. (2014). What did I miss? A qualitative assessment of the impact of patient suicide on hospice clinical staff. *Journal of Palliative Medicine*, *17*(7), 832-836. https://doi.org/10.1089/jpm.2013.0391
- Faulkner, D., Sharam, A., James, A., Tually, S., & Barrie, H. (2023). *Inquiry into housing policies and practices for precariously housed older Australians; AHURI Final Report No. 406*. Australian Housing and Urban Research Institute Limited. Retrieved 26 January 2024 from https://www.ahuri.edu.au/research/final-reports/406

- Fernando, G., & Hughes, S. (2019). Team approaches in palliative care: a review of the literature. International journal of palliative nursing, 25(9), 444-451. <u>https://doi.org/10.12968/ijpn.2019.25.9.444</u>
- Fetters, M. D., Curry, L. A., & Creswell, J. W. (2013). Achieving integration in mixed methods designs-principles and practices. *Health Serv Res*, *48*(6 Pt 2), 2134-2156. <u>https://doi.org/10.1111/1475-6773.12117</u>
- Flieger, S. P., Chui, K., & Koch-Weser, S. (2020). Lack of Awareness and Common Misconceptions About Palliative Care Among Adults: Insights from a National Survey. *J Gen Intern Med*, 35(7), 2059-2064. <u>https://doi.org/10.1007/s11606-020-05730-4</u>
- Florczak, K. L. (2021). Best Available Evidence or Truth for the Moment: Bias in Research. *Nursing Science Quarterly*, *35*(1), 20-24. <u>https://doi.org/10.1177/08943184211051350</u>
- Follmann, A. (2022). Geographies of peri-urbanization in the global south. *Geography Compass*, *16*(7), e12650. <u>https://doi.org/https://doi.org/10.1111/gec3.12650</u>
- Foster, H., Towers, B., Whittaker, J., Handmer, J., & Lowe, T. (2013). Peri-urban Melbourne in 2021 : changes and implications for the Victorian emergency management sector.
 Australian journal of emergency management, 28(3), 6-11.
- Frey, R., Robinson, J., Wong, C., & Gott, M. (2018). Burnout, compassion fatigue and psychological capital: Findings from a survey of nurses delivering palliative care. *Applied nursing research*, 43, 1-9. <u>https://doi.org/10.1016/j.apnr.2018.06.003</u>
- Friedrichsen, M., Hajradinovic, Y., Jakobsson, M., Brachfeld, K., & Milberg, A. (2021). Cultures that collide: an ethnographic study of the introduction of a palliative care consultation team on acute wards. *BMC Palliat Care*, 20(1), 180. <u>https://doi.org/10.1186/s12904-021-00877-1</u>
- Fringer, A., Hechinger, M., & Schnepp, W. (2018). Transitions as experienced by persons in palliative care circumstances and their families – a qualitative meta-synthesis. *BMC palliative care*, 17(1), 22. <u>https://doi.org/10.1186/s12904-018-0275-7</u>
- Galekop, M. M. J., van Dijk, H. M., van Exel, J., & Cramm, J. M. (2019). Views of professionals and volunteers in palliative care on patient-centred care: a Qmethodology study in the Netherlands. *BMC palliative care*, *18*(1), 97. https://doi.org/10.1186/s12904-019-0479-5
- Gardiner, C., Robinson, J., Connolly, M., Hulme, C., Kang, K., Rowland, C., Larkin, P., Meads, D., Morgan, T., & Gott, M. (2020). Equity and the financial costs of informal caregiving in palliative care: a critical debate. *BMC palliative care*, *19*(1), 71. <u>https://doi.org/10.1186/s12904-020-00577-2</u>
- Gardiner, F., Bishop, L., deGraaf, Campbell, J., Gale, L., & Quinlan, F. (2020). Equitable patient access to primary healthcare in Australia.

https://files.flyingdoctor.org.au/dd/files/RN081 Equitable Health Research Report P 4 Digital.725b.pdf

- Garg, R., Chauhan, V., & Sabreen, B. (2018). Coping Styles and Life Satisfaction in Palliative Care. *Indian J Palliat Care*, 24(4), 491-495. <u>https://doi.org/10.4103/ijpc.ljpc_63_18</u>
- Garrard, E. (1996). Palliative care and the ethics of resource allocation. *Int J Palliat Nurs*, 2(2), 91-94. <u>https://doi.org/10.12968/ijpn.1996.2.2.91</u>
- Gehman, J., Glaser, V. L., Eisenhardt, K. M., Gioia, D., Langley, A., & Corley, K. G. (2018).
 Finding Theory–Method Fit: A Comparison of Three Qualitative Approaches to Theory Building. *Journal of management inquiry*, *27*(3), 284-300.
 https://doi.org/10.1177/1056492617706029
- Gerber, K., Hayes, B., & Bryant, C. (2019). 'It all depends!': A qualitative study of preferences for place of care and place of death in terminally ill patients and their family caregivers. *Palliative Medicine*, 33(7), 802-811. <u>https://doi.org/10.1177/0269216319845794</u>
- Ghandourh, W. A. (2016). Palliative care in cancer: managing patients' expectations. *Journal of medical radiation sciences*, 63(4), 242-257. https://doi.org/https://doi.org/10.1002/jmrs.188
- Ghin, P., & Ainsworth, S. (2022). The great migration: leaving our cities for the regions part
 1: internal migration patterns, motivations and experiences. University of Melbourne.
 Retrieved 2 August 2022 from https://apo.org.au/node/317669
- Gilbert, A., Owusu-Addo, E., Feldman, P., Mackell, P., Garratt, S., Brijnath, B. (2020). Models of Integrated Care, Health and Housing: Report prepared for the Royal Commission into Aged Care Quality and Safety. National Ageing Research Institute Ltd. Retrieved 8 October 2022 from

https://agedcare.royalcommission.gov.au/sites/default/files/2020-08/Research%20Paper%207%20-

%20Models%20of%20integrated%20care%2C%20health%20and%20housing.pdf

- Gile, K. J., Beaudry, I. S., Handcock, M. S., & Ott, M. Q. (2018). Methods for Inference from Respondent-Driven Sampling Data. *Annual review of statistics and its application*, 5(1), 65-93. https://doi.org/10.1146/annurev-statistics-031017-100704
- Gioia, D. (2021). A Systematic Methodology for Doing Qualitative Research. *The Journal of applied behavioral science*, 57(1), 20-29. <u>https://doi.org/10.1177/0021886320982715</u>
- Gioia, D. A., Corley, K. G., & Hamilton, A. L. (2013). Seeking Qualitative Rigor in Inductive Research: Notes on the Gioia Methodology. *Organizational research methods*, *16*(1), 15-31. <u>https://doi.org/10.1177/1094428112452151</u>

- Glare, P. A., & Virik, K. (2001). Can we do better in end-of-life care? The mixed management model and palliative care. *Medical Journal of Australia*, *175*(10), 530-533. <u>https://doi.org/10.5694/j.1326-5377.2001.tb143711.x</u>
- Glenister, K., & Archbold, T. (2022). Keeping Rural Australians Out of Hospital. University of Melbourne. Retrieved 27 December 2022 from <u>https://aci.health.nsw.gov.au/__data/assets/pdf_file/0009/181935/HS13-</u> 034 Framework-DevelopMoC D7.pdf
- Glover, L. (2020). The Australian Health Care System. In R. Tikkanen, R. Osborn, E.
 Mossialos, A. Djordjevic, & G. Wharton (Eds.), *International Profiles of Health Care* Systems 2020. The Commonwealth Fund.

Go Gentle Australia. (2022). Suicide by people with terminal illness in NSW. Go Gentle Australia. Retrieved 15 December 2023 from <u>https://assets.nationbuilder.com/gogentleaustralia/pages/2605/attachments/original/1</u> <u>671596324/Suicide_by_people_with_terminal_ilness_in_NSW.pdf?1671596324</u>

- Gomes, B., Calanzani, N., Curiale, V., McCrone, P., & Higginson, I. J. (2013). Effectiveness and cost-effectiveness of home palliative care services for adults with advanced illness and their caregivers. *Cochrane database of systematic reviews*, 2016(3), CD007760-CD007760. <u>https://doi.org/10.1002/14651858.CD007760.pub2</u>
- Gomes, B., & Higginson, I. J. (2006). Factors influencing death at home in terminally ill patients with cancer: systematic review. *BMJ*, *332*(7540), 515-521. <u>https://doi.org/10.1136/bmj.38740.614954.55</u>
- Gomez-Batiste, X., Connor, S. (Ed.). (2017). *Building Integrated Palliative Care Programs and Services*. Liberduplex. <u>https://www.iccp-portal.org/resources/building-integrated-palliative-care-programs-and-services</u>.
- Gómez-Urquiza, J. L., Albendín-García, L., Velando-Soriano, A., Ortega-Campos, E., Ramírez-Baena, L., Membrive-Jiménez, M. J., & Suleiman-Martos, N. (2020).
 Burnout in Palliative Care Nurses, Prevalence and Risk Factors: A Systematic Review with Meta-Analysis. *Int J Environ Res Public Health*, *17*(20). https://doi.org/10.3390/ijerph17207672
- Goodson, L., & Vassar, M. (2011). An overview of ethnography in healthcare and medical education research. J Educ Eval Health Prof, 8, 4. <u>https://doi.org/10.3352/jeehp.2011.8.4</u>
- Gorod, A., Hallo, L., & Merchant, S. (2021). Governance of patient-centred care: A systemic approach to cancer treatment. *Systems research and behavioral science*, *38*(2), 257-271. <u>https://doi.org/https://doi.org/10.1002/sres.2728</u>

- Gottero, E., Cassatella, C., & Larcher, F. (2021). Planning Peri-Urban Open Spaces: Methods and Tools for Interpretation and Classification. *Land (Basel)*, *10*(8), 802. <u>https://doi.org/10.3390/land10080802</u>
- Gregory, G. (2009). Impact of rurality on health practices and services: summary paper to the inaugural rural and remote health scientific symposium. *Aust J Rural Health*, 17(1), 49-52. https://doi.org/10.1111/j.1440-1584.2008.01037.x
- Grief Australia. (2022). *A guiding voice through the universal human experience of grief*. Grief Australia. Retrieved 5 December 2023 from <u>https://www.grief.org.au/</u>
- Groeneveld, E. I., Cassel, J. B., Bausewein, C., Csikos, A., Krajnik, M., Ryan, K., Haugen, D.
 F., Eychmueller, S., Gudat Keller, H., Allan, S., Hasselaar, G. J., García-Baquero
 Merino, T., Swetenham, K., Piper, K., Furst, C. J., & Murtagh, F. E. (2017). Funding
 models in palliative care: Lessons from international experience. *Palliative Medicine*, 31(4), 296-305. <u>https://doi.org/10.1177/0269216316689015</u>
- Groot, M. M., Derksen, E. W. C., Crul, B. J. P., Grol, R. P. T. M., & Vernooij-Dassen, M. J. F. J. (2007). Living on borrowed time: Experiences in palliative care. *Patient education and counseling*, 65(3), 381-386.

https://doi.org/https://doi.org/10.1016/j.pec.2006.09.009

- Gu, D., & Wang, N. (2020). Place of Death. In D. Gu & M. E. Dupre (Eds.), *Encyclopedia of Gerontology and Population Aging* (pp. 1-10). Springer International Publishing. <u>https://doi.org/10.1007/978-3-319-69892-2_1119-1</u>
- Guaralda, M., Hearn, G., Foth, M., Yigitcanlar, T., Mayere, S., & Law, L. (2020). Towards Australian Regional Turnaround: Insights into Sustainably Accommodating Post-Pandemic Urban Growth in Regional Towns and Cities. *Sustainability (Basel, Switzerland)*, *12*(24), 1-13. https://doi.org/10.3390/su122410492
- Guenther, J., Smede, B., & Young, M. (2020). Digital inclusion in central Australia: what is it and what makes it different? *Rural Society*, *29*(3), 154-170. <u>https://doi.org/10.1080/10371656.2020.1819524</u>
- Guetterman, T. C. (2019). Basics of statistics for primary care research. *Fam Med Community Health*, 7(2). <u>https://doi.org/10.1136/fmch-2018-000067</u>
- Gurran, N., Forsyth, A., Darcy, M., Searle, G., Buckle, C., & Zou, S. (2021). AHURI Final Report 362: Population growth, regional connectivity, and city planning international lessons for Australian practice. Australian Housing and Urban Research Institute. Retrieved 2 August 2022 from <u>https://apo.org.au/node/313772</u>
- Haj-Younes, J., Abildsnes, E., Kumar, B., & Diaz, E. (2022). The road to equitable healthcare: A conceptual model developed from a qualitative study of Syrian refugees in Norway.
 SOCIAL SCIENCE & MEDICINE, 292, 114540. https://doi.org/https://doi.org/10.1016/j.socscimed.2021.114540

- Halcomb, E., Fernandez, R., Ashley, C., McInnes, S., Stephen, C., Calma, K., Mursa, R.,
 Williams, A., & James, S. (2021). The impact of COVID-19 on primary health care
 delivery in Australia. *Journal of advanced nursing*. <u>https://doi.org/10.1111/jan.15046</u>
- Hall, C., Hudson, P., & Bughey, A. (2012). Bereavement support standards for specialist palliative care services. Department of Health. <u>https://grief.org.au/Common/Uploaded%20files/Bereavement%20support%20standar</u> ds.pdf
- Hambleton, S. J., & Aloizos AM, J. (2019). Australia's digital health journey. *Medical Journal* of Australia, 210(S6), S5-S6. <u>https://doi.org/https://doi.org/10.5694/mja2.50039</u>
- Han, J. H., Kim, J. Y., & Kim, J. (2017). Dynamics of Housing Mobility in Australian Metropolitan Areas, 2001-2010: A Longitudinal Study. *Urban Policy and Research*, 35(2), 122-136. <u>https://doi.org/10.1080/08111146.2016.1146583</u>
- Hanna, T. P., King, W. D., Thibodeau, S., Jalink, M., Paulin, G. A., Harvey-Jones, E.,
 O'Sullivan, D. E., Booth, C. M., Sullivan, R., & Aggarwal, A. (2020). Mortality due to cancer treatment delay: systematic review and meta-analysis. *BMJ*, *371*, m4087.
 https://doi.org/10.1136/bmj.m4087
- Harman, B. P., Pruetz, R., & Houston, P. (2015). Tradeable development rights to protect peri-urban areas: lessons from the United States and observations on Australian practice. *Journal of environmental planning and management*, *58*(2), 357-381. https://doi.org/10.1080/09640568.2013.859130
- Hart, L. G., Larson, E. H., & Lishner, D. M. (2005). Rural definitions for health policy and research. Am J Public Health, 95(7), 1149-1155. https://doi.org/10.2105/ajph.2004.042432
- Hart, N. H., Crawford-Williams, F., Crichton, M., Yee, J., Smith, T. J., Koczwara, B., Fitch, M. I., Crawford, G. B., Mukhopadhyay, S., Mahony, J., Cheah, C., Townsend, J., Cook, O., Agar, M. R., & Chan, R. J. (2022). Unmet supportive care needs of people with advanced cancer and their caregivers: A systematic scoping review. *Critical Reviews in Oncology/Hematology*, *176*, 103728.

https://doi.org/https://doi.org/10.1016/j.critrevonc.2022.103728

- Hassan, Z. A., Schattner, P., & Mazza, D. (2006). Doing A Pilot Study: Why Is It Essential? *Malays Fam Physician*, 1(2-3), 70-73.
- Hawley, P. (2017). Barriers to Access to Palliative Care. *Palliat Care*, *10*, 1178224216688887. https://doi.org/10.1177/1178224216688887
- Healthdirect. (2022). *Primary health networks (PHNs)*. Australian Government. Retrieved 8 April 2023 from <u>https://www.healthdirect.gov.au/primary-health-networks-phns</u>

- Heinrich, M., Hofmann, L., Baurecht, H., Kreuzer, P. M., Knüttel, H., Leitzmann, M. F., & Seliger, C. (2022). Suicide risk and mortality among patients with cancer. *Nature Medicine*, 28(4), 852-859. https://doi.org/10.1038/s41591-022-01745-y
- Hii, A., Ko, K., Jones, S., & Lee, J. (Eds.). (2022). *Digital Health; Australia*. Lexology. <u>https://cdn.bfldr.com/FM3YDCO2/at/m8b7fkkx797thnxw98gbf7mn/2022 Digital Healt</u> <u>h - Australia.pdf</u>.
- Holland, J. M., & Neimeyer, R. A. (2010). An Examination of Stage Theory of Grief among Individuals Bereaved by Natural and Violent Causes: A Meaning-Oriented Contribution. OMEGA - Journal of Death and Dying, 61(2), 103-120.
 https://doi.org/10.2190/OM.61.2.b
- Holloway, I., & Galvin, K. (2017). *Qualitative research in nursing and healthcare* (Fourth edition. ed.). Wiley.
- Holt, N. R., Neumann, J. T., McNeil, J. J., & Cheng, A. C. (2020). Implications of COVID-19 for an ageing population. *The Medical Journal of Australia*, *213 (8)*, 342-344e341. <u>https://www.mja.com.au/journal/2020/213/8/implications-covid-19-ageing-</u> <u>population#:~:text=Australians%20aged%20over%2065%20years,care%20delivery%</u> 20to%20digital%20platforms.
- Howie, L., & Peppercorn, J. (2013). Early palliative care in cancer treatment: rationale, evidence and clinical implications. *Ther Adv Med Oncol*, *5*(6), 318-323. https://doi.org/10.1177/1758834013500375
- Hu, S. (2014). Pretesting. In A. C. Michalos (Ed.), *Encyclopedia of Quality of Life and Well-Being Research* (pp. 5048-5052). Springer Netherlands. <u>https://doi.org/10.1007/978-94-007-0753-5_2256</u>
- Hu, X., Ma, J., Jemal, A., Zhao, J., Nogueira, L., Ji, X., Yabroff, K. R., & Han, X. (2023).
 Suicide Risk Among Individuals Diagnosed With Cancer in the US, 2000-2016. JAMA Network Open, 6(1), e2251863-e2251863.
 https://doi.org/10.1001/jamanetworkopen.2022.51863
- Hudson, B. F., Best, S., Stone, P., & Noble, T. B. (2019). Impact of informational and relational continuity for people with palliative care needs: a mixed methods rapid review. *BMJ Open*, 9(5), e027323. <u>https://doi.org/10.1136/bmjopen-2018-027323</u>
- Hudson, P. (2003). Home-based support for palliative care families: challenges and recommendations. *Medical Journal of Australia*, *179*(6), S35-S17. <u>https://doi.org/10.5694/j.1326-5377.2003.tb05576.x</u>
- Hudson, P., Hall, C., Boughey, A., & Roulston, A. (2018). Bereavement support standards and bereavement care pathway for quality palliative care. *Palliative & supportive care*, *16*(4), 375-387. <u>https://doi.org/10.1017/S1478951517000451</u>

- Hudson, P., Remedios, C., Zordan, R., Thomas, K., Clifton, D., Crewdson, M., Hall, C.,
 Trauer, T., Bolleter, A., Clarke, D., & Bauld, C. (2012). Guidelines for the Psychosocial and Bereavement Support of Family Caregivers of Palliative Care Patients. *Journal of Palliative Medicine*, *15*(6), 696-702. https://doi.org/10.1089/jpm.2011.0466
- Hudson, P. L., Remedios, C., & Thomas, K. (2010). A systematic review of psychosocial interventions for family carers of palliative care patients. *BMC palliative care*, 9(1), 17. <u>https://doi.org/10.1186/1472-684X-9-17</u>
- Hugo, G. (2002). Changing Patterns of Population Distribution in Australia. *Journal of population research (Canberra, A.C.T.)*, Special ed. 2002, 1-21.
- Hui, D., & Bruera, E. (2020). Models of Palliative Care Delivery for Patients With Cancer. Journal of Clinical Oncology, 38(9), 852-865. <u>https://doi.org/10.1200/jco.18.02123</u>
- Hui, D., Heung, Y., & Bruera, E. (2022). Timely Palliative Care: Personalizing the Process of Referral. *Cancers (Basel)*, *14*(4). <u>https://doi.org/10.3390/cancers14041047</u>
- Hulme, C., Carmichael, F., & Meads, D. (2016). What About Informal Carers and Families? In J. Round (Ed.), *Care at the End of Life: An Economic Perspective* (pp. 167-176).
 Springer International Publishing. https://doi.org/10.1007/978-3-319-28267-1 12
- Hunt, M. R. (2009). Strengths and Challenges in the Use of Interpretive Description: Reflections Arising From a Study of the Moral Experience of Health Professionals in Humanitarian Work. *Qualitative Health Research*, *19*(9), 1284-1292.
 https://doi.org/10.1177/1049732309344612
- IAHPC. (2024). *Palliative Care Definition*. International Association for Hospice and Palliative Care. Retrieved 2 February 2024 from <u>https://hospicecare.com/what-we-</u>do/projects/consensus-based-definition-of-palliative-care/definition/#ftn1
- Independent Health and Aged Care Pricing Authority (IHACPA). (2024). *Sub-acute and non-acute care*. IHACPA. Retrieved 25 January 2024 from <u>https://www.ihacpa.gov.au/</u>

Infrastructure Australia. (2018). *Future Cities: Planning for our growing population*. Commonwealth Government. Retrieved 2 August 2022 from <u>https://www.infrastructureaustralia.gov.au/publications/future-cities-planning-our-growing-population</u>

- Infrastructure Victoria. (2021). *The post-pandemic commute; The effects of more working from home in Victoria*. Infrastructure Victoria. Retrieved 2 August 2022 from https://www.infrastructurevictoria.com.au/wp-content/uploads/2021/11/The-post-pandemic-commute.pdf
- Jackson, C. L. (2022). Living with COVID-19 in 2022: the impact of the pandemic on Australian general practice. *Medical Journal of Australia*, *216*(9), 442-444. <u>https://doi.org/10.5694/mja2.51512</u>

- James, S., Ashley, C., Williams, A., Desborough, J., Mcinnes, S., Calma, K., Mursa, R., Stephen, C., & Halcomb, E. J. (2021). Experiences of Australian primary healthcare nurses in using telehealth during COVID-19: a qualitative study. *BMJ Open*, *11*(8), e049095. <u>https://doi.org/10.1136/bmjopen-2021-049095</u>
- Janzen, J. A., Silvius, J., Jacobs, S., Slaughter, S., Dalziel, W., & Drummond, N. (2006).
 What is a health expectation? Developing a pragmatic conceptual model from psychological theory. *Health expectations : an international journal of public participation in health care and health policy*, 9(1), 37-48.
 https://doi.org/10.1111/j.1369-7625.2006.00363.x
- Javanparast, S., Anaf, J., & Tieman, J. (2022). Equity consideration in palliative care policies, programs, and evaluation: an analysis of selected federal and South Australian documents. *BMC palliative care*, *21*(1), 1-109. <u>https://doi.org/10.1186/s12904-022-00997-2</u>
- Jick, T. D. (1979). Mixing Qualitative and Quantitative Methods: Triangulation in Action. *Administrative Science Quarterly*, 24(4), 602-611. <u>https://doi.org/10.2307/2392366</u>
- Johns, L. (2015). Bereavement Care in Rural and Remote Communities in Australia. *Austral-Asian Journal of Cancer*, *14*(1). <u>https://research-</u> <u>repository.griffith.edu.au/bitstream/handle/10072/99943/JohnsPUB1839.pdf?sequenc</u> <u>e=1</u>
- Johns, L. L., McAuliffe, D., & Dorsett, P. (2019). Psychosocial care provision for terminally ill clients in rural Australian communities: the role of social work. *Rural and Remote Health*, 19(3), 5285-5285. <u>https://doi.org/10.22605/RRH5285</u>
- Johnson, C. E., Girgis, A., Paul, C. L., & Currow, D. C. (2011). Palliative care referral practices and perceptions: The divide between metropolitan and non-metropolitan general practitioners. *Palliative & supportive care*, 9(2), 181-189. <u>https://doi.org/https://doi.org/10.1017/S1478951511000058</u>
- Johnson, C. E., Senior, H., McVey, P., Team, V., Ives, A., & Mitchell, G. (2020). End-of-life care in rural and regional Australia: Patients', carers' and general practitioners' expectations of the role of general practice, and the degree to which they were met. *Health & social care in the community*, 28(6), 2160-2171. https://doi.org/10.1111/hsc.13027
- Jończy, R., Śleszyński, P., Dolińska, A., Ptak, M., Rokitowska-Malcher, J., & Rokita-Poskart, D. (2021). Environmental and Economic Factors of Migration from Urban to Rural Areas: Evidence from Poland. *Energies (Basel)*, *14*(24), 8467. <u>https://doi.org/10.3390/en14248467</u>
- Jones, R. (2000). Development of a common definition of, and approach to data collection on, the geographic location of students to be used for nationally comparable reporting

of outcomes of schooling within the context of the "National Goals for Schooling in the *Twenty-First Century*". National Education Performance Monitoring Taskforce of the Ministerial Council on Education, Employment, Training and Youth Affairs. http://www.curriculum.edu.au/verve/ resources/geographiclocation file.pdf

- Kamal, A. H. M. D., Currow, D. C. B. M. P. H., Ritchie, C. S. M. D., Bull, J. M. D., & Abernethy, A. P. M. D. (2013). Community-Based Palliative Care: The Natural Evolution for Palliative Care Delivery in the U.S. *Journal of pain and symptom management*, *46*(2), 254-264. https://doi.org/10.1016/j.jpainsymman.2012.07.018
- Kellehear, A. (1999). Health-promoting palliative care: Developing a social model for practice. *Mortality (Abingdon, England)*, *4*(1), 75-82. <u>https://doi.org/10.1080/713685967</u>
- Kenny, P., Street, D. J., Hall, J., Agar, M., & Phillips, J. (2021). Valuing End-of-Life Care for Older People with Advanced Cancer: Is Dying at Home Important? *The Patient -Patient-Centered Outcomes Research*, *14*(6), 803-813. https://doi.org/10.1007/s40271-021-00517-z
- Kent, J. L., Mulley, C., & Stevens, N. (2019). Transport and wellbeing in a newly constructed greenfield estate: A quantitative exploration of the commuting experience [Article]. *Journal of Transport & Health*, *13*, 210-223. <u>https://doi.org/10.1016/j.jth.2019.04.007</u>
- Khalil, H., Leach, M. J., Bolton, E., Spargo, M., Byrne, A., & Ristevski, E. (2022). Rural nurses' self-rated knowledge and skills in pain, medication, symptom and emergency management in community-based palliative care: A cross-sectional survey. *Collegian (Royal College of Nursing, Australia)*, *30*(1), 84-93. https://doi.org/10.1016/j.colegn.2022.07.008
- Khanassov, V., Pluye, P., Descoteaux, S., Haggerty, J. L., Russell, G., Gunn, J., & Levesque, J. F. (2016). Organizational interventions improving access to community-based primary health care for vulnerable populations: a scoping review. *International Journal for Equity in Health*, *15*(1), 1-34. <u>https://doi.org/10.1186/s12939-016-0459-9</u>
- Khashoggi, B. F., & Murad, A. (2020). Issues of Healthcare Planning and GIS: A Review. *ISPRS International Journal of Geo-Information*, *9*(6), 352. <u>https://www.mdpi.com/2220-9964/9/6/352</u>
- Kistin, C., & Silverstein, M. (2015). Pilot Studies: A Critical but Potentially Misused Component of Interventional Research. Jama, 314(15), 1561-1562. <u>https://doi.org/10.1001/jama.2015.10962</u>
- Koenig, B. A., Back, A. L., & Crawley, L. M. (2003). Qualitative Methods in End-of-Life Research: Recommendations to Enhance the Protection of Human Subjects. *Journal* of pain and symptom management, 25(4), S43-S52. <u>https://doi.org/10.1016/S0885-3924(03)00060-5</u>

- Kovan, L., & de Vries, K. (2010). 'Getting it right': an ethnographic study on how palliative care nurses discuss death. *International Journal of Work Organisation and Emotion*, 3. <u>https://doi.org/10.1504/IJWOE.2010.032921</u>
- Kozikowski, A., Morton-Rias, D., Mauldin, S., Jeffery, C., Kavanaugh, K., & Barnhill, G.
 (2022). Choosing a Provider: What Factors Matter Most to Consumers and Patients? *J Patient Exp*, 9, 23743735221074175. <u>https://doi.org/10.1177/23743735221074175</u>
- KPMG. (2020). Investing to save: the economics of increased investment in palliative care in Australia. Palliative Care Australia. Retrieved 2 August 2022 from https://apo.org.au/node/305285
- Králová, J. (2015). What Is Social Death? *Contemporary Social Science*, *10*(3), 235-248. https://doi.org/https://doi.org/10.1080/21582041.2015.1114407
- Krikorian, A., Maldonado, C., & Pastrana, T. (2020). Patient's Perspectives on the Notion of a Good Death: A Systematic Review of the Literature. *Journal of pain and symptom management*, 59(1), 152-164.

https://doi.org/https://doi.org/10.1016/j.jpainsymman.2019.07.033

- Kristjanson, L. J., & Aoun, S. (2004). Palliative Care for Families: Remembering the Hidden Patients. *Canadian Journal of Psychiatry*, *49*(6), 359-365. <u>https://doi.org/10.1177/070674370404900604</u>
- Kurpas, D., Mroczek, B., & Bielska, D. (2014). Rural and urban disparities in quality of life and health-related behaviors among chronically ill patients. *Rural and Remote Health*, 14(2), 24-37. <u>https://doi.org/10.3316/INFORMIT.351119729015973</u>
- Kuruvilla, L., Weeks, G., Eastman, P., & George, J. (2018). Medication management for community palliative care patients and the role of a specialist palliative care pharmacist: A qualitative exploration of consumer and health care professional perspectives. *Palliative Medicine*, *32*(8), 1369-1377. https://doi.org/10.1177/0269216318777390
- Kwame, A., & Petrucka, P. M. (2021). A literature-based study of patient-centered care and communication in nurse-patient interactions: barriers, facilitators, and the way forward. *BMC nursing*, 20(1), 1-10. <u>https://doi.org/10.1186/s12912-021-00684-2</u>
- Labott, S. M., Johnson, T. P., Fendrich, M., & Feeny, N. C. (2013). Emotional risks to respondents in survey research. *Journal of empirical research on human research ethics*, 8(4), 53-66. <u>https://doi.org/10.1525/jer.2013.8.4.53</u>
- Labott, S. M., Johnson, T. P., Fendrich, M., & Feeny, N. C. (2016). Evaluating and Addressing Emotional Risks in Survey Research. *Survey Practice*, *9*(1). <u>https://doi.org/10.29115/SP-2016-0006</u>.
- Lai, A. G., Pasea, L., Banerjee, A., Hall, G., Denaxas, S., Chang, W. H., Katsoulis, M., Williams, B., Pillay, D., Noursadeghi, M., Linch, D., Hughes, D., Forster, M. D.,

Turnbull, C., Fitzpatrick, N. K., Boyd, K., Foster, G. R., Enver, T., Nafilyan, V., . . . Hemingway, H. (2020). Estimated impact of the COVID-19 pandemic on cancer services and excess 1-year mortality in people with cancer and multimorbidity: near real-time data on cancer care, cancer deaths and a population-based cohort study. *BMJ Open*, *10*(11), e043828. <u>https://doi.org/10.1136/bmjopen-2020-043828</u>

- Lalani, N. (2022). Palliative care for rural growth and wellbeing: identifying perceived barriers and facilitators in access to palliative care in rural Indiana, USA. *BMC palliative care*, *21*(25 (2022)), 1. <u>https://doi.org/https://doi.org/10.1186/s12904-022-00913-8</u>
- Lally, K., Kematick, B. S., Gorman, D., & Tulsky, J. (2021). Rapid Conversion of a Palliative Care Outpatient Clinic to Telehealth. *JCO Oncology Practice*, *17*(1), e62-e67. <u>https://doi.org/10.1200/op.20.00557</u>
- Landauer, T. K., Foltz, P. W., & Laham, D. (1998). An introduction to latent semantic analysis. *Discourse Processes*, 25(2-3), 259-284. <u>https://doi.org/10.1080/01638539809545028</u>
- Latif, A. S. (2020). The Importance of Understanding Social and Cultural Norms in Delivering Quality Health Care-A Personal Experience Commentary. *Trop Med Infect Dis*, 5(1). <u>https://doi.org/10.3390/tropicalmed5010022</u>
- Latter, S., Campling, N., Birtwistle, J., Richardson, A., Bennett, M. I., Ewings, S., Meads, D., & Santer, M. (2020). Supporting patient access to medicines in community palliative care: on-line survey of health professionals' practice, perceived effectiveness and influencing factors. *BMC palliative care*, *19*(1), 148-148. https://doi.org/10.1186/s12904-020-00649-3
- Le, B., Eastman, P., Vij, S., McCormack, F., Duong, C., & Philip, J. (2017). Palliative care in general practice: GP integration in caring for patients with advanced cancer. *Australian family physician*, 46(1/2), 51-55. <u>https://doi.org/10.3316/informit.609427035167544</u>
- Lee, J. Y., Rocco, T. S., & Shuck, B. (2020). What Is a Resource: Toward a Taxonomy of Resources for Employee Engagement. *Human Resource Development Review*, 19(1), 5-38. <u>https://doi.org/10.1177/1534484319853100</u>
- Levesque, J. F., Breton, M., Senn, N., Levesque, P., Bergeron, P., & Roy, D. A. (2013a). The Interaction of Public Health and Primary Care: Functional Roles and Organizational Models that Bridge Individual and Population Perspectives. *Public health reviews*, 35(1), 1-27. <u>https://doi.org/10.1007/BF03391699</u>
- Levesque, J. F., Harris, M., & Russell, G. (2013b). Patient-centred access to health care: Conceptualising access at the interface of health systems and populations. *International Journal for Equity in Health*, *12*(1), 18-18. <u>https://doi.org/10.1186/1475-9276-12-18</u>

- Liberati, A., Altman, D. G., Tetzlaff, J., Mulrow, C., Gøtzsche, P. C., Ioannidis, J. P. A., Clarke, M., Devereaux, P. J., Kleijnen, J., & Moher, D. (2009). The PRISMA statement for reporting systematic reviews and meta-analyses of studies that evaluate healthcare interventions: explanation and elaboration. *BMJ*, *339*(2009), b2700-b2700. https://doi.org/10.1136/bmj.b2700
- Liu, Z., & Robinson, G. M. (2016). Residential development in the peri-urban fringe: The example of Adelaide, South Australia. *Land use policy*, *57*, 179-192. <u>https://doi.org/10.1016/j.landusepol.2016.05.026</u>
- Loscalzo, M. J. (2008). Palliative Care: An Historical Perspective. *Hematology*, 2008(1), 465-465. <u>https://doi.org/10.1182/asheducation-2008.1.465</u>
- Low Choy, D., Sutherland, D. C., Gleeson, B., Sipe, N., & Dodson, J. (2008). Change and Continuity in Peri-Urban Australia: Peri-Urban Futures & Sustainable Development. *Griffith University, Brisbane*. <u>https://parksleisure.com.au/parc-library/176-change-andcontinuity-in-peri-urban-australia-peri-urban-futures-and-sustainable-development/</u>
- Luckett, T., Donkor, A., Phillips, J., Currow, D. C., Parker, D., Lobb, E., & Agar, M. R. (2021). Australian specialist palliative care's response to COVID-19: an anonymous online survey of service providers. *Annals of Palliative Medicine*, *10*(3), 2747-2757. <u>https://apm.amegroups.org/article/view/60420</u>
- Luckett, T., Phillips, J. L., Agar, M., Virdun, C., Green, A., & Davidson, P. M. (2014). Elements of effective palliative care models: a rapid review. *BMC health services research*, *14*(136). <u>https://www.biomedcentral.com/1472-6963/14/136</u>
- Luo, Q., O'Connell, D. L., Yu, X. Q., Kahn, C., Caruana, M., Pesola, F., Sasieni, P., Grogan, P. B., Aranda, S., Cabasag, C. J., Soerjomataram, I., Steinberg, J., & Canfell, K. (2022). Cancer incidence and mortality in Australia from 2020 to 2044 and an exploratory analysis of the potential effect of treatment delays during the COVID-19 pandemic: a statistical modelling study. *The Lancet Public Health*, 7(6), e537-e548. https://doi.org/10.1016/S2468-2667(22)00090-1
- Macauley, R. C. (2019). The Limits of "Life-Limiting". *Journal of pain and symptom management*, 57(6), 1176-1181.

https://doi.org/https://doi.org/10.1016/j.jpainsymman.2019.03.010

- Macintyre, S. (2018). Owners and Tenants: The Commonwealth Housing Commission and Post-War Housing, 1943–1949. *Australian Economic History Review*, *58*(3), 265-282. <u>https://doi.org/10.1111/aehr.12160</u>
- Malden, L. T., Sutherland, C., Tattersall, M. H. N., Morgan, J., Forsyth, S., Levi, J. A., & Gunz,
 F. W. (1984). Dying of cancer: Factors influencing the place of death of patients. *Medical Journal of Australia*, *141*(3), 147-150.
 https://doi.org/https://doi.org/10.5694/j.1326-5377.1984.tb113060.x

- Mangan, P. A., Taylor, K. L., Yabroff, K. R., Fleming, D. A., & Ingham, J. M. (2003).
 Caregiving near the end of life: Unmet needs and potential solutions. *Palliative & supportive care*, *1*(3), 247-259. https://doi.org/10.1017/S1478951503030414
- MapPorn. (2019). *Population Density Map of Australia*. Reddit. Retrieved 26 November 2023 from

https://www.reddit.com/r/MapPorn/search/?q=population+density+map+of+australia&t ype=link&cld=d90092f9-4433-4d95-866b-151ee7583b7a&ild=a8ea73e1-0622-45acb765-7defa8c5193f

- Marshall, A., Babacan, H., & Dale, A. (2021). *Leveraging digital development in regional and rural Queensland: Policy Discussion Paper*. Rural Economies Centre of Excellence. <u>https://www.ruraleconomies.org.au/top/research-papers/leveraging-digital-</u> <u>development-in-regional-and-rural-queensland-policy-discussio</u>
- Martin, A. K., Green, T. L., McCarthy, A. L., Sowa, P. M., & Laakso, E. L. (2022). Healthcare Teams: Terminology, Confusion, and Ramifications. *Journal of Multidisciplinary Healthcare*, *15*, 765-772. https://doi.org/10.2147/jmdh.S342197
- Masman, A. D., van Dijk, M., Tibboel, D., Baar, F. P., & Mathôt, R. A. (2015). Medication use during end-of-life care in a palliative care centre. *Int J Clin Pharm*, 37(5), 767-775. <u>https://doi.org/10.1007/s11096-015-0094-3</u>
- Mason, H., Derubeis, M. B., & Hesseltine, B. (2021). Early Palliative Care for Oncology Patients: How APRNs Can Take the Lead. *J Adv Pract Oncol*, *12*(5), 477-484. <u>https://doi.org/10.6004/jadpro.2021.12.5.3</u>
- Mathew, C., Hsu, A. T., Prentice, M., Lawlor, P., Kyeremanteng, K., Tanuseputro, P., & Welch,
 V. (2020). Economic evaluations of palliative care models: A systematic review. *Palliative Medicine*, 34(1), 69-82. <u>https://doi.org/10.1177/0269216319875906</u>
- Matsumoto, M., Kashima, S., Ogawa, T., & Takeuchi, K. (2013). Do rural and remote areas really have limited accessibility to health care? Geographic analysis of dialysis patients in Hiroshima, Japan. *Rural Remote Health*, *13*(3), 2507.
- Maxwell, T. L., Hanlon, A. L., & Naylor, M. D. (2022). A Systematized Approach to Advancing the Quality of Community-Based Palliative Care. *American journal of hospice & palliative medicine*, 39(8), 951-955. <u>https://doi.org/10.1177/10499091211065173</u>
- McBride, K., & Moucheraud, C. (2022). Rural-Urban Differences: Using Finer Geographic Classifications to Reevaluate Distance and Choice of Health Services in Malawi.
 Health Syst Reform, 8(1), e2051229. <u>https://doi.org/10.1080/23288604.2022.2051229</u>
- McDougall, A., & Maharaj, V. (2011). Closing gaps on the urban fringe of Australian capital cities: an investment worth making [Article]. *Australian Planner*, *48*(3), 131-140. https://doi.org/10.1080/07293682.2011.581244

- McFarland, P. (2015). The Peri-urban Land-Use Planning Tangle: An Australian Perspective. *International Planning Studies*, *20*(3), 161-179. <u>https://doi.org/10.1080/13563475.2014.965250</u>
- McGrail, M. R., & Humphreys, J. S. (2009). Geographical classifications to guide rural health policy in Australia. *Australia and New Zealand Health Policy*, 6(1), 28. https://doi.org/10.1186/1743-8462-6-28
- McGrail, M. R., Russell, D. J., & Humphreys, J. S. (2017). Index of Access: a new innovative and dynamic tool for rural health service and workforce planning. *Australian health review*, *41*(5), 492-498. <u>https://doi.org/https://doi.org/10.1071/AH16049</u>
- McIlfatrick, S., Slater, P., Beck, E., Bamidele, O., McCloskey, S., Carr, K., Muldrew, D., Hanna-Trainor, L., & Hasson, F. (2021). Examining public knowledge, attitudes and perceptions towards palliative care: a mixed method sequential study. *BMC palliative care*, 20(1), 44-44. <u>https://doi.org/10.1186/s12904-021-00730-5</u>
- McManus, P. (2022). Counterurbanisation, demographic change and discourses of rural revival in Australia during COVID-19. *Australian Geographer*. https://doi.org/10.1080/00049182.2022.2042037
- Meeus, S., & Gulinck, H. (2008). Semi-Urban Areas in Landscape Research: A Review. *Living Reviews in Landscape Research*, 2. <u>https://doi.org/10.12942/lrlr-2008-3</u>
- Meier, E. A. P. D., Gallegos, J. V. M. A., Thomas, L. P. M. P. D., Depp, C. A. P. D., Irwin, S. A. M. D. P. D., & Jeste, D. V. M. D. (2016). Defining a Good Death (Successful Dying): Literature Review and a Call for Research and Public Dialogue. *The American journal of geriatric psychiatry*, 24(4), 261-271. <u>https://doi.org/10.1016/j.jagp.2016.01.135</u>
- Michael, N., O'Callaghan, C., Brooker, J. E., Walker, H., Hiscock, R., & Phillips, D. (2016).
 Introducing a model incorporating early integration of specialist palliative care: A qualitative research study of staff's perspectives. *Palliative Medicine*, *30*(3), 303-312.
 https://doi.org/10.1177/0269216315598069
- Microsoft Corporation. (2018). Microsoft Excel. https://office.microsoft.com/excel

Microsoft Corporation. (2019). *Microsoft Teams (Version 4.2.4.0)*. In <u>https://www.microsoft.com/en-au</u>

- Milch, V., Wang, R., Der Vartanian, C., Austen, M., Hector, D., Anderiesz, C., & Keefe, D. (2021). Cancer Australia consensus statement on COVID-19 and cancer care: embedding high value changes in practice. *Medical Journal of Australia*, 215(10), 479-484. <u>https://doi.org/10.5694/mja2.51304</u>
- Miller, E. M., & Porter, J. E. (2021). Understanding the Needs of Australian Carers of Adults Receiving Palliative Care in the Home: A Systematic Review of the Literature. SAGE Open Nursing, 7, 2377960820985682-2377960820985682. https://doi.org/10.1177/2377960820985682

- Miller, E. M., Porter, J. E., & Barbagallo, M. S. (2023). Simplifying Qualitative Case Study Research Methodology: A Step-By-Step Guide Using a Palliative Care Example. *The Qualitative Report*, 28(8), 2363-2379. <u>https://doi.org/https://doi.org/10.46743/2160-</u> <u>3715/2023.6478</u>
- Miller, E. M., Porter, J. E., & Peel, R. (2021). Palliative and End-of-Life Care in the Home in Regional/Rural Victoria, Australia: The Role and Lived Experience of Primary Carers. SAGE Open Nursing, 7, 23779608211036284-23779608211036284. https://doi.org/10.1177/23779608211036284
- Mills, J., Fox, J., Damarell, R., Tieman, J., & Yates, P. (2021). Palliative care providers' use of digital health and perspectives on technological innovation: a national study. *BMC palliative care*, 20(1), 1-124. <u>https://doi.org/10.1186/s12904-021-00822-2</u>
- Mishra, S., & Dey, A. K. (2022). Understanding and Identifying 'Themes' in Qualitative Case Study Research. South Asian Journal of Business and Management Cases, 11(3), 187-192. <u>https://doi.org/10.1177/22779779221134659</u>
- Mitchell, G. (2021). Rational prescribing in community palliative care. *Aust Prescr*, *44*(2), 45-48. <u>https://doi.org/10.18773/austprescr.2021.001</u>
- Mitchell, S., Oliver, P., Gardiner, C., Chapman, H., Khan, D., Boyd, K., Dale, J., Barclay, S., & C, R. M. (2021). Community end-of-life care during the COVID-19 pandemic: findings of a UK primary care survey. *BJGP Open*, *5*(4). https://doi.org/10.3399/bjgpo.2021.0095
- Moorhouse, C., George, M., & Smith, B. (2000). Palliative Care in Rural Australia: Involving the Community in Multidisciplinary Coordinated Care. *Australian journal of primary health*, 6(4), 141. <u>https://doi.org/10.1071/PY00047</u>
- Morris, S. M., King, C., Turner, M., & Payne, S. (2015). Family carers providing support to a person dying in the home setting: A narrative literature review. *Palliat Med*, 29(6), 487-495. <u>https://doi.org/10.1177/0269216314565706</u>
- Mortel, T. F., Marr, K., Burmeister, E., Koppe, H., Ahern, C., Walsh, R., Tyler-Freer, S., & Ewald, D. (2017). Reducing avoidable admissions in rural community palliative care: a pilot study of care coordination by General Practice registrars. *The Australian journal of rural health*, 25(3), 141-147. <u>https://doi.org/10.1111/ajr.12309</u>
- Mroz, S., Dierickx, L., Deliens, L., Cohen, J., & Chambaere, K. (2017). International Association for Hospice and Palliative Care Position Statement: Euthanasia and Physician-Assisted Suicide. *Journal of Palliative Medicine*, 20(1), 8-14. <u>https://doi.org/10.1089/jpm.2016.0290</u>
- Munday, D., Dale, J., & Murray, S. (2007). Choice and place of death: individual preferences, uncertainty, and the availability of care. *J R Soc Med*, *100*(5), 211-215. <u>https://doi.org/10.1177/014107680710000509</u>

- Murray, J. (2023). Grief and Loss. In N. Beel, C. Chinchen, T. Machin, & C. duPlessis (Eds.), *Common Client Issues in Counselling: An Australian Perspective* (pp. 135-148). University of Southern Queensland. <u>https://usq.pressbooks.pub/counselling/</u>
- Nagel, L. (2020). The influence of the COVID-19 pandemic on the digital transformation of work. International journal of sociology and social policy, 40(9/10), 861-875. <u>https://doi.org/10.1108/IJSSP-07-2020-0323</u>
- Namasivayam, P., Bui, D. T., Low, C., Barnett, T., Bridgman, H., Marsh, P., & Lee, S. (2022). Use of telehealth in the provision of after-hours palliative care services in rural and remote Australia: A scoping review protocol. *PloS one*, *17*(1), e0261962-e0261962. <u>https://doi.org/10.1371/journal.pone.0261962</u>
- Näppä, U., & Björkman-Randström, K. (2020). Experiences of participation in bereavement groups from significant others' perspectives; a qualitative study. *BMC palliative care*, *19*(1), 124. <u>https://doi.org/10.1186/s12904-020-00632-y</u>
- Näppä, U., Nässén, E., & Winqvist, I. (2023). Palliative care in rural areas collaboration between district nurses and doctors: an interview study. *BMC palliative care*, 22(1), 73. <u>https://doi.org/10.1186/s12904-023-01190-9</u>
- National Geographic. (2024). *Rural Area*. National Geographic Society. Retrieved 21 March 2024 from <u>https://education.nationalgeographic.org/resource/rural-area/</u>
- National Health and Medical Research Council. (2007). *National Statement on Ethical Conduct in Human Research*. Commonwealth of Australia.

https://www.nhmrc.gov.au/about-us/publications/national-statement-ethical-conducthuman-research-2007-updated-2018

- National Health Funding Body. (2022). *Local Health Networks*. National Health Funding Pool. Retrieved 8 April 2023 from <u>https://www.publichospitalfunding.gov.au/local-hospital-networks</u>
- National Rural Health Alliance (NRHA). (2022). *Geographic Classifications*. National Rural Health Alliance Inc. Retrieved 19 June 2023 from

https://www.ruralhealth.org.au/book/geographic-

classifications#:~:text=Using%20geographical%20classification%20systems%20for% 20health%20purposes%20has,have%20been%20used%20by%20the%20Departmen t%20since%20then.

National Suicide Prevention Project Reference Group. (2020). *National suicide prevention strategy for Australia's health system: 2020–2023*. Department of Health and Human Services. Retrieved 9 December 2023 from

https://www.health.gov.au/resources/publications/national-suicide-preventionstrategy-for-australias-health-system-2020-2023

- Nelson, D., Inghels, M., Kenny, A., Skinner, S., McCranor, T., Wyatt, S., Phull, J., Nanyonjo, A., Yusuff, O., & Gussy, M. (2023). Mental health professionals and telehealth in a rural setting: a cross sectional survey. *BMC health services research*, *23*(1), 200. https://doi.org/10.1186/s12913-023-09083-6
- Neuberger, J. (2003). A healthy view of dying. *BMJ*, 327(7408), 207-208. <u>https://doi.org/10.1136/bmj.327.7408.207</u>
- Nevin, M., Smith, V., & Hynes, G. (2019). Non-specialist palliative care: A principle-based concept analysis. *Palliative Medicine*, 33(6), 634-649. https://doi.org/10.1177/0269216319840963
- Newton, J. C., Hohnen, H., Johnson, C. E., Ives, A., McKiernan, S., Platt, V., Saunders, C., & Slavova-Azmanova, N. (2020). ' ... If I don't have that sort of money again, what happens?': adapting a qualitative model to conceptualise the consequences of out-ofpocket expenses for cancer patients in mixed health systems [Article]. *Australian health review*, *44*(3), 355-364. <u>https://doi.org/10.1071/ah18250</u>
- Newton, P., Meyer, D., & Glackin, S. (2017). Becoming urban: Exploring the transformative capacity for a suburban-to-urban transition in Australia's low-density cities. *Sustainability (Basel, Switzerland)*, *9*(10), 1718. https://doi.org/10.3390/su9101718
- Newton, P. W. (2010). Beyond Greenfield and Brownfield: The Challenge of Regenerating Australia's Greyfield Suburbs. *Built environment (London. 1978)*, *36*(1), 81-104. <u>https://doi.org/10.2148/benv.36.1.81</u>
- Nguyen, H., Nguyen, H., Dao, A., Nguyen, T., Nguyen, P., Le, P., Vu, K., Tran, A., Dao, P., Nguyen, C., & Debattista, J. (2022). The COVID-19 pandemic in Australia: Public health responses, opportunities and challenges. *The International journal of health planning and management*, *37*(1), 5-13. <u>https://doi.org/10.1002/hpm.3326</u>
- Norinder, M., Årestedt, K., Lind, S., Axelsson, L., Grande, G., Ewing, G., Holm, M., Öhlén, J., Benkel, I., & Alvariza, A. (2021). Higher levels of unmet support needs in spouses are associated with poorer quality of life – a descriptive cross-sectional study in the context of palliative home care. *BMC palliative care*, *20*(1), 132. https://doi.org/10.1186/s12904-021-00829-9
- North Western Melbourne Primary Health Network (NWMPHN). (2021). Access and Equity Framework: A framework for improving health equity in the North Western Melbourne PHN region. Australian Government Department of Health. Retrieved 23 November 2023 from <u>https://nwmphn.org.au/wp-</u> content/uploads/2021/10/NWMPHN Access Equity Framework FINAL-2.pdf
- Nwosu, A. C., McGlinchey, T., Sanders, J., Stanley, S., Palfrey, J., Lubbers, P., Chapman, L., Finucane, A., & Mason, S. (2022). Identification of Digital Health Priorities for

Palliative Care Research: Modified Delphi Study. *JMIR aging*, *5*(1), e32075-e32075. https://doi.org/10.2196/32075

- Nygaard, C. A., & Parkinson, S. (2021). Analysing the impact of COVID-19 on urban transitions and urban-regional dynamics in Australia. *The Australian Journal of Agricultural and Resource Economics*, 65(4), 878-899. <u>https://doi.org/10.1111/1467-8489.12449</u>
- Nysæter, T. M., Olsson, C., Sandsdalen, T., Wilde-Larsson, B., Hov, R., & Larsson, M. (2022). Preferences for home care to enable home death among adult patients with cancer in late palliative phase a grounded theory study. *BMC Palliat Care*, *21*(1), 49. https://doi.org/10.1186/s12904-022-00939-y
- O'Sullivan, A., Alvariza, A., Öhlén, J., & Larsdotter, C. (2021). Support received by family members before, at and after an ill person's death. *BMC palliative care*, *20*(1), 92. https://doi.org/10.1186/s12904-021-00800-8
- Ogi, M., Campling, N., Birtwistle, J., Richardson, A., Bennett, M., Santer, M., & Latter, S. (2021). Community access to palliative care medicines—patient and professional experience: systematic review and narrative synthesis. *BMJ Supportive & amp; Palliative Care*, bmjspcare-2020-002761. <u>https://doi.org/10.1136/bmjspcare-2020-002761</u>
- Olson, J. L., March, S., Clough, B., Biddle, S. J. H., & Ireland, M. (2019). Not quite city and not quite rural: Active lifestyle beliefs in peri-urban Australians. *Health promotion journal of Australia*, *30*(S1), 72-84. <u>https://doi.org/10.1002/hpja.234</u>
- Olvera, C. E., Levin, M. E., & Fleisher, J. E. (2023). Chapter 4 Community-based neuropalliative care. In J. M. Miyasaki & B. M. Kluger (Eds.), *Handbook of Clinical Neurology* (Vol. 191, pp. 49-66). Elsevier. <u>https://doi.org/https://doi.org/10.1016/B978-0-12-824535-4.00001-X</u>
- Organisation for Economic Co-operation and Development (OECD). (2020). *Digital Transformation in the Age of COVID-19, Building Resilience and Bridging Divides; Digital Economy Outlook 2020 Supplement.* Organisation for Economic Co-operation and Development. Retrieved 2 August 2022 from <u>www.oecd.org/digital/digital-</u> <u>economy-outlook-covid.pdf</u>
- Österlind, J., & Henoch, I. (2021). The 6S-model for person-centred palliative care: A theoretical framework. *Nursing Philosophy*, *22*(2), e12334. https://doi.org/10.1111/nup.12334
- Page, M. J., McKenzie, J. E., Bossuyt, P. M., Boutron, I., Hoffmann, T. C., Mulrow, C. D.,
 Shamseer, L., Tetzlaff, J. M., Akl, E. A., Brennan, S. E., Chou, R., Glanville, J.,
 Grimshaw, J. M., Hróbjartsson, A., Lalu, M. M., Li, T., Loder, E. W., Mayo-Wilson, E.,
 McDonald, S., McGuinness, L., Stewart, L., Thomas, J., Tricco, A., Welch, A., Whiting,

P., Moher, D. . (2021). The PRISMA 2020 statement: an updated guideline for reporting systematic reviews. *BMJ (Online)*(372), n71-n71. https://doi.org/https://doi.org/10.1136/bmj.n71

- Page, M. J., Moher, D., Bossuyt, P. M., Boutron, I., Hoffmann, T. C., Mulrow, C. D., Shamseer, L., Tetzlaff, J. M., Akl, E. A., Brennan, S. E., Chou, R., Glanville, J., Grimshaw, J. M., Hróbjartsson, A., Lalu, M. M., Li, T., Loder, E. W., Mayo-Wilson, E., McDonald, S., . . . McKenzie, J. E. (2021). PRISMA 2020 explanation and elaboration: updated guidance and exemplars for reporting systematic reviews. *BMJ* (*Online*), 372, n160-n160. <u>https://doi.org/10.1136/bmj.n160</u>
- PalliAged. (2017). *Models of Care Synthesis*. CareSearch. Retrieved 3 October 2022 from <u>https://www.palliaged.com.au/tabid/4502/Default.aspx</u>
- PalliAged. (2022). *A Palliative Approach*. CareSearch. Retrieved 26 November 2023 from <u>https://www.palliaged.com.au/tabid/5748/Default.aspx</u>
- Palliative Care Australia (PCA). (2005). A Guide to Palliative Care Service Development: A population based approach. Palliative Care Australia. Retrieved 30 August 2022 from https://palliativecare.org.au/wp-content/uploads/2015/05/A-guide-to-palliative-care-service-development-a-population-based-approach.pdf
- Palliative Care Australia (PCA). (2015). *What is Palliative Care?* Palliative Care Australia Ltd. Retrieved 26 November 2023 from <u>https://palliativecare.org.au/resource/what-is-palliative-care/</u>
- Palliative Care Australia (PCA). (2018a). *National Palliative Care Standards 5th Edition*. PCA. Retrieved 6 August 2023 from

https://palliativecare.org.au/publication/standards/

- Palliative Care Australia (PCA). (2018b). *Palliative Care Service Development Guidelines*. Palliative Care of Australia. Retrieved 2 November 2023 from <u>https://palliativecare.org.au/wp-content/uploads/dlm_uploads/2018/02/PalliativeCare-Service-Delivery-2018_web-1.pdf</u>
- Palliative Care Australia (PCA). (2018c). *Policy Statement Grief and Bereavement*. Palliative Care Australia. Retrieved 19 June 2023 from <u>https://palliativecare.org.au/wp-</u> <u>content/uploads/dlm_uploads/2018/08/Grief-and-Bereavement-Policy-Statement-FINAL.pdf</u>
- Palliative Care Australia (PCA). (2020). *Palliative Care and COVID-19: Grief, Bereavement and Mental Health*. Palliative Care Australia. Retrieved 15 December 2023 from https://www.croakey.org/wp-content/uploads/2020/12/Palliative-Care-Australia-Grief-Bereavement-and-Mental-Health-December-2020-final-1.pdf

- Palliative Care Australia (PCA). (2023). *Celebrating 30 years of advocating for quality palliative care*. Retrieved 25 November 2023 from <u>https://palliativecare.org.au/pca-30th-anniversary/</u>
- Palliative Care Victoria. (2017). *Living, dying & grieving well: A guide to palliative care.* Palliative Care Victoria. <u>https://www.pallcarevic.asn.au/store/products/3/living-dying-grieving-well-a-guide-to-palliative-care-booklet#:~:text=It%20is%20an%20easy%2Dto,of%20people%20receiving%20palliative-ve%20care.</u>
- Park, S. (2017). Digital inequalities in rural Australia: A double jeopardy of remoteness and social exclusion. *Journal of Rural Studies*, *54*, 399-407. https://doi.org/10.1016/j.jrurstud.2015.12.018
- Parreira, P., Santos-Costa, P., Neri, M., Marques, A., Queirós, P., & Salgueiro-Oliveira, A. (2021). Work Methods for Nursing Care Delivery. *Int J Environ Res Public Health*, *18*(4). <u>https://doi.org/10.3390/ijerph18042088</u>
- Pask, S., Pinto, C., Bristowe, K., van Vliet, L., Nicholson, C., Evans, C. J., George, R., Bailey, K., Davies, J. M., Guo, P., Daveson, B. A., Higginson, I. J., & Murtagh, F. E. M. (2018). A framework for complexity in palliative care: A qualitative study with patients, family carers and professionals. *Palliative Medicine*, *32*(6), 1078-1090. https://doi.org/10.1177/0269216318757622
- PCA. (2022). Palliative Care Australia Roadmap 2022-2027. https://palliativecare.org.au/publication/palliative-care-australia-roadmap-2022-2027/
- Pearse, J., Mazevska, D., & Sheehan, J. (2011). *Demographic and utilisation changes for rural and remote populations—subacute admitted care*. National Rural Health Alliance. Retrieved 30 January 2024 from https://ruralhealth.org.au/11nrhc/papers/11th%20NRHC%20Pearse_Jim_A3.pdf
- Penchansky, R., & Thomas, J. W. (1981). The Concept of Access: Definition and Relationship to Consumer Satisfaction. *Medical Care*, *19*(2), 127-140. <u>http://www.jstor.org.elibrary.jcu.edu.au/stable/3764310</u>
- Pereira, J., & Chasen, M. R. (2016). Early palliative care: taking ownership and creating the conditions. *Curr Oncol*, *23*(6), 367-370. <u>https://doi.org/10.3747/co.23.3461</u>
- Peri-Urban Councils Victoria. (2022). *Understanding who we are*. Peri-Urban Councils Victoria. Retrieved 1 March 2022 from <u>https://periurban.org.au/members</u>
- Perry, S. J., Richter, J. P., & Beauvais, B. (2018). The Effects of Nursing Satisfaction and Turnover Cognitions on Patient Attitudes and Outcomes: A Three-Level Multisource Study. *Health Serv Res*, 53(6), 4943-4969. <u>https://doi.org/10.1111/1475-6773.12997</u>

- Pesut, B., Hooper, B., Jacobsen, M., Nielsen, B., Falk, M., & O 'Connor, B. P. (2017). Nurseled navigation to provide early palliative care in rural areas: a pilot study. *BMC palliative care*, *16*(1), 37-37. https://doi.org/10.1186/s12904-017-0211-2
- Pesut, B., Robinson, C. A., & Bottorff, J. L. (2014). Among neighbors: An ethnographic account of responsibilities in rural palliative care. *Palliative & supportive care*, *12*(2), 127-138. <u>https://doi.org/10.1017/S1478951512001046</u>
- Petrie, S., Carson, D., Peters, P., Hurtig, A. K., LeBlanc, M., Simpson, H., Barnabe, J., Young, M., Ostafichuk, M., Hodge, H., Gladman, J., Smale, M., & Gonzalez Garcia, M. (2021). What a Pandemic Has Taught Us About the Potential for Innovation in Rural Health: Commencing an Ethnography in Canada, the United States, Sweden, and Australia. *Front Public Health*, *9*, 768624. https://doi.org/10.3389/fpubh.2021.768624
- Philip, J., Le Gautier, R., Collins, A., Nowak, A. K., Le, B., Crawford, G. B., Rankin, N., Krishnasamy, M., Mitchell, G., McLachlan, S. A., M, I. J., Hudson, R., Rischin, D., Sousa, T. V., & Sundararajan, V. (2021). Care plus study: a multi-site implementation of early palliative care in routine practice to improve health outcomes and reduce hospital admissions for people with advanced cancer: a study protocol. *BMC Health Serv Res*, *21*(1), 513. https://doi.org/10.1186/s12913-021-06476-3
- Phillips, J. L., Davidson, P. M., Newton, P. J., & DiGiacomo, M. (2008). Supporting Patients and Their Caregivers After-Hours at the End of Life: The Role of Telephone Support. *Journal of pain and symptom management*, 36(1), 11-21. <u>https://doi.org/https://doi.org/10.1016/j.jpainsymman.2007.08.017</u>
- Phyo, A. Z. Z., Freak-Poli, R., Craig, H., Gasevic, D., Stocks, N. P., Gonzalez-Chica, D. A., & Ryan, J. (2020). Quality of life and mortality in the general population: a systematic review and meta-analysis. *BMC Public Health*, 20(1), 1596. <u>https://doi.org/10.1186/s12889-020-09639-9</u>
- Polit, D. F., & Beck, C. T. (2017). *Nursing research : generating and assessing evidence for nursing practice* (Tenth edition. ed.). Wolters Kluwer Health.
- Pope, C., & Mays, N. (2020). *Qualitative research in health care* (Fourth edition. ed.). Wiley Blackwell.
- Prem, V., Jaykumar, S., Karthikbabu, S., Karvannan, H., Kumar, S. P., Sisodia, V., & Syed, N. (2012). Study of nurses' knowledge about palliative care: A quantitative crosssectional survey. *Indian journal of palliative care*, *18*(2), 122-127. <u>https://doi.org/10.4103/0973-1075.100832</u>
- Prigerson, H. G., Bierhals, A. J., Kasl, S. V., Reynolds Iii, C. F., Shear, M. K., Newsom, J. T., & Jacobs, S. (1996). Complicated grief as a disorder distinct from bereavement-
related depression and anxiety: a replication study. *The American journal of psychiatry*, *153*(11), 1484-1486. <u>https://doi.org/10.1176/ajp.153.11.1484</u>

- Private Healthcare Australia. (2023). *Australians sign up to private health insurance in record numbers*. <u>https://www.privatehealthcareaustralia.org.au/australians-sign-up-to-</u> <u>private-health-insurance-in-record-</u> <u>numbers/#:~:text=A%20record%2014.42%20million%20Australians,persons%20takin</u> g%20out%20private%20cover.
- Productivity Commission. (2021). *Working from home: Research Paper*. <u>https://www.pc.gov.au/research/completed/working-from-home/working-from-home.pdf</u>
- Proulx, T., Inzlicht, M., & Harmon-Jones, E. (2012). Understanding all inconsistency compensation as a palliative response to violated expectations. *Trends in Cognitive Sciences*, 16(5), 285-291. <u>https://doi.org/https://doi.org/10.1016/j.tics.2012.04.002</u>

Qualtrics. (2020). Qualtrics. Retrieved 17 January 2023 from https://www.qualtrics.com

- Radbruch, L., De Lima, L., Knaul, F., Wenk, R., Ali, Z., Bhatnaghar, S., Blanchard, C., Bruera, E., Buitrago, R., Burla, C., Callaway, M., Munyoro, E. C., Centeno, C., Cleary, J., Connor, S., Davaasuren, O., Downing, J., Foley, K., Goh, C., . . . Pastrana, T. (2020).
 Redefining Palliative Care-A New Consensus-Based Definition. *J Pain Symptom Manage*, 60(4), 754-764. <u>https://doi.org/10.1016/j.jpainsymman.2020.04.027</u>
- Ragusa, A. T. (2010). Country landscapes, private dreams? Tree change and the dissolution of rural Australia. *Rural Society*, *20*(s), 137-150. <u>https://doi.org/10.5172/rsj.20.s.137</u>
- Rainsford, S. (2019). Palliative Care in Rural Settings. In R. D. MacLeod & L. VandenBlock (Eds.), *Textbook of Palliative Care* (1st 2019. ed., pp. 952 to 965). Springer International Publishing. https://doi.org/10.1007/978-3-319-77740-5
- Ransom, H., & Olsson, J. M. (2017). Allocation of Health Care Resources: Principles for Decision-making. *Pediatrics In Review*, 38(7), 320-329. <u>https://doi.org/10.1542/pir.2016-0012</u>
- Ray, R. A., Fried, O., & Lindsay, D. (2014). Palliative care professional education via video conference builds confidence to deliver palliative care in rural and remote locations.
 BMC health services research, *14*(1), 272-272. <u>https://doi.org/10.1186/1472-6963-14-272</u>
- Reeves, S., Kuper, A., & Hodges, B. D. (2008). Qualitative research methodologies: ethnography. *BMJ*, 337, a1020. <u>https://doi.org/10.1136/bmj.a1020</u>
- Regional Australia Institute. (2021). *Regional lifestyle continues to appeal* <u>https://regionalaustralia.org.au/Web/Web/Media/Media-Releases/2021/Regional-</u> <u>Lifestyle-Continues-to-Appeal.aspx</u>

- Reid, C., Bennetts, S. K., Nicholson, J. M., Amir, L. H., & Chamberlain, C. (2023). Rural primary care workforce views on trauma-informed care for parents experiencing complex trauma: A descriptive study. *Australian Journal of Rural Health*, *31*(1), 98-113. https://doi.org/https://doi.org/10.1111/ajr.12922
- Reinhard, J., & Buckland, M. (2022). Addressing Infrastructure Inequality: A path to equality for Melbourne's outer suburbs. The McKell Institute. <u>https://mckellinstitute.org.au/wp-content/uploads/2022/08/Victorian-Infrastructure-Inequality.pdf</u>
- Renjith, V., Yesodharan, R., Noronha, J. A., Ladd, E., & George, A. (2021). Qualitative Methods in Health Care Research. *Int J Prev Med*, *12*, 20. <u>https://doi.org/10.4103/ijpvm.IJPVM_321_19</u>
- Reymond, L., Parker, G., Gilles, L., & Cooper, K. (2018). Home-based palliative care. *Australian Journal for General Practitioners*, *47*, 747-752. https://www1.racqp.org.au/ajqp/2018/november/home-based-palliative-care
- Rich, J. L., Chojenta, C., & Loxton, D. (2013). Quality, rigour and usefulness of free-text comments collected by a large population based longitudinal study - ALSWH. *PloS* one, 8(7), e68832. https://doi.org/10.1371/journal.pone.0068832
- Riiskjær, E., Ammentorp, J., & Kofoed, P.-E. (2012). The value of open-ended questions in surveys on patient experience: number of comments and perceived usefulness from a hospital perspective. *International Journal for Quality in Health Care*, *24*(5), 509-516. <u>https://doi.org/10.1093/intqhc/mzs039</u>
- Robinson, J., Gott, M., Gardiner, C., & Ingleton, C. (2016). The 'problematisation' of palliative care in hospital: an exploratory review of international palliative care policy in five countries. *BMC Palliat Care*, *15*, 64. <u>https://doi.org/10.1186/s12904-016-0137-0</u>
- Robinson, J., Gott, M., Gardiner, C., & Ingleton, C. (2018). The impact of the environment on patient experiences of hospital admissions in palliative care. *BMJ supportive* & *palliative care*, *8*(4), 485-492. <u>https://doi.org/10.1136/bmjspcare-2015-000891</u>
- Rome, R. B., Luminais, H. H., Bourgeois, D. A., & Blais, C. M. (2011). The role of palliative care at the end of life. *Ochsner J*, *11*(4), 348-352.
- Roufeil, L., & Battye, K. (2008). *Effective regional, rural and remote family and relationships services delivery*. Australian Government. Retrieved 2 August 2022 from <u>https://inform.regionalaustralia.org.au/population-and-people/family-community-and-disability/item/effective-regional-rural-and-remote-family-and-relationships-service-delivery</u>
- Royal Australian College of General Practitioners (RACGP). (2016). *GP-led palliative care in rural Australia*. RACGP. Retrieved 3 July 2023 from https://www.racgp.org.au/FSDEDEV/media/documents/RACGP/Position%20stateme https://www.racgp.org.au/FSDEDEV/media/documents/RACGP/Position%20stateme https://www.racgp.org.au/FSDEDEV/media/documents/RACGP/Position%20stateme https://www.racgp.org.au/FSDEDEV/media/documents/RACGP/Position%20stateme https://www.racgp.org.au/FSDEDEV/media/documents/RACGP/Position%20stateme

- Royal Australian College of General Practitioners (RACGP). (2023). Part A. Palliative and end-of-life care. In *RACGP aged care clinical guide (Silver Book) 5th edition*. <u>https://www.racgp.org.au/getattachment/55aea74c-fbe9-4d01-8cbc-</u> 425948225cb8/Palliative-care.aspx
- Rucinska, M., & Nawrocki, S. (2022). COVID-19 Pandemic: Impact on Cancer Patients. *Int J* Environ Res Public Health, 19(19). <u>https://doi.org/10.3390/ijerph191912470</u>
- Ruiz-Martínez, I., & Esparcia, J. (2020). Internet access in rural areas: Brake or stimulus as post-covid-19 opportunity? *Sustainability (Basel, Switzerland)*, *12*(22), 1-17. https://doi.org/10.3390/su12229619
- Rumbold, B., & Aoun, S. M. (2021). Palliative and end-of-life care service models: To what extent are consumer perspectives considered? *Healthcare (Basel)*, *9*(10), 1286. <u>https://doi.org/10.3390/healthcare9101286</u>
- Rural Health Standing Committee. (2018). *National Strategic Framework for Rural and Remote Health*. Government of Australia. Retrieved 28 January 2023 from <u>https://apo.org.au/node/167991</u>
- Russell, A., & Ball, J. (2003). *ARIA Scores: Index of Accessibility/Remoteness*. The Australian Longitudinal Study on Women's Health (ALSWH). Retrieved 5 February 2023 from <u>https://alswh.org.au/wp-content/uploads/2020/08/DDSSection5ARIA.pdf</u>
- SA Health. (2021). COVID-19 Acute Care Response Strategy. Government of SA. Retrieved 20 July 2022 from https://www.sahealth.sa.gov.au/wps/wcm/connect/f628ddc3-a4e2-4be5-92a1-bee79091ef55/OFFICIAL+SAH+COVID-19+Acute+Care+Response+Strategy+v1.0+031221.pdf?MOD=AJPERES&CACHEID =ROOTWORKSPACE-f628ddc3-a4e2-4be5-92a1-bee79091ef55-nR-zc30
- Saarinen, J., Mishina, K., Soikkeli-Jalonen, A., & Haavisto, E. (2023). Family members' participation in palliative inpatient care: An integrative review. *Scandinavian Journal of Caring Sciences*, 37(4), 897-908. <u>https://doi.org/https://doi.org/10.1111/scs.13062</u>
- Sandsdalen, T., Grøndahl, V. A., Hov, R., Høye, S., Rystedt, I., & Wilde-Larsson, B. (2016).
 Patients' perceptions of palliative care quality in hospice inpatient care, hospice day care, palliative units in nursing homes, and home care: a cross-sectional study. *BMC palliative care*, *15*(1), 79-79. <u>https://doi.org/10.1186/s12904-016-0152-1</u>
- Sapeta, P., Centeno, C., Belar, A., & Arantzamendi, M. (2022). Adaptation and continuous learning: integrative review of coping strategies of palliative care professionals. *Palliat Med*, 36(1), 15-29. <u>https://doi.org/10.1177/02692163211047149</u>
- Sarradon-Eck, A., Besle, S., Troian, J., Capodano, G., & Mancini, J. (2019). Understanding the Barriers to Introducing Early Palliative Care for Patients with Advanced Cancer: A Qualitative Study. *Journal of Palliative Medicine*, *22*(5), 508-516. https://doi.org/10.1089/jpm.2018.0338

- Saunders, C. (2001). The evolution of palliative care. *Journal of the Royal Society of Medicine*, 94(9), 430-432. https://doi.org/10.1177/014107680109400904
- Saunders, C. H., Sierpe, A., von Plessen, C., Kennedy, A. M., Leviton, L. C., Bernstein, S. L., Goldwag, J., King, J. R., Marx, C. M., Pogue, J. A., Saunders, R. K., Citters, A. V., Yen, R. W., Elwyn, G., & Leyenaar, J. K. (2023). Practical thematic analysis: a guide for multidisciplinary health services research teams engaging in qualitative analysis. *BMJ*, *381*, e074256. https://doi.org/10.1136/bmj-2022-074256
- Saurman, E., Allingham, S., Draper, K., Edwards, J., Moody, J., Hooper, D., Kneen, K., Connolly, J., & Eagar, K. (2022). Preferred Place of Death—A Study of 2 Specialist Community Palliative Care Services in Australia. *Journal of Palliative Care*, *37*(1), 26-33. <u>https://doi.org/10.1177/08258597211018059</u>
- Schaeffer, N. C., & Dykema, J. (2011). Questions for Surveys: Current Trends and Future Directions. *Public Opin Q*, *75*(5), 909-961. <u>https://doi.org/10.1093/poq/nfr048</u>
- Scheil-Adlung, X. (2015). Global Evidence on Inequities in Rural Health Protection : New Data on Rural Deficits in Health Coverage for 174 Countries. <u>https://www.ilo.org/secsoc/information-resources/publications-and-</u> tools/Workingpapers/WCMS_383890/lang--en/index.htm
- Schneider, R. M. (2006). Group Bereavement Support for Spouses Who Are Grieving the Loss of a Partner to Cancer. *Social work with groups (New York. 1978)*, *29*(2-3), 259-278. <u>https://doi.org/10.1300/J009v29n02_17</u>
- Scholl, I., LaRussa, A., Hahlweg, P., Kobrin, S., & Elwyn, G. (2018). Organizational- and system-level characteristics that influence implementation of shared decision-making and strategies to address them a scoping review. *Implementation Science*, *13*(1), 40. <u>https://doi.org/10.1186/s13012-018-0731-z</u>
- Schoonenboom, J., & Johnson, R. B. (2017). How to Construct a Mixed Methods Research Design. Kolner Zeitschrift fur Soziologie und Sozialpsychologie, 69(Suppl 2), 107-131. <u>https://doi.org/10.1007/s11577-017-0454-1</u>
- Schulte, A., Majerol, M., & Nadler, J. (2022). *Narrowing the rural-urban health divide; Bringing virtual health to rural communities*. <u>https://www2.deloitte.com/us/en/insights/industry/public-sector/virtual-health-telemedicine-rural-areas.html</u>
- Sealey, M., O'Connor, M., Aoun, S. M., & Breen, L. J. (2015). Exploring barriers to assessment of bereavement risk in palliative care: perspectives of key stakeholders. *BMC palliative care*, 14(1), 49. <u>https://doi.org/10.1186/s12904-015-0046-7</u>
- Semino, E., Demjén, Z., & Koller, V. (2014). 'Good' and 'bad' deaths: Narratives and professional identities in interviews with hospice managers. *Discourse studies*, 16(5), 667-685. <u>https://doi.org/10.1177/1461445614538566</u>

- Senate Standing Committee on Community Affairs. (2007). *Highway to health: better access for rural and remote patients*. <u>https://apo.org.au/node/3377</u>
- Senate Standing Committee on Community Affairs. (2012). *Palliative care in Australia* (ISBN 978-1-74229-707-1).

https://www.aph.gov.au/parliamentary_business/committees/senate/community_affair s/completed_inquiries/2010-13/palliativecare/report/c03

- Seow, H., Bainbridge, D., Brouwers, M., Bryant, D., Tan Toyofuku, S., & Kelley, M. L. (2020).
 Common care practices among effective community-based specialist palliative care teams: a qualitative study. *BMJ supportive & palliative care*, *10*(1), e3-e3.
 https://doi.org/10.1136/bmjspcare-2016-001221
- Seow, H., Sutradhar, R., Burge, F., McGrail, K., Guthrie, D., Lawson, B., Erbas Oz, U., Chan, K., Peacock, S., & Barbera, L. (2021). End-of-life outcomes with or without early palliative care: a propensity score matched, population-based cancer cohort study. BMJ Open, 11(e041432), 1-8. <u>https://doi.org/10.1136/bmjopen-2020-041432</u>
- Shakir, M., & ur Rahman, A. (2022). Conducting Pilot Study In A Qualitative Inquiry: Learning Some Useful Lessons. *Journal of Positive School Psychology*, 6(10), 1620-1624. <u>https://journalppw.com/index.php/jpsp/article/view/13459/8727</u>
- Shao, J., Hu, Z., Li, B., Luo, J., & Xi, J. (2019). A sustainable urban design framework for the suburbanisation of coastal southeastern Australia. *Environmental Science and Pollution Research International*, 26(14), 13931-13947. https://doi.org/10.1007/s11356-019-04879-8
- Shukla, N., Pradhan, B., Dikshit, A., Chakraborty, S., & Alamri, A. M. (2020). A Review of Models Used for Investigating Barriers to Healthcare Access in Australia. *Int J Environ Res Public Health*, 17(11). <u>https://doi.org/10.3390/ijerph17114087</u>
- Simon, D. (2008). Urban environments: Issues on the peri-urban fringe. Annual review of environment and resources, 33(1), 167-185. <u>https://doi.org/10.1146/annurev.environ.33.021407.093240</u> (Annual Review of Environment and Resources)
- Simon, D. (2020). Peri-urbanization. In *The Palgrave Encyclopedia of Urban and Regional Futures* (pp. 1-5). Springer International Publishing. <u>https://doi.org/10.1007/978-3-030-51812-7_192-1</u>
- Simone. (2022). What is a peri-urban area and how are they changing? Retrieved 18 May 2022 from https://blog.id.com.au/2017/population/what-is-a-peri-urban-area-and-how-are-they-changing/
- Simundić, A. M. (2013). Bias in research. *Biochem Med (Zagreb)*, *23*(1), 12-15. <u>https://doi.org/10.11613/bm.2013.003</u>

- Sinclair, S. (2011). Impact of death and dying on the personal lives and practices of palliative and hospice care professionals. *Canadian Medical Association Journal*, *183*(2), 180-187. https://doi.org/10.1503/cmaj.100511
- Siouta, N., Van Beek, K., Van Der Eerden, M. E., Preston, N., Hasselaar, J. G., Hughes, S., Garralda, E., Centeno, C., Csikos, A., Groot, M., Radbruch, L., Payne, S., & Menten, J. (2016). Integrated palliative care in Europe: A qualitative systematic literature review of empirically-tested models in cancer and chronic disease. *BMC palliative care*, *15*(1), 56-56. <u>https://doi.org/10.1186/s12904-016-0130-7</u>
- Slater, L. (2006). Person-centredness: A concept analysis. Contemporary nurse : a journal for the Australian nursing profession, 23(1), 135-144. <u>https://doi.org/10.5172/conu.2006.23.1.135</u>
- Slavko, B., Glavatskiy, K., & Prokopenko, M. (2020). City structure shapes directional resettlement flows in Australia. *Scientific Reports*, *10*(1), 8235-8235. <u>https://doi.org/10.1038/s41598-020-65208-5</u>
- Slocum-Gori, S., Hemsworth, D., Chan, W. W., Carson, A., & Kazanjian, A. (2013). Understanding Compassion Satisfaction, Compassion Fatigue and Burnout: A survey of the hospice palliative care workforce. *Palliative Medicine*, *27*(2), 172-178. <u>https://doi.org/10.1177/0269216311431311</u>
- Smith, K. B., Humphreys, J. S., & Wilson, M. G. A. (2008). Addressing the health disadvantage of rural populations: How does epidemiological evidence inform rural health policies and research? *The Australian journal of rural health*, *16*(2), 56-66. <u>https://doi.org/10.1111/j.1440-1584.2008.00953.x</u>
- Snoswell, C. L., Taylor, M. L., Comans, T. A., Smith, A. C., Gray, L. C., & Caffery, L. J. (2020). Determining if Telehealth Can Reduce Health System Costs: Scoping Review. J Med Internet Res, 22(10), e17298. <u>https://doi.org/10.2196/17298</u>
- Sørstrøm, A. K., Ludvigsen, M. S., & Kymre, I. G. (2023). Home care nurses facilitating planned home deaths. A focused ethnography. *BMC palliative care*, *22*(1), 175. <u>https://doi.org/10.1186/s12904-023-01303-4</u>
- South Australian Productivity Commission. (2019). *Inquiry into local government costs and efficiency*. Government of South Australia. Retrieved 10 April 2023 from https://www.sapc.sa.gov.au/inquiries/inquiries/inquiries/local-government-inquiry/final-report
- Spelten, E., Timmis, J., Heald, S., & Duijts, S. F. A. (2019). Rural palliative care to support dying at home can be realised; experiences of family members and nurses with a new model of care. *The Australian journal of rural health*, *27*(4), 336-343. https://doi.org/10.1111/ajr.12518
- Spilsbury, K., & Rosenwax, L. (2017). Community-based specialist palliative care is associated with reduced hospital costs for people with non-cancer conditions during

the last year of life. *BMC palliative care*, *16*(1), 68-68. <u>https://doi.org/10.1186/s12904-017-0256-2</u>

- St Clair, M., & Murtagh, D. (2019). Barriers to telehealth uptake in rural, regional, remote Australia: What can be done to expand telehealth access in remote areas? *Studies in Health Technology and Informatics*, 266, 174-182. https://doi.org/10.3233/SHTI190791
- State Planning Commission. (2020, 20th June 2023). *Rural Living: Fact Sheet*. State Government of South Australia. <u>https://plan.sa.gov.au/__data/assets/pdf_file/0003/587442/Fact_Sheet_-</u> Rural_Living.pdf
- Stebbins, R. A. (2001). Exploratory research in the social sciences: what is exploration? *Exploratory Research in the Social Sciences*, 2-18. <u>https://www.researchgate.net/publication/309967516_Exploratory_research_in_the_s</u> <u>ocial_sciences_what_is_exploration</u>
- Steele, R., & Davies, B. (2015). Supporting Families in Palliative Care. In N. Coyle & B. R. Ferrell (Eds.), *Social Aspects of Care* (pp. 0). Oxford University Press. <u>https://doi.org/10.1093/med/9780190244132.003.0003</u>
- Stjernswärd, J. (2007). Palliative Care: The Public Health Strategy. *Journal of Public Health Policy*, 28(1), 42-55. <u>https://doi.org/10.1057/palgrave.jphp.3200115</u>
- Stockdill, M. L., Barnett, M. D., Taylor, R. A., Dionne-Odom, J. N., & Bakitas, M. (2021, 2021/02//). Telehealth in Palliative Care: Communication strategies from the COVID-19 pandemic. *Clinical journal of oncology nursing*, 25(1), 17+. https://link.gale.com/apps/doc/A650494079/AONE?u=james_cook&sid=bookmark-AONE&xid=f22b8feb
- Strunk, K. K., & Mwavita, M. (2021). *Design and analysis in educational research using Jamovi : ANOVA designs*. Routledge. <u>https://www-taylorfrancis-</u> <u>com.elibrary.jcu.edu.au/books/mono/10.4324/9781003154297/design-analysis-</u> <u>educational-research-using-jamovi-kamden-strunk-mwarumba-mwavita</u>
- Swarte, N. B., van der Lee, M. L., van der Bom, J. G., van den Bout, J., & Heintz, A. P. (2003). Effects of euthanasia on the bereaved family and friends: a cross sectional study. *BMJ*, 327(7408), 189. <u>https://doi.org/10.1136/bmj.327.7408.189</u>
- Swerissen, H., & Duckett, S. J. (2015). What can we do to help Australians die the way they want to? *Medical Journal of Australia*, *202*(1), 10-11. https://doi.org/10.5694/mja14.01580
- Sypniewska, B., Baran, M., & Kłos, M. (2023). Work engagement and employee satisfaction in the practice of sustainable human resource management – based on the study of

Polish employees. *International Entrepreneurship and Management Journal*, *19*(3), 1069-1100. <u>https://doi.org/10.1007/s11365-023-00834-9</u>

- Szumer, R. T. O., & Arnold, M. (2023). The Ethics of Overlapping Relationships in Rural and Remote Healthcare. A Narrative Review. *J Bioeth Inq*, *20*(2), 181-190. https://doi.org/10.1007/s11673-023-10243-w
- Taherdoost, H. (2022). What are Different Research Approaches? Comprehensive Review of Qualitative, Quantitative, and Mixed Method Research, Their Applications, Types, and Limitations. *IDEAS Working Paper Series from RePEc*. https://doi.org/10.30564/jmser.v5i1.4538
- Tait, P., Horwood, C., Hakendorf, P., & To, T. (2020). Improving community access to terminal phase medicines through the implementation of a 'Core Medicines List' in South Australian community pharmacies. *BMJ supportive & palliative care*, *10*(1), e4-e4. <u>https://doi.org/10.1136/bmjspcare-2016-001191</u>
- Tan, A. C. W., Emmerton, L. M., & Hattingh, H. L. (2012). Issues with medication supply and management in a rural community in Queensland. *Australian Journal of Rural Health*, 20(3), 138-143. <u>https://doi.org/https://doi.org/10.1111/j.1440-1584.2012.01269.x</u>
- Tanner, E. C., Vann, R. J., & Kizilova, E. (2020). Consumer-Level Perceived Access to Health Services and Its Effects on Vulnerability and Health Outcomes. *Journal of public policy & marketing*, 39(2), 240-255. <u>https://doi.org/10.1177/0743915620903299</u>
- Taylor, A., Caffery, L. J., Gesesew, H. A., King, A., Bassal, A.-R., Ford, K., Kealey, J., Maeder, A., McGuirk, M., Parkes, D., & Ward, P. R. (2021a). How Australian Health Care Services Adapted to Telehealth During the COVID-19 Pandemic: A Survey of Telehealth Professionals. *Frontiers in public health*, *9*, 648009-648009. https://doi.org/10.3389/fpubh.2021.648009
- Taylor, D., Lange, J., Laurence, C., Beilby, J., Kitson, A., Barrie, H., & Visvanathan, R. (2021b). General practice access in regional and remote Australia for ageing populations. *Geographical Research*, *59*(1), 6-15. <u>https://doi.org/10.1111/1745-5871.12447</u>
- Taylor, E., Butt, A., & Amati, M. (2017). Making the Blood Broil: Conflicts Over Imagined Rurality in Peri-Urban Australia. *Planning, practice & research*, 32(1), 85-102. <u>https://doi.org/10.1080/02697459.2015.1028252</u>
- Terry, D., Le, Q., Nguyen, U., & Hoang, H. (2015). Workplace health and safety issues among community nurses: a study regarding the impact on providing care to rural consumers. *BMJ Open*, 5(8), e008306. <u>https://doi.org/10.1136/bmjopen-2015-008306</u>
- The Committee for Sydney. (2017). *The future of Australian cities: Submission to the Standing Committee on Infrastructure, Transport and Cities*. Retrieved 19 July 2022 from <u>http://www.aph.gov.au</u>>documentstore

The Senate Community Affairs Committee Secretariat. (2012). Chapter 3 - The federal funding of palliative care in Australia. In *Palliative Care in Australia*. The Senate Community Affairs References Committee.

https://www.aph.gov.au/parliamentary_business/committees/senate/community_affair s/completed inquiries/2010-13/palliativecare/report/index

- Thomas, J., Barraket, J., Wilson, C., Holcombe-James, I. K., Jenny, Rennie, E., Ewing, S., & MacDonald, T. (2020). *Measuring Australia's digital divide: the Australian digital inclusion index 2020*. RMIT University. <u>https://doi.org/10.25916/5f6eb9949c832</u>
- Thomas, S. L., Wakerman, J., & Humphreys, J. S. (2015). Ensuring equity of access to primary health care in rural and remote Australia what core services should be locally available? *International Journal for Equity in Health*, *14*(1), 111-111. <u>https://doi.org/10.1186/s12939-015-0228-1</u>
- Thompson, C., Williams, K., Masso, M., Morris, D., Kobel, C., Duncan, K., Samsa, P. (2017). Research into services and needs for people experiencing complicated grief: Final report. *Australian Health Services Research Institute*, 903. Retrieved 8 October 2022, from <u>https://ro.uow.edu.au/ahsri/903</u>
- Thomson, K., Williams, L., & Malam, K. (2019). An Introduction to where Australians Live. Department of Infrastructure, Regional Development and Cities. Retrieved 2 August 2022 from <u>https://www.bitre.gov.au/publications/2019/is_96</u>
- Thornton, L. E., Schroers, R.-D., Lamb, K. E., Daniel, M., Ball, K., Chaix, B., Kestens, Y., Best, K., Oostenbach, L., & Coffee, N. T. (2022). Operationalising the 20-minute neighbourhood. *International Journal of Behavioral Nutrition and Physical Activity*, 19(1), 15. <u>https://doi.org/10.1186/s12966-021-01243-3</u>
- Tieman, J. J., Swetenham, K., Morgan, D. D., To, T. H., & Currow, D. C. (2016). Using telehealth to support end of life care in the community: a feasibility study. *BMC palliative care*, *15*(1), 1-7. <u>https://doi.org/10.1186/s12904-016-0167-7</u>
- Urban Development Institute of Australia (UDIA). (2023). UDIA National Residential Greenfield and Apartment Market Study: State of the Land 2023. UDIA. Retrieved 20 November 2023 from https://udia.com.au/research/udia-state-of-the-land-2023/
- Verhoef, M. J., & Casebeer, A. L. (1997). Broadening horizons: Integrating quantitative and qualitative research. *Can J Infect Dis*, 8(2), 65-66. https://doi.org/10.1155/1997/349145
- Virdun, C., Luckett, T., Lorenz, K., Davidson, P. M., & Phillips, J. (2020). Hospital patients' perspectives on what is essential to enable optimal palliative care: A qualitative study. *Palliative Medicine*, 34(10), 1402-1415. <u>https://doi.org/10.177/0269216320947570</u>
- WA Country Health Service (WACHS). (2021). *Medication Safety for Carers of Palliative Care Patients at Home Procedure*. Government of Western Australia.

https://www.wacountry.health.wa.gov.au/~/media/WACHS/Documents/Aboutus/Policies/Medication-Safety-for-Carers-of-Palliative-Care-Patients-at-Home-Procedure.pdf?thn=0

- Waeraas, A. (2022). Thematic Analysis: Making Values Emerge from Texts. In G. Espedal, B. Jelstad Løvaas, S. Sirris, & A. Wæraas (Eds.), *Researching Values: Methodological Approaches for Understanding Values Work in Organisations and Leadership* (pp. 153-170). Springer International Publishing. <u>https://doi.org/10.1007/978-3-030-90769-3_9</u>
- Wakerman, J., Humphreys, J. S., Wells, R., Kuipers, P., Entwistle, P., & Jones, J. (2008).
 Primary health care delivery models in rural and remote Australia a systematic review. *BMC health services research*, 8(1), 276. <u>https://doi.org/10.1186/1472-6963-8-276</u>
- Wall, S. (2015). Focused Ethnography: A Methodological Adaptation for Social Research in Emerging Contexts. Forum, qualitative social research, 16(1). <u>https://doi.org/10.17169/fqs-16.1.2182</u>
- Walters, P. (2021). The Vanishing Suburban Dream in Australia. *disP The Planning Review*, 57(3), 4-21. <u>https://doi.org/10.1080/02513625.2021.2026646</u>
- Wang, F. (2020). Why Public Health Needs GIS: A Methodological Overview. *Ann GIS*, 26(1), 1-12. <u>https://doi.org/10.1080/19475683.2019.1702099</u>
- Wang, T., Molassiotis, A., Chung, B. P. M., & Tan, J. Y. (2018). Unmet care needs of advanced cancer patients and their informal caregivers: a systematic review. *BMC palliative care*, *17*(1), 1-29. <u>https://doi.org/10.1186/s12904-018-0346-9</u>
- Warr, D., Luscombe, G., & Couch, D. (2021). Hype, evidence gaps and digital divides: Telehealth blind spots in rural Australia. *Health (London, England : 1997)*, 136345932110607-13634593211060763.

https://doi.org/10.1177/13634593211060763

- Webster, D. R., & Muller, L. (2009). Peri-Urbanization : Zones of Rural-Urban Transition. In S. Sassan (Ed.), *Human settlement development* (Vol. 1, pp. 280-309).
- Webster, J., & Kristjanson, L. J. (2002). "But isn't it depressing?" The Vitality of Palliative Care. *Journal of Palliative Care*, *18*(1), 15-24. https://doi.org/10.1177/082585970201800104
- Webster, R. (1985). Palliative care for cancer patients. *Medical Journal of Australia*, 143(12-13), 547-550. <u>https://doi.org/https://doi.org/10.5694/j.1326-5377.1985.tb119947.x</u>
- Weng, K., Shearer, J., & Grangaard Johnson, L. (2021). Developing Successful Palliative Care Teams in Rural Communities: A Facilitated Process. *Journal of Palliative Medicine*, 0(0), null. <u>https://doi.org/10.1089/jpm.2021.0287</u>

- Weng, K., Shearer, J., & Grangaard Johnson, L. (2022). Developing Successful Palliative Care Teams in Rural Communities: A Facilitated Process. *J Palliat Med*, 25(5), 734-741. <u>https://doi.org/10.1089/jpm.2021.0287</u>
- Wenham, S., Cumming, M., & Saurman, E. (2020). Improving palliative and end-of-life care for rural and remote Australians. *Public Health Research & Practice*, *30*(1). <u>https://doi.org/10.17061/phrp3012001</u>
- White, B. P., Willmott, L., Close, E., Hewitt, J., Meehan, R., Greaves, L. L., Parker, M. H., & Yates, P. (2021). Development of Voluntary Assisted Dying Training in Victoria, Australia: A Model for Consideration. *Journal of Palliative Care*, *36*(3), 162-167. <u>https://doi.org/10.1177/0825859720946897</u>
- Wilkes, L. M., & Beale, B. (2001). Palliative care at home: Stress for nurses in urban and rural New South Wales, Australia. *International journal of nursing practice*, 7(5), 306-313. https://doi.org/10.1046/j.1440-172X.2001.00267.x
- Wilson, D. M., Underwood, L., & Errasti-Ibarrondo, B. (2021). A scoping research literature review to map the evidence on grief triggers. *Social science & medicine (1982)*, 282, 114109-114109. https://doi.org/10.1016/j.socscimed.2021.114109
- Wilson, F., Ingleton, C., Gott, M., & Gardiner, C. (2014). Autonomy and choice in palliative care: time for a new model? *Journal of advanced nursing*, *70*(5), 1020-1029. <u>https://doi.org/https://doi.org/10.1111/jan.12267</u>
- Wolff, S., Mdemu, M., & Lakes, T. (2021). Defining the Peri-Urban: A Multidimensional Characterization of Spatio-Temporal Land Use along an Urban–Rural Gradient in Dar es Salaam, Tanzania. *Land*, *10*, 177. <u>https://doi.org/10.3390/land10020177</u>
- Wollondilly Shire Council. (2022). *Sydney Peri Urban Network (SPUN)*. Wollondilly Shire Council. Retrieved 1 March 2022 from <u>https://www.wollondilly.nsw.gov.au/shire-projects/growth-in-wollondilly/sydney-peri-uban-network-spun/</u>
- Woo, J. A., Maytal, G., & Stern, T. A. (2006). Clinical Challenges to the Delivery of End-of-Life Care. *Prim Care Companion J Clin Psychiatry*, 8(6), 367-372. <u>https://doi.org/10.4088/pcc.v08n0608</u>
- Wood, S. M., Alston, L., Beks, H., Mc Namara, K., Coffee, N. T., Clark, R. A., Wong Shee, A., & Versace, V. L. (2023). The application of spatial measures to analyse health service accessibility in Australia: a systematic review and recommendations for future practice [24 June 2023]. *BMC health services research*, 23(1), 330. https://doi.org/10.1186/s12913-023-09342-6
- Woodman, C., Baillie, J., & Sivell, S. (2016). The preferences and perspectives of family caregivers towards place of care for their relatives at the end-of-life. A systematic review and thematic synthesis of the qualitative evidence. *BMJ supportive & palliative care*, 6(4), 418-429. <u>https://doi.org/10.1136/bmjspcare-2014-000794</u>

- Woods, D. (2001). Models of care for rural palliative care services. 6th National Rural Health Conference, Canberra.
- World Health Organization (WHO). (1987). *The Ottawa Charter for Health Promotion*. WHO. Retrieved 18 February 2024 from <u>https://iris.who.int/bitstream/handle/10665/53166/WH-1987-May-p16-17-</u> eng.pdf?sequence=1
- World Health Organization (WHO). (2013). Strengthening of palliative care as a component of comprehensive care throughout the life course. *Provisional Agenda Item* (2 August 2022). Retrieved 2 August 2022, from https://apps.who.int/gb/ebwha/pdf files/EB134/B134_28-en.pdf
- World Health Organization (WHO). (2020a). *Palliative Care*. World Health Organization. Retrieved 29th April 2023 from <u>https://www.who.int/news-room/fact-sheets/detail/palliative-care</u>
- World Health Organization (WHO). (2020b). *The true death toll of COVID-19: Estimating global excess mortality*. Retrieved 5 November 2023 from https://www.who.int/data/stories/the-true-death-toll-of-covid-19-estimating-global-excess-mortality
- World Health Organization (WHO). (2023). *Coronavirus COVID-19 Dashboard*. Retrieved 5 November 2023 from <u>https://covid19.who.int/region/wpro/country/au</u>
- World Health Organization [WHO]. (2021). Assessing the development of palliative care worldwide: a set of actionable indicators. World Health Organization. <u>https://apps.who.int/iris/handle/10665/345532</u>
- Wynne, L., Ruoso, L.-E., Cordell, D., & Jacobs, B. (2020). 'Locationally disadvantaged': planning governmentalities and peri-urban agricultural futures. *Australian Geographer*, 51(3), 377-397. <u>https://doi.org/10.1080/00049182.2020.1790134</u>
- Yu, J., Soh, K. L., He, L., Wang, P., Soh, K. G., & Cao, Y. (2023). The Experiences and Needs of Hospice Care Nurses Facing Burnout: A Scoping Review. *American Journal* of Hospice and Palliative Medicine®, 40(9), 1029-1039. https://doi.org/10.1177/10499091221141063
- Zolkefli, Y. (2017). Evaluating the Concept of Choice in Healthcare. *Malays J Med Sci*, 24(6), 92-96. <u>https://doi.org/10.21315/mjms2017.24.6.11</u>

Appendices

Appendix A: Geographical Classification

The ABS uses two definitions of Australia

- Geographic Australia, used for social and demographic statistics, and
- Economic Australia, used for economic statistics (ABS, 2016a).

For ASGC purposes, the ABS uses the definition of Australia as set out in section 17(a) of the Acts Interpretation Act 1901 which currently defines Australia or the Commonwealth as meaning:

"...the Commonwealth of Australia and, when used in a geographical sense, includes the Territory of Christmas Island and the Territory of Cocos (Keeling) Islands, but does not include any other external Territory." (ABS, 2009a).

The ASGC included the Territories of Christmas Island and Cocos (Keeling) Island into geographic Australia from 1 July 1993, following the Territories Law Reform Act, No. 104, 1992 which amended the Acts Interpretation Act 1901 (ABS, 2009a). The treatment of Jervis Bay Territory also changed in the ASGC from July 1993 with other external territories, including Norfolk Island, remaining excluded (ABS, 2009a). For ASGS purposes, the ABS uses the Geographic definition of Australia, as set out in section 2B of the Acts Interpretation Act 1901 and as amended by the Territories Law Reform Act, No. 104, 1992 (ABS, 2016a). The geographic definition of Australia or the Commonwealth is interpreted as meaning:

"...the Commonwealth of Australia and, when used in a geographical sense, includes Norfolk Island, the Territory of Christmas Island and the Territory of Cocos (Keeling) Islands, but does not include any other external Territory." (ABS, 2016a).

Included in this definition of Geographic Australia are the:

- States of New South Wales, Victoria, Queensland, South Australia, Western Australia, and Tasmania
- Northern Territory
- Australian Capital Territory (ACT)
- Territory of Cocos (Keeling) Islands
- Territory of Christmas Island

- Jervis Bay Territory
- Territory of Norfolk Island (ABS, 2016a)

Macquarie Island is excluded although legally part of Tasmania, as it is extremely isolated and with no permanent population (ABS, 2016a). Any population on the Island for Census of Population and Housing is included in the Migratory – Offshore – Shipping Statistical Area Level 2 for Tasmania (ABS, 2016a). Historically, the Territory of Norfolk Island has been excluded from the definition of Geographic Australia however the definition was updated to include the island following Australian Government reforms in 2015 (ABS, 2016a). Subsequently, the 2016 ASGS included the Territory of Norfolk Island within the 'Other Territories' category along with Jervis Bay Territory, the Territory of Cocos (Keeling) Islands and the Territory of Christmas Island (ABS, 2016a). The economic definition of Australia is not used by the ASGC or ASGS (ABS, 2016a).

Australian Standard Geographical Classification (ASGC)

The Australian Standard Geographical Classification (ASGC) was used by the ABS for the collection and dissemination of geographically classified statistics between 1984 and 2011 (ABS, 2021). The geographical areas that had been used by the ABS prior to 1984 were adopted into the ASGC (ABS, 2011a). The ASGC was designed to meet the needs of users for social, demographic and economic statistics by providing a common framework of statistical geography which enabled production of statistics which were comparable and could be spatially integrated (ABS, 2006a, 2023g).

The ASGC was updated and released annually to update Statistical Local Areas (SLA) and to reflect any changes in gazetted Local Government Authorities (LGA) (ABS, 2006a; 2011b). The ASGC classified statistical units such as households and businesses to a geographical area in one of seven hierarchies of geographical areas, as shown in Figure 21 (ABS, 2006a). Each structure was designed for a different statistical purpose and were as follows:

- Main Structure
- LGA Structure
- Statistical District Structure
- Statistical Region Structure
- Urban Centre/Locality Structure
- Section of State Structure
- Remoteness Structure (ABS, 2006a)



Figure 21 ASGC Structural Chart (ABS, 2009b)

The Main Structure, the Statistical Region Structure, the Section of State Structure, and the Remoteness Structure covered the whole of Australia while the remaining structures covered only parts of Australia (ABS, 2011b). In Census of Population and Housing years all seven structures of the ASGC were defined however, in intercensal years only the first four structures were defined (ABS, 2009a). Classification structures were composed of different numbers of levels, made up of one type of geographical spatial unit (ABS, 2011b). The hierarchy is shown in Figure 22. The spatial units or geographical areas used in ASGC were listed as follows:

- Census Collection District (CD)
- Statistical Local Area (SLA)
- Statistical Subdivision (SSD)
- Statistical Division (SD)
- State and Territory (S/T)
- Statistical District (S Dist.)
- Local Government Area (LGA)
- Statistical Region Sector (SRS)
- Statistical Region (SR)

- Major Statistical Region (MSR)
- Urban Centre/Locality (UC/L)
- Section(s) of State (SOS)
- Remoteness Area (RA) (ABS, 2006a).

ASGC Structure	Hierarchical LevelsComponent Spatial Units	Covers whole of Australia?
Local Government Area	4CD,SLA,LGA,S/T (a)	No
Statistical District	4CD,SLA,SSD,S Dist (b)	No
Main	5CD,SLA,SSD,SD,S/T	Yes
Statistical Region	6CD,SLA,SRS,SR,MSR,S/T	Yes
Section of State	3CD,SOS,S/T	Yes
Remoteness	3CD,RA,S/T	Yes
Urban Centre/Locality	2CD,UC/L (c)	No

Figure 22 Hierarchical Levels ASGC Structure (ABS, 2006a)

In Census of Population and Housing Census years, the smallest spatial unit used to build the classification structures was the Census Collection District (CD) however, in intercensal years the smallest spatial unit was the Statistical Local Area (SLA) (ABS, 2009b). CDs aggregate to form the larger spatial units of SLAs in the Main, Statistical Region, Statistical District and LGAs, SOS in the SOS Structure, Urban Centres and Localities in the Structure and Remoteness Areas in the Remoteness Structure (ABS, 2001a).

The original definition of a CD was an area that one census collector could physically cover, that is deliver and collect census forms, in a period of approximately ten days (ABS, 2001a, 2006b). CDs were created in response to changes within any given area, including significant increase in population or population growth resulting in expansion of locality or urban centre boundary (ABS, 2006b).

Division of CDs into areas of increased population growth to reflect urban growth in Census results is a process referred to as fragmentation (ABS, 2006b). Population in existing CDs which increased to the point that it was not possible for one collector to manage would result in one CD being split into two or more (ABS, 2006b). CDs in urban areas averaged 220 dwellings in 2001 and 225 in 2006 with the number of dwellings per CD declining in rural areas as population densities decreased (ABS, 2001a, 2006b). CD boundaries were designed to not cross SLA or LGA boundaries and adjusted to conform with changes to LGA boundaries (ABS, 2006b). An aggregation of CDs cover the administrative area of a local government entity to enable as much comparability between Censuses as possible (ABS, 2001a, 2006b). CDs are defined for each Census and remain current only at Census time (ABS, 2006b).

ASGC Urban and Rural Classification

Section of State Structure (SOS)

The SOS Structure of the ASGC uses population counts from the Census of Population and Housing to class structures as urban and those that are not urban, as rural (ABS, 2001a). SOS categories of Major Urban and Other Urban are regarded as urban and the categories of Bounded Locality and Rural Balance are regarded as rural (ABS, 2001a). SOS categories are further broken down into different categories called Sections of State Range (SOSR) based on population ranges from the UC/L Structure (ABS, 2009b). The SOS Structure was defined by population data from the Census of Population and Housing, therefore the classification structure was current only at the time of the Census (ABS, 2009b). The SOS categories based on population ranges are:

- Major Urban: provides for three SOSR categories of urban areas
- Other Urban: provides for five SOSR categories of urban areas
- Bounded Locality: provides for two SOSR categories of rural areas
- Rural Balance: equivalent to the SOSR of the remainder of the State/Territory
- Migratory: equivalent to the SOSR of offshore, shipping and migratory (AIHW, 2004; McGrail & Humphreys, 2009)

The following diagram demonstrates the breakdown of SOS categories into SOSR categories.



Figure 24 SOSR Categories (ABS, 2001b; 2009b)

Urban Centre/Locality (UC/L)

The UC/L Structure is a separate but interrelated structure to the SOS Structure in the ASGC and aggregates CDs to form areas with similar characteristics by population (ABS, 2001a). The resulting areas formed by one or more contiguous CDs are called Urban Centres or Localities. The areas within the UC/L Structure refer to smaller geographic areas than the broad categories in SOS Structure in the ASGC (ABS, 2001a). People living in urban centres are classified as urban for statistical purposes and generally live in a population centre with a 'core urban population' of 1,000 or more people (ABS, 2001a). People living in localities are classified as rural for statistical purposes and generally live in a population centre of between 200 and 999 people (ABS, 2001a). However, a Locality may contain a population in excess of 1,000 people which would then meet the criteria for an Urban Centre (ABS, 2001a). Urban Centres/Localities are redefined at each Population Census and are comprised of one or more Census Collection Districts (CD) (ABS, 2001a).

Census Collection District (CD)

- CDs classified as urban include the following:
 - All adjoining CDS with a population density of 200 or more persons per square kilometre are classified as urban. State, SD, LGA and other administrative boundaries will not be included in determining if a CD is included within an Urban Centre.
 - A CD with the majority of the land used for factories, airports, small sports areas, cemeteries, hostels, institutions, prisons, military camps or certain research stations is classified as urban if adjoining urban CDs.
 - A CD with most of the land used for large sporting areas, large parks, explosives handling and munitions areas, or holding yards associated with meatworks and abattoirs is classified as urban only if it is bordered on three sides by CDs classified as urban.
 - Any area which is surrounded by urban CDs must be classified as urban.
 - Where an Urban Centre of 20,000 or more population is separated from another urban area by a gap in urban development of less than three kilometres, the gap shall be classified as an urban CD.
 - Large peripheral CDs in growth areas containing at least one hundred persons, may be fragmented and regarded as CDs (ABS, 2001a).

Remoteness Structure

The final structure in the ASGC is the Remoteness Structure which bases remoteness of an area by population and accessibility to goods and services (ABS, 2009a). The Remoteness Structure aggregates all CDs within Australia with common characteristics into broad geographical regions called Remoteness Areas (RAs) (ABS, 2009a). RAs do not align with other spatial units within the ASGC, therefore the Remoteness Structure is maintained as a separate structure in the ASGC (ABS, 2009a). There are three hierarchical levels of the Remoteness Structure; CDs aggregate to RAs which aggregate to S/Ts and six areas within the structure.

- Major Cities of Australia
- Inner Regional Australia
- Outer Regional Australia
- Remote Australia
- Very Remote Australia
- Migratory (ABS, 2006a)

Regional areas of Australia are those areas that are located outside of major cities where geographic distance restricts accessibility to the widest range of goods, services and opportunities for social interaction (AIHW, 2004). Inner Regional Australia and Outer Regional Australia are categories within the ASGC Remoteness Structure and viewed as Regional Australia (ABS, 2006a; AIHW, 2004).

Population data aggregates CDs to RAs and S/Ts, with accessibility to services measured using the Accessibility and Remoteness Index of Australia (ARIA) (ABS, 2006a).

Accessibility/Remoteness Index of Australia (ARIA)

The Accessibility/Remoteness Index of Australia (ARIA) was developed by the National Key Centre for Social Applications of GIS (GISCA) in 1997 (ABS, 2001b; DoHAC, 2001). ARIA was commissioned to measure remoteness in terms of access and the distance using road networks from any locality to one of four categories of service centres (ABS, 2001b; ALSWH, 2003). Actual road distance is calculated rather than the shorter, straight line distance from the populated locality GPO to the GPO of the nearest service centre in each category (Jones, 2000). ARIA Service Centres are ABD defined urban centres grouped into four categories by population size:

Class A: 250,000 or more

Class B: 48,000 to 249,999

Class C: 18,000 to 47,999

Class D: 5,000 to 17,999 (AIHW, 2004; 2000)

Service Centres are categorised according to a relationship existing between the population size of service centres and the number and range of goods, services and social interaction opportunities accessible (ABS, 2001b). Research during the development of the ARIA, identified that the number, choice and level of services increased with the size of the population at the service centres (ABS, 2001b). However, differences were evident in levels of health services available in service centres of the same category (ABS, 2001b).

The concept of remoteness in the ARIA is the measurement of physical road distance between where people live and the locations where people need to travel to obtain goods, services and social interaction (ABS, 2001b). Socioeconomic factors, time taken to travel and road conditions are not factored into the measurement (ABS, 2001b). The shortest road distance to the mean shortest distance for each category of Service Centre is calculated and capped at three. The five individual values are summed to arrive at a single ARIA Score for the Populated Locality, with the highest total ARIA Score a Populated Locality can have been twelve and the lowest zero as demonstrated in Figure 23.

ARIA score 0 - 1.84	Category - Description Highly accessible <i>Relatively unrestricted accessibility to a wide range of goods and</i> <i>services and opportunities for social interaction</i>
>1.84 - 3.51	Accessible Some restrictions to accessibility of some goods, services and opportunities for social interaction
>3.51 - 5.80	Moderately accessible Significantly restricted accessibility of goods, services and opportunities for social interaction
>5.80 - 9.08	Remote Very restricted accessibility of goods, services and opportunities for social interaction
>9.08 – 12	Very Remote Locationally disadvantaged - very little accessibility of goods, services and opportunities for social interaction

Figure 23 Five categories of remoteness (AIHW, 2004)

Accessibility/Remoteness Index of Australia (ARIA+)

ARIA is continuously developed and in 2003, GISCA released a new version of the continuous varying index of remoteness called ARIA+. The new version continues to be given geographical areas a score based on road distance from populated localities to the nearest Service Centres in categories based on population size. The major difference between ARIA

and ARIA+ is the inclusion of an additional Service Centre category to include all urban centres with populations greater than 1,000 and less than 5,000. This inclusion of a fifth Service Centre category resulted in an index ranging from 0-15 in ARIA+, instead of 0-12 as in ARIA.

There are 5 categories of ARIA+ Service Categories which are detailed in the following figure:



Figure 24 ARIA+ Service Centres by population (ABS, 2001b)

ARIA+ assigns a score between zero and fifteen based on the road distance to service centres. Scores for regions are calculated by averaging scores of 1 km square grid, with index scores classified into various categories. Two of the main categories and their ARIA+ scores are as follows:

Remoteness Areas:

- Major Cities of Australia (ARIA score 0 <=0.20) relatively unrestricted accessibility to a wide range of goods, services, and opportunities for social interaction,
- Inner Regional Australia (ARIA score 0.20 <=2.40) some restrictions to accessibility to some goods, services, and opportunities for social interaction,
- Outer Regional Australia (ARIA score 2.40 <=5.92) significantly restricted accessibility to goods, services, and opportunities for social interaction,
- Remote Australia (ARIA score 5.92 <=10.53) very restricted accessibility to goods, services, and opportunities for social interaction,

 Very Remote Australia (ARIA score 10.53 <=15) – very little accessibility to goods, services, and opportunities for social interaction.

Remoteness Classes:

- Highly Accessible (ARIA score 0 <=0.20) relatively unrestricted accessibility to a wide range of goods, services, and opportunities for social interaction,
- Accessible (ARIA score 0.20 <=2.40) some restrictions to accessibility to some goods, services, and opportunities for social interaction,
- Moderately Accessible (ARIA score 2.40 <=5.92) significantly restricted accessibility to goods, services, and opportunities for social interaction,
- Remote (ARIA score 5.92 <=10.53) very restricted accessibility to goods, services, and opportunities for social interaction,
- Very Remote (ARIA score 10.53 <=15) very little accessibility to goods, services, and opportunities for social interaction.

In ARIA, distance was measured from populated locality GPO to the GPO of the nearest Service Centre. In ARIA+, distance is measured from populated locality to the edge of the builtup area of the nearest service centre per the ABS UC/L boundary. The AIHW believes that this change models accessibility at the urban fringe more effectively as it reduces the distance to the nearest service centre, particularly in major service centres with urban sprawl.

In summary, ARIA, and ARIA+ define remoteness in terms of a measure of relative access to a particular service. This definition uses the average distance of a location from a service centre, which assumes the greater the distance, the more remote the area and the less opportunities for social interaction and supply of goods and services. Figure 25 provides a visual of remoteness areas. Accessibility becomes questionable when the requirement for a service centre is that it contains at least one identified service, usually a government primary school or an Australian Post Office with no requirement for specialist health services. Defining accessibility more broadly would lead to consideration of the dimensions of accessibility being affordability, availability, accommodation, acceptability, and appropriateness and the dimensions of abilities. The abilities of the population are interrelated with dimensions of access, to realise access.

ARIA and ARIA+ define remoteness on two elements and do not consider the complexities involved in access to services. Complexities include the location of the individual, the infrastructure involved in the distance, the service involved and its affordability and appropriateness in the situation.



Figure 25 Map of 2016 Remoteness Areas of Australia (ABS, 2017, 2018a)

Australian Statistical Geography Standard (ASGS)

The ASGS is a classification of the whole of Australia into a hierarchical framework of statistical areas. The framework brings all the regions which ABS collects, disseminates and publishes geographically classified statistics, within the one framework (ABS, 2022a). The ASGS is a social geography, reflecting the location of people and communities and updated every five years to account for growth and change (ABS, 2016b). The ASGS replaced the ASGC in 2011 following a review of the ASGC (ABS, 2011a). The aim of the review was to create a new Australian statistical geography framework to both address some of the identified shortcomings of the ASGC and to meet the future needs of users (ABS, 2011a).

Initially adopted for the 2011 Census of Population and Housing and shown in Figure 26, the ASGC and the ASGS both operated in 2011, however the ASGS has been the sole ABS statistical geography framework since 2012 (ABS, 2023a). The ASGS was developed to be a more comprehensive, flexible and consistent way of defining Australia's statistical

geography than the previous ASGC framework (ABS, 2011a). It was expected that the new classification framework would lead to improvements in the quality of small area data and enable better comparison of data from the Census and other sources adopting the ASGS (ABS, 2012). The development of a statistical geography to better meet the needs of users resulted in changes including the replacement of spatial units used in the ASGC such as CDs, SLAs and SDs (ABS, 2016d). Due to potential issues with time series comparisons, 2011 Census data was available for both ASGS and SLAs from 2011 ASGC (ABS, 2016d).



Figure 26 Comparison main structures by spatial units (ABS, 2012)

In summary, the ASGS took over from the ASGC in the collection and dissemination of geographical statistical data used to classify locations in Australia as urban or non-urban. The ASGS is the common boundary set for analysis, display and recoding of socioeconomic information within the Statistical Framework for Australia, providing a consistent, hierarchical set of geographies (ABS, 2020). The core ASGS boundary hierarchies include approximately equal population numbers within each area to enable comparisons and to aggregate hierarchies to define areas as urban or non-urban (rural). The boundaries range in size from small geographical areas to larger areas, but small areas are designed to ensure a size large enough to provide privacy and confidentiality to residents (ABS, 2020).

The smallest spatial unit used by health services is a SA2 with a population of 3,000 to 25,000 people. Challenges to this size of geographical area becomes evident in peri-urban areas due to the diversity of population and housing density that may be evident within each SA2. Definitions of Other Urban and Bounded Locality as urban or rural are too simplistic for use in peri-urban areas and fail to capture the needs of the community without additional local input.

ABS and Non-ABS Structures

Development of the ASGS involved the inclusion of all regions used by the ABS to output data under the one statistical framework, which is divided into two broad categories, as demonstrated in Figure 27:

- ABS Structures regions defined and maintained by the ABS
- Non-ABS Structures regions for which ABS supplies statistical data but defined and maintained by other organisations (ABS, 2016c; 2016d).



Figure 27 Interrelatedness ABS and Non-ABS Structures (ABS, 2022a)

ABS Structures

ABS Structures comprise six interrelated hierarchies of regions, consistent in population size and designed to remain stable between Censuses (as shown in Figure 28).



Figure 28 ASGS ABS Structures (ABS, 2016c)

Main Structure

The structure has seven hierarchical levels comprising in ascending order: Mesh Block (MB), Statistical Area Level 1 (SA1), Statistical Area Level 2 (SA2), Statistical Area Level 3 (SA3), Statistical Area Level 4 (SA4), State and Territory (S/T), and Australia (AUS) (ABS, 2016a).

- Mesh Blocks (MBs) are the smallest geographical areas defined by the ABS and aggregate to form the building blocks for the larger regions of the ASGS (ABS, 2016c; 2016d). A MB contains an average of thirty dwellings and are both urban and rural in nature (ABS, 2016a; 2023g). The purpose of using Mesh Blocks is to support more accurate statistics for commonly used regions including postcodes and suburbs and to reflect the location of people and communities in Australia (ABS, 2016a).
- Statistical Areas Level 1 (SA1s) were adopted in the 2011 Census as the new base unit of output geography with SA1s being smaller than CDS (ABS, 2012). SA1s are the second smallest geographic area defined by the ABS and are composed of aggregates of building block geographical units called a Mesh Block (MB) (ABS,

2016d). SA1s are designed to be consistent in population size and either urban or rural in character, with a population between 200 and 800 persons or an average of 400 persons (ABS, 2016d).

- SA1s replace CDs which consisted of approximately 200 dwellings and were primarily designed to enable one Census collector to deliver and retrieve Census forms (ABS, 2016d). This resulted in geographic areas that were inconsistent in population size and character (ABS, 2016d).
- Statistical Areas Level 2 (SA2s) consist of one or more SA1s and are designed to represent functional areas outside of major urban and regional Australia (ABS, 2016d). SA2s are based on gazetted suburbs and localities with an average population 10,000 persons (ABS, 2016d). SA2s are consistent in population and are designed to be stable over several Censuses (ABS, 2016d).
- SA2s replace SLAs which were inconsistent in population size and do not reflect functional areas or settlement patterns (ABS, 2006a). SLAs were subject to change following review of gazetted LGAs as they are contained within LGAs (ABS, 2006a).
- Statistical Areas Level 3 (SA3s) represent regions between approximately 15,000 to 150,000 people, built from aggregations of whole SA2s (ABS, 2016d). SA3s represent a combination of informal regions, both rural and urban and fit within SA4 boundaries (ABS, 2016d).
- Statistical Areas Level 4 (SA4s) are built from whole SA3s and have large populations over 100,000 people and fit within State/Territory boundaries (ABS, 2016d). In regional areas, SA4s have populations closer to the minimum (100,000 to 300,000 people), in metropolitan areas, the SA4s tend to have larger populations (300,000 to 500,000 people) (ABS, 2016d).
- **States and Territories (S/Ts)** are the largest spatial units in the ASGS Main Structure and have fixed boundaries due to being politically governed entities (ABS, 2016c). S/Ts consist of one or more SA4s and cover Australia without gaps or overlaps (ABS, 2016c).
- **Australia** For ABS Structure purposes, the ABS uses the Geographic Australia definition of Australia as set out in section 2B of the Acts Interpretation Act 1901 and as amended by the Territories Law Reform Act, No. 104, 1992 (ABS, 2016a).
- Indigenous Structure
 - Indigenous Regions (IREGs) are large geographical units based on Aboriginal and Torres Strait Islander Commission boundaries and consisting of one or more Indigenous Areas (IREAs) (ABS, 2016c). IREGs do not cross state and territory borders and aggregate to S/T level (ABS, 2016c).

- Indigenous Areas (IREAs) are geographical units consisting of one or more Indigenous Locations (ILOCs) and generally have a minimum of 250 Aboriginal and Torres Strait Islander residents (ABS, 2016c).
- Indigenous Locations (ILOCs) aggregate to IAREs and consist of one or more SA1s and identify data on particular Aboriginal and Torres Strait Islander communities with a minimum population of 90 usual residents (ABS, 2016c).
- Greater Capital City Statistical Area Structure (GCCSAs) replace the ASGC Capital City Statistical Divisions and provide a stable geographical area to represent each of the eight state and territory capital cities (ABS, 2016a; 2016c). GCCSAs are built up from whole SA4s with the areas not defined as included in the greater capital city, represented by a Rest of State region (ABS, 2016a).

• Significant Urban Area Structure

Significant Urban Areas (SUAs) represent Urban Centres (ABS, 2022a). They are areas based on Urban Centres and Localities (UCLs) and built from Statistical Areas Level 2 (SA2) (ABS, 2022a). SUAs represent an Urban Centre or cluster of Urban Centres that contain the urban population and closely associated populations of major urban concentrations of over 10,000 people (ABS, 2016a). SUAs may contain one or more associated Urban Centre and/or Locality and areas in between as they are designed to include projected growth patterns (ABS, 2016a). SUAs can cross State/Territory boundaries (ABS, 2022a).

Areas defined as SUAs combine one or more adjacent SA2s based on the following criteria:

- SA2s containing an Urban Centre of 10,000 persons or more are considered SUAs
- Other SA2s containing Urban Centres within a five-kilometre distance via sealed road of large Urban Centres may be aggregated into the same SUA
- SUAs containing Urban Centres of 7,000 persons or more may be considered a SUA if there are other Urban Centres within five-kilometre distance via sealed road and the Urban Centres have a combined population over 10,000
- SA2s included in an SUA should be within the same Labour Market as defined by GCCSA classification (ABS, 2022a).

• Urban Centres and Localities/Section of State Structures

The 2011 Urban Centres and Localities and Section of State are broadly comparable to ASGC Urban Centres, Bounded Localities and Rural Balance (ABS, 2016d). Section of State Structures and Urban Centres and Localities are built from SA1s in the ASGC rather than Census Collection Districts, used in ASGC (ABS, 2022a).

- Urban Centre and Locality (UCL) classification is aggregated from SA1s which meet population density criteria or contain urban infrastructure (ABS, 2022a). Areas not within Urban Centres or Localities are labelled as Remainder of State/Territory (ABS, 2022a). Urban Centres are areas of concentrated urban populations that are constantly changing by growing, absorbing nearby centres or declining (ABS, 2022a).
- Section of State (SOS) groups the UCLs into four classes of urban areas based on population size, with the remainder considered to be rural (ABS, 2022a). The classes are an aggregation of the more detailed SOSR classification, with SOS regions covering the whole of Australia.

Remoteness Area Structure

 Remoteness Areas (RAs) comprise the six categories identifying a region in Australia used in the ASGC ranging from Major Cities (highly accessible) to Very Remote (very little accessibility) (ABS, 2022a). RAs are built up from SA1s which share a degree of remoteness which is determined using the ARIA or ARIA+ (ABS, 2022a).

The AGS builds Remoteness Areas from SA1s which share a degree of remoteness rather than Census Collection Districts used in the ASGC (ABS, 2022a).

 Section of State Range (SOSR) provides a more detailed classification than SOS to enable statistical comparison of differently sized Urban Centres and rural areas (ABS, 2022a).

Non-ABS Structures

The Non ABS Structures are hierarchies of regions which are not defined or maintained by the ABS, but for which the ABS provides a range of statistical data (ABS, 2022a) (refer to Figure 29). The ABS approximates these regions by Mesh Blocks (MBs), Statistical Areas Level 1 (SA1s) or Statistical Areas Level 2 (SA2s) from the main structure of the ASGS (ABS, 2022a). The use of whole geographical units enables the ABS to release data which represents commonly used administrative and environmental units such as Postcode and LGA (ABS, 2016e).



Figure 29 ASGS Non-ABS Structures (ABS, 2016a)

LGAs, Commonwealth Electoral Divisions and State Electoral Divisions are updated annually where redistributions have occurred with the remainder of the structures updated every 5 years (ABS, 2016e).

Defining Urban and Rural Classifications in ASGS

The ASGS defines and classifies areas as urban in several different ways to make a wide range of statistical geographical data available.

Urban Centres and Localities (UCLs)

Urban Centres represent areas of concentrated urban development with populations of 200 people or more. UCLs are defined using Statistical Areas Level 1 (SA1s) that meet the criteria of Urban Character.

UCLs are not official definitions of towns, with small towns less than 200 people and other urban areas represented by State Government Gazetted Localities. To enable comparisons, medium towns are those with populations of 10,000 to 50,000, large towns have populations of 50,000 to 100,000 and major cities are those with populations of 100,000 or more. Census data on these localities are provided through the State Suburbs (SSCs) which are part of the ASGC non-ABS Structures.

For statistical analysis, people living in areas defined as Urban Centres are viewed as urban, areas not defined by Urban Centres or Localities are considered to be 'rural'. UCLs include a category of Rural Balance which ensures that they cover the whole of Australia. Statistical data for Urban Centres are updated every Census and ABS does not provide population projections for Urban Centres, therefore data may be five years out of date.

SA1s are considered as urban using population and dwelling criteria based on 'place of usual residence' from Census of Population and Housing. Mesh Blocks (MBs) are the building blocks of SA1s and an Urban Mesh Block is defined as a MB with a population density of 200 persons or more per square kilometre. Criteria for SA1s to be considered as urban are as follows:

- Have an Urban Mesh Block (MB) population greater or equal to 45% of the total population and a dwelling density greater or equal to 45 dwellings per square kilometre, or
- Have a population density greater or equal to 100 persons per square kilometre and a dwelling density greater or equal to 50 dwellings per square kilometre, or
- Have a population density greater or equal to 200 persons per square kilometre

SA1s adjacent to SA1s classified as 'urban' that contain substantial 'urban infrastructure/ land use' and those located within 500 metres road distance, are also considered to be 'urban' SA1s. A cluster of SA1s classified as urban with a total population of 1,000 persons or more is defined as a separate Urban Centre. Clusters with total populations between 200 and 999 are considered Localities.

Localities

Localities represent configurations of small settlements, including villages, towns, clusters of peri-urban development and areas with significant tourism, with populations between 200 and 1,000 people. Localities are defined by grouping together adjacent SA1s that meet the following 'locality' criteria and for statistical purposes, people living in Localities are viewed as rural.

SA1s are considered to meet the 'locality' criteria if:

- They are not included in an Urban Centre, and
- Have at least one Urban Mesh Block, and
- Have a population density greater or equal to 200 persons per square kilometre.

Localities can be defined in any one of the following ways:

- One or more adjacent SA1s that meet the urban criteria with a population between 200 and 1,000 people.
- One or more adjacent 'locality' SA1s with a total population of at least 200 people.
- One or more adjacent SA1s representing a Discrete Aboriginal and Torres Strait Islander community with a total population between 200 and 999 people.
- One or more adjacent SA1s with a total population of at least 200 people in a concentration of housing and a discernible community centre containing community facilities.
- May contain a population exceeding 999 persons if it contains an 'urban' SA1 that does not meet the criteria for an Urban Centre.

Section of State (SOS) and Section of State Range (SOSR)

Section of State (SOS) classifies UCLs into four classes within each State and Territory (S/T) based on population size. The four classes are an aggregation of Section of State Range (SOSR) classifications with the categories being Major Urban, Other Urban, Bounded Locality and Rural Balance (Outlined in Figure 30).

Identifier	Name	Definition	Urban or Rural		
0	Major Urban	Major Urban represents a combination of all Urban Centres with a population of 100,000 or more	Urban		
1	Other Urban	Other Urban represents a combination of all Urban Centres with a population between 1,000 and 99,999	Urban		
2	Bounded Locality (a)	Bounded Localities represents a combination of all Localities	Rural		
3	Rural Balance	Rural Balance represents the Remainder of State/Territory	Rural		

Figure 30 SOS Classification (ABS, 2017)

Section of State Range (SOSR) are aggregates of UCLs and aggregate to SOS to cover the whole of Australia. SOSR provide more details than SOS which enables

statistical comparison of different sized Urban Centres and the remaining rural areas. SOSR classify UCLs into eleven classes within each State and Territory (S/T). Classes are made up of eight population size classes grouping Urban Centres, two population size classes grouping Localities and a Rural Balance (as outlined in Figure 31).

	SOSR Identifiers and Names		
SOSR Identifier	SOSR Name	SOS Name	
01	1 million or more	Major Urban	
02	250,000 to 999,999	Major Urban	
03	100,000 to 249,999	Major Urban	
11	50,000 to 99,999	Other Urban	
12	20,000 to 49,999	Other Urban	
13	10,000 to 19,999	Other Urban	
14	5,000 to 9,999	Other Urban	
15	1,000 to 4,999	Other Urban	
21	500 or more (a)	Bounded Locality	
22	200 or 499	Bounded Locality	
31	Remainder of State / Territory	Rural Balance	

Figure 31 SOSR Classification (ABS, 2017)

Significant Urban Areas (SUAs) represent significant towns and cities of 10,000 people or more and are classified as urban.

Greater Capital City Statistical Areas (GCCSA) provide the broadest urban extent and represent the functional extent of each of the eight State and Territory capital cities. This classification includes not only the population within the urban city area, but also those who live in rural areas on the fringe of the city and commute. The term urban extent can be measured differently depending on user needs and available data. GCCSA boundaries provide a definition of capital cities and do not align with the builtup edge of the cities. GCCSA are built from SA4s and the area not defined as included in the GCCSA is represented by a Rest of State region.

Rural and remote classifications

Rural, Remote and Metropolitan Area (RRMA)

The Rural, Remote and Metropolitan Area (RRMA) classification is a general purpose tool based on ABS Census of Population and Housing data (as shown in Figure 32) (Department of Primary Industries and Energy and Department of Human Services and Health, 1994). RRMA structure consists of three broad zones, metropolitan, rural, and remote and seven classifications. The RRMA classification allocated each Statistical Local Area (SLA) within capital cities and metropolitan centres (having a population of 100,000 or more) to the Metropolitan zone. All other SLAs were allocated to either the Rural or Remote zone based on the SLA's score on an 'Index of remoteness'. The index score was calculated by combining a personal distance index (relating to the SLA's population density) and distance indices (relating to the distance of the centroid of an SLA to the nearest urban centres in each of four categories). The SLA was then allocated a class (e.g. 'small rural centres') within the zone, based on the population of the urban centre within the SLA (DPIE & DHSH 1994). RRMA classifies SLAs as metropolitan ('capital cities' or 'other metropolitan areas'), rural ('large rural centres', 'small rural centres' and 'other rural areas'), and remote ('remote centres' and 'other remote areas'). The RRMA measure of remoteness is based on population estimates from the 1991 census (Senate Standing Committee on Community Affairs, 2007).



Figure 32 RRMA Structure (Department of Primary Industries and Energy and Department of Human Services and Health, 1994)

In summary the RRMA was a geographical classification of remoteness using Census data to classify areas by population. It has gradually become outdated in health use due to a lack of updates to the population data and is not relevant for use in current classifications of peri-urban locations.

229

Modified Monash Model (MMM)

The MMM is used by Health providers to define a location as metropolitan, rural, remote or very remote based on the ASGS-RA framework of geographical remoteness and town size (DoHAC, 2021). Transition to the new MMM commenced in 2019 and continues to aid various health workforce programs outside major cities (DoHAC, 2021). Remoteness and population size are measured on a Modified Monash category MM 1 to MM 7.

- MM 1 All areas categorised ASGS-RA1 (major city)
- MM 2 Areas categorised ASGS-RA2 and 3 that are in, or within 20km distance by road of a town with population greater than 50,000
- MM 3 Areas categorised ASGS-RA2 and 3 that are not in MM 2 and are in, or within 15km distance by road of a town with a population between 15,000 and 50,000
- MM 4 Areas categorised ASGS-RA2 and ASGS-RA3 that are not in MM 2 or MM 3 and are in or with 10km distance by road of a town with a population between 5,000 and 15,000
- MM 5 All other areas in ASGS-RA2 and 3
- MM 6 All areas categorised ASGS-RA4 that are not on a populated island that is more than 5km offshore
- MM 7 All other areas (DoHAC, 2021)

In summary, the model is used to determine eligibility for health workforce programs, including rural Bulk Billing and attracting health professionals to work in more remote and smaller communities and is not relevant for peri-urban areas due to proximity to major urban centres.

Other classifications

Primary Health Networks

Australia has 31 PHN areas covering the whole of Australia which replaced the sixty one Medicare Locals in 2015 (AIHW, 2022b). Medical Locals were designed to improve regional primary health care with a focus on General Practitioner services and boundaries closely aligned with LHNs (AIHW, 2016). PHNs are independent organisations funded by the Australian Government with the focus on improving patient outcomes by encouraging primary, community and specialist health sectors to collaborate in their geographical area (Healthdirect, 2022). These networks are managed by a board and clinical committees and work directly with

General Practitioners, allied and community health professionals and hospitals (Healthdirect, 2022). PHNs are designed to work closely with LHNs to integrate care between primary and acute local hospital networks, with boundaries closely aligned with LHN boundaries (Healthdirect, 2022).

ASGS SA3 level data forms the basis for any sub-regions developed by a PHN for demographic analysis. Circumstances may require the use for LHN sub-divisions LGAs. This data is used in conjunction with needs analysis conducted by the PHNs to understand resource allocation in the region. PHN areas are classed into two PHN groups, metropolitan or regional (AIHW, 2022b). PHNs with at least 85% of the population residing in the ASGS-RA classification of Major Cities are classified as metropolitan, with the remainder classified as Regional (AIHW, 2022b).

As an independent primary health care organisation, PHNs were established to meet the needs of the local community. However, the degree that consumer experience and feedback is integrated into healthcare services is subject to the identification and priority placed by the PHN on needs.

Local Health Networks or Local Health Districts (LHDs)

Every public hospital in Australia is part of a local network, known as either networks or districts, depending upon the State or Territory involved (National Health Funding Body, 2022). LHNs comprise a range of sites and services, including services which could be reasonably attributed to a public hospital. These include emergency departments, inpatient and outpatient services including some community based health services and residential aged care services which have a geographical connection (Booth et al., 2016). Each LHN has a governing board and a chief executive officer who reports to the board. Boundaries are set by State and Territory governments who classify LHNs as metropolitan or country.

As local networks, LHNs are like PHNs with a core requirement being to meet the needs of the local community. However, meeting local needs requires input from community members and the priority that the LHN places on the feedback provided.

Population Health Areas (PHA)

Population Health Areas (PHAs) were developed following the introduction of the ASGS when SLAs, the previous dataset used in the ASGC, were phased out. It became evident that the number of cases in some health population datasets would be too small to map in some datasets (AIHW, 2021b). The equivalent area level to SLAs are SA2s which average half the population of the previous SLA units which would increase the difficulty of
maintaining confidentiality in datasets. Rather than use the next hierarchical level of SA3, a new Population Health Area (PHA) was constructed, comprised of a combination of whole SA2s and aggregates of SA2s (AIHW, 2021b). In 2016, there were 1,165 PHAs in Australia.

PHAs are used in several health datasets including chronic disease dashboards.

State Suburbs (SSC)

State Suburbs (SSC) are created by allocating one or more MBs to enable the release of ABS data on areas that approximate Gazetted Localities (ABS, 2016b). Gazetted Localities are officially recognised boundaries of suburbs and localities.

Postal Areas (POAs) and Postcodes

Postcodes are four digit numeral codes allocated to geographic areas to assist with processing and delivery of mail to specific addresses (ABS, 2016f). Postcodes are easily accessible and therefore used by many researchers to link data to a geographic area. The ABS provides Postcode Indexes to link postcode data to standard ABS geographic areas including SAs and GCCSAs within the ASGS (ABS, 2016f). Postal Areas (POAs) differ to postcodes being an ABS approximation of postcodes to enable release of data that closely approximates postcodes. POAs are approximated using one or more SA1s from the ASGS and are incorporated as a non-ABS structure (ABS, 2016f). Postcode and POA data have the benefit of providing easily accessible health data to assist with development and implementation of health services that meet the needs of local communities.

Cities, Towns and Villages (CTVs)

CTV geography is designed to assist with analysis of spatial distribution of service access in regional Australia. ABS data is sourced to provide an estimate of population density by square kilometre to identify population centres or clusters where people both live and where services are location. CTV geography identifies population and service centres and includes locations with populations of fewer than 200 persons, thereby removing a limitation of the ASGS. Service centres include health, education, postal, retail and government services (BITRE, 2019). The use of CTVs to represent population centres, has been superseded using GCCS, SUA and UCL, however, none represent the nexus of population and service centre required for service access analysis.

Local Government Areas

The term 'LGA' is used by the ABS to refer to local government administrative zones which have several varying designations. LGAs comprise a diverse range of communities and are designated areas for which incorporated local governing bodies have responsibility (ABS, 2022a). LGAs are not defined by the ABS and are therefore identified as non-ABS structures in the ASGS. LGA boundaries are an ABS MB approximation of gazetted local government boundaries as defined by state and territory governments and are used for statistical purposes only (ABS, 2022a). Every LGA in Australia can be assigned to a group based on similar population, population density and the proportion of the population that is classified as urban, as determined by ABS data (ABS, 2016b). These groups are based on the Australian Classification of Local Government (ACLG) framework which was a three step hierarchical framework with twenty two categories and deemed too detailed to enable meaningful comparisons (South Australian Productivity Commission, 2019). Groups of similar LGAs are determined by the States or Territories involved with general groupings being metropolitan, metropolitan fringe, regional city, large rural and rural based on ABS data.

Data provided at a LGA level provides a smaller spatial unit than State or Territory wide, although the area has the potential to cover a diverse range of communities. It can be difficult to define peri-urban areas within LGA boundaries using LGA geography due to the variety of land uses that may exist. Classifying LGAs as peri-urban or metropolitan fringe due to their general proximity to urban centres has the potential to oversimplify challenges to accessing community health services that may exist to some residents within the LGA.

Appendix B: Literature Review summary papers

Author/s	Title	Study Design	Location	Location classification	Theme	Quotes
(Hart et al., 2005)	Rural definitions for health policy and research	Qualitative discussion and comparison rural taxonomies	Rural	Different taxonomies – population size and density, commuting patterns	Definition or rural for health policy	"By treating these diverse types of rural cities and towns and the problems they confront similarly, policy analysts may fail to identify each site's distinct health care concerns"
(Humphreys, 1998)	Delimiting 'rural': implications of an agreed 'rurality' index for healthcare planning and resource allocation	Qualitative discussion rurality	Rural	Geographical units	Definition of rural for health policy	"rural Australia comprises a complex mosaic"
(Racher et al., 2004)	Conceptualizations of 'rural:' challenges and implications for nursing research	Qualitative discussion re urban/rural	Rural	Population density and size, isolation, census data	Definition of rural for health research	"Defining rural is a complex and multifaceted process that changes, according to the purpose for which the definition is being designed"
(Amirinejad et al., 2018)	Ambiguity at the peri-urban interface in Australia	Qualitative review	Peri-urban	Complex set of land use relationships	Peri-urban definition more than rural-urban dichotomy	"peri-urban area amalgamates urban, rural and natural features in such a way that each feature is influenced by the other two"
(Žlender, 2021)	Developing a spatially explicit method for delineating peri- urban landscape	Qualitative literature review	Peri-urban	Land uses – agriculture, leisure, sealed land, infrastructure, agricultural	Peri-urban definition for planning	'spatial delineation should be based on variables that reflect peri-urban land use as well as other relevant variables like population density.'
(Kaminski et al., 2021)	Using landscape metrics to characterize towns along an urban-rural gradient	Quantitative cluster analysis	Rural	Distance, population size and density, landscape pattern metrics	Definition of rural for sustainable planning	"Understanding where a town occurs along the urban-rural gradient could aid local decision- makers"
(Çörek Öztaş, 2021)	How to Best Classify Rural in Metropolitan Areas? The Turkish Case	Quantitative analysis	Rural	Population density	Definition of rural for policies and planning	'developing a clear understanding of what is rural, including a quantifiable and mapable definition, planners will be better prepared to improve strategies'
(Saxena, 2019)	Peri-Urban Areas: Sustainability Dimension	Qualitative review	Peri-urban	Population size and density, employment	Peri-urban definition for planning	"Peri-urban area is no a recognised entity for both studies and research"
(Miller, 2013)	More than "not urban": Seeking a quantifiable definition of rural	Qualitative review	Rural	Census data	Definition of rural for policy and planning	To "best inform policy specific to planning outcomes"
(Smith et al., 2013)	The utility of rural and underserved designations in geospatial assessments of distance traveled to healthcare services: Implications for public health research and practice	Qualitative review	Rural	Geography, population size and density, distance to services	Definition of rural for planning	"discrepancies among conclusions that may be drawn based upon the application of the different definitions of rurality when applied to health data"
(Allen, 2003)	Environmental planning and management of the peri-urban interface: Perspectives on an emerging field	Qualitative review	Peri-urban	Population size and density, infrastructural characteristics, administrative boundaries and economic activities	Peri-urban definition for planning	"construction of an approach that responds to the specific environment, social, economic and institutional aspects of the peri-urban interface"
(laquinta & Drescher, 2000)	Defining the peri-urban: Rural- urban linkages and institutional connections	Qualitative review	Peri-urban	Distance to city centres and local culture	Peri-urban definition for food production planning	'definitions are largely situational and case specific. They provide little basis for a unified understanding of what constitutes periurban.'

(McGrail & Humphreys, 2009)	Geographical classifications to guide rural health policy in Australia	Qualitative debate	Rural	GIS classifications	Definition of rural for health policy	"way in which populations and communities are delimitedhas important implications for health care planning and policy"
(Booza et al., 2010)	Incorporating Geographic Information Systems (GIS) into program evaluation: lessons from a rural medicine initiative	Qualitative review	Rural	3 measures: percentage of population, internal level continuous measure and census data	Definition to assist with addressing GP shortage	'outcome of student placement in rural practices depended on the definition of rural.'
(Nixon et al., 2021)	Developing the geographic classification for health, a rural-urban classification for New Zealand health research and policy: A research protocol	Mixed methods	Rural	Develop existing GIS classification using population and spatial	Definition for rural health policy	'Rural-urban definitions are most complex and elusive for small towns in rural regions, peri- urban communities in commuter zones and the rural- urban fringe.'
(Williams & Cutchin, 2002)	The rural context of health care provision	Qualitative review	Rural	Land use, population density, demographic structure, environmental characteristics, population characteristics, non- metropolitan areas and community patterns	Definition of rural to address rural health care staff provision	"may provide the basis to best understand the attributes and challenges of rural places"
(Bennett et al., 2019)	What Is Rural? Challenges And Implications Of Definitions That Inadequately Encompass Rural People And Places	Qualitative review	Rural	Population density, travel (distance, roads, geography, availability), resources, socio-economics, local perception, culture and amenities	Definition of rural for health policy and planning	"definition of rural could be operationalized as an index and incorporate measures from a variety of areas"
(Taylor et al., 2021)	General practice access in regional and remote Australia for ageing populations	Quantitative analysis	Rural	Population size and density, spatial and distance to GP clinics	Definition of rural for health policy (GP clinic placement)	'General practice access in regional and remote Australia for ageing populations'

Appendix C: Ethics Approval

This administrative form has been removed

Appendix D: Survey questions

Peri-urban survey

Q1.1 Thankyou for participating in this research.

Community based palliative care services aim to optimise quality of life and wellbeing through a person-centred, multidisciplinary approach to home-based care. Palliative care is provided to people diagnosed with a life-limiting illness and their family members. Every person has the right to access all health services including palliative care with barriers to healthcare services from rural and remote locations recognised. Increasing numbers of Australians are relocating from densely populated, urban areas to peri-urban locations offering a rural lifestyle within commuting distance to major cities. This research aims to understand if this counterurbanisation trend to a peri-urban location, is influencing access to healthcare services involved with providing community palliative care.

Access is a complex concept and includes availability of affordable and effective services, provided in a manner which considers diverse perspectives and individual needs. Therefore questions in the following survey will include, how people identify, seek, reach and use the health services required and if the holistic needs of those facing end-of-life care are being fulfilled by the services provided and the method of service delivery. Providing geographic location will enable researcher to identify if peri-urban location influences access to care provision. Health providers views on service availability will provide valuable insights and enable comparison of the views of those receiving care.

Your views may assist in recommendations for future policies.

You are encouraged to read the Grief Supports page and the Participant Information Sheet in full, before consenting to participate in this research.

- o Participant Information Sheet
- o Grief Supports

${\tt Q1.4}$ I have had the opportunity to read the Participant Information Sheet and ask questions about this research

I have had the opportunity to read the Grief Supports page

I understand the possible risks to my involvement in this research

I understand that I may not answer questions and/or leave the survey without submitting

I am over 18 years of age and voluntarily consent to this research

- o Yes
- o **No**

Q1.5 Please select the statement that best describes you

- o I am a person receiving community palliative care following a diagnosis of a life-limiting illness
- o I am a family member or care giver of a person diagnosed with a life-limiting illness
- o I am a Health Professional providing community palliative care

Family member or care giver survey questions

Q2.1 Your age group?

- o 18 to 34 years
- o 35 49 years
- o 50 64 years
- o 65 years and over
- Prefer not to answer

Q2.2 How long have you been receiving community palliative care?

Palliative care support and services are provided during the illness period and extends to family members and caregivers after death

o Less than 1 month

- o 1 to 6 months
- o 6 months to 1 year
- o Over 1 year
- o Unsure
- o Prefer not to answer

Q2.3 What is your relationship to the person diagnosed with a life limiting illness?

- o Spouse
- o Sibling
- o Child
- o Extended family member
- o Friend
- o Other

Q2.4 In what State/Territory are services provided?

If you are receiving in-home care in more than one location, please consider only the main location

- o Queensland
- o South Australia
- Northern Territory
- New South Wales
- o Western Australia
- o Victoria
- o ACT
- o **Tasmania**
- o Outside of Australia

Q2.5 How would you describe the area where community palliative care is provided?

- o Metropolitan (densely populated, urban centres)
- Peri-urban (fringe areas of major cities, a mix of rural and new residential development, experiencing population growth, often within 150km of major city and commuting distance)
- o Rural (predominantly farming land, less densely populated, often over 150km from major city)
- o Regional
- o Remote
- o Unsure
- Other (please describe) _____

Q2.6 In what local government area (council or shire) is the in-home (community) care being provided? To maintain your privacy, please do not provide your full address

Understanding the LGA will provide further detail about the type of location and influence on access

Q2.7 The term Health services in this survey refers to all health services/supports that may be needed for those people receiving palliative care and include pharmacy, medical clinic, pathology, radiology, specialist and community services

How much do you agree with the following statements?

	Strongly Disagree	Somewhat Disagree	Neither agree nor disagree	Somewhat agree	Strongly agree
Existing health services were considered when selecting the home location	0	0	0	0	0
The location was selected as services were in the development stage	0	0	0	0	0

Conside facility)	eration was given to accessing services for possible future health needs (eg hospice, aged care when selecting the home location	0	0	0	0	0
There w	vas an expectation that health services were accessible	0	0	0	0	0
Q2.8 Ho	w did you receive/find information about available services? select all that apply Health Professional (GP, Specialist, clinic staff)					
	Referred to Palliative care					
	Friends/family					
	Pamphlets/booklets					
	Online forums					
	Support groups					
	Google					
	Personal knowledge/experience					
	I was not involved in accessing services					
	Other					
Q2.9 Di	d you find it difficult to find information about service/s and identify what was nee Yes	ded?	•			
-						

- o No
- o N/A

Q2.10 Did you find it difficult to access identified services when required?

- o Yes
- o No
- o N/A

Q2.11 Please select all the services that you know are being received or have been received in the home

- Unsure
- Specialist Palliative Care
- Community Nursing
- Personal Care
- □ Transport Assistance
- Occupational Therapist
- Psychiatrist/Psychologist
- Social Worker/Counsellor
- Physiotherapy
- Dietician/Nutritionist/Speech Pathology
- Palliative Care Specialist Physician
- □ Spiritual Support
- □ Volunteers
- Complementary Therapies art, music or massage
- Housework/Gardening Assistance
- Home maintenance/modifications
- Equipment
- Assistance with Advance Care Directive
- Financial/legal assistance
- Regular General Practitioner
- □ End-of-life care
- Bereavement support
- □ Family/friends
- Other _____

Q2.12 Do services received meet needs and expectations?

- o Yes
- o Partially (please provide detail)_
- o No (provide detail)
- o Prefer not to answer

Q2.13 If you are not using all services that are available, why not? select all that apply

- □ Not required
- □ Family provide

- Not happy with the service previously
- Not able to access service when I want
- Not able to access service as often as I want
- Service not available in this location
- □ Local providers do not have the resources to provide the service
- □ Service is too expensive
- Using all services we knew about
- Personal choice
- □ Have tried to access but unable unsure why
- Other_

Q2.14 Please select any of the following that apply:

- One or more hospital stays have been extended as services unavailable in-home to support discharge
- Needed to attend clinic or hospital outpatients for medication administration as in-home services unavailable
- Needed to attend clinic or hospital outpatients for assistance with wound/catheter/stoma care as in-home services unavailable
- Needed to call an ambulance for assistance with pain/symptom control as unable to contact community services
- Respite in aged care facility was required as in-home services unavailable to support hospital discharge
- Will need to transfer to hospital for end-of-life care to remain in local area

Q2.15 What is the longest distance your family member has needed to travel for healthcare services?

- consider services such as scans, radiotherapy, chemotherapy, dialysis, specialist appointments
- o all services are local and within 10km from home
- o 10 to 50km
- o 50 to 100km
- o 100 to 150km
- o 150km to 250km
- o over 250km
- o Unsure

Q2.16 What barriers to accessing services do you believe apply in your situation?

- □ There are no barriers to access
- Person requiring services is under 65 years of age
- Home location and the distance involved for service providers
- Lack of resources by local providers
- Time taken for services to commence
- Distance and time involved in travelling to some health services not available locally
- Cultural or gender diversity
- □ Language
- Stigma associated with palliative care
- Challenging behaviours
- □ Financial
- Personal choice
- Care needs beyond scope of existing services
- No identified need
- □ Other

Q2.17 Thinking about transport from home to the location of some health services, please select statements that apply to your family

- □ The time/distance involved has influenced decisions about attending appointments, treatment or tests
- The time/distance has influenced the home location of the person requiring care in the home
- □ Travel time/distance is not a problem
- Travel any distance is difficult (exhausting, pain, discomfort, etc)
- Private transport is difficult due to cost of petrol, parking, lost wages and time involved
- Public transport is not an option
- Community car is not available/expensive/not an option
- □ All health services are provided in the home
- □ All health services are provided locally
- Need to relocate to be closer to services that will provide a higher level of care
- Other_____

 Ω^2 18 How much do you agree with the following statements

dz. 10 now much do you agree with the following state	Strongly disagree	Somewhat disagree	Neither agree nor disagree	Somewhat agree	Strongly agree
The choice to remain in own home has been supported	\bigcirc	0	\bigcirc	0	0
Personal choices, preferences and goals of care have been respected	\bigcirc	0	\bigcirc	\bigcirc	\bigcirc
All communication and information provided has been respectful and appropriate for our needs	\bigcirc	0	\bigcirc	\bigcirc	0
Access to support is available 24 hours a day 7 days a week	0	0	\bigcirc	\bigcirc	0
Quality of life has been improved by community services	0	0	0	0	0
The need for ambulance transport and admission to hospital has been reduced by in-home services	\bigcirc	0	\bigcirc	0	0
Family members have been included in all decision- making about care and services	\bigcirc	\bigcirc	\bigcirc	\bigcirc	0
Physical symptoms are managed in the home	0	0	0	0	0
Support has been provided with advance care planning, completing will and grief counselling	0	\bigcirc	0	0	0
Financial assistance needed has been provided	\bigcirc	0	0	0	\bigcirc
All support and education required by family members to provide safe, quality care, including end-of-life care and/or home death has been provided	\bigcirc	0	0	0	0
All services have been provided promptly and when needed	\bigcirc	0	0	0	0
Medication is available locally	0	0	0	0	0
Choices for end-of-life care including hospice, home death and voluntary assisted dying, have been supported and have enabled informed decisions	0	0	0	0	0
Available services are affordable	\bigcirc	0	0	0	0
Staff providing care have the training, knowledge and expertise to provide the quality of care needed	\bigcirc	0	\bigcirc	0	0
Choice of provider and/or staff who attend home visits is provided	\bigcirc	0	0	0	0
Family/caregiver stress and burnout has been reduced with service provision	0	0	\bigcirc	\bigcirc	0
Patient/family have good relationships with health professionals involved in care	0	0	\bigcirc	0	

Q2.19 Digital health, telehealth, eHealth and telemedicine are broad terms used interchangeably. This survey will use the term digital health to refer to the all the tools and services that are available to diagnose, treat, collect and share

health information, including consultations with a healthcare provider that are conducted by phone or video rather than face-to-face.

How are palliative care services provided in your home? select all that apply

- Face-to-face
- Phone consultation
- Video consultation \Box
- \square Fmail
- Text
- Apps
- Other

Q2.20 Have you experienced any problems with digital health options? Please select all that apply

- No problems using digital health
- Lack of computer access/skills \Box
- I have not used digital options \Box
- poor mobile coverage
- Unreliable/non-existent internet access
- Digital options not offered by providers \Box
- \Box Other_

Q2.21 How much you agree with the following

Q2.2 Thow much you agree with the following	Strongly disagree	Somewhat disagree	Neither agree nor disagree	Somewhat agree	Strongly agree
Digital health is an important tool in ensuring regular communication with the palliative care team	0	\bigcirc	0	0	0
The method palliative care provides support in the home either face-to-face or digital is appropriate for our needs	0	\bigcirc	\bigcirc	\bigcirc	0
I have the resources and confidence to use digital health effectively	0	0	0	0	0
Digital health consultations are an acceptable alternative to face-to-face consultations with medical professionals	0	\bigcirc	\bigcirc	\bigcirc	0
Digital health means that access to medical support is available that I would not be able to access due to distance	0	\bigcirc	0	0	0
I feel that the choice of place of care is supported with digital health	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I am comfortable with my family member remaining at home knowing that support can be accessed at any time	0	\bigcirc	0	\bigcirc	0

Q2.22 Overall are you satisfied with the access to community palliative care services in your location?

- Yes о
- No 0
- 0 Unsure
- Prefer not to comment o

End of Block: Family member or care giver survey

Start of Block: Final questions

Q3.1 Do you have comments you would like to add?

Please remember that your opinion is important o

o Yes o No If = Yes Q3.2 Please provide details_

Q3.3 Thankyou for participating in this research

Would you like details of the completed research emailed to you.

o Yes o No

o If = Yes

Q3.4 Please provide your email

These details will not be linked to survey responses

Email

End of Block: Final questions

Start of Block: Submission

Q4.1 If you experience any upset or distress because of your participation in this research project, please seek assistance from available services in your local area, talk to someone, or

- Call Lifeline on 13 11 14
- 13 Yarn support for Aboriginal and Torres Strait Islander people 13 92 76
- Beyond Blue 1300 224 636
- Lifeline Australia 13 11 14
- or contact your palliative care service for support, GP to discuss the possibility of a Mental Health Care Plan, employer or education institution for available employee assistance or health and wellbeing programs that offer confidential counselling and support.

Do not forget your community support network which includes; friends, family members, church minister or religious leader, a school, university or TAFE counsellor, coach, work colleague, community leaders.

If you or someone near you is in immediate danger: Call Emergency Services on 000; or go to a hospital emergency department

End of Block: Submission

Start of Block: Patient survey

Q5.1 Your age group?

- o 18 to 34 years
- o 35 49 years
- o 50 64 years
- o 65 years and over
- o Prefer not to answer

Q5.2 How long have you been receiving community palliative care?

please consider if you wish to continue with this research if you recently commenced receiving palliative care

- o less than 1 month
- o 1 to 6 months
- o 6 months to 1 year
- o over 1 year
- o Unsure
- o Prefer not to answer

Q5.3 In what State/Territory do you live? care provision varies by location

- o Queensland
- o South Australia
- o Northern Territory
- o New South Wales
- o Western Australia
- o Victoria
- o ACT
- o Tasmania
- o Outside of Australia

Q5.4 How would you describe the area where you live and receive care?

- o Metropolitan (densely populated, urban centres)
- o Peri-urban (fringe areas of major cities, a mix of rural and new residential development, experiencing high population growth, often up to 150km from adjoining metropolitan centre)
- o Rural (predominantly farming land, less densely populated, often over 150km from metropolitan centre)
- o Regional
- o Remote
- o Unsure

Q5.5 In what local government area (council or shire) do you receive care? to maintain your privacy, please do not provide your full address understanding the LGA will provide further detail about the type of location and influence on access

Q5.6 The term Health services in this survey, refers to all services/supports that may be required for palliative care and include pharmacy, medical clinic, pathology, radiology, specialist and community services **How strongly do you feel about the following statements?**

	Strongly disagree	Somewhat disagree	Neither agree nor disagree	Somewhat agree	Strongly agree
I chose this location as health services I need are close by	0	0	0	0	0
I chose this location as services I need/may need, are in the planning or development stage	0	\bigcirc	0	0	0
I chose this location as health services that provide for increased care needs are close by (eg hospice, aged care facility, specialist care)	0	\bigcirc	0	0	0
There was an expectation that health services were accessible	0	\bigcirc	\bigcirc	\bigcirc	0

Q5.7 How did you identify, contact or access services and information? select all that apply

- □ I was not involved with seeking services or information
- Health Professional providing treatment (GP, Specialist, clinic staff)
- □ I was referred to Palliative care
- □ Friends/family
- Pamphlets/booklets
- Online forums
- □ Support groups
- □ Google
- Personal knowledge/experience
- □ Other

Q5.8 Did you find it difficult to find information about service/s and identify what was needed?

- o Yes
- o No
- o n/a

Q5.9 Did you find it difficult to access identified services when required?

- o Yes
- o No
- o n/a

Q5.10 Please select all community services you are using/have used

- Unsure
- Specialist Palliative Care
- □ Community nursing
- Personal Care
- Transport Assistance
- Occupational Therapist
- Psychiatrist/Psychologist
- □ Social Worker/Counsellor
- Physiotherapy
- Dietician/Nutritionist/Speech Pathology
- Palliative Care Specialist Physician

- □ Spiritual Support
- O Volunteers
- Complementary therapies art, music or massage
- □ Housework/Gardening assistance
- □ Home maintenance/modifications
- Equipment
- Assistance with Advance Care Directive
- □ Financial/legal assistance
- Regular General Practitioner
- □ Family/friends
- Other____

Q5.11 Do the services you receive meet your needs and expectations?

- o Yes
- o Partially (please provide detail)
- o No (please provide detail)
- o Prefer not to answer

Q5.12 If you are not using all available community services, why not?

- □ I don't need the service at this time
- □ Family provide
- □ Not happy with the service previously
- Not able to access when I want
- Not able to access service as often as I want
- □ I am under 65 years of age and unable to access
- Service not available in this location
- Local providers do not have the resources to provide the service
- □ Too expensive
- Didn't know about all the services available
- Personal choice
- Unsure why unable to access
- □ Other_

Q5.13 Please select any of the following that apply to you

- I have had one or more hospital stays extended as services were unavailable to support my discharge home
- I have needed to attend a clinic or hospital outpatients for medication administration as in-home services were unavailable
- □ I have needed to attend a clinic or hospital outpatients for assistance with wound/catheter/stoma care as inhome services were unavailable
- □ I have needed to call an ambulance for assistance with pain/symptom control as unable to contact local services
- Respite in aged care facility has been required as in-home services unavailable to support hospital discharge
- U Will need to transfer to hospital for end-of-life care to remain in this location

Q5.14 What is the longest distance you have needed to travel for healthcare services? consider services such as scans, radiotherapy, chemotherapy, dialysis, specialist appointments

- o all services are local and within 10km from home
- o 10 to 50km
- o 50 to 100km
- o 100 to 150km

- o over 150km
- o unsure

Q5.15 What barriers to accessing services do you believe apply in your situation?

- □ There are no barriers to access
- □ I am under 65 years of age
- □ Home location and the distance involved for service providers
- □ Lack of resources by local providers
- Time taken for services to commence
- Distance and time involved in travelling to some health services not available locally
- Cultural or gender diversity
- □ Language
- I do not want palliative care
- Challenging behaviours related to a family member/caregiver
- □ Financial
- Personal choice
- Care needs beyond scope of existing services
- □ No identified need
- □ Other_

Q5.16 Thinking about distance to health services, please select statements that apply to you

- □ The distance required to travel to some health services has influenced my decisions about attending appointments, treatments or tests
- □ The distance required to travel to some health services has influenced where I live
- Travel any distance to health services is difficult (exhausting, pain, discomfort, etc)
- Private transport to some health services is difficult due to cost of petrol, parking, lost wages and time involved for my family
- Public transport is not an option in this location
- Community car is not available/expensive/not an option
- Health services are provided in my home
- Health services are provided locally
- I will need to relocate to be closer to services that will provide the level of care I need/will need
- □ Other _

Q5.17 How much do you agree with the following statements about the services you receive?

	Strongly disagree	Somewhat disagree	Neither agree nor disagree	Somewhat agree	Strongly agree
I have been supported in my choice to remain in my own home	0	0	\bigcirc	\bigcirc	0
Personal choices, preferences and goals of care are being respected	0	\bigcirc	0	\bigcirc	0
All communication and care provided has been culturally and gender inclusive and respectful	0	0	\bigcirc	\bigcirc	0
I have access to support 24 hours a day and 7 days a week	0	0	\bigcirc	0	0
My quality of life has been improved by remaining in my own home	0	0	0	0	0
Access to community services have reduced my need for ambulance transport and hospital admissions	0	0	0	\bigcirc	0
I have been included in all decision-making about care and services provided	0	0	0	\bigcirc	0
I am comfortable with the level of symptom management that is available	0	0	\bigcirc	0	0
I have received support with advance care planning, completing my will and grief counselling	0	0	0	0	0
I have received financial assistance as needed	0	0	\bigcirc	\bigcirc	0
I am satisfied that my family and caregivers are receiving the support they need	0	0	0	0	0
All the services I need have been provided promptly	0	0	0	0	0
I am able to access the medication that I need easily	0	0	\bigcirc	0	0
My personal choices for end-of-life care including hospice, home death and voluntary assisted dying, have been supported and I have been able to make informed decisions	0	0	0	0	0
All services I need are affordable	0	0	\bigcirc	0	\bigcirc
I believe the staff providing my care have the training, knowledge and expertise to provide the quality of care I need	0	0	0	0	0
I have choice of provider and/or staff who attend my home	0	\bigcirc	\bigcirc	\bigcirc	0
Family/caregiver stress and burnout has been reduced with service provision	0	\bigcirc	\bigcirc	0	0
I have a good relationship with health professionals involved in my care	0	0	\bigcirc	0	0

Q5.18 Digital health, telehealth, eHealth and telemedicine are broad terms used interchangeably. This survey will use the term digital health to refer to all the tools and services that are available to diagnose, treat, collect and share

health information, including those consultations with a healthcare provider that are conducted by phone or video rather than face-to-face.

Digital health options improve access to health services including palliative care. How do you receive palliative care services in your home? select all that apply

- □ Face-to-face
- Phone consultation
- Video consultation
- Email
- □ Text
- □ Apps
- □ Other

Q5.19 Have you experienced any problems with digital health options? Please select all that apply

- No problems using digital health
- □ Lack of computer access/skills
- I have not used digital options
- Poor mobile coverage
- Unreliable/non-existent internet access
- Digital options not offered by providers
- Other

Q5.20 How much you agree with the following

	Strongly disagree	Somewhat disagree	Neither agree nor disagree	Somewhat agree	Strongly agree
Digital health is an important tool in ensuring I have regular communication with my palliative care team	0	0	0	\bigcirc	0
I have the resources and confidence to use digital health effectively	\bigcirc	0	0	\bigcirc	\bigcirc
Digital health consultations are an acceptable alternative to face-to-face consultations with medical professionals	0	0	0	0	0
Digital health means that I can access medical support that I would not be able to access due to distance	\bigcirc	0	0	\bigcirc	0
I am comfortable to remain at home knowing that I can access support at any time	\bigcirc	0	0	0	0

Q5.21 Overall, are you satisfied with the access to community palliative care services in your location?

- o Yes
- o No
- o Unsure
- o Prefer not to answer

End of Block: Patient survey

Start of Block: Health Professionals survey

Q6.1 Your age group?

- o 18 to 34 years
- o 35 49 years
- o 50 64 years
- o 65 years and over
- o Prefer not to answer

Q6.2 How long have you been providing community palliative care?

- o less than 1 year
- o 1 to 5 years
- o over 5 years
- o prefer not to answer

Q6.3 What is your current role in providing community palliative care?

- o Nurse Practitioner
- o Clinical Nurse
- o Registered Nurse
- o Enrolled Nurse
- o AIN/Personal Care attendant
- o Occupational Therapist
- o General Practitioner
- o Psychiatrist
- o Psychologist
- o Social Worker
- o Physiotherapist
- o Dietician
- o Speech Therapist
- o Palliative Care Physician
- o Chaplain
- o Pharmacist
- o Volunteer
- o Other

Q6.4 How would you describe the area where you provide community palliative care?

If you provide care in more than one location, consider the location where you provide the majority of care

- o Metropolitan (densely populated, urban centres)
- o Peri-urban (fringe areas of major cities, a mix of rural and new residential development, experiencing population growth, often within 150km of major city)
- o Rural (predominantly farming land, less densely populated, often over 150km from major city)
- o Regional
- o Remote
- o Unsure
- o Other (please describe)

Q6.5 **In what local government area (council or shire) do you provide community palliative care?** If multiple areas, please consider the area where you provide majority of care

Understanding the LGA will provider further detail about the type of location and influence on access

Q6.6 In what State/Territory?

- o Queensland
- o South Australia

- Northern Territory 0
- New South Wales 0
- Western Australia о
- Victoria 0
- ACT 0
- Tasmania 0
- Outside of Australia 0

Q6.7 In what type of organisation do you provide community palliative care? If you work in more than one service/organisation, please think about where you work the most hour

- о Specialist Palliative Care team
- о Generalist Community Health
- **Public Hospital** о
- **Private Hospital** 0
- **General Practitioner Clinic** о
- 0 Home Care Provider
- о **Residential Aged Care Facility**
- Other 0

Q6.8 Palliative care teams are multidisciplinary What health professions are employed to support community palliative care in your organisation? select all that apply in full or part-time roles

- Nurse Practitioner \square
- **Clinical Nurse**
- **Registered Nurse**
- Enrolled Nurse
- AIN/Personal Care attendant
- **Occupational Therapist** General Practitioner
- \Box
- Psychiatry
- Psychology
- Social Work
- \Box Physiotherapy Dietician
- Speech Therapy
- Palliative Care Specialist
- \Box Chaplain
- Pharmacist \Box
- Other

Q6.9 If you have a choice, are there any health profession roles you would like added to your team? consider part-time and full-time roles

- Yes (please detail) о
- No о
- Unsure о

Q6.10 Palliative patients often require support services from multiple organisations to remain in their home. What services are accessible to all homes in the area where you work? please select all services that you are aware of, whether or not service is provided by your organisation

- Specialist palliative care
- **Occupational Therapist**
- Physiotherapy
- Daily medication administration

- □ Community nursing
- □ Assistance with personal care
- Equipment
- Home modifications
- Advance Care planning
- □ Housekeeping, gardening, home maintenance
- □ Counselling
- End-of-life care in the home
- Bereavement care
- Transport assistance (public transport, community car)
- Dietician, speech pathology, nutrition
- General Practitioner
- Pharmacy
- Complementary therapies art, music, massage
- □ Volunteer support
- Pathology
- □ Radiology
- Specialist physicians
- □ Other

Q6.11 Access is influenced by multiple factors. What barriers to accessing services do you believe apply in your location?

- No barriers to client access
- Clients aged under 65 years
- □ Home location and the distance involved in providing in-home service
- □ Lack of resources/capacity from providers
- Time taken for services to commence
- Distance and time involved in clients having to travel to services that are not local
- Cultural or gender diversity
- Language
- Client refusing due to stigma associated with palliative care
- Challenging behaviours
- □ Financial services expensive
- □ No identified need
- Personal choice
- □ Care needs increase beyond capacity of existing services
- Lack of information about available services
- Unsure
- □ Other _

Q6.12 Thinking about the organisation in which you provide community palliative care, are there any changes you would like to see in your role?

- No changes, service meets the needs of clients
- □ Increase frequency services could be available in the home
- Enable choice of provider
- Provide choice of day/time of service
- Provide 24/7 contact for support
- Provide 24/7 in-home care options
- Increase in staff numbers to support service availability
- Remove restrictions on time allocated to each home visit to ensure needs are met
- Increase in available equipment
- Reduce cost to consumer

- □ Improve training opportunities for staff
- □ Other

Q6.13 Are you aware of any of the following in your area?

- Hospital admissions extended due to in-home services unable/unavailable to support discharge
- Clients needing to attend medical clinics or outpatient clinics as no in-home services to provide medication administration
- Clients needing to attend medical clinics or outpatient clinics as no/insufficient in-home services available to assist with wound/catheter/stoma care
- Ambulance being called for assistance with pain/symptom control as community services unable to be contacted
- Respite in aged care facility as no/insufficient services available in-home to support hospital discharge
- End-of-life care provided in hospital as appropriate services unable to be provided in the home
- People relocating due to services being unable to provide the level of support needed in the home
- None apply

Q6.14 How does your organisation provide information about relevant services to the community?

- rely on staff member to discuss face-to-face
- □ staff member to discuss over phone
- pamphlets/booklets
- □ website
- advertise in local health and community centres
- □ community events
- rely on local community members to discuss
- □ other _

Q6.15 Consider the area that you provide community palliative care services to. Select how much you agree with the following statements

	Strongly disagree	Disagree	Neither agree nor disagree	Agree	Strongly agree
Existing health services meet the needs and expectations of people in this location	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
There is a need for people in this area to travel over 100km distances for some healthcare services	0	\bigcirc	\bigcirc	\bigcirc	0
The need to travel long distances for some healthcare services influences decisions about treatment	0	0	0	0	0
The need to travel long distances for some healthcare services results in people relocating closer to services	0	0	0	0	0
People are supported to remain in their own home	0	0	0	\bigcirc	0
Home location does not influence access to services	0	0	\bigcirc	0	0
Communication and care provided is culturally and gender inclusive and respectful	0	\bigcirc	0	0	0
Personal choices, preferences and goals of care are respected	0	0	\bigcirc	\bigcirc	0
Access is available for support 24/7	0	0	\bigcirc	0	0
Quality of life is improved by remaining in own home	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Ambulance transport and hospital admissions are reduced by community services provided	0	0	\bigcirc	0	0
Patients and family members are included in all decision-making about care and services	0	\bigcirc	0	0	0
Physical, emotional, spiritual and social support is provided	0	0	0	\bigcirc	0
Services are provided promptly and when needed	0	0	0	0	0
Medication required can be accessed easily	0	0	\bigcirc	0	\bigcirc
The choice for end-of-life in the home and a home death is supported regardless of the location of the home	0	0	0	\bigcirc	0
All services are affordable	0	0	0	0	0
Staff providing care have the training, knowledge and expertise to provide quality care	0	\bigcirc	\bigcirc	0	\bigcirc
Choices for end-of-life care including hospice, home death and voluntary assisted dying are supported and clients are able to make	0	0	0	0	0
Informed decisions	0	0	\bigcirc	U	0
שבובמיפוווכווג נמוב וז מימוומטוב מז ובעטוובט	0	\bigcirc	0	0	\bigcirc
Family/caregiver stress and burnout is reduced with in-home services	0	\bigcirc	0	\bigcirc	\bigcirc

Q6.16 Digital health, telehealth, eHealth and telemedicine are broad terms used interchangeably. This survey will use the term digital health to refer to the all the tools and services that are available to diagnose, treat, collect and share health information, including consultations with a healthcare provider that are conducted by phone or video rather than face-to-face.

How do you provide palliative care in the community? select all that apply

- □ Face-to-face in-home or clinic
- □ phone
- □ video
- email
- □ text
- □ other _

Q6.17 Why do you use digital health options? Select all options that apply

- □ To reduce infection risk
- □ Improve support provided to patients/family members
- Reduces travel time for clinician
- Personal choice of patients/family members
- □ Service provided is not available face-to-face
- □ After hours support
- Personal choice of clinician/patient
- □ Face-to-face visit not required
- □ Family are able to be present during consultation
- □ Specialist input required but distance provides a barrier for timely input
- □ Other ___

Q6.18 Have you experienced any problems with digital health options? Please select all that apply

- □ No problems using digital health
- □ Lack of computer access/skills by patient/family members
- D Personal preference of patient/family members for face-to-face service delivery only
- D Poor mobile coverage in locations where I work
- Unreliable/non-existent internet access in locations where I work
- Digital options not offered by this organisation/serviced
- □ the service I provide requires a face-to-face visit
- □ Other _

Q6.19 How much you agree with the following

	Strongly disagree	Somewhat disagree	Neither agree nor disagree	Somewhat agree	Strongly agree
Digital health is an important tool in ensuring regular communication with patients	0	0	\bigcirc	0	0
I have the resources and confidence to use digital health options effectively	0	0	\bigcirc	0	0
Telehealth consultations are an acceptable alternative to face-to-face consultations with medical professionals	0	\bigcirc	\bigcirc	0	0
Telehealth means that patients can access medical support that would not be readily accessible due to distance	\bigcirc	\bigcirc	0	0	0
I feel that patient's choice of place of care is supported with digital health options	0	0	\bigcirc	0	\bigcirc
Patients and family members are comfortable to remain at home knowing that they can access support at any time	0	0	\bigcirc	0	0

Q6.20 Overall, are you satisfied with the access to community palliative care provided in your location?

- o Yes
- o No
- o Unsure
- o Prefer not to answer

End of Block: Health Professionals survey

Appendix E: PCA Consent

This administrative form has been removed

Appendix F: PCA e-newsletter link



Research spotlight

Participate in palliative care research:

James Cook University PhD Researcher is conducting a study to investigate the influence of counter-urbanisation on access to community palliative care services. Understanding issues to accessibility may contribute to recommendations to future policy. You will be required to complete a short survey which will take approximately 10 minutes to complete. All information collected is anonymous. <u>Take the survey now.</u>

To be eligible to participate you need to be over 18 years of age, a person receiving community palliative care, a family member or caregiver or a health professional providing community palliative care in any location in Australia. JCU HREC approval #H8936

If there are any questions/issues please contact shirley.papavasiliou@my.jcu.edu.au.



Appendix G: Participant Information Sheet

PROJECT TITLE:

"Access and equity in Australian community palliative care services: the influence of peri-urban location"

You are invited to take part in a research project which aims to understand influences to community palliative care in those areas on the fringe of Australian major cities which are experiencing high population growth, termed periurban areas.

What is the research about?

Australians have shown a preference for receiving care in their own home, including end-of-life care and a home death. In home care has been found to improve quality of life for patients and assists family members and friends with the grieving process. In-home or community care includes face-to-face and digital health options. There are known challenges in provision of end-of-life care in a home setting, particularly in a rural area. Recent trends show people moving from metropolitan cities, with high growth evident in those areas within commuting distance to major cities. These peri-urban areas have the potential for increased challenges to accessing services due to location and corresponding influence on service availability and digital access.

Who is doing the research?

The study is being conducted by Principal Investigator Shirley Papavasiliou and will contribute to completion of a PhD by publication at James Cook University.

Who can take part in this research?

Health Professionals, including paid and unpaid, over 18 years of age, in any role which provides palliative care in the community.

Patients, family members and support persons, over 18 years of age, who have received, or are receiving, palliative care in the community.

What do I have to do?

If you agree to be involved in the study, you will be invited to complete an online survey. The survey will take approximately 10-15 minutes to complete.

This survey involves questions about how you access or provide access to community palliative care services.

Accessibility to services is complex and includes identifying palliative care needs, seeking services, reaching, obtaining, using, paying for and engaging with services. Access involves services being appropriate, that is quality services that meet the needs of users, when they are needed, and are culturally and socially acceptable. Services may be face-to-face or virtual.

Do I have to take part in this research?

Taking part in this study is completely voluntary. Any information obtained in connection with this research project will be anonymous. You will not be asked for your name or contact details (unless you request a copy of the survey results).

Any decision about participation in this study will not affect any current or future relationships with any health care professionals or organisations you are involved with as data is anonymous.

By completing the consent form at the start of the online survey, you are consenting to the use of your survey responses in this research. You may select not to answer any question you choose (not including the mandatory questions that ensure your eligibility) and you may withdraw from the survey at any time, without providing a reason, with no risk of repercussion. Please note that once you submit the survey, the researcher will not be able to delete your responses as the survey data is anonymous.

Are there any risks or inconveniences to taking part in this research?

You may find some questions about in-home or community palliative care may trigger an emotional response.

Prior to participating in this survey please remember that grief has no timeline. It can begin long before death arrives. It is not unusual for grief to be felt over an extended period of time, even for many years. Sometimes we may experience feelings of anxiety, dread, sadness, longing, regret, loneliness, thoughts of 'if only' and more.

Grief is not just accepting death, but the losses occurring as an illness progresses. Feelings may spring up unexpectedly or be triggered by an event or a memory. These triggers can re-kindle emotions and be upsetting or bring memories of great joy and gratitude for happier times.

If this survey is likely to trigger feelings that you may find upsetting, then please consider if you wish to continue. Please do not participate if you have a diagnosed mental health condition, are recently bereaved, experiencing chronic grief or are dealing with multiple traumas, due to possible risk of distress from participation.

If you do participate then you may decide to withdraw from the research at any time you choose. If you do not wish to answer a question, you may choose to not answer and go to the next question. You may also stop taking part in the study at any time and not submit, without explanation or prejudice.

If you experience any upset or distress because of your participation in this research project, please seek assistance from available services in your local area, or contact your

- palliative care service for support.
- GP to discuss the possibility of a Mental Health Care Plan.
- employer for available employee assistance programs or health and wellbeing programs that offer confidential counselling and support.
- education institution for available health and wellbeing programs that offer students confidential counselling and support.

Do not forget your community support network which includes; friends, family members, religious leader, coach, work colleague, community leaders.

If you or someone near you is in immediate danger:

- Call Emergency Services on 000; or
- Go to a hospital emergency department

If you are having suicidal thoughts and need someone to talk to:

- Call Lifeline on 13 11 14
- Call 13 Yarn for support for Aboriginal and Torres Strait Islander people at 13 92 76
- Talk to someone

Alternatively, you can contact several free services directly, including

- The Black Dog Institute
 Online resources
 www.blackdoginstitute.org.au/
- Beyond Blue 1300 224 636 www.beyondblue.org.au
- Lifeline Australia
 13 11 14
 <u>www.lifeline.org.au/</u>
- Palliative Care Australia (PCA) Online resources <u>https://palliativecare.org.au/</u>
- Trauma & Grief Network
 Focus on Indigenous support
 <u>https://tgn.anu.edu.au/</u>
- PCA Grief & Loss booklet
 <u>https://palliativecare.org.au//wp-</u>
 <u>content/uploads/dlm_uploads/2015/05/PCA002_Understanding-Grief_W02.pdf</u>

Are there any benefits to taking part in this research?

The researcher cannot guarantee or promise that you will receive any clear benefits to you personally from participating in this research. There is no payment for taking part in this research. However, a possible benefit may include feeling a sense of value by providing meaningful insights into your journey through palliative care provision which may assist in shaping future palliative care provision for individuals and families. This may or may not directly or indirectly affect health professionals role in palliative care provision.

What happens to the information I provide?

The intention is to publish and report the results of this research project. All information published will done in a way that will not identify any participant.

You have the right to receive feedback about the study results. There will be a question in the survey asking if you wish to receive results at the completion of the survey. Alternatively you could email the researcher and request results at <u>Shirley.papavasiliou@my.jcu.edu.au</u>. You will receive a response once the study is completed and data has been analysed.

Anonymous survey data will be collected and stored electronically on password protected, external hard drive and secure James Cook University (JCU) cloud-based facilities (OneDrive). Data will be retained for 5 years post publication and then permanently deleted from all storage devices.

All research in Australia involving humans is reviewed by a Human Research Ethics Committee (HREC). This research project has been approved by the JCU HREC.

Who do I contact if I have any questions, require further information or wish to provide feedback about this research?

Please contact the following people:

Principal Investigator: Shirley Papavasiliou College: Business, Law & Governance James Cook University Email: <u>Shirley.papavasiliou@my.jcu.edu.au</u> Supervisor: Associate Professor Carmen Reaiche College: Business, Law & Governance James Cook University Email: <u>carmen.reaiche@jcu.edu.au</u>

Secondary Supervisor: Dr Reza Akbari College: Business, Law & Governance James Cook University Email: <u>reza.akbari@jcu.edu.au</u>

If you have any concerns regarding the ethical conduct of the study, please contact: Human Ethics, Research Office James Cook University, Townsville, Qld, 4811 Phone: (07) 4781 5011 (<u>ethics@jcu.edu.au</u>)

Appendix H: Grief Supports

Q1.2

Grief Supports If the subject or any questions in this survey are likely to trigger feelings that you may find upsetting, then please consider if you wish to continue. Triggers may include some mental health diagnoses, being recently bereaved or receiving bereavement support for an extended period of time, or involved in other recent traumatic events such as bushfires or flooding. If you do participate then you may decide to withdraw from the research at any time you choose. If you do not wish to answer a question, you may choose to not answer and go to the next question. The only mandatory questions are the initial questions used to ensure your eligibility. You may also stop taking part in the study at any time and not submit, without explanation or prejudice. Prior to participating in this survey please remember that grief has no timeline. It can begin long before death arrives. It is not unusual for grief to be felt over an extended period of time, even for many years. Sometimes we may experience feelings of anxiety, dread, sadness, longing, regret, loneliness, thoughts of 'if only' and more. Grief is not just accepting death, but the losses occurring as an illness progresses. Feelings may spring up unexpectedly or be triggered by an event or a memory. These triggers can re-kindle emotions and be upsetting or bring memories of great joy and gratitude for happier times.

If you or someone near you is in immediate danger:

- Call Emergency Services on 000; or
- Go to a hospital emergency department If you are having suicidal thoughts and need someone to talk to:
- Call Lifeline on 13 11 14
- Call 13 Yarn for support for Aboriginal and Torres Strait Islander people at 13 92 76
- Talk to someone

If you become upset or distressed because of your participation in this research project, please seek assistance from available services in your local area, or contact your

• palliative care service for support.

• GP to discuss the possibility of a Mental Health Care Plan. This will entitle you to up to 10 Medicare rebated mental health sessions with a mental health professional, or

• employer for available employee assistance programs or health and wellbeing programs that offer confidential counselling and support.

• education institution for available health and wellbeing programs that offer confidential counselling and support.

• do not forget your community support network which includes; friends, family members, church minister or religious leader, coach, work colleague or community leader.

Alternatively, you can contact several free services directly, including

- <u>The Black Dog Institute</u>
- Beyond Blue 1300 224 636
- Lifeline Australia 13 11 14
- Palliative Care Australia (PCA)
- Indigenous Trauma and Grief Network
- PCA Grief & Loss booklet

Appendix I: Quantitative Results

Summary survey access

Table 11 Summary Survey Access

Detail	n	n
Accessed survey link		182
Accessed Participant Information Sheet	21	
Accessed Grief Supports	32	
Declined Informed Consent		6
Opted out during survey		4
Removed when identified as working outside of Australia		1
Total submitted survey for analysis		171

Participants by group

Table 12 Participants by group

Group	Peri-urban	Rural	Regional	Metropolitan	Total
	n	n	n	n	n
Patient	14	9	4	3	30
Family/caregiver	57	12	7	15	91
Health Professional	35	5	3	7	50
Total	106	26	14	25	171

Data analysis conducted using actual responses received for each question which is reflected in the findings, as:

- survey questions not required to identify participants and gain informal consent, were voluntary,
- responses identified as 'unsure' or 'prefer not to answer' were not included,
- responses identified as 'other' that provided qualitative data, were included in thematic analysis, and
- multiple responses were allowed for some questions.

Family Members/Care-givers group

Table 13 Question 2.1

Q2.1 Family - Your age group?										
Age	Peri-	urban	rban Rural		Regional		Metropolitan		Total responses	
	%	n	%	n	%	n	%	n	%	n
18-34 years	14.5	5	5.6	1	25.0	1	12.5	2	12.5	9
35-49 years	29.4	10	33.3	6	25.0	1	25.0	4	29.2	21
50-64 years	38.2	13	38.9	7	25.0	1	50.0	8	40.3	29
65+	17.7	6	22.2	4	25.0	1	12.5	2	18.0	13
Total	47.2	34	25.0	18	5.6	4	22.2	16	100.0	72

Table 14 Question 2.2

Q2.2 Family - How long have you been receiving community palliative care?										
Receiving care	Peri-u	ırban	n Rural		Regional		Metropolitan		Total responses	
	%	n	%	n	%	n	%	n	%	n
<1 month	25.6	11	9.1	2	14.3	1	6.2	1	17.0	15
1-6 months	48.8	21	59.1	13	28.6	2	50.0	8	50.0	44
6-12 months	18.6	8	22.7	5	14.3	1	25.0	4	12.5	11
>1 years	7.0	3	9.1	2	42.8	3	18.8	3	12.5	11
Total	48.9	43	25.0	22	7.9	7	18.2	16	100.0	88

Table 15 Question 2.3

Q2.3 Family - What is your relationship to the person diagnosed with a life-limiting illness?											
Relationship	Peri-urban		tionship Peri-urban Rura		al	l Regional		Metropolitan		Total responses	
	%	n	%	n	%	n	%	n	%	n	
Spouse	40.0	18	52.2	12	42.9	3	25.0	4	40.7	37	
Child/Step-child	44.4	20	21.7	5	42.8	3	62.5	10	41.8	38	
Sibling/extended family/ friend	15.6	7	26.1	6	14.3	1	12.5	2	17.6	16	
Total	49.4	45	25.3	23	7.7	7	17.6	16	100.0	91	

Table 16 Question 2.4

Q2.4 Family - In what State/Territory are services provided?										
State/Territory	Peri-urban		Rur	Rural		onal	Metro	politan	Total responses	
-	%	n	%	n	%	n	%	n	%	n
Queensland	8.9	4	0	0	14.2	1	6.2	1	6.6	6
South Australia	66.7	30	91.3	21	28.6	2	37.5	6	64.8	59
NT and WA	0	0	0	0	0	0	0	0	0	0
ACT	4.4	2	0	0	0	0	0	0	2.2	2
New South Wales	8.9	4	0	0	42.9	3	25.0	4	12.1	11
Victoria	11.1	5	8.7	2	14.3	1	25.0	4	13.2	12
Tasmania	0	0	0	0	0	0	6.3	1	1.1	1
Total	49.4	45	25.3	23	7.7	7	17.6	16	100.0	91

Table 17 Question 2.5

Q2.5 Family - How would you describe the area where community palliative care is provided?							
	%	n					
Peri-urban	49.4	45					
Rural	25.3	23					
Regional	7.7	7					
Metropolitan	17.6	16					
Total	100.0	91					

Table 18 Question 2.7

Q2.7 (Family) How much do you agree with the following statements?										
	Peri-	urban	Rur	al	Regio	onal	Metro	politan	Total r	esponses
	%	n	%	n	%	n	%	n	%	n
2.7(1) (Family) Existing he	alth se	rvices	were c	onsid	dered w	vhen	selecting	g the hoi	me locati	ion
Strongly disagree	28.9	13	39.1	9	28.6	2	18.7	3	29.6	27
Somewhat disagree	8.9	4	34.8	8	14.3	1	25.0	4	18.7	17
Neither agree/disagree	6.7	3	0	0	0	0	0	0	3.3	3
Somewhat agree	28.9	13	26.1	6	42.8	3	50.0	8	33.0	30
Strongly agree	26.6	12	0	0	4.3	1	6.3	1	15.4	14
Total	49.4	45	25.3	23	7.7	7	17.6	16	100.0	91
2.7(2) (Family) The locatio stage	n was	selecte	d as he	alth	service	es we	ere in the	plannin	g develo	opment
Strongly disagree	60.0	27	60.9	14	42.9	3	75.0	3	61.5	56
Somewhat disagree	15.6	7	21.7	5	28.6	2	18.8	3	18.7	17
Neither agree/disagree	4.4	2	8.7	2	14.3	1	6.2	1	6.6	6
Somewhat agree	17.8	8	8.7	2	14.3	1	0	0	12.1	11
Strongly agree	2.2	1	0	0	0	0	0	0	1.1	1
Total	49.4	45	25.3	23	7.7	7	17.6	16	100.0	91
2.7(3) (Family) Considerat	ion wa cility) w	s given vhen se	to acc	essii the	ng serv home l	vicers ocati	for pos	sible fut	ure healt	h needs
Strongly disagree	53.5	24	56.5	13	42.8	3	56.3	9	53.8	49
Somewhat disagree	22.2	10	26.2	6	14.3	1	12.5	2	20.9	19
Neither agree/disagree	0	0	4.3	1	0	0	0	0	1.1	1
Somewhat agree	11.2	5	13.0	3	28.6	2	18.7	3	14.3	13
Strongly agree	13.3	6	0	0	14.3	1	12.5	2	9.9	9
Total	49.4	45	25.3	23	7.7	7	17.6	16	100.0	91
2.7(4) (Family) There wa	s an e	cpecta	tion th	at h	ealth s	ervic	es were	accessi	ble	
Strongly disagree	0	0	0	0	0	0	0	0	0	0
Somewhat disagree	0	0	0	0	0	0	0	0	0	0
Neither agree/disagree	0	0	4.3	1	0	0	0	0	1.1	1
Somewhat agree	20.0	9	52.2	12	28.6	2	12.5	2	27.5	25
Strongly agree	80.0	36	43.5	10	71.4	5	87.5	14	71.4	65
Total	49.4	45	25.3	23	7.7	7	17.6	16	100.0	91

Table 19 Question 2.8

Q2.8 Family - How did you receive/find information about available services? Select all that apply.										
Source	Peri-	Peri-urban		Rural		Regional		oolitan	Total responses	
	%	n	%	n	%	n	%	n	%	n
Health Professional	64.4	29	56.5	13	42.9	3	81.3	13	63.7	58
Referral	77.8	35	73.9	17	57.1	4	7.5	12	74.7	68
Friends/family	22.2	10	21.8	5	42.9	3	18.8	3	23.1	21
Pamphlets	4.4	2	0	0	0	0	0	0	2.2	2
Online forums	6.7	3	4.3	1	42.9	3	18.8	3	11.0	10
Support group	15.6	7	4.3	1	28.6	2	43.8	7	18.7	17
Google	15.6	7	8.7	2	0	0	31.2	5	15.4	14
Experience	4.4	2	4.3	1	0	0	0	0	3.3	3
Not involved	0	0	4.3	1	0	0	0	0	1.1	1
Other	26.7	12	17.4	4	28.6	2	12.5	2	22.0	20
Total responses	50.0	107	21.0	45	8.0	17	21.0	45	100.0	214

Table 20 Question 2.9

Q2.9 Family - Did you find it difficult to find information about service/s and identify what was needed?										
	Peri-urban		Rur	al	Regio	onal	Metropolitan		Total responses	
	%	n	%	n	%	n	%	n	%	n
Yes	60.0	27	65.0	13	71.4	5	66.7	10	63.2	55
No	40.0	18	35.0	7	28.6	2	33.3	5	36.8	32
Total	51.7	45	23.0	20	8.0	7	17.3	15	100.0	87

Table 21 Question 2.10

Q2.10 Family - Did you find it difficult to access identified services when required?										
	Peri-	urban	Rur	al	Regio	onal	Metro	oolitan	Total r	esponses
	%	n	%	n	%	n	%	n	%	n
Yes	72.7	32	80.0	16	71.4	5	60.0	9	72.1	62
No	27.2	12	20.0	4	28.6	2	40.0	6	27.9	24
Total	51.2	44	23.3	20	8.1	7	17.4	15	100.0	86

Table 22 Question 2.10

Q2.10 Family - Did you find it difficult to access identified services when required?								
	Minimum	Maximum	Mean	Std Deviation	Variance	Count		
False	1.00	2.00	1.75	0.43	0.19	4		
True	1.00	2.00	1.26	0.44	0.19	82		

Table 23 Question 2.11

Q2.11 Family - Please select all the services that you kn	ow are being received or have been received in
the home. Select all that apply	multiple responses
allowed	

Services	Peri-urban		Rural		Regional		Metropolitan		Total responses	
	%	n	%	n	%	n	%	n	%	n
Unsure	11.1	5	8.7	2	0	0	0	0	7.7	7
Specialist Palliative Care (SPC)	73.3	33	47.8	11	42.9	3	43.7	7	59.3	54
Community Nursing	24.4	11	39.1	9	28.6	2	43.7	7	31.9	29
Personal care	33.3	15	34.8	8	57.1	4	62.5	10	40.7	37
Transport	8.9	4	0	0	14.3	1	37.5	6	12.1	11
Occupational Therapy	29.9	13	39.1	9	14.3	1	25.0	4	29.7	27
Psychologist	4.4	2	4.3	1	14.3	1	12.5	2	6.6	6
Social Worker	29.3	13	34.8	8	28.6	2	12.5	2	27.5	25
Physiotherapy	4.4	2	21.7	5	14.3	1	6.2	1	9.9	9
Dietician/Speech Pathology	11.1	5	21.7	5	28.6	2	12.5	2	15.4	14
Palliative Care Specialist	8.9	4	21.7	5	0	0	12.5	2	12.1	11
Spiritual	0	0	2.2	1	0	0	6.2	1	2.2	2
Volunteers	6.7	3	0	0	0	0	0	0	3.3	3

Complementary	0	0	0	0	0	0	6.2	1	1.1	1
House/Garden care	37.8	17	17.4	4	57.1	4	56.2	9	37.4	34
Home maintenance	8.9	4	8.7	2	14.3	1	43.7	7	15.4	14
Equipment	42.2	19	47.8	11	42.9	3	56.2	9	46.2	42
ACD assist	13.3	6	17.4	4	0	0	12.5	2	13.2	12
Financial	2.2	1	18.0	3	14.3	1	12.5	2	7.7	7
GP	11.1	5	13.0	3	42.9	3	31.2	5	17.6	16
End-of-life care	11.1	5	17.4	4	0	0	18.7	3	2.7	12
Bereavement	13.3	6	4.3	1	0	0	12.5	2	9.9	9
Family/friends	26.7	12	39.1	9	85.7	6	25.0	4	34.1	31
Other	40.0	18	47.8	11	47.1	4	25.0	4	40.6	37
Total responses	45.1	203	25.8	116	8.7	39	20.4	92	100.0	450
Total respondents	49.4	45	25.3	23	7.7	7	17.6	16	100.0	91

Table 24 Question 2.12

Q2.12 Family - Do services received meet needs and expectations?											
	Peri-urban		Rural		Regional		Metrop	olitan	Total responses		
	%	n	%	n	%	n	%	n	%	n	
Yes	24.4	11	21.7	5	28.6	2	18.7	3	23.1	21	
Partially	35.6	16	47.8	11	0	0	43.8	7	37.4	34	
No	40.0	18	30.4	7	71.4	5	37.5	6	39.5	36	
Total	49.4	45	25.3	23	7.7	7	17.6	16	100.0	91	

Table 25 Question 2.13

Select all that apply Multiple responses allowed												
Reasons	Peri-urban		Rural		Regional		Metropolitan		Total responses			
	%	n	%	n	%	n	%	n	%	n		
Not required	13.3	6	13.0	3	14.3	1	18.7	3	14.3	13		
Family provide	33.3	15	30.0	7	57.1	4	12.5	2	30.8	28		
Not happy with services previously	2.2	1	0	0	14.3	1	6.2	1	3.3	3		
Not available when I want	42.2	19	26.1	6	28.6	2	31.2	5	35.2	32		
Not available as often as I want	40.0	18	30.4	7	42.9	3	25.0	4	35.2	32		
Not available home location	22.2	10	60.9	14	42.9	3	0	0	29.7	27		
No resources available	22.2	10	39.1	9	42.9	3	0	0	24.2	22		
Too expensive	20.0	9	13.0	3	14.3	1	12.5	2	16.5	15		
Using all we know about	11.1	5	4.3	1	28.6	2	12.5	2	11.0	10		
Personal choice to use/not use	6.7	3	0	0	0	0	0	0	3.3	3		
Unsure reason why	26.7	12	17.4	4	28.6	2	18.7	3	23.1	21		
Total responses	52.4	108	26.2	54	10.7	22	10.7	22	100.0	206		
Total respondents	49.4	45	25.3	23	7.7	7	17.6	16	100.0	91		
Table 26 Question 2.14

Q2.14 Family - Please select any of the following that apply											
	Peri-	urban	Rui	ral	Region	al	Metropolita n		Total responses		
	%	n	%	n	%	n	%	n	%	n	
Q2.14(1) (Family) One or more hospital stays have been extended as services unavailable in-home to support discharge	85.7	30	38.9	7	66.7	4	90.9	10	72.9	51	
Q2.14(2) (Family) Needed to attend clinic or hospital outpatients for medication administration as in-home services unavailable	17.1	6	11.1	2	33.3	2	0	0	14.3	10	
Q2.14(3) (Family) Needed to attend clinic or hospital outpatients for assistance with wound/catheter/ stoma care as in-home services unavailable	17.1	6	22.2	4	16.7	1	9.1	1	17.1	12	
Q2.14(4) (Family) Needed to call an ambulance for assistance with pain/symptom control as unable to contact community services	51.4	18	55.6	10	50.0	3	27.3	3	48.6	34	
Q2.14(5) (Family) Respite in aged care facility was required as in-home services unavailable to support hospital discharge	20.0	7	11.1	2	0	0	0	0	12.9	9	
Q2.14(6) (Family) Will need to transfer to hospital for end-of-life care to remain in local area	51.4	18	77.8	14	100.0	6	27.3	3	58.6	41	
Total respondents	50.0	35	25.7	18	8.6	6	15.7	11	100.0	70	

Table 27 Question 2.15

Q2.15 Family - What is the longest distance your family member has needed to travel for healthcare services?												
Distance	Peri-u	rban	Rural		Region		nal Metropo		Total res	sponses		
	%	n	%	n	%	n	%	n	%	n		
Services local	0	0	0	0	0	0	37.5	6	6.6	6		
10-50 km	11.1	5	4.3	1	28.6	2	56.2	9	18.7	17		
50-100 km	71.1	32	34.8	8	0	0	0	0	44.0	40		
100-150 km	8.9	4	34.8	8	14.3	1	0	0	14.3	13		
150-250 km	2.2	1	26.1	6	0	0	0	0	7.7	7		
Over 250 km	6.7	3	0	0	57.1	4	6.3	1	8.8	8		
Unsure	0	0	0	0	0	0	0	0	0	0		
Total	49.4	45	25.3	23	7.7	7	17.6	16	100.0	91		

Table 28 Question 2.16

Q2.16 Family - What barriers to accessing services do you believe apply in your situation?													
Select all that apply Multiple responses allowed													
Barriers	Peri-	urban	ırban Ru		Regional		Metropolitan		Total responses				
	%	n	%	n	%	n	%	n	%	n			
No barriers	8.9	4	4.3	1	0	0	25.9	4	9.9	9			
Under 65 years	22.2	10	17.4	4	28.6	2	37.5	6	24.2	22			
Distance travel to home	33.3	15	82.6	19	57.1	4	0	0	41.8	38			
Lack resources	35.5	16	52.2	12	28.6	2	6.2	1	34.1	31			
Time taken to start	22.2	10	4.3	1	0	0	0	0	12.1	11			
Distance travel from	28.9	13	65.2	15	57.1	4	0	0	35.2	32			
Diversity	0	0	0	0	0	0	6.2	1	1.1	1			
Language	2.2	1	0	0	0	0	6.2	1	2.2	2			
Palliative care stigma	6.7	3	0	0	0	0	6.2	1	4.4	4			
Behavioural challenges	11.1	5	0	0	28.6	2	0	0	7.7	7			
Financial	8.9	4	21.7	5	0	0	6.2	1	11.0	10			
Personal choice	8.9	4	8.7	2	0	0	0	0	6.6	6			
Beyond scope of local providers	13.3	6	39.1	9	42.8	3	0	0	19.8	18			
No need for services	0	0	0	0	0	0	6.2	1	1.1	1			
Total responses	47.5	91	35.4	68	8.8	17	8.3	16	1.0	192			
Total respondents	49.4	45	25.3	23	7.7	7	17.6	16	100.0	91			

Table 29 Question 2.17

Q2.17 Family - Thinking about transport from home to the location of some health services,												
please select all statements that apply to your family. What is your relationship to the person diagnosed with a life-limiting illness?												
diagnocoa mina a no mina	Peri-u	urban	Rur	al	Regio	onal	Metrop	olitan	Total re	esponses		
	%	n	%	n	%	n	%	n	%	n		
Q2.17(1) Family - The time/distance involved has influenced decisions about attending appointments, treatment or tests	55.8	24	69.6	16	71.4	5	6.3	1	51.7	46		
Q2.17(2) Family - The time/distance has influenced the home location of the person requiring care in the home	32.6	14	17.4	4	14.3	1	6.3	1	22.5	20		
Q2.17(3) Family - Travel time/distance is not a problem	9.3	4	4.3	1	0	0	18.8	3	9.0	8		
Q2.17(4) Family - Travel any distance is difficult (exhausting, pain, discomfort, etc)	79.1	34	82.6	19	100	7	62.5	10	78.7	70		
Q2.17(5) Family - Private transport is difficult due to cost of petrol, parking, lost wages and time involved	81.4	35	78.3	18	100	7	43.8	7	75.3	67		
Q2.17(6) Family - Public transport is not an option	86.0	37	96.3	21	85.7	6	56.3	9	82.0	73		
Q2.17(7) Family - Community car is not available/expensive/not an option	46.5	20	60.9	14	57.1	4	25.0	4	47.2	42		

Q2.17(8) Family - All health services are provided in the home	2.3	1	4.3	1	0	0	6.3	1	3.4	3
Q2.17(9) Family - All health services are provided locally	2.3	1	4.3	1	42.9	3	31.3	5	11.2	10
Q2.17(10) Family - Need to relocate to be closer to services that will provide a higher level of care	39.5	17	21.7	5	42.9	3	0	0	28.1	25
Total respondents	48.3	43	25.8	23	7.9	7	18.0	16	100.0	89

Table 30 Question 2.18

Q2.18 Family - How much do	o you a	gree w	ith the	followi	ng stat	ement	s?			
	Peri-	urban	Ru	iral	Regio	onal	Metrop	olitan	Total respon	ses
	%	n	%	n	%	n	%	n	%	n
2.18(1) (Family) The choice t	o rema	in in o	wn hon	ne has	been s	uppor	ted			
Strongly disagree	34.1	15	13.0	3	0	0	7.6	1	1.8	19
Somewhat disagree	13.6	6	17.4	4	57.1	4	23.1	3	19.5	17
Neither agree/disagree	11.4	5	8.7	2	0	0	23.1	3	11.5	10
Somewhat agree	9.1	4	21.7	5	42.9	3	23.1	3	17.2	15
Strongly agree	31.8	14	39.1	9	0	0	23.1	3	29.9	26
Total	50.6	44	26.4	23	8.1	7	14.9	13	100.0	87
2.18(2) (Family) Personal che	oices, p	orefere	nces a	nd goa	ls of ca	are hav	e been r	espected	b	
Strongly disagree	23.3	10	13.0	3	0	0	7.7	1	16.3	14
Somewhat disagree	18.6	8	21.7	5	42.9	3	30.8	4	23.3	20
Neither agree/disagree	9.3	4	8.8	2	0	0	15.4	2	9.3	8
Somewhat agree	20.9	9	13.0	3	57.1	4	23.1	3	22.1	19
Strongly agree	27.9	12	43.5	10	0	0	23.0	3	29.0	25
Total	50.0	45	26.8	23	8.1	7	15.1	13	100.0	86
2.18(3) (Family) All commun	ication	and in	formati	ion pro	vided	has be	en respe	ctful and	l approp	riate
for our needs		_		-	-	-	-	-		
Strongly disagree	16.3	7	8.7	2	0	0	0	0	10.6	9
Somewhat disagree	29.3	4	13.0	3	28.6	2	16.7	2	12.9	11
Neither agree/disagree	13.8	6	21.7	5	28.6	2	41.6	5	21.2	18
Somewhat agree	32.6	14	8.7	2	28.6	2	16.7	2	23.5	20
Strongly agree	27.9	12	47.8	11	4.2	1	25.0	3	31.8	27
Total	50.6	43	27.1	23	8.2	7	14.1	12	100.0	85
2.18(4) (Family) Access to su	upport	is avai	able 24	4 hours	s a day	7 days	a week			
Strongly disagree	31.8	14	17.4	4	0	0	0	0	20.9	18
Somewhat disagree	36.4	16	34.8	8	71.4	5	33.3	4	38.4	33
Neither agree/disagree	15.9	7	8.7	2	0	0	16.7	2	12.8	11
Somewhat agree	4.5	2	21.7	5	28.6	2	25.0	3	13.9	12
Strongly agree	11.4	5	17.4	4	0	0	25.0	3	14.0	12
Total	51.2	44	26.7	23	8.1	7	14.0	12	100.0	86
2.18(5) (Family) Quality of lif	e has b	een im	proved	d by co	mmun	ity serv	/ices			
Strongly disagree	25.0	11	8.7	2	0	0	0	0	14.9	13
Somewhat disagree	13.6	6	17.4	4	42.8	3	23.1	3	18.4	16
Neither agree/disagree	13.6	6	21.7	5	14.3	1	38.5	5	19.6	17
Somewhat agree	25.0	11	17.4	4	42.3	3	15.4	2	23.0	20
Strongly agree	22.7	10	34.8	8	0	0	23.1	3	24.1	21
Total	50.6	44	26.4	23	8.1	7	14.9	13	100.0	87
2.18(6) (Family) The need for in-home services	r ambu	lance t	ranspo	rt and	admiss	sion to	hospital	has bee	n reduce	ed by
Strongly disagree	23.8	10	18.2	4	14.3	1	0	0	18.1	15
Somewhat disagree	31.2	13	36.4	8	42.9	3	41.7	5	34.9	29
Neither agree/disagree	19.0	8	13.6	3	28.6	2	25.0	3	19.3	16

Somewhat agree	9.5	4	18.2	4	14.3	1	8.3	1	12.0	10
Strongly agroo	16.7	7	13.6	3	0	0	25.0	3	15.7	12
	50.C	1	10.0 26 E	2	0 1	7	20.0	12	10.7	02
101ai	50.0	42	20.5		0.4	1	14.5	12	100.0	03
2.18(7) (Family) Family mem	bers na	ive bee	en incit	idea in	all dec	cision-	making a	about ca	re and	
Strongly disagree	23.8	10	87	2	1/1 3	1	0	0	15.5	13
Somowhat disagree	16.7	7	12.0	2	17.0	2	16.7	2	17.0	15
Noither agree/diaggree	10.7	/ E	13.0	5	42.9	3	10.7	2	20.2	17
Neither agree/disagree	11.9	5	20.1	0	14.3	1	41.7	5	20.2	17
Somewhat agree	28.0	12	13.0	3	14.3		10.7	2	21.4	18
Strongly agree	19.0	8	39.1	9	14.3	1	25.0	3	25.0	21
	50.0	42	27.4	23	8.3	7	14.3	13	100.0	84
2.18(8) (Family) Physical syn	nptoms	s are m	anageo	d in the	home			-		
Strongly disagree	25.6	11	21.7	5	0	0	16.7	2	21.2	18
Somewhat disagree	30.2	13	17.4	4	57.1	4	25.0	3	28.2	24
Neither agree/disagree	13.9	6	8.7	2	14.3	1	25.0	3	14.1	12
Somewhat agree	13.9	6	39.1	9	14.3	1	0	0	18.8	16
Strongly agree	16.3	7	13.0	3	14.3	1	33.3	4	17.7	15
Total	50.6	43	27.1	23	8.2	7	14.1	12	100.0	85
2.18(9) (Family) Support has	been p	orovide	d with	advan	ce care	plann	ing, com	pleting	will and g	grief
counselling	-					-	•			-
Strongly disagree	34.9	15	9.1	2	0	0	7.7	1	21.2	18
Somewhat disagree	30.2	13	22.7	5	57.2	4	46.2	6	32.9	28
Neither agree/disagree	6.9	3	9.1	2	14.3	1	15.4	2	9.4	8
Somewhat agree	14.0	6	40.9	9	28.5	2	7.7	1	21.2	18
Strongly agree	14.0	6	18.2	4	0	0	23.0	3	15.3	13
Total	50.6	43	25.9	22	8.2	7	15.3	13	100.0	85
2.18(10) (Family) Financial as	ssistan	ce nee	ded ha	s been	provid	ded				
Strongly disagree	40.9	18	17.4	4	0	0	16 7	2	27.9	24
Somewhat disagree	27.3	12	17.4	4	71.0	5	50.0	6	31.4	27
Neither agree/disagree	15.0	7	26.1	6	1/1.3	1	83	1	17 /	15
Somewhat agree	6.8	3	34.8	8	14.3	1	0.0	0	17.4	12
Strongly agroo	0.0	1	12	1	0	0	25.0	3	0.2	0 0
	51 2	4	4.0 26 7	1	01	7	23.0	12	100 0	0
2 19(11) (Eamily) All aupport	ond or	44 Jucatia	20.7	23 ired by	0.1 fomily	/ momb	14.0	12 rovido or		00
care including end-of-life ca	ro and	lucatio	no doat	h hae	hoon n	rovido	d d	loviue se	ale, quai	Ly
Strongly disagree	32.6	14	13.0	3	14.3	1	15.4	2	23.2	20
Somewhat disagree	27.9	12	21.7	5	42.0	3	30.7	4	27.9	24
Neither agree/disagree	16	2	17 /	4	28.6	2	23.1	т 2	12.8	11
Somewhat agree	20.0	0	17.4	4	1/ 3	1	77	1	17.0	15
Strongly agroo	14.0	6	30.4	7	0	0	7.7 22.1	3	19.6	16
	14.0 50.0	42	30.4	1	04	7	45.0	12	10.0	10
	50.0	43	20.7	23	0.1	/	15.2	13	100.0	00
2.10(12) (Family) All Services		been p			puyan			ג ר ו	20.2	24
	30.0 27.2	10	10.2	4	14.3	1 2	10.7	<u>ک</u>	20.2	∠4 26
Somewhat disagree	27.3	12	31.8	1	42.8	3	33.3	4	30.0	20
iveither agree/disagree	4.5	2	4.5	1	14.3	1	10.7	2	1.1	0
Somewhat agree	11.4	5	27.3	6	28.6	2	8.3	1	16.5	14
Strongly agree	18.2	8	18.2	4	0	0	25.0	3	17.6	15
Total	51.8	44	25.9	22	8.2	7	14.1	12	100.0	85
2.18(13) (Family) Medication	is avai	lable lo	ocally							
Strongly disagree	2.4	1	8.7	2	0	0	0	0	3.5	3
Somewhat disagree	0	0	30.4	7	14.3	1	7.7	1	10.6	9
Neither agree/disagree	16.7	7	17.4	4	0	0	0	0	12.9	11
Somewhat agree	57.1	24	17.4	8	85.7	6	38.5	5	50.6	43
Strongly agree	23.8	10	8.7	2	0	0	53.8	7	22.4	19
Total	49.4	42	27.1	23	8.2	7	15.3	13	100.0	85
2.18(14) (Family) Choices for	end-o	f-life ca	are incl	uding	hospic	e, hom	e death	and volu	intary	
assisted dying, have been so	upporte	ed and	have e	nabled	inform	ned de	cisions		-	-
Strongly disagree	34.1	15	17.4	4	14.3	1	23.1	3	26.4	23
Somewhat disagree	25.0	11	34.8	8	71.4	5	46.1	6	34.5	30

Neither agree/disagree	4.6	2	8.7	2	0	0	0	0	4.6	4
Somewhat agree	13.6	6	17.4	4	14.3	1	7.7	1	13.8	12
Strongly agree	22.7	10	21.8	5	0	0	23.1	3	20.7	18
Total	50.6	44	26.4	23	8.1	7	14.9	13	100.0	87
2.18(15) (Family) Available so	ervices	are af	fordabl	e						
Strongly disagree	11.9	5	9.1	2	0	0	8.3	1	9.7	8
Somewhat disagree	16.7	7	27.3	6	42.9	3	8.3	1	20.5	17
Neither agree/disagree	40.5	17	36.4	8	28.6	2	41.7	5	38.5	32
Somewhat agree	19.0	8	9.1	2	28.6	2	16.7	2	16.9	14
Strongly agree	1.9	5	18.2	4	0	0	25.0	3	14.4	12
Total	50.6	42	26.5	22	8.4	7	14.5	12	100.0	83
2.18(16) (Family) Staff provid	ling ca	re have	e the tra	aining,	knowle	edge a	nd exper	tise to p	rovide	
quality of care needed			l		l			I	I	
Strongly disagree	9.5	4	13.6	3	0	0	0	0	8.4	7
Somewhat disagree	4.8	2	18.2	4	14.3	1	8.3	1	9.6	8
Neither agree/disagree	23.8	10	13.6	3	28.6	2	41.7	5	24.1	20
Somewhat agree	40.5	17	22.7	5	42.9	3	25.0	3	33.8	28
Strongly agree	21.4	9	31.8	7	14.3	1	25.0	3	24.1	20
Total	50.6	42	26.5	22	8.4	7	14.5	12	100.0	83
2.18(17) (Family) Choice of p	rovide	r and/o	r staff	who at	tend he	ome vi	sits is pr	ovided		
Strongly disagree	19.0	8	27.3	6	14.3	1	8.3	1	19.3	16
Somewhat disagree	33.3	14	31.8	7	57.1	4	50.0	6	37.4	31
Neither agree/disagree	31.0	13	22.7	5	0	0	16.7	2	24.1	20
Somewhat agree	4.8	2	18.2	4	28.6	2	0	0	9.6	8
Strongly agree	11.9	5	0	0	0	0	25.0	3	9.6	8
Total	50.6	42	26.5	22	8.4	7	14.5	12	100.0	83
2.18(18) (Family) Family/care	giver s	tress a	and bui	rnout h	as bee	n redu	ced with	service	provisio	n
Strongly disagree	29.5	13	18.2	4	14.3	1	23.1	3	24.4	21
Somewhat1 disagree	31.8	14	22.7	5	57.1	4	38.5	5	32.6	28
Neither agree/disagree	6.8	3	9.1	2	0	0	15.4	2	8.1	7
Somewhat agree	6.8	3	18.2	4	28.6	2	0	0	10.5	9
Strongly agree	25.0	11	31.8	7	0	0	23.1	3	24.4	21
Total	51.2	44	25.6	22	8.1	7	15.1	13	100.0	86
2.18(19) (Family) Patient/fam	ily hav	e good	relatio	onships	s with h	ealth p	orofessio	onals inv	olved in	care
Strongly disagree	14.0	6	9.1	2	0	0	0	0	9.4	8
Somewhat disagree	9.3	4	13.6	3	0	0	15.4	2	10.6	9
Neither agree/disagree	14.0	6	18.2	4	28.6	2	30.8	4	18.8	16
Somewhat agree	30.2	13	9.1	2	42.9	3	15.4	2	23.6	20
Strongly agree	32.5	14	50.0	11	28.6	2	38.5	5	37.6	32
Total	50.6	43	25.9	22	8.2	7	15.3	13	100.0	85

Table 31 Question 2.18 (Descriptive Statistics)

Q2.18 Family - How much do you agree with the following statements?

otatomonio							
						Variance	Confidence
	n	min	max	mean	SD		Interval
2.18(1) (Family) The choice	to remai	n in own	home h	as been	support	ed	
Peri-Urban	44	1	5	2.9	1.7	2.8	0.25
Rural	23	1	5	3.6	1.5	2.1	0.30
Regional	7	2	4	2.8	1.0	1.0	0.40
Metro	13	1	5	3.3	1.3	1.6	0.40
Total	87	1	5	3.2	1.4	1.9	0.40
2.18(2) (Family) Personal ch	noices, p	referenc	es and g	joals of o	care hav	e been respe	cted
Peri-Urban	45	1	5	3.1	1.6	2.4	0.23
Rural	23	1	5	3.5	1.5	2.3	0.31
Regional	7	2	4	3.1	1.0	1.0	0.40
Metro	13	1	5	3.2	1.3	1.3	0.40
Total	86	1	5	3.2	1.4	1.8	0.30

2.18(3) (Family) All communication and information provided has been respectful and											
appropriate for our needs	40	4	-	0.5		0.0	0.00				
Peri-Urban	43	1	5	3.5	1.4	2.0	0.20				
	23	1	5	3.7	1.4	1.9	0.30				
Regional	1	2	5	3.2	1.0	1.1	0.40				
Metro	12	2	5	3.5	1.0	1.1	0.30				
Total	85	1	5	3.5	1.2	1.5	0.30				
2.18(4) (Family) Access to s	support is	s availab	ole 24 ho	urs a da	y 7 days	a week					
Peri-Urban	44	1	5	2.3	1.3	1.6	0.20				
Rural	23	1	5	2.9	1.4	1.9	0.30				
Regional	7	2	4	3.3	1.0	1.1	0.40				
Metro	12	2	5	3.5	1.1	1.0	0.30				
Total	86	1	5	3.0	1.2	1.4	0.30				
2.18(5) (Family) Quality of li	fe has be	en impr	oved by	commu	nity serv	ices					
Peri-Urban	44	1	5	3.1	1.5	2.3	0.20				
Rural	23	1	5	3.5	1.3	1.8	0.30				
Regional	7	2	4	3.1	0.9	0.9	0.30				
Metro	13	2	5	3.5	1.1	1.3	0.30				
Total	87	1	5	3.3	1.2	1.6	0.30				
2.18(6) (Family) The need for	or ambula	ance trar	nsport a	nd admis	ssion to	hospital has	been reduced				
by in-home services				-	-						
Peri-Urban	42	1	5	2.6	1.4	1.9	0.20				
Rural	22	1	5	2.5	1.2	1.4	0.30				
Regional	7	1	4	2.7	1.3	1.6	0.50				
Metro	12	2	5	3.5	1	1.1	0.30				
Total	83	1	5	2.8	1.2	1.5	0.30				
2.18(7) (Family) Family men	nbers hav	ve been	includec	d in all de	ecision-n	naking about	care and				
services						-					
Peri-Urban	42	1	5	3.0	1.5	2.2	0.20				
Rural	23	1	5	3.6	1.3	1.8	0.30				
Regional	7	1	5	2.7	1.3	1.6	0.50				
Metro	13	2	5	3.5	1	1.4	0.30				
Total	84	1	5	3.2	1.3	1.8	0.30				
2.18(8) (Family) Physical sy	mptoms	are man	aged in	the hom	е						
Peri-Urban	43	1	5	2.7	1.4	2.0	0.20				
Rural	23	1	5	3.0	1.4	2.0	0.30				
Regional	7	2	5	2.9	1.1	1.2	0.40				
Metro	12	1	5	3.1	1.4	1.9	0.40				
Total	85	1	5	2.9	1.3	1.8	0.30				
2.18(9) (Family) Support has	s been pr	rovided	with adv	ance car	re planni	ng, completii	ng will and				
grief counselling						-	-				
Peri-Urban	43	1	5	2.4	1.4	2.0	0.20				
Rural	22	1	5	3.4	1.3	1.6	0.30				
Regional	7	2	5	2.7	0.9	0.8	0.30				
Metro	13	1	5	2.9	1.3	1.8	0.40				
Total	85	1	5	2.8	1.2	1.6	0.30				
2.18(10) (Family) Financial a	assistanc	e neede	d has be	een prov	ided						
Peri-Urban	44	1	5	2.2	1.3	1.6	0.20				
Rural	23	1	5	2.9	1.2	1.4	0.20				
Regional	7	2	4	2.4	0.7	0.5	0.30				
Metro	12	1	5	2.7	1.4	2.0	0.40				
Total	86	1	5	2.6	1.2	1.4	0.30				
2.18(11) (Family) All suppor	t and edu	ucation	required	by famil	y memb	ers to provid	e safe, quality				
care, including end-of-life c	are and/o	or home	death ha	as been	provided	· ·					
Peri-Urban	43	1	5	2.6	1.5	2.2	0.20				
Rural	23	1	5	3.3	1.4	2.0	0.30				
Regional	7	1	4	2.4	0.9	0.8	0.30				
Metro	13	1	5	2.9	1.4	1.9	0.40				
Total	86	1	5	2.8	1.3	1.7	0.30				
2.18(12) (Family) All service	s have b	een prov	vided pro	omptly a	nd when	needed					
2.18(12) (Family) All service Peri-Urban	es have b 44	een prov	vided pro 5	omptly a 2.4	nd when 1.5	2.3	1.50				
2.18(12) (Family) All service Peri-Urban Rural	es have b 44 22	een prov 1 1	vided pro 5 5	omptly a 2.4 2.9	nd when 1.5 1.3	2.3	1.50 0.20				
2.18(12) (Family) All service Peri-Urban Rural Regional	es have b 44 22 7	een prov 1 1 1	vided pro 5 5 4	omptly a 2.4 2.9 2.4	nd when 1.5 1.3 0.9	2.3 1.5 0.8	1.50 0.20 0.30				

Total	85	1	5	2.7	1.3	1.7	0.60
2.18(13) (Family) Medication	ı is avail	able loca	ally				
Peri-Urban	42	1	5	4.0	0.8	0.6	0.10
Rural	23	1	5	3.0	1.2	1.3	0.20
Regional	7	1	4	3.7	0.7	0.5	0.30
Metro	13	1	5	4.4	0.8	0.7	0.20
Total	85	1	5	3.8	0.9	0.8	0.20
2.18(14) (Family) Choices for	or end-of	-life care	includi	ng hospi	ce, hom	e death and v	oluntary
assisted dying, have been s	supporte	d and ha	ve enab	led infor	med dec	isions	,
Peri-Urban	44	1	5	2.7	1.6	2.5	0.20
Rural	23	1	5	2.9	1.4	2.1	0.30
Regional	7	1	4	2.1	0.8	0.7	0.30
Metro	12	1	5	3.4	1.2	1.4	0.30
Total	83	1	5	2.8	1.3	17	0.30
2 18(15) (Family) Available	services	are affor	rdahle	2.0	1.0		0.00
Peri-I Irban	42	1	5	3.0	11	13	0.20
Rural	22	1	5	3.0	1.1	1.0	0.20
Regional	7	2	4	2.0	0.8	0.7	0.00
Metro	12	1	5	2.5	1.2	1.1	0.30
Total	83	1	5	3.4	1.2	1.4	0.30
2 19(16) (Eamily) Staff provi	ding oor	n hava ti	bo troiniu		I.I	I.Z	o provido
2.10(10) (Failing) Start provi	ung car	e nave li		ig, know	leuge al	iu expertise t	o provide
Pori Urban	10	1	5	26	10	1 2	0.20
Dural	42	1	5	3.0	1.2	1.3	0.20
Ruiai	- 22	1	5	3.4	1.4	2.0	0.30
Netro	10	2	5	3.0	0.9	0.3	0.90
	12	<u> </u>	5	3.7	0.9	0.9	0.30
	83	I	5	3.0	1.1	1.2	0.40
2.18(17) (Family) Choice of	provider	and/or s	staπ who	attend	nome vis	its is provide	a
Peri-Urban	42	1	5	2.6	1.2	1.4	0.20
Rural	- 22	1	4	2.3	1.1	1.1	0.30
Regional	1	1	4	2.4	1	1.1	0.40
Metro	12	1	4	3.7	0.9	0.9	0.30
Total	83	1	5	2.8	1.1	1.1	0.30
2.18(18) (Family) Family/car	egiver s	tress and	d burnou	it has be	en reduc	ced with serv	ce provision
Peri-Urban	44	1	5	2.7	1.6	2.5	0.20
Rural	22	1	5	2.3	1.1	1.1	0.20
Regional	7	1	4	2.4	1	1.1	0.40
Metro	13	1	5	2.6	1.4	2.0	0.40
Total	86	1	5	2.5	1.3	1.7	0.30
2.18(19) (Family) Patient/far	nily have	e good re	elationsh	nips with	health p	rofessionals	involved in
care							
Peri-Urban	43	1	5	3.6	1.4	1.9	0.20
Rural	22	1	5	3.8	1.4	2.0	0.30
Regional	7	3	5	4.0	0.8	0.6	0.30
Metro	13	2	5	3.8	1.1	1.2	0.30
Total	85	1	5	3.8	1.2	1.4	0.30

Table 32 Question 2.19

Q2.19 Family - How are palliative care services provided in your home? Select all that apply Multiple responses allowed													
Methods	Peri-u	ırban	Rural		Regional		Metropolitan		Total responses				
	%	n	%	n	%	Ν	%	n	%	n			
Face-to-face	86.0	37	78.3	18	85.7	6	68.7	11	80.9	72			
Phone	60.5	26	78.3	18	71.4	5	26.1	6	61.8	55			
Video	18.6	8	43.5	10	42.9	3	18.8	3	27.0	24			
Email	0	0	4.3	1	0	0	0	0	1.1	1			
Text	11.6	5	17.4	4	0	0	12.5	2	12.3	11			
Apps	0	0	4.3	1	0	0	0	0	1.1	1			
Total respondents	48.3	43	25.8	23	7.9	7	18.0	16	100.0	89			
Total responses	46.3	76	31.8	23	7.9	7	18.0	16	100.0	164			

Table 33 Question 2.20

Q2.20 Family - Have you experienced any problems with digital health options? Select all that apply Multiple responses allowed													
Methods	Peri-u	rban	Rural	Rural		Regional		Metropolitan		sponses			
	%	n	%	n	%	n	%	n	%	n			
No problems	44.2	19	17.4	4	57.1	4	68.8	11	42.7	38			
No access/skill	11.6	5	8.7	2	0	0	0	0	7.9	7			
Do not use	9.3	4	8.7	2	0	0	6.2	1	7.9	7			
Poor mobile	11.6	5	60.9	14	0	0	0	0	21.3	19			
Poor internet	9.3	4	47.8	11	0	0	0	0	16.9	15			
Digital not offered	23.3	10	13.0	3	14.3	1	18.8	3	19.1	17			
Total respondents	48.3	43	25.8	23	7.9	7	18.0	16	100.0	89			
Total responses	45.6	47	34.9	36	4.9	5	14.6	15	100.0	103			

Table 34 Question 2.21

Q2.21 Family - How much do you agree with the following statements?												
	Peri-	urban	Rur	al	Regio	onal	Metrop	olitan	Total r	esponses		
	%	n	%	n	%	n	%	n	%	n		
2.21(1) (Family) Digital heap palliative care team	alth is a	an imp	ortant t	ool ii	n ensu	ring r	egular c	ommuni	cation w	ith the		
Strongly disagree	2.7	1	5.0	1	0	0	0	0	2.7	2		
Somewhat disagree	5.4	2	10.0	2	16.7	1	0	0	6.7	5		
Neither agree/disagree	54.1	20	30.0	6	33.3	2	50.0	6	45.3	34		
Somewhat agree	18.9	7	40.0	8	50.0	3	33.3	4	29.3	22		
Strongly agree	18.9	7	15.0	3	0	0	16.7	2	16.0	12		
Total	49.3	37	26.7	20	8.0	6	16.0	12	100.0	75		
2.21(2) (Family) The methodigital is appropriate for o	od palli our nee	ative c ds	are pro	ovide	s supp	ort in	the hon	ne, eithei	r face-to	-face or		
Strongly disagree	8.3	3	10.0	2	0	0	0	0	6.7	5		
Somewhat disagree	5.6	2	5.0	1	16.7	1	0	0	5.4	4		
Neither agree/disagree	36.1	13	20.0	4	33.3	2	33.4	4	31.1	23		
Somewhat agree	22.2	8	35.0	7	16.7	1	33.3	4	27.1	20		
Strongly agree	27.8	10	30.0	6	33.3	2	33.3	4	29.7	22		
Total	48.6	36	27.1	20	8.1	6	16.2	12	100.0	74		
2.21(3) (Family) I have the	resour	ces an	d conf	idenc	e to us	se dig	jital heal	th effect	ively			
Strongly disagree	0	0	0	0	0	0	0	0	0	0		
Somewhat disagree	5.4	2	0	0	0	0	0	0	2.7	2		
Neither agree/disagree	16.2	6	10.0	2	16.7	1	8.4	1	13.3	10		
Somewhat agree	21.6	8	55.0	11	66.6	4	33.3	4	36.0	26		
Strongly agree	56.8	21	35.0	7	16.7	1	58.3	7	48.0	36		
Total	49.3	37	26.7	20	8.0	6	16.0	12	100.0	75		
2.21(4) (Family) Digital here consultations with medica	alth coi al profe	nsultat ssiona	ions ar Is	e an	accept	able	alternati	ve to fac	e-to-face	9		
Strongly disagree	2.6	1	5.0	1	0	0	0	0	2.6	2		
Somewhat disagree	10.5	4	5.0	1	50.0	3	8.4	1	11.8	9		
Neither agree/disagree	36.8	14	30.0	6	16.7	1	33.3	4	33.0	25		
Somewhat agree	39.5	15	55.0	11	33.3	2	50.0	6	44.7	34		
Strongly agree	10.5	4	5.0	1	0	0	8.3	1	7.9	6		
Total	50.0	38	26.3	20	7.9	6	15.8	12	100.0	76		
2.21(5) (Family) Digital head be able to access due to a	alth me listanc	ans th e	at acce	ss to	medic	al su	pport is	available	e that I w	ould not		
Strongly disagree	2.7	1	0	0	0	0	0	0	1.3	1		
Somewhat disagree	5.4	2	10.5	2	16.7	1	25.0	3	10.8	8		
Neither agree/disagree	40.6	15	21.1	4	33.3	2	50.0	6	36.5	27		
Somewhat agree	43.2	16	47.4	9	50.0	3	8.3	1	39.2	29		

Strongly agree	8.1	3	21.0	4	0	0	16.7	2	12.2	9		
Total	50.0	37	25.7	19	8.1	6	16.2	12	100.0	74		
2.21(6) (Family) I feel that	the cho	oice of	place o	of car	e is su	ppor	ted with	digital h	ealth			
Strongly disagree	8.3	3	5.0	1	0	0	0	0	5.4	4		
Somewhat disagree	13.9	5	15.0	3	16.7	1	16.7	2	14.9	11		
Neither agree/disagree	41.7	15	30.0	6	50.0	3	50.0	6	40.6	30		
Somewhat agree	27.8	10	35.0	7	33.3	2	16.6	2	28.3	21		
Strongly agree	8.3	3	15.0	3	0	0	16.7	2	10.8	8		
Total	48.7	36	27.1	20	8.1	6	16.2	12	100.0	74		
2.21(7) (Family) I am comfortable with my family member remaining at home knowing that												
2.21(7) (Family) I am comf	ortable	with n	ny tami	iy me	emper	rema	ming at i	iome kn	owing th	at		
support can be accessed	at any	time	ny tami	iy me	emper	rema	ining at i	Iome kn	owing th	at		
support can be accessed a Strongly disagree	at any 1 12.1	time 4	10.0	1 у т е 2	0	0	8.3	1	9.9	at 7		
2.21(7) (Family) Fam communications support can be accessed a Strongly disagree Somewhat disagree	at any 1 12.1 21.2	with n time 4 7	10.0 20.0	2 4	0 66.7	0 4	8.3 41.7	1 5	9.9 28.2	7 20		
2.21(7) (Family) Fam commission support can be accessed a Strongly disagree Somewhat disagree Neither agree/disagree	12.1 21.2 30.3	time 4 7 10	10.0 20.0 40.0	2 4 8	0 66.7 33.3	0 4 2	8.3 41.7 25.0	1 5 3	9.9 28.2 32.4	7 20 23		
2.21(7) (Family) Fam communications support can be accessed a Strongly disagree Somewhat disagree Neither agree/disagree Somewhat agree	at any 1 12.1 21.2 30.3 21.2	with n time 4 7 10 7	10.0 20.0 40.0 20.0	2 4 8 4	0 66.7 33.3 0	0 4 2 0	8.3 41.7 25.0 8.3	1 5 3 1	9.9 28.2 32.4 16.9	7 20 23 12		
2.21(7) (Family) Fam communication support can be accessed a Strongly disagree Somewhat disagree Neither agree/disagree Somewhat agree Strongly agree	at any 12.1 12.1 21.2 30.3 21.2 15.2	with n time 4 7 10 7 5	10.0 20.0 40.0 20.0 10.0	2 4 8 4 2	0 66.7 33.3 0 0	0 4 2 0 0	8.3 41.7 25.0 8.3 16.7	1 5 3 1 2	9.9 28.2 32.4 16.9 12.6	7 20 23 12 9		

Table 35 Question 2.21 (Descriptive Statistics)

Q2.21 Family - How much do you agree with the following statements?												
		_				Variance	Confidence					
	n	min	max	mean	SD .		Interval					
2.21(1) (Family) Digital heal	th is an i	mportan	it tool in	ensuring	g regular	communicat	tion with the					
Dari Linhan	07	4	F	25	0.0	0.0	0.20					
Peri-Orban	37	1	5	3.5	0.9	0.9	0.20					
Rural	20	1	5	3.5	1.0	1.1	0.20					
Regional	0	2	4	3.3	0.7	0.0	0.30					
	12	3	5	3.7	0.7	0.0	0.20					
2 21(2) (Eamily) The method	C/ I nalliati		C vrovidos	3.0	U.8	0.0						
2.21(2) (Family) The method palliative care provides support in the home, either face-to-face or digital is appropriate for our needs												
Peri-Urban	36	1	5	3.6	1.2	1.4	0.20					
Rural	20	1	5	3.7	1.2	1.5	0.30					
Regional	6	2	4	3.7	1.1	1.2	0.40					
Metro	12	2	4	4.0	0.8	0.7	0.20					
Total	74	1	5	3.8	1.1	1.2	0.30					
2.21(3) (Family) I have the r	esources	s and co	nfidence	to use o	digital he	alth effective	ly					
Peri-Urban	36	1	5	4.3	0.9	0.9	0.20					
Rural	20	1	5	4.2	0.6	0.4	0.10					
Regional	6	2	5	4.0	0.6	0.3	0.20					
Metro	12	2	5	4.5	0.6	0.4	0.20					
Total	74	1	5	4.3	0.7	0.5	0.20					
2.21(4) (Family) Digital heal	th consu	Itations	are an a	cceptabl	e alterna	tive to face-t	o-face					
Consultations with medical	protessi	onais		2.4	0.0	0.9	0.15					
Perl-Urban	38	1	5	3.4	0.9	0.8	0.15					
Rural	20	1	5	3.5	0.9	0.8	0.20					
Regional	0	2	4	2.8	0.9	0.0	0.40					
	12	<u> </u>	5 F	3.0	0.8	0.0	0.20					
2 21(5) (Family) Digital heal	/0	s that ac	COSS to r	3.3 nodical (0.9	0.0 s available th	0.20					
not be able to access due t	o distanc	S that ac	Cess 10 I	neuicai	support	is available li						
Peri-Urban	37	1	5	3.5	0.8	0.7	0.10					
Rural	19	2	5	3.8	0.9	0.8	0.20					
Regional	6	2	4	3.3	0.7	0.6	0.30					
Metro	12	2	5	3.2	1.0	1.0	0.30					
Total	74	1	5	3.5	0.9	0.8	0.20					
2.21(6) (Family) I feel that the	ne choice	e of plac	e of care	is supp	orted wit	h digital heal	th					

Peri-Urban	36	1	5	3.1	1.0	1.1	0.20					
Rural	20	1	5	3.4	1.1	1.1	0.20					
Regional	6	2	4	3.2	0.7	0.5	0.30					
Metro	12	2	5	3.3	0.9	0.9	0.30					
Total	74	1	5	3.3	0.9	0.9	0.30					
2.21(7) (Family) I am comfortable with my family member remaining at home knowing that												
support can be accessed at any time												
Peri-Urban	33	1	5	3.1	1.2	1.5	0.20					
Peri-Urban Rural	33 20	1	5 5	3.1 3.0	1.2 1.2	1.5 1.2	0.20 0.20					
Peri-Urban Rural Regional	33 20 6	1 1 2	5 5 3	3.1 3.0 2.3	1.2 1.2 0.5	1.5 1.2 0.2	0.20 0.20 0.20					
Peri-Urban Rural Regional Metro	33 20 6 12	1 1 2 1	5 5 3 5	3.1 3.0 2.3 2.8	1.2 1.2 0.5 1.2	1.5 1.2 0.2 1.5	0.20 0.20 0.20 0.40					

Table 36 Question 2.22

Q2.22 Family - Overall are you satisfied with the access to community palliative care services in your location?

Opinion	Peri-u	rban	Rur	al	Regio	nal	Metropo	olitan	Total resp	onses
	%	n	%	n	%	n	%	n	%	n
Yes	36.4	12	57.1	8	25.0	1	37.5	3	40.7	24
No	63.6	21	42.9	6	75.0	3	62.5	5	59.3	35
Total	55.9	33	23.7	14	6.8	4	13.6	8	100.0	59

Health Care Professionals

Table 37 Question 6.1

Q6.1 Professional - Your age group?													
Age	Peri-urban		Ru	Rural		Regional		Metropolitan		sponses			
	%	n	%	n	%	n	%	n	%	n			
18-34 years	9.1	2	33.3	1	0	0	33.3	2	14.7	5			
35-49 years	40.9	9	33.3	1	66.7	2	33.3	2	41.2	14			
50-64 years	50.0	22	33.4	1	33.3	1	33.4	2	44.1	15			
65+	0	0	0	0	0	0	0	0	0	0			
Total	64.7	22	8.8	3	8.8	3	17.7	6	100.0	34			

Table 38 Question 6.2

Q6.2 Professional - How long have you been providing community palliative care?													
Providing care	Peri-u	rban	an Rura		al Regiona		I Metropolitan		Total responses				
	%	n	%	n	%	n	%	n	%	n			
<1 year	5.7	2	16.7	1	0	0	14.3	1	7.8	4			
1-5 years	34.3	12	16.7	1	0	0	57.1	4	33.3	17			
>5 years	60.0	21	66.7	4	100	3	28.6	2	58.8	30			
Total	68.6	35	11.8	6	5.9	3	13.7	7	100.0	51			

Q6.3 Professionals - What is your current role in providing community palliative care?													
Role	Peri-u	rban	Ru	ral	Regio	nal	Metrop	olitan	Total res	ponses			
	%	n	%	n	%	n	%	n	%	n			
Nurse Practitioner	2.8	1	0	0	0	0	14.3	1	4.0	2			
Clinical Nurse	20.0	7	20.0	1	33.3	1	0	0	18.0	9			
Registered Nurse	22.8	8	20.0	1	0	0	71.4	5	28.0	14			
AIN/PCA	14.3	5	20.0	1	0	0	0	0	12.0	6			
OT/Physiotherapist	5.8	2	0	0	0	0	14.3	1	6.0	3			
GP	2.8	1	0	0	0	0	0	0	2.0	1			
Psychologist	2.8	1	0	0	33.3	1	0	0	4.0	2			
Social Worker	2.8	1	20.0	1	0	0	0	0	4.0	2			
Volunteer	8.6	3	0	0	0	0	0	0	6.0	3			
Nurse Navigator	5.8	2	20.0	1	33.4	1	0	0	8.0	4			
Equipment/Transport	5.7	2	0	0	0	0	0	0	4.0	2			
Admin/Referral	5.8	2	0	0	0	0	0	0	4.0	2			
Total	67.3	35	13.5	5	5.7	3	13.5	7	100.0	50			

Table 39 Question 6.3

Table 40 Question 6.4

Q6.4 Professionals - How would you describe the area where community palliative care is provided?									
	%	n							
Peri-urban	67.3	35							
Rural	13.5	7							
Regional	5.7	3							
Metropolitan	13.5	7							
Total	100.0	52							

Table 41 Question 6.6

Q6.6 Professionals - In what State/Territory are services provided?														
State/Territory	Peri-	urban	Rur	al	Regio	onal	Metro	politan	Total responses					
	%	n	%	n	%	n	%	n	%	n				
Queensland	5.7	2	20.0	1	66.7	2	14.3	1	12.0	6				
South Australia	94.3	33	80.0	4	33.3	1	57.1	4	84.0	42				
NT, WA and ACT	0	0	0	0	0	0	0	0	0	0				
New South Wales	0	0	0	0	0	0	28.6	2	4.0	2				
Victoria and Tasmania	0	0	0	0	0	0	0	0	0	0				
Total	70.0	35	10.0	5	6.0	3	14.0	7	100.0	30				

Table 42 Question 6.7

Q6.7 (Professionals) In what type of organisation do you provide community palliative care?														
Organisation	anisation Peri-urb		ban Rural		Regio	nal	Metropo	litan	Total responses					
	%	n	%	n	%	n	%	n	%	n				
Specialist palliative care team (SPC)	31.4	11	40.0	2	66.7	2	14.3	1	32.0	16				
Public Hospital	8.6	3	20.0	1	33.3	1	0	0	10.0	5				
Private Hospital	0	0	0	0	0	0	14.4	1	2.0	1				
GP clinic	2.9	1	0	0	0	0	0	0	2.0	1				

Home Care Provider	20.0	7	0	0	0	0	42.9	3	20.0	10
RACF Community	2.8	1	20.0	1	0	0	14.4	1	6.0	3
Community Health	22.9	8	20.0	1	0	0	0	0	18.0	9
Mental Health	2.8	1	0	0	0	0	0	0	2.0	1
Retired	2.9	1	0	0	0	0	0	0	2.0	1
Palliative Care Unit	2.8	1	0	0	0	0	14.4	1	4.0	2
Cancer charity	2.9	1	0	0	0	0	0	0	2.0	1
Total	70.0	35	10.0	5	6.0	3	14.0	7	100.0	50

Table 43 Question 6.8

Q6.8 If answered yes to SP palliative care in your orga	PC team	- What า?	health p	orofes	sions a	re em	ployed to	suppor Select	rt communit all that app	y ly
Role	Peri-u	rban	Rur	al	Regio	nal	Metropo	olitan	Total resp	onses
	%	n	%	n	%	n	%	n	%	n
Nurse Practitioner	18.2	2	0	0	11.1	1	12.5	1	25.0	4
Clinical Nurse	90.9	10	25.0	1	22.2	2	0	0	81.2	13
Registered Nurse	27.3	3	0	0	0	0	12.5	1	25.0	4
AIN/PCA	9.1	1	0	0	0	0	0	0	6.2	1
OT	72.7	8	0	0	0	0	12.5	1	56.2	9
GP	0	0	0	0	11.1	1	0	0	6.2	1
Psychologist	0	0	0	0	11.2	1	0	0	6.2	1
Social Worker	54.5	6	25.0	1	22.2	2	12.5	1	62.5	10
Physiotherapist	18.2	2	0	0	0	0	12.5	1	18.7	3
Palliative Care Specialist	27.3	3	25.0	1	22.2	2	12.5	1	43.8	7
Chaplain	0	0	0	0	0	0	12.5	1	6.2	1
Pharmacist	0	0	0	0	0	0	12.5	1	6.2	1
Bereavement Coordinator	9.1	1	0	0	0	0	0	0	6.2	1
Nurse Navigator	0	0	25.0	1	0	0	0	0	6.2	1
Total responses	63.2	36	7.0	4	15.8	9	14.0	8	100.0	57
Total respondents	68.8	11	12.5	2	12.5	2	6.2	1	100.0	16

Table 44 Question 6.9

Q6.9 If yes as member of SPC team- If you have a choice, are there any health profession roles you would like added to your team?													
	Peri-urban Rural Regional Metropolitan Total responses												
	%	n	%	n	%	n	%	n	%	n			
Yes	100.0	11	50.0	1	100.0	2	100.0	1	93.8	15			
No	0	0	50.0	1	0	0	0	0	6.2	1			
Total	68.7	11	12.5	2	12.5	2	6.3	1	100.0	16			

Table 45 Question 6.9

Q6.9 If yes - please pro	ovide de	etail		Multiple responses allowed						
Role	Peri-u	rban	Rura	d 👘	Regio	nal	Metropo	litan	Total resp	onses
	%	n	%	n	%	n	%	n	%	n
Nurse Practitioner	0	0	0	0	50.0	1	0	0	6.7	1
Clinical Nurse	18.2	2	0	0	0	0	0	0	13.3	2
Registered Nurse	0	0	100.0	1	0	0	0	0	100.0	1
Enrolled Nurse	54.5	6	0	0	50.0	1	0	0	46.7	7
AIN/PCA	72.7	8	0	0	0	0	0	0	53.8	8
OT	18.2	2	100.0	1	100	2	0	0	33.3	5
GP	18.2	2	100.0	1	50.0	1	0	0	26.7	4
Psychologist	27.2	3	0	0	0	0	0	0	20.0	3

Social Worker	54.5	6	0	0	0	0	0	0	40.0	6
Physiotherapist	45.4	5	0	0	50.0	1	0	0	40.0	6
Palliative Care	9.1	1	0	0	50.0	1	0	0	13.3	2
Specialist										
Chaplain/Spiritual	9.1	1	0	0	0	0	0	0	6.7	1
Pharmacist	9.1	1	100.0	1	0	0	0	0	13.3	2
Grief/Bereavement	45.4	5	0	0	50.0	1	0	0	40.0	6
support										
Speech Pathologist	18.2	2	0	0	0	0	0	0	13.3	2
Equipment support	36.3	4	0	0	0	0	0	0	26.7	4
Dietician	9.1	1	0	0	0	0	100	1	13.3	2
Volunteer	9.1	1	0	0	0	0	0	0	6.7	1
Handyman	9.1	1	0	0	0	0	0	0	6.7	1
Total responses	79.7	51	6.2	4	12.5	8	1.6	1	100.0	64
Total respondents	73.3	11	6.7	1	13.3	2	6.7	1	100.0	15

Table 46 Question 6.10

Q6.10 – Professionals	- Pleas	e select	t all the	servio	es that a	are ac	cessible	to all h	omes in the	area
Services		at appiy	/ Dur	'al	Pagior	al	Motropo	respon	Total rosp	oneoe
Sel VICES	%	n	%	n	%	n	weitopt %	n	%	n
Unsure	57	2	0	0	0	0	0	0	4.0	2
Specialist Palliative	77.1	27	80.0	4	100.0	3	85.7	6	80.0	40
Care Team (SPC)										
Community Nursing	85.7	30	60.0	3	100.0	3	71.4	5	82.0	41
Personal care	88.6	31	40.0	2	100.0	3	85.7	6	84.0	42
Transport Assistance	80.0	28	40.0	2	100.0	3	71.4	5	34.0	38
Occupational Therapy	80.0	28	60.0	3	100.0	3	100	7	82.0	41
Psychologist	0	0	0	0	0	0	0	0	0	0
Social Worker	25.7	9	20.0	1	66.7	2	71.4	5	34.0	17
Physiotherapy	65.7	23	40.0	2	100.0	3	85.7	6	68.0	34
Dietician/Speech Therapy	65.7	23	80.0	4	66.7	2	71.4	5	68.0	34
Palliative Care Specialist	25.7	9	20.0	1	33.3	1	85.7	6	34.0	17
Spiritual	0	0	20.0	1	0	0	28.6	2	6.0	3
Volunteers	20.0	7	20.0	1	33.3	1	57.1	4	26.0	13
Complementary	14.3	5	0	0	33.3	1	71.4	5	22.0	11
House/Garden	82.8	29	40.0	2	100.0	3	71.4	5	78.0	39
Homo modification	77 1	27	40.0	2	100.0	3	71 /	5	74.0	37
	90.0	21	40.0	۲ ۲	100.0	3	05.7	5	74.0	40
	00.0	20 12	100	ວ ວ	66.7	<u>。</u> っ	05.7	6	04.0 79.0	42
Directive assistance	57.1	15	40.0	2	00.7	2	00.7	0	70.0	23
Financial	0	0	0	0	0	0	0	0	0	0
GP	65.7	23	40.0	2	100.0	3	100	7	70.0	35
End-of-life care	88.6	31	80.0	4	100.0	3	85.7	6	88.0	44
Bereavement	28.6	10	40.0	2	66.7	2	85.7	6	40.0	20
Family/friends	0	0	0	0	0	0	0	0	0	0
Daily medication assistance	25.7	9	0	0	100.0	3	71.4	5	34.0	17
Pharmacy	54.3	19	60.0	3	66.7	2	85.7	6	60.0	30
Pathology	62.8	22	40.0	2	33.3	1	100	7	64.0	32
Radiology	42.8	15	40.0	2	66.7	2	85.7	6	50.0	25
Wigs	2.8	1	0	0	0	0	0	0	2.0	1
Meal Delivery	0	0	0	0	0	0	14.3	1	2.0	1
Overnight respite	0	0	0	0	0	0	14.3	1	2.0	1
Doula	2.8	1	0	0	0	0	0	0	2.0	1

Anything that helps	5.7	2	0	0	0	0	0	0	4.0	2
Depends on location	8.6	3	0	0	0	0	14.3	1	6.0	4
Other	22.8	8	0	0	6.7	2	42.9	3	26.0	13
Total responses	66.1	463	7.1	50	7.8	54	19.0	133	100.0	700
Total respondents	70.0	35	10.0	5	6.0	3	14.0	7	100.0	700

Table 47 Question 6.11

Q6.11 Professionals - Select all that apply	What b	arriers	to acces	ssing	services	do y	ou believ	e apply	in your loc	ation?
Barriers	Peri-u	rban	Rui	ral	Regior	nal	Metropo	olitan	Total resp	onses
	%	n	%	n	%	n	%	n	%	n
No barriers to access	2.9	1	0	0	0	0	28.6	2	6.0	3
Clients under 65 yrs	82.9	29	60.0	3	100.0	3	28.6	2	74.0	37
Distance for providers	88.6	31	60.0	3	100.0	3	0	0	74.0	37
Providers lack resources	94.3	33	60.0	3	100.0	3	28.6	2	82.0	41
Time taken to start	80.0	28	60.0	3	33.3	1	28.6	2	68.0	34
Distance travel to	82.9	29	60.0	3	66.7	2	0	0	68.0	34
Diversity	31.4	11	0	0	0	0	14.3	1	24.0	12
Language	40.0	14	40.0	2	33.3	1	14.3	1	36.0	18
Palliative care stigma	60.0	21	60.0	3	100.0	3	28.6	2	58.0	29
Behaviours	25.7	9	20.0	1	33.3	1	0	0	22.0	11
Financial	57.1	20	40.0	2	66.7	2	42.9	3	54.0	27
Personal choice	45.7	16	20.0	1	0	0	28.6	2	38.0	19
Care needs exceed scope	85.7	30	60.0	3	66.7	2	28.6	2	74.0	37
No need identified	11.4	4	0	0	0	0	0	0	8.0	4
Lack of information	57.1	20	0	0	66.7	2	42.9	3	50.0	25
Total respondents	70.0	35	10.0	5	6.0	3	14.0	7	100.0	50

Table 48 Question 6.12

Q6.12 Professionals – Are there any changes you would like to see in your role? Select all that apply

Reason	Peri-u	rban	Ru	ral	Regio	nal	Metrop	olitan	Total res	ponses
	%	n	%	n	%	n	%	n	%	n
No changes	0	0	20.0	1	0	0	14.3	1	4.0	2
Increase frequency	85.7	30	80.0	4	33.3	1	28.6	2	74.0	37
Enable choice provider	31.4	11	40.0	2	33.3	1	0	0	28.0	14
Enable choice day/time	71.4	25	60.0	3	33.3	1	28.6	2	62.0	31
Provide 24/7 contact	77.1	27	80.0	4	66.7	2	28.6	2	70.0	35
24/7 in-home care	82.8	29	80.0	4	33.3	1	42.9	3	74.0	37
Increase staff	100	35	80.0	4	66.7	2	57.1	4	88.0	44
Remove time restriction	74.3	26	60.0	3	33.3	1	42.9	3	66.0	33
Increase avail equip	62.9	22	60.0	3	33.3	1	28.6	2	36.0	28
Reduce service cost	51.4	18	40.0	2	33.3	1	42.9	3	48.0	25
Improve staff training	51.4	18	60.0	3	0	0	42.9	3	48.0	24
Total respondents	70.0	35	10.0	5	6.0	3	14.0	7	100.0	50
Total responses	77.7	241	10.6	33	3.6	11	8.1	25	100.0	310

Table 49 Question 6.13

Q6.13 Professionals - Please sel	ect any	of the	follow	ing tl	hat apply					
	Peri-	urban	Rui	ral	Region	al	Metropo	litan	Total respon	ISes
	%	n	%	n	%	n	%	n	%	n
Q6.13(1) (Professionals) Hospital admissions extended due to in- home services unable/ unavailable to support discharge	85.7	30	57.1	4	100.0	3	28.6	2	78.0	39
Q6.13(2) (Professionals) Clients needing to attend medical clinics or outpatient clinics as no in- home services to provide medication administration	65.7	23	57.1	4	0	0	28.6	2	58.0	29
Q6.13(3) (Professionals) Clients needing to attend medical clinics or outpatient clinics as no/ insufficient in-home services available to assist with wound/ catheter/stoma care	71.4	25	28.6	2	33.3	1	42.9	3	62.0	31
Q6.13(4) (Professionals) Ambulance being called for assistance with pain/symptom control as community services unable to be contacted	82.9	29	42.9	3	66.7	2	57.1	4	76.0	38
Q6.13(5) (Professionals) Respite in aged care facility as no/insufficient services available in-home to support hospital discharge	80.0	28	42.9	3	66.7	2	57.1	4	74.0	37
Q6.13(6) (Professionals) End-of- life care provided in hospital as appropriate services unable to be provided in the house	88.6	31	57.1	4	33.3	1	57.1	4	76.0	38
Q6.13(7) (Professionals) None apply	5.7	2	0	0	0	0	42.8	3	10.0	5
Total responses	70.0	35	10.0	5	6.0	3	14.0	7	100.0	50

Table 50 Question 6.14

the community? Select all Information source	that app Peri-	oly urban	Ru	iral	Regio	onal	Metrop	olitan	Total respon	ses
	%	n	%	n	%	n	%	n	%	n
Rely staff face-to-face	29.4	10	20.0	1	100	3	28.6	2	32.6	16
Referral process	47.0	16	40.0	2	0	0	57.1	4	44.9	22
Pamphlets/booklets	26.5	9	40.0	2	33.3	1	57.1	4	32.6	16
Website	38.2	13	40.0	2	0	0	57.1	4	38.8	19
Advertise locally	23.5	8	0	0	0	0	28.6	2	20.4	10
Community events	14.7	5	0	0	0	0	28.6	2	14.3	7
Rely community talk	23.5	8	0	0	33.3	1	42.9	3	24.0	12
Staff over phone	20.6	34	60.0	3	100	3	42.9	3	32.6	16
Total respondents	69.3	34	10.2	5	6.1	3	14.4	7	100.0	49
Total responses	64.4	76	8.5	10	6.8	8	20.3	24	100.0	118

O6 14 Professionals - How does your organisation provide information about relevant services to

Table 51 Question 6.15

Q6.15 (Professionals) How much	do yo	u agree	with th	ie fol	lowing	state	ments?			
	Peri-	urban	Rur	ral	Regio	onal	Metrop	olitan	Total	
	0/		0/					1	respon	ises
	%	<u>n</u>	%	n	%	n	%	n	%	n
6.15(1) (Professionals) Existing i	nealth s	services	meet	the n	eeds al	na ex	pectatio	ns of pe	opie in ti	nis
Strongly disagree	86	3	0	0	33.3	1	14.3	1	10.2	5
Disagree	37.1	13	50.0	2	0	0	14.3	1	32.6	16
Neither agree/disagree	17.1	6	25.0	1	0	0	0	0	14.3	7
Agree	34.3	12	25.0	1	66.7	2	14.3	1	32.7	16
Strongly agree	2.9	1	0	0	0	0	57.1	4	10.2	5
Total	71.4	35	8.2	4	6.1	3	14.3	7	100.0	49
6.15(2) (Professionals) There is a	need	for peop	ole in th	nis ar	ea to tr	avel	over 100)km dista	ances fo	r
some healthcare services										
Strongly disagree	2.9	1	0	0	0	0	14.3	1	4.1	2
Disagree	8.6	3	0	0	0	0	57.1	4	14.3	7
Neither agree/disagree	8.6	3	0	0	0	0	0	0	6.1	3
Agree	37.1	13	25.0	1	66.7	2	14.3	1	34.7	17
Strongly agree	42.8	15	75.0	3	33.3	1	14.3	1	40.8	20
Total	71.4	35	8.2	4	6.1	3	14.3	7	100.0	49
6.15(3) (Professionals) The need decisions about treatment	to trav	el long	distand	ces fo	or some	e hea	Ithcare s	services	influenc	es
Strongly disagree	0	0	0	0	0	0	0	0	0	0
Disagree	0	0	0	0	0	0	28.6	2	4.1	2
Neither agree/disagree	14.3	5	0	0	0	0	28.6	2	14.3	7
Agree	57.1	20	50.0	2	66.7	2	28.5	2	53.1	26
Strongly agree	28.6	10	50.0	2	33.3	1	14.3	1	28.5	14
Total	71.4	35	8.2	4	6.1	3	14.3	7	100.0	49
6.15(4) (Professionals) The need	to trav	el long	distan	ces fo	or som	e hea	Ithcare s	services	results i	n
Strongly disagree	0	0	0	0	0	0	0	0	0	0
Disagree	0	0	0	0	0	0	14.3	1	20	1
Neither agree/disagree	57	2	25.0	1	0	0	28.5	2	10.2	5
Agree	54.3	19	25.0	1	66.7	2	28.6	2	49.0	25
Strongly agree	40.0	14	50.0	2	33.3	1	28.6	2	38.8	19
Total	71.4	35	8.2	4	6.1	3	14.3	7	100.0	49
6.15(5) (Professionals) People ar	e supp	orted to	remai	n in t	heir ov	n ho	me			
Strongly disagree	0	0	0	0	0	0	14.3	1	2.0	1
Disagree	14.3	5	25.0	1	33.3	1	14.3	1	16.3	8
Neither agree/disagree	2.9	1	0	0	0	0	0	0	2.0	1
Agree	68.6	5	25.0	1	0	0	57.1	4	20.5	10
Strongly agree	14.2	5	25.0	1	0	0	57.1	4	20.5	10
Total	71.4	35	8.2	4	6.1	3	14.3	7	100.0	49
6.15(6) (Professionals) Home loc	ation d	oes not	influe	nce a	iccess	to se	rvices			
Strongly disagree	51.4	18	50.0	2	33.3	1	42.8	3	49.0	24
Disagree	28.6	10	0	0	66.7	2	14.3	1	26.5	13
Neither agree/disagree	14.3	5	50.0	2	0	0	0	0	14.3	7
Agree	5.7	2	0	0	0	0	28.6	2	8.2	4
Strongly agree	0	0	0	0	0	0	14.3	1	2.0	1
Total	71.4	35	8.2	4	6.1	3	14.3	7	100.0	49
6.15(7) (Professionals) Communi respectful	ication	and car	e prov	ided	is cultu	Irally	and ger	der inclu	usive and	d
Strongly disagree	0	0	0	0	0	0	0	0	0	0
Disagree	8.6	3	0	0	0	0	0	0	6.1	3
Neither agree/disagree	11.4	4	0	0	0	0	14.3	1	10.2	5
Agree	51.4	18	50.0	2	66.7	2	42.9	3	51.0	25

Strongly agree	28.6	10	50.0	2	33.3	1	42.8	3	32.7	16
Total	71.4	35	8.2	4	6.1	3	14.3	7	100.0	49
6.15 (8) (Professionals) Personal	choice	es. prefe	erences	and	goals	of ca	re are re	spected		
Strongly disagree	0	0	0	0	0	0	0	0	0	0
Disagree	5.9	2	0	0	0	0	0	0	4.2	2
Neither agree/disagree	8.8	3	25.0	1	0	0	57.1	4	8.3	4
Agree	55.9	19	25.0	1	66.7	2	42.9	3	54.2	26
Strongly agree	29.4	10	50.0	2	33.3	1	14.6	7	33.3	16
Total	70.8	34	8.3	4	6.3	3	14.6	7	100.0	48
6.15(9) (Professionals) Access is	availa	ble for s	suppor	t 24/7	7	1	1	1	1	1
Strongly disagree	11.8	4	25.0	1	66.7	2	14.2	1	16.7	8
Disagree	61.8	21	50.0	2	33.3	1	0	0	50.0	24
Neither agree/disagree	14.7	5	25.0	1	0	0	0	0	12.5	6
Agree	11.7	4	0	0	0	0	42.9	3	14.6	7
Strongly agree	0	0	0	0	0	0	42.9	3	6.2	3
Total	70.8	34	8.3	4	6.3	3	14.6	7	100.0	48
6.15(10) (Professionals) Quality of	of life is	s improv	/ed by	rema	ining i	n ow	n home			
Strongly disagree	0	0	0	0	0	0	0	0	0	0
Disagree	0	0	0	0	0	0	0	0	0	0
Neither agree/disagree	2.9	1	0	0	66.7	2	14.3	1	8.2	4
Agree	37.1	13	25.0	1	0	0	14.3	1	30.6	15
Strongly agree	60.0	21	75.0	3	33.3	1	71.4	5	61.2	30
Total	71.4	35	8.2	4	6.1	3	14.3	7	100.0	49
6.15(11) (Professionals) Ambula	nce trai	nsport a	ind hos	spital	admis	sions	s are red	uced by	commur	nity
service provided	0	0		•	00.0	4		•	0.4	
Strongly disagree	0	0	0	0	33.3	1	0	0	2.1	1
Disagree	14.7	5	0	0	0	0	0	0	10.4	5
Neither agree/disagree	32.3	11	50.0	2	0	0	28.6	2	31.2	15
Agree	41.2	14	50.0	2	33.4	1	14.3	1	37.5	18
Strongly agree	11.8	4	0	U	33.3	1	57.1	4	18.8	9
Total	70.0	24	0 2	4	6.2	2	446	7	400.0	40
Total	70.8	34 milu ma	8.3	4	6.3	3 d in a	14.6	7 on moki	100.0	48
Total 6.15(12) (Professionals) Patients and services	70.8 and fa	34 mily me	8.3 mbers	4 are i	6.3 nclude	3 d in a	14.6 all decisi	7 on-maki	100.0 ng about	48 t care
Total 6.15(12) (Professionals) Patients and services Strongly disagree	70.8 and fa	34 mily me	8.3 mbers	4 are i 0	6.3 nclude	3 d in a 0	14.6 all decisi	7 on-maki 0	100.0 ng about 0	48 t care
Total 6.15(12) (Professionals) Patients and services Strongly disagree Disagree	70.8 and fa 0 2.9	34 mily me 0 1	8.3 mbers 0 0	4 are i 0 0	6.3 nclude 0 0	3 d in a 0 0	14.6 all decisi 0 14.2	7 on-maki 0 1	100.0 ng about 0 4.2	48 t care 0 2
Total 6.15(12) (Professionals) Patients and services Strongly disagree Disagree Neither agree/disagree	70.8 and fa 0 2.9 14.8	34 mily me 0 1 5	8.3 mbers 0 0 25.0	4 are i 0 0	6.3 nclude 0 0 0	3 d in a 0 0 0	14.6 all decisi 0 14.2 14.3	7 on-maki 0 1 1	100.0 ng about 0 4.2 14.6	48 t care 0 2 7
Total 6.15(12) (Professionals) Patients and services Strongly disagree Disagree Neither agree/disagree Agree	70.8 and fa 0 2.9 14.8 52.9	34 mily me 0 1 5 18	8.3 mbers 0 25.0 25.0	4 are i 0 0 1 1	6.3 nclude 0 0 0 66.7	3 d in a 0 0 0 2	14.6 all decisi 0 14.2 14.3 28.6	7 on-maki 0 1 1 2	100.0 ng about 0 4.2 14.6 47.9	48 t care 0 2 7 23
Total6.15(12) (Professionals) Patients and servicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agree	70.8 and fa 0 2.9 14.8 52.9 29.4	34 mily me 0 1 5 18 10	8.3 mbers 0 25.0 25.0 50.0	4 are i 0 0 1 1 2	6.3 nclude 0 0 66.7 33.3	3 d in a 0 0 2 1	14.6 all decisi 0 14.2 14.3 28.6 57.1	7 on-maki 0 1 1 2 3	100.0 ng about 0 4.2 14.6 47.9 33.3	48 t care 0 2 7 23 16
Total6.15(12) (Professionals) Patients and servicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal	70.8 and fa 0 2.9 14.8 52.9 29.4 70.8	34 mily me 0 1 5 18 10 34	8.3 mbers 0 25.0 25.0 50.0 8.3	4 are i 0 0 1 1 2 4	6.3 nclude 0 0 66.7 33.3 6.3	3 d in a 0 0 2 1 3	14.6 all decisi 0 14.2 14.3 28.6 57.1 14.6	7 on-maki 0 1 2 3 7	100.0 ng about 0 4.2 14.6 47.9 33.3 100.0	48 t care 0 2 7 23 16 48
Total6.15(12) (Professionals) Patients and servicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(13) (Professionals) Physica	70.8 and fa 0 2.9 14.8 52.9 29.4 70.8 , emoti	34 mily me 0 1 5 18 10 34 onal, sp	8.3 mbers 0 25.0 25.0 50.0 8.3 Diritual	4 are i 0 1 1 2 4 and s	6.3 nclude 0 0 66.7 33.3 6.3 social s	3 d in a 0 0 2 1 3 supp	14.6 all decisi 0 14.2 14.3 28.6 57.1 14.6 ort is pro	7 on-maki 0 1 2 3 7 ovided	100.0 ng about 0 4.2 14.6 47.9 33.3 100.0	48 t care 0 2 7 23 16 48
Total6.15(12) (Professionals) Patients and servicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(13) (Professionals) PhysicaStrongly disagree	70.8 and fa 0 2.9 14.8 52.9 29.4 70.8 I, emoti 29.0	34 mily me 0 1 5 18 10 34 onal, sp 1	8.3 mbers 0 25.0 25.0 50.0 8.3 Diritual 0	4 are i 0 1 1 2 4 and : 0	6.3 nclude 0 0 66.7 33.3 6.3 social s	3 d in a 0 0 2 1 3 supp 0	14.6 0 14.2 14.3 28.6 57.1 14.6 ort is pro 14.2	7 on-maki 0 1 2 3 7 ovided 1	100.0 ng about 0 4.2 14.6 47.9 33.3 100.0 4.2	48 t care 0 2 7 23 16 48 2
Total6.15(12) (Professionals) Patients and servicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(13) (Professionals) PhysicaStrongly disagreeDisagreeDisagree	70.8 and fa 0 2.9 14.8 52.9 29.4 70.8 I, emoti 29.0 29.4	34 mily me 0 1 5 18 10 34 ional, sp 1 10	8.3 mbers 0 25.0 25.0 50.0 8.3 obiritual 0 50.0	4 are i 0 1 1 2 4 and 5 0 2	6.3 nclude 0 0 66.7 33.3 6.3 social s 0 33.3	3 d in a 0 0 2 1 3 supp 0 1	14.6 0 14.2 14.3 28.6 57.1 14.6 ort is pro 14.2 0	7 on-maki 0 1 2 3 7 ovided 1 0	100.0 ng about 0 4.2 14.6 47.9 33.3 100.0 4.2 27.1	48 t care 0 2 7 23 16 48 2 13
Total6.15(12) (Professionals) Patients and servicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(13) (Professionals) PhysicaStrongly disagreeDisagreeNeither agree/disagreeNeither agree/disagreeNeither agree/disagree	70.8 and fa 0 2.9 14.8 52.9 29.4 70.8 1, emoti 29.0 29.4 14.8	34 mily me 0 1 5 18 10 34 0 nal, sp 1 10 5	8.3 mbers 0 25.0 25.0 50.0 8.3 Diritual 0 50.0 0	4 are i 0 1 1 2 4 and 5 0 2 0	6.3 nclude 0 0 66.7 33.3 6.3 social s 0 33.3 0	3 d in a 0 0 2 1 3 supp 0 1 0	14.6 0 14.2 14.3 28.6 57.1 14.6 ort is pro 14.2 14.3 14.3 28.6 57.1 14.6 ort is pro 14.2 0 14.2	7 on-maki 1 2 3 7 ovided 1 0 1	100.0 ng about 0 4.2 14.6 47.9 33.3 100.0 4.2 27.1 12.5	48 care 0 2 7 23 16 48 2 13 6
Total6.15(12) (Professionals) Patients and servicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(13) (Professionals) PhysicaStrongly disagreeDisagreeNeither agree/disagreeAgree	70.8 and fa 0 2.9 14.8 52.9 29.4 70.8 , emoti 29.0 29.4 14.8 35.3	34 mily me 0 1 5 18 10 34 0 nal, sp 1 10 5 12	8.3 mbers 0 25.0 25.0 50.0 8.3 Diritual 0 50.0 0 25.0	4 are i 0 1 1 2 4 and 5 0 2 0 0 1	6.3 nclude 0 0 66.7 33.3 6.3 social s 0 33.3 0 33.4	3 d in a 0 0 2 1 3 supp 0 1 0 1 0 1	14.6 0 14.2 14.3 28.6 57.1 14.6 ort is pro 14.2 0 14.2 28.6	7 on-maki 0 1 2 3 7 ovided 1 0 1 2	100.0 ng about 0 4.2 14.6 47.9 33.3 100.0 4.2 27.1 12.5 33.3	48 care 0 2 7 23 16 48 2 13 6 16
Total6.15(12) (Professionals) Patients and servicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(13) (Professionals) PhysicaStrongly disagreeDisagreeNeither agree/disagreeStrongly disagreeStrongly disagreeStrongly disagreeStrongly disagreeStrongly disagreeStrongly disagreeStrongly disagreeStrongly agreeStrongly agree	70.8 and fa 0 2.9 14.8 52.9 29.4 70.8 70.8 9.0 29.4 14.8 35.3 17.6	34 mily me 0 1 5 18 10 34 ional, sp 1 10 5 12 6	8.3 mbers 0 25.0 25.0 50.0 8.3 Diritual 0 50.0 0 25.0 25.0	4 are i 0 1 2 4 and 3 0 2 0 1 1 1	6.3 nclude 0 0 66.7 33.3 6.3 social s 0 33.3 0 33.4 33.3	3 d in a 0 0 2 1 3 supp 0 1 0 1 1 1	14.6 0 14.2 14.3 28.6 57.1 14.6 ort is pro 14.2 0 14.2 0 14.2 0 14.2 0 14.2 28.6 42.9	7 on-maki 0 1 2 3 7 ovided 1 0 1 2 3	100.0 ng about 0 4.2 14.6 47.9 33.3 100.0 4.2 27.1 12.5 33.3 22.9	48 0 2 7 23 16 48 2 13 6 16 11
Total6.15(12) (Professionals) Patients and servicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(13) (Professionals) PhysicaStrongly disagreeDisagreeNeither agree/disagreeStrongly disagreeStrongly disagreeStrongly disagreeStrongly disagreeStrongly disagreeStrongly agreeAgreeStrongly agreeTotal	70.8 and fa 0 2.9 14.8 52.9 29.4 70.8 9.4 70.8 9.4 14.8 35.3 17.6 70.8	34 mily me 0 1 5 18 10 34 ional, sp 1 10 5 12 6 34	8.3 mbers 0 25.0 25.0 50.0 8.3 Diritual 0 50.0 0 25.0 25.0 8.3	4 are i 0 0 1 1 2 4 and 5 0 2 0 1 1 1 4	6.3 nclude 0 0 66.7 33.3 6.3 social s 0 33.3 0 33.4 33.3 6.3	3 d in a 0 0 2 1 3 supp 0 1 0 1 1 3 3	14.6 0 14.2 14.3 28.6 57.1 14.6 ort is pro 14.2 0 14.2 0 14.2 0 14.2 0 14.2 14.2 14.2 14.2 14.2 14.2 14.2 14.2 14.2 14.2 14.2 14.2 14.2 14.5	7 on-maki 0 1 2 3 7 ovided 1 0 1 2 3 7	100.0 ng about 0 4.2 14.6 47.9 33.3 100.0 4.2 27.1 12.5 33.3 22.9 100.0	48 0 2 7 23 16 48 2 13 6 11 48
Total6.15(12) (Professionals) Patients and servicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(13) (Professionals) PhysicaStrongly disagreeDisagreeNeither agree/disagreeStrongly disagreeStrongly disagreeStrongly disagreeStrongly disagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(14) (Professionals) Services	70.8 and fa 0 2.9 14.8 52.9 29.4 70.8 9, emoti 29.0 29.4 14.8 35.3 17.6 70.8 a re pr	34 mily me 0 1 5 18 10 34 ional, sp 1 10 5 12 6 34 ovided	8.3 mbers 0 25.0 25.0 50.0 8.3 Diritual 0 50.0 0 25.0 25.0 8.3 prompt	4 are i 0 0 1 1 2 4 and 2 0 2 0 1 1 1 4 tly an	6.3 nclude 0 0 66.7 33.3 6.3 social s 0 33.3 0 33.4 33.3 6.3 d when	3 d in a 0 0 2 1 3 supp 0 1 0 1 1 3 1 1 3 3	14.6 0 14.2 14.3 28.6 57.1 14.6 ort is pro 14.2 28.6 41.2 14.2 14.2 14.2 14.2 14.2 14.2 14.2 14.2 28.6 42.9 14.6 oded	7 on-maki 0 1 2 3 7 ovided 1 0 1 2 3 7	100.0 ng about 0 4.2 14.6 47.9 33.3 100.0 4.2 27.1 12.5 33.3 22.9 100.0	48 t care 0 2 7 23 16 48 2 13 6 16 48
Total6.15(12) (Professionals) Patients and servicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(13) (Professionals) PhysicaStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly disagreeStrongly agreeTotal6.15(14) (Professionals) ServicesStrongly disagree	70.8 and fa 0 2.9 14.8 52.9 29.4 70.8 1, emoti 29.0 29.4 14.8 35.3 17.6 70.8 5 are pr 0	34 mily me 0 1 5 18 10 34 0 0 5 12 6 34 0 0	8.3 mbers 0 25.0 25.0 50.0 8.3 0 50.0 0 25.0 25.0 8.3 prompt 0	4 are i 0 0 1 1 2 4 and 5 0 2 0 1 1 1 4 stly an 0	6.3 nclude 0 0 66.7 33.3 6.3 social s 0 33.3 0 33.4 33.3 6.3 6.3 d when 0	3 d in a 0 0 2 1 3 3 supp 0 1 0 1 1 3 3 n nee 0	14.6 all decisi 0 14.2 14.3 28.6 57.1 14.6 ort is pro 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.6 oded 0	7 on-maki 0 1 2 3 7 ovided 1 0 1 2 3 7 7 0	100.0 ng about 0 4.2 14.6 47.9 33.3 100.0 4.2 27.1 12.5 33.3 22.9 100.0 0	48 t care 0 2 7 23 16 48 2 13 6 11 48 0
Total6.15(12) (Professionals) Patients and servicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(13) (Professionals) PhysicaStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(14) (Professionals) ServicesStrongly disagreeDisagreeDisagreeDisagreeDisagreeDisagreeDisagree	70.8 and fa 0 2.9 14.8 52.9 29.4 70.8 70.8 35.3 17.6 70.8 35.3 17.6 70.8 35.3 17.6 70.8 31.4	34 mily me 0 1 5 18 10 34 0 0 1 1 0 5 12 6 34 0 0 11	8.3 mbers 0 25.0 25.0 50.0 8.3 0 50.0 0 25.0 25.0 8.3 prompt 0 50.0	4 are i 0 0 1 1 2 4 and 3 0 2 0 1 1 1 4 tily an 0 2	6.3 nclude 0 0 66.7 33.3 6.3 social s 0 33.4 33.3 6.3 6.3 d when 0 66.7	3 d in a 0 0 2 1 3 3 supp 0 1 0 1 1 3 n nee 0 2	14.6 0 14.2 14.3 28.6 57.1 14.6 ort is pro 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 28.6 42.9 14.6 oded 0 28.6	7 on-maki 0 1 2 3 7 ovided 1 0 1 2 3 7 7 0 2	100.0 ng about 0 4.2 14.6 47.9 33.3 100.0 4.2 27.1 12.5 33.3 22.9 100.0 0 34.7	48 1 care 0 2 7 23 16 48 2 13 6 16 11 48 0 17
Total6.15(12) (Professionals) Patients and servicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(13) (Professionals) Physical Strongly disagreeDisagreeNeither agree/disagreeAgreeStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(14) (Professionals) ServicesStrongly disagreeDisagreeNeither agree/disagreeNeither agree/disagreeNeither agree/disagreeNeither agree/disagree	70.8 and fa 0 2.9 14.8 52.9 29.4 70.8 , emoti 29.0 29.4 14.8 35.3 17.6 70.8 are pr 0 31.4 31.4	34 mily me 0 1 5 18 10 34 0 0 1 1 0 5 12 6 34 0 0 11 11 11	8.3 mbers 0 25.0 25.0 50.0 8.3 0 50.0 0 25.0 25.0 8.3 prompt 0 50.0 0 0	4 are i 0 0 1 1 2 4 and 2 0 2 0 1 1 1 4 tly an 0 2 0 0 1 1 1 2 0 0 1 1 1 2 0 0 2 0 0 1 1 1 2 2 0 0 0 1 1 1 2 4 4 0 0 0 0 1 1 1 2 1 0 0 0 1 1 1 2 1 0 0 0 1 1 1 1	6.3 nclude 0 0 66.7 33.3 6.3 social s 0 33.4 33.3 6.3 6.3 d when 0 66.7 0	3 d in a 0 0 2 1 3 supp 0 1 0 1 1 3 1 1 3 0 2 0	14.6 0 14.2 14.3 28.6 57.1 14.6 ort is pro 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 28.6 42.9 14.6 oded 0 28.6 0 28.6 0 28.6 0 28.6	7 on-maki 0 1 2 3 7 ovided 1 0 1 2 3 7 7 0 2 0	100.0 ng about 0 4.2 14.6 47.9 33.3 100.0 4.2 27.1 12.5 33.3 22.9 100.0 0 34.7 22.4	48 1 care 0 2 7 23 16 48 2 13 6 16 48 0 17 11
Total6.15(12) (Professionals) Patients and servicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(13) (Professionals) PhysicaStrongly disagreeDisagreeNeither agree/disagreeStrongly disagreeStrongly disagreeStrongly agreeTotal6.15(13) (Professionals) PhysicaStrongly disagreeDisagreeAgreeStrongly agreeTotal6.15(14) (Professionals) ServicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly disagreeAgreeStrongly disagreeDisagreeNeither agree/disagreeAgree	70.8 and fa 2.9 14.8 52.9 29.4 70.8 9.0 29.4 14.8 35.3 17.6 70.8 5 are pr 0 31.4 31.4 34.3	34 mily me 0 1 5 18 10 34 ional, sp 1 10 5 12 6 34 ovided 0 11 11 12	8.3 mbers 0 25.0 25.0 50.0 8.3 Diritual 0 50.0 25.0 25.0 8.3 prompl 0 50.0 0 50.0	4 are i 0 0 1 1 2 4 and 2 0 2 0 1 1 1 4 tly an 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 2 2 0 0 2 2 1 2 1	6.3 nclude 0 0 66.7 33.3 6.3 social s 0 33.4 33.3 6.3 6.3 d when 0 66.7 0 33.3	3 d in a 0 0 2 1 3 supp 0 1 1 0 1 1 3 0 1 1 3 0 2 0 1	14.6 0 14.2 14.3 28.6 57.1 14.6 ort is pro 14.2 28.6 42.9 14.6 o 14.2 28.6 42.9 14.6 o 28.6 0 28.6 0 28.6 0 28.6	7 on-maki 0 1 2 3 7 ovided 1 0 1 2 3 7 0 2 0 2	100.0 ng about 0 4.2 14.6 47.9 33.3 100.0 4.2 27.1 12.5 33.3 22.9 100.0 0 34.7 22.4 34.7	48 0 2 7 23 16 48 2 13 6 16 11 48 0 17 11 17
Total6.15(12) (Professionals) Patients and servicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(13) (Professionals) PhysicaStrongly disagreeDisagreeNeither agree/disagreeStrongly disagreeDisagreeStrongly disagreeStrongly agreeTotal6.15(14) (Professionals) ServicesStrongly disagreeDisagreeNeither agree/disagreeDisagreeNeither agree/disagreeStrongly disagreeStrongly disagreeStrongly disagreeStrongly disagreeStrongly disagreeStrongly disagreeStrongly disagreeStrongly disagreeStrongly agreeStrongly agreeStrongly agree	70.8 and fa 0 2.9 14.8 52.9 29.4 70.8 9.9 29.4 70.8 29.4 14.8 35.3 17.6 70.8 31.4 31.4 34.3 2.9	34 mily me 0 1 5 18 10 34 ional, sp 1 10 5 12 6 34 ovided 0 11 11 12 1	8.3 mbers 0 25.0 25.0 50.0 8.3 Diritual 0 50.0 0 25.0 8.3 prompt 0 50.0 0 50.0 0 0	4 are i 0 0 1 1 2 4 0 2 0 0 1 1 1 4 4 tly an 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 1 1 1 2 0 0 2 0 0 0 1 1 1 2 4 1 2 2 4 0 0 0 0 1 1 1 2 1 2 1 1 1 2 2 1 1 1 1 2 2 1 1 1 1 2 2 1 1 1 1 2 2 1 1 1 1 2 2 1 1 1 1 2 2 1 1 1 1 2 2 1 1 1 1 2 2 1 1 1 1 2 2 1 1 1 1 2 2 1 1 1 1 2 2 1 1 1 1 2 2 1 1 1 1 2 2 1 1 1 1 1 2 2 1 1 1 1 1 1 1 2 2 1 1 1 1 1 1 1 2 2 1 1 1 1 1 2 2 1	6.3 nclude 0 0 66.7 33.3 6.3 social s 0 33.4 33.3 6.3 d when 0 66.7 0 33.3 0 0 33.4 33.3 6.3 0 0 33.4 33.3 0	3 d in a 0 0 2 1 3 supp 0 1 1 0 1 1 3 0 1 1 3 0 0 1 1 0 0 1 0 0 1 0 0 1 0 0 0 0	14.6 0 14.2 14.3 28.6 57.1 14.6 ort is pro 14.2 28.6 42.9 14.6 o 28.6 42.9 14.6 o 28.6 42.9 14.6 o 28.6 42.9 14.6 o 28.6 42.8	7 on-maki 0 1 2 3 7 ovided 1 0 1 2 3 7 0 2 0 2 3	100.0 ng about 0 4.2 14.6 47.9 33.3 100.0 4.2 27.1 12.5 33.3 22.9 100.0 0 34.7 22.4 34.7 8.2	48 0 2 7 23 16 48 2 13 6 11 48 0 17 14
Total6.15(12) (Professionals) Patients and servicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(13) (Professionals) PhysicaStrongly disagreeDisagreeNeither agree/disagreeStrongly disagreeStrongly disagreeStrongly agreeTotal6.15(13) (Professionals) PhysicaStrongly disagreeDisagreeStrongly agreeTotal6.15(14) (Professionals) ServicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotalTotal	70.8 and fa 0 2.9 14.8 52.9 29.4 70.8 29.4 70.8 29.0 29.4 14.8 35.3 17.6 70.8 31.4 31.4 31.4 34.3 2.9 71.4	34 mily me 0 1 5 18 10 34 ional, sp 1 10 5 12 6 34 ovided 0 11 11 12 1 1 35	8.3 mbers 0 25.0 25.0 50.0 8.3 Diritual 0 50.0 0 25.0 8.3 prompt 0 50.0 0 50.0 0 50.0 0 50.0 0 6.1	4 are i 0 0 1 1 2 4 and 2 0 0 1 1 1 4 tly an 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 1 4 4 4 4 4 6 0 0 1 1 1 2 4 4 6 0 0 1 1 1 1 2 4 4 1 1 1 2 4 4 1 1 1 1 2 4 4 1 1 1 1	6.3 nclude 0 0 66.7 33.3 6.3 social s 0 33.4 33.3 6.3 d when 0 66.7 0 33.3 0 6.3 d when 0 66.7 0 66.7	3 d in a 0 0 2 1 3 supp 0 1 1 0 1 1 3 0 1 1 2 0 0 1 1 0 2 0 1 1 0 3	14.6 0 14.2 14.3 28.6 57.1 14.6 ort is pro 14.2 28.6 42.9 14.6 o 14.2 28.6 42.9 14.6 o 28.6 42.9 14.6 o 28.6 42.8 14.3	7 on-maki 0 1 2 3 7 ovided 1 0 1 2 3 7 0 2 0 2 0 2 3 7	100.0 ng about 0 4.2 14.6 47.9 33.3 100.0 4.2 27.1 12.5 33.3 22.9 100.0 0 34.7 22.4 34.7 8.2 100.0	48 0 2 7 23 16 48 2 13 6 16 11 48 0 17 11 17 4 48
Total6.15(12) (Professionals) Patients and servicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(13) (Professionals) PhysicaStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(14) (Professionals) ServicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(15) (Professionals) Medication	70.8 and fa 0 2.9 14.8 52.9 29.4 70.8 . emoti 29.0 29.4 14.8 35.3 17.6 70.8 35.3 17.6 70.8 31.4 31.4 31.4 34.3 2.9 71.4 on req	34 mily me 0 1 5 18 10 34 0 34 0 5 12 6 34 0 0 11 11 11 12 1 35 uired ca	8.3 mbers 0 25.0 25.0 50.0 8.3 0 50.0 0 25.0 8.3 prompt 0 50.0 0 50.0 0 50.0 0 6.1 m be a	4 are i 0 0 1 1 2 4 and 1 2 0 2 0 1 1 1 4 4 tily an 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 1 1 1 5 2 4 0 2 0 0 2 0 0 1 1 1 2 4 4 2 0 0 2 0 0 1 1 1 2 4 4 1 1 1 2 2 4 4 1 1 1 2 1 2 1	6.3 nclude 0 0 66.7 33.3 6.3 social s 0 33.4 33.3 6.3 0 33.4 33.3 6.3 d when 0 66.7 0 33.3 0 6.3 sed eas	3 d in a 0 0 2 1 3 3 supp 0 1 0 1 0 1 1 3 0 2 0 0 1 0 2 0 1 0 3 3 sily	14.6 0 14.2 14.3 28.6 57.1 14.6 ort is pro 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 28.6 0 28.6 0 28.6 0 28.6 14.3	7 on-maki 0 1 2 3 7 ovided 1 0 1 2 3 7 0 2 0 2 3 7	100.0 ng about 0 4.2 14.6 47.9 33.3 100.0 4.2 27.1 12.5 33.3 22.9 100.0 0 34.7 22.4 34.7 8.2 100.0	48 1 0 2 7 23 16 48 2 13 6 16 11 48 0 17 11 17 4 48
Total6.15(12) (Professionals) Patients and servicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(13) (Professionals) PhysicaStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(14) (Professionals) ServicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(15) (Professionals) MedicatiStrongly disagreeTotal6.15(15) (Professionals) MedicatiStrongly disagree	70.8 and fa 0 2.9 14.8 52.9 29.4 70.8 , emoti 29.0 29.4 14.8 35.3 17.6 70.8 3 are pr 0 31.4 31.4 31.4 34.3 2.9 71.4 on req 0	34 mily me 0 1 5 18 10 34 0 34 0 5 12 6 34 0 0 11 11 12 1 12 1 35 uired ca 0	8.3 mbers 0 25.0 25.0 50.0 8.3 0 50.0 0 25.0 25.0 8.3 prompt 0 50.0 0 50.0 0 50.0 0 50.0 0 6.1 m be a 0	4 are i 0 0 1 1 2 4 and 2 0 2 0 1 1 1 4 tly an 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 1 1 1 2 4 4 3 6 6 9 1 1 1 2 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	6.3 nclude 0 0 66.7 33.3 6.3 social s 0 33.4 33.3 6.3 0 33.4 33.3 6.3 d when 0 66.7 0 33.3 0 66.7 0 56.7 0 56.7 0 56.7 0 56.7 0 56.7 57 57 57 57 57 57 57 57 57 57 57 57 57	3 d in a 0 0 2 1 3 supp 0 1 0 1 0 1 1 0 1 1 3 0 2 0 1 0 2 0 1 0 3 3 silly 0	14.6 0 14.2 14.3 28.6 57.1 14.6 ort is prod 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 28.6 42.9 14.6 oded 0 28.6 42.8 14.3 0	7 on-maki 0 1 2 3 7 ovided 1 0 1 2 3 7 0 2 0 2 3 7 7 0 2 3 7	100.0 ng about 0 4.2 14.6 47.9 33.3 100.0 4.2 27.1 12.5 33.3 22.9 100.0 0 34.7 22.4 34.7 8.2 100.0	48 48 care 0 2 7 23 16 48 2 13 6 16 11 48 0 17 11 17 4 48 0
Total6.15(12) (Professionals) Patients and servicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(13) (Professionals) PhysicalStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(14) (Professionals) ServicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(15) (Professionals) MedicatiStrongly disagreeDisagreeDisagreeDisagreeDisagreeDisagreeDisagreeDisagreeDisagreeDisagreeDisagree	70.8 and fa 0 2.9 14.8 52.9 29.4 70.8 , emoti 29.0 29.4 14.8 35.3 17.6 70.8 are pr 0 31.4 31.4 34.3 2.9 71.4 on req 0 35.3	34 mily me 0 1 5 18 10 34 0 0 1 1 0 5 12 6 34 0 0 11 11 12 1 35 uired ca 0 12	8.3 mbers 0 25.0 25.0 50.0 8.3 0 50.0 0 25.0 25.0 25.0 8.3 prompt 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 100	4 are i 0 0 1 1 2 4 and 2 0 2 0 0 1 1 1 4 5 1 9 a 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 1 1 1 2 4 4 0 0 0 2 0 0 0 1 1 1 2 4 0 0 0 2 0 0 0 1 1 1 2 4 0 0 0 2 0 0 0 1 1 1 2 0 0 0 1 1 1 1 2 0 0 0 1 1 1 2 0 0 0 1 1 1 1	6.3 nclude 0 0 66.7 33.3 6.3 social s 0 33.4 33.3 0 33.4 33.3 6.3 d when 0 66.7 0 33.3 0 6.3 sed eas 0 66.7	3 d in a 0 0 2 1 3 supp 0 1 0 1 0 1 1 0 1 1 3 0 2 0 1 1 0 2 0 1 1 0 2 0 1 1 3 3 sily 0 0 2 2 1 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3	14.6 0 14.2 14.3 28.6 57.1 14.6 ort is pro 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 0 14.2 28.6 42.9 14.6 o 28.6 0 28.6 14.3 0 28.6 14.3	7 on-maki 0 1 2 3 7 ovided 1 0 1 2 3 7 0 2 0 2 3 7 7 0 2 3 7 7	100.0 ng about 0 4.2 14.6 47.9 33.3 100.0 4.2 27.1 12.5 33.3 22.9 100.0 0 34.7 22.4 34.7 8.2 100.0 0 33.3	48 48 care 0 2 7 23 16 48 2 13 6 16 48 0 17 11 17 4 48 0 16
Total6.15(12) (Professionals) Patients and servicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(13) (Professionals) PhysicalStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(14) (Professionals) ServicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(15) (Professionals) MedicatiStrongly disagreeDisagreeNeither agree/disagreeDisagreeNeither agree/disagreeNeither agree/disagreeNeither agree/disagree	70.8 and fa 0 2.9 14.8 52.9 29.4 70.8 29.4 70.8 29.4 14.8 35.3 17.6 70.8 35.3 17.6 70.8 31.4 31.4 31.4 34.3 2.9 71.4 0 35.3 41.2	34 mily me 0 1 5 18 10 34 ional, sp 1 10 5 12 6 34 ovided 0 11 11 12 1 35 uired ca 0 12 14	8.3 mbers 0 25.0 25.0 50.0 8.3 Diritual 0 50.0 0 25.0 25.0 8.3 prompi 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 100 0	4 are i 0 0 1 1 2 4 and 2 0 2 0 1 1 1 4 5 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 1 1 1 2 4 0 0 0 1 1 1 2 4 0 0 0 1 1 1 2 4 0 0 0 1 1 1 2 2 4 0 0 0 1 1 1 2 0 0 0 1 1 1 2 0 0 0 1 1 1 2 0 0 0 1 1 1 2 0 0 0 1 1 1 2 0 0 0 1 1 1 1	6.3 nclude 0 0 66.7 33.3 6.3 social s 0 33.4 33.3 6.3 0 33.4 33.3 6.3 d when 0 66.7 0 33.3 0 6.3 sed eas 0 66.7	3 d in a 0 0 2 1 3 supp 0 1 1 0 1 1 3 0 1 1 0 2 0 1 0 3 3 sily 0 2 0 0	14.6 0 14.2 14.3 28.6 57.1 14.6 ort is pro 14.2 28.6 42.9 14.6 o 14.2 28.6 42.9 14.6 o 28.6 42.9 14.6 o 28.6 0 28.6 0 28.6 0 28.6 14.3	7 on-maki 0 1 2 3 7 ovided 1 0 1 2 3 7 0 2 0 2 3 7 7	100.0 ng about 0 4.2 14.6 47.9 33.3 100.0 4.2 27.1 12.5 33.3 22.9 100.0 0 34.7 22.4 34.7 8.2 100.0 0 33.3 27.5	48 48 care 0 2 7 23 16 48 2 13 6 16 11 48 0 17 11 17 4 48 0 16 18
Total6.15(12) (Professionals) Patients and servicesStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(13) (Professionals) PhysicaStrongly disagreeDisagreeNeither agree/disagreeObisagreeStrongly disagreeDisagreeNeither agree/disagreeAgreeStrongly agreeTotal6.15(14) (Professionals) ServicesStrongly disagreeDisagreeNeither agree/disagreeDisagreeStrongly disagreeDisagreeStrongly agreeTotal6.15(15) (Professionals) MedicatiStrongly disagreeDisagreeNeither agree/disagreeDisagreeNeither agree/disagreeAgreeStrongly disagreeDisagreeNeither agree/disagreeAgreeNeither agree/disagreeAgreeNeither agree/disagreeAgree	70.8 and fa 2.9 14.8 52.9 29.4 70.8 29.4 70.8 29.4 14.8 35.3 17.6 70.8 31.4 31.4 31.4 34.3 2.9 71.4 0 35.3 41.2 23.5	34 mily me 0 1 5 18 10 34 ional, sp 1 10 5 12 6 34 ovided 0 11 11 12 1 35 uired ca 0 12 14 8	8.3 mbers 0 25.0 25.0 50.0 8.3 Diritual 0 50.0 0 25.0 25.0 8.3 prompt 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 0 50.0 0 0 50.0 0 50.0 0 25.0 25.	4 are i 0 0 1 1 2 4 and 2 0 2 0 1 1 1 4 2 0 0 1 1 1 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 1 1 1 2 0 0 2 0 0 1 1 1 2 0 0 0 1 1 2 0 0 0 1 1 1 2 0 0 0 1 1 1 2 0 0 0 1 1 1 2 0 0 0 1 1 1 2 0 0 0 1 1 1 2 0 0 0 1 1 1 2 0 0 0 1 1 1 2 0 0 0 1 1 1 1	6.3 nclude 0 0 66.7 33.3 6.3 social s 0 33.4 33.3 6.3 0 33.4 33.3 6.3 d when 0 66.7 0 33.3 0 6.3 sed eas 0 66.7 0 33.3 0 6.3	3 d in a 0 0 2 1 3 supp 0 1 1 0 1 1 3 0 1 1 0 2 0 1 0 3 3 sily 0 2 0 1 1 0 1 1 0 2 0 1 1 0 1 1 3 0 0 0 0 0 0 0 0 0 0 0 0 0	14.6 0 14.2 14.3 28.6 57.1 14.6 ort is pro 14.2 28.6 42.9 14.6 o 28.6 42.9 14.6 o 28.6 0 28.6 0 28.6 0 28.6 0 28.6 0 28.6 0 28.6 0 28.6	7 on-maki 0 1 2 3 7 ovided 1 0 1 2 3 7 0 2 3 7 0 2 3 7 0 2 3 7	100.0 ng about 0 4.2 14.6 47.9 33.3 100.0 4.2 27.1 12.5 33.3 22.9 100.0 0 34.7 22.4 34.7 8.2 100.0 0 33.3 27.5 22.9	48 1 0 2 7 23 16 48 2 13 6 16 11 48 0 17 11 17 4 48 0 16 18 11

Total	70.8	34	8.3	4	6.3	3	14.3	7	100.0	48
6.15(16) (Professionals) The choir regardless of the location of the	ce for home	end-of-l	ife in tl	ne ho	me an	d a h	ome dea	th is sup	ported	
Strongly disagree	0	0	25.0	1	0	0	0	0	2.1	1
Disagree	35.3	12	25.0	1	33.3	1	28.6	2	33.3	16
Neither agree/disagree	23.5	8	0	0	0	0	0	0	16.7	8
Agree	35.3	12	25.0	1	66.7	2	28.6	2	35.4	17
Strongly agree	5.9	2	25.0	1	0	0	42.8	3	12.5	6
Total	70.8	34	8.3	4	6.3	3	14.6	7	100.0	48
6.15(17) (Professionals) All servi	ces are	afforda	ble							
Strongly disagree	5.9	2	0	0	0	0	14.3	1	6.3	3
Disagree	47.1	16	25.0	1	33.3	1	42.8	3	43.7	21
Neither agree/disagree	26.5	9	25.0	1	0	0	0	0	20.8	10
Agree	17.6	6	50.0	2	66.7	2	14.3	1	22.9	11
Strongly agree	2.9	1	0	0	0	0	28.6	2	6.3	3
Total	70.8	34	8.3	4	6.3	3	14.6	7	100.0	48
6.15(18) (Professionals) Staff pro	viding	care ha	ve the	train	ing, kn	owlee	dge and	expertis	e to prov	vide
quality care	-				•		•	-	-	
Strongly disagree	5.7	2	0	0	33.3	1	0	0	6.1	3
Disagree	11.4	4	50.0	2	0	0	28.6	2	16.3	8
Neither agree/disagree	25.8	9	25.0	1	0	0	14.3	1	22.5	11
Agree	45.7	16	25.0	1	33.4	1	14.3	1	38.8	19
Strongly agree	11.4	4	0	0	33.3	1	42.8	3	16.3	8
Total	71.4	35	8.2	4	6.1	3	14.3	7	100.0	49
6.15(19) (Professionals) Choices	for end	d-of-life	care in	Iclud	ing hos	spice	, home d	eath and	d volunta	ary
assisted dying are supported and	d client	ts are at	ole to n	nake	inform	ed de	cisions			
Strongly disagree	0	0	0	0	0	0	0	0	0	0
Disagree	26.5	9	50.0	2	0	0	28.6	2	27.1	13
Neither agree/disagree	32.3	11	50.0	2	33.4	1	0	0	29.2	14
Agree	35.3	12	0	0	33.3	1	57.1	4	35.4	17
Strongly agree	5.9	2	0	0	33.3	1	14.3	1	8.3	4
Total	70.8	34	8.3	4	6.3	3	14.6	7	100.0	48
6.15(20) (Professionals) Bereave	ment c	are is av	ailable	e as r	equire	b				
Strongly disagree	12.1	4	0	0	66.7	2	0	0	12.8	6
Disagree	27.3	9	50.0	2	0	0	0	0	23.4	11
Neither agree/disagree	42.4	14	50.0	2	0	0	0	0	34.0	16
Agree	15.2	5	0	0	0	0	71.4	5	21.3	10
Strongly agree	3.0	1	0	0	33.3	1	28.6	2	8.5	4
Total	70.2	33	8.5	4	6.4	3	14.9	7	100.0	47
6.15(21) (Professionals) Family/c	aregive	er stress	and b	urno	ut is re	duce	d with in	-home s	ervices	
Strongly disagree	0	0	0	0	0	0	0	0	0	0
Disagree	11.4	4	25.0	1	0	0	0	0	10.2	5
Neither agree/disagree	20.0	7	0	0	33.3	1	14.3	1	18.4	9
Agree	54.3	19	50.0	2	66.7	2	57.1	4	55.1	27
Strongly agree	14.3	5	25.0	1	0	0	28.6	2	16.3	8
Total	71.4	35	8.2	4	6.1	3	14.3	7	100.0	49
6.15(22) (Professionals) Staff ma care	intain g	good rel	ations	hips	with pa	tient	s/family	member	s receivi	ng
Strongly disagree	0	0	0	0	0	0	0	0	0	0
Disagree	0	0	0	0	0	0	0	0	0	0
Neither agree/disagree	8.6	3	0	0	0	0	0	0	6.1	3
Agree	57.1	20	25.0	1	66.7	2	57.1	4	55.1	27
Strongly agree	34.3	12	75.0	3	33.3	1	42.9	3	38.8	19
Total	71.4	35	8.2	4	6.1	3	14.3	7	100.0	49

Table 52 Questions 6.15 (Descriptive Statistics)

Q6.15 (Professionals) How much do you agree with the following statements?

_		_					Confidence
Statement	n	min	max	mean	SD	Variance	Interval
6.15(1) (Professionals) Existing health							
services meet the needs and	40	1	5	2.0	1 0	15	0.20
6 15(2) (Professionals) There is a need	49	1	5	5.0	1.2	1.5	0.20
for people in this area to travel over							
100km distances for some healthcare							
services	49	1	5	3.9	1.2	1.4	0.16
6.15(3) (Professionals) The need to							
travel long distances for some							
healthcare services influences							
decisions about treatment	49	2	5	4.2	0.7	0.5	0.10
6.15(4) (Professionals) The need to							
travel long distances for some							
nealthcare services results in people	40	1	F	2.0	1.0	1.0	0.14
relocating closer to services	49	1	5	3.8	1.0	1.0	0.14
6.15(5) (Professionals) People are							
home	49	1	5	19	10	1 1	0 15
6.15(6) (Professionals) Home location		1	0	1.0	1.0	1.1	0.10
does not influence access to services	49	1	5	4.1	0.8	0.7	0.11
6.15(7) (Professionals)			-				
Communication and care provided is							
culturally and gender inclusive and							
respectful	48	2	5	4.2	0.7	0.5	0.10
6.15(8) (Professionals) Personal							
Choices, Preferences and goals of			_				
care are respected	48	2	5	2.4	1.1	1.2	0.16
6.15(9) (Professionals) Access is	40	4	-	4 5	0.0	0.4	0.00
available for support 24/7	48	1	5	4.5	0.6	0.4	0.09
6.15(10) (Floressionals) Quality of the							
home	49	2	5	3.6	1.0	1.0	0.14
6.15(11) (Professionals) Ambulance				0.0			
transport and hospital admissions are							
reduced by community services							
provided	48	1	5	4.1	0.8	0.6	0.11
6.15(12) (Professionals) Patients and							
family members are included in all							
decision-making about care and	40	0	-	0.4	10	4 5	0.00
Services	48	2	5	3.4	1.2	1.5	0.20
omotional spiritual and social support							
is provided	48	1	5	12	10	1.0	0 10
6.15(14) (Professionals) Services are	10		Ű		1.0	1.0	0.10
provided promptly and when needed	49	2	5	3.0	0.9	0.8	0.10
6.15(15) (Professionals) Medication							
required can be accessed readily	48	2	5	3	0.9	0.8	0.10
6.15(16) (Professionals) The choice for							
end-of-life in the home and a home							
death is supported regardless of the	10		_				o 1 =
location of the home	48	1	5	3.3	1.0	1.1	0.15
6.15(1/) (Professionals) All services	40	1	E	2.0	1 1	1 0	0.20
are anonuable 6 15(18) (Professionals) Staff	40		5	2.9	1.1	1.3	0.20
nroviding care have the training							
knowledge and expertise to provide							
quality care	49	1	5	3.4	1.1	1.3	0.20
6.15(19) (Professionals) Choices for		· ·					0.20
end-of-life care including hospice.							

home death and voluntary assisted							
dying are supported and clients are							
able to make informed decisions	48	2	5	3.2	1.0	0.9	0.10
6.15(20) (Professionals) Bereavement							
care is available as required	47	2	5	3.2	1.0	1.3	0.20
6.15(21) (Professionals)							
Family/caregiver stress and burnout is							
reduced with in-home services	49	2	5	3.8	0.8	0.6	0.11
6.15(22) (Professionals) Staff maintain							
good relationships with							
patients/family members receiving							
care	49	3	5	4.3	0.6	0.9	0.10

Table 53 Question 6.16

Q6.16 Professionals – How do you provide palliative care in the community? Select all that apply												
Methods	Peri-urban		Rural		Regional		Metropo	olitan	Total res	ponses		
	%	n	%	n	%	n	%	n	%	n		
Face-to-face	100.0	35	100.0	4	100.0	3	100.0	7	100.0	49		
Phone	51.4	18	75.0	3	100.0	3	85.7	6	61.2	30		
Video	28.6	10	0	0	100.0	3	28.6	2	30.6	15		
Email	8.6	3	25.0	1	100.0	3	0	0	14.3	7		
Text	22.9	8	50.0	2	100.0	3	14.3	1	28.6	14		
Apps	0	0	0	0	0	0	0	0	0	0		
Role does not contact	2.9	1	0	0	0	0	0	0	2.0	1		
Total respondents	71.4	35	8.2	4	6.1	3	14.3	7	100.0	49		
Total responses	64.6	75	8.7	10	12.9	15	13.8	16	100.0	116		

Table 54 Question 6.17

Q6.17 Professional – Why do you use digital health options? Select all that apply													
Reasons	Peri-u	rban	Rur	al	Region	nal	Metrop	olitan	Total res	ponses			
	%	n	%	n	%	n	%	n	%	n			
Reduce infection	38.7	12	0	0	0	0	50.0	3	34.9	15			
Improve support	29.0	9	66.7	2	66.7	2	83.3	5	41.9	18			
Reduce travel time for clinician	29.0	9	66.7	2	66.7	2	50.0	3	37.2	16			
Patient/family choice	25.8	8	0	0	33.3	1	66.7	4	30.2	13			
Face-to-face not required	9.7	2	33.3	1	66.7	2	0	0	13.9	6			
After hours support	16.1	5	0	0	0	0	66.7	4	20.9	9			
Professional choice	41.9	13	66.7	2	33.3	1	66.7	4	46.5	20			
Home visit not required	29.0	9	33.3	1	0	0	83.3	5	34.9	15			
Family present	12.9	4	0	0	66.7	2	33.3	2	18.6	8			
Specialist input needed	32.2	10	33.3	1	100.0	3	33.3	2	37.2	16			
Total respondents	72.1	31	7.0	3	7.0	3	13.9	6	100.0	43			
Total responses	60.3	82	6.6	9	9.6	13	23.5	32	100.0	136			

Problems	Peri-u	ırban	Ru	ral	Regior	nal	Metrop	olitan	Total res	sponses
	%	n	%	n	%	n	%	n	%	n
No problems using	15.1	5	0	0	0	0	50.0	3	17.0	8
Patient lack access/skill	30.3	10	0	0	100.0	3	50.0	3	34.0	16
Choice patient/family	24.2	8	20.0	1	33.3	1	33.3	2	25.5	12
Poor mobile coverage	42.4	14	60.0	3	100.0	3	16.7	1	44.7	21
Unreliable/no internet	33.3	11	60.0	3	100.0	3	0	0	36.2	17
No digital option offered	6.1	2	0	0	0	0	16.7	1	6.4	3
Face-to-face required	45.4	15	40.0	2	0	0	50.0	3	42.6	20
Patient does not answer phone	9.1	3	0	0	0	0	0	0	6.4	3
Hearing/language barrier	3.0	1	0	0	0	0	0	0	2.1	1
Sometimes/not used	6.1	2	0	0	0	0	0	0	4.2	2
Total respondents	70.2	33	10.6	5	6.4	3	12.8	6	100.0	47
Total responses	68.9	71	8.8	9	9.7	10	12.6	13	100.0	103

Table 55 Question 6.18

Table 56 Question 6.19

Q6.19 Professionals - How	/ much	do you	agree	with	the follo	win	g statem	nents?		
	Peri-	urban	Rur	al	Region	nal	Metrop	olitan	Total r	esponses
	%	n	%	n	%	n	%	n	%	n
6.19(1) (Professional) Digi	tal hea	lth is an	i impor	tant f	tool in e	nsu	ring regu	ular com	municati	ion with
patients										
Strongly disagree	0	0	0	0	0	0	0	0	0	0
Somewhat disagree	9.4	3	0	0	33.3	1	0	0	8.7	4
Neither agree/disagree	28.1	9	25.0	1	0	0	14.3	1	23.9	11
Somewhat agree	50.0	16	25.0	1	66.7	2	42.8	3	47.8	22
Strongly agree	12.5	4	50.0	2	0	0	42.9	3	19.6	9
Total	69.6	32	8.7	4	6.5	3	15.2	7	100.0	46
6.19(2) (Professional) I have	ve the	resourc	es and	conf	idence t	o us	se digital	health	options e	effectively
Strongly disagree	0	0	0	0	0	0	0	0	0	0
Somewhat disagree	9.1	3	0	0	33.3	1	28.6	2	12.8	6
Neither agree/disagree	9.1	3	25.0	1	0	0	0	0	8.5	4
Somewhat agree	48.0	16	25.0	1	33.4	1	28.6	2	42.5	20
Strongly agree	33.3	11	50.0	2	33.3	1	42.8	3	36.2	17
Total	70.2	33	8.5	4	6.4	3	14.9	7	100.0	47
6.19(3) (Professional) Tele consultations with medica	health	consult ssional	ations s	are a	in accep	tabl	e alterna	ative to f	ace-to-fa	ice
Strongly disagree	0	0	0	0	0	0	0	0	0	0
Somewhat disagree	18.7	6	50.0	2	33.3	1	14.3	1	21.7	10
Neither agree/disagree	28.1	9	25.0	1	0	0	28.6	2	26.1	12
Somewhat agree	46.9	15	25.0	1	66.7	2	42.9	3	45.6	21
Strongly agree	6.3	2	0	0	0	0	14.2	1	6.6	3
Total	69.6	32	8.7	4	6.5	3	15.2	7	100.0	46
6.19(4) (Professional) Tele be readily accessible due	health to dist	means ance	that pa	tient	s can ac	ces	s medica	al suppo	rt that w	ould not
Strongly disagree	0	0	0	0	0	0	0	0	0	0
Somewhat disagree	0	0	0	0	0	0	0	0	0	0
Neither agree/disagree	24.2	8	50.0	2	0	0	28.6	2	25.5	12
Somewhat agree	54.6	18	25.0	1	66.7	2	14.3	1	46.8	22
Strongly agree	21.2	7	25.0	1	33.3	1	57.1	4	27.7	13
Total	70.2	33	8.5	4	6.4	3	14.9	7	100.0	47

6.19(5) (Professional) I fee options	l that p	atient's	choice	e of p	place of o	care	is supp	orted wi	th digita	health
Strongly disagree	0	0	0	0	0	0	0	0	0	0
Somewhat disagree	6.1	2	0	0	0	0	14.3	3	6.4	3
Neither agree/disagree	30.3	10	50.0	2	0	0	42.9	1	31.9	15
Somewhat agree	45.4	15	50.0	2	100.0	3	0	0	42.6	20
Strongly agree	18.2	6	0	0	0	0	42.8	3	19.1	9
Total	70.2	33	8.5	4	6.4	3	14.9	7	100.0	47
6.19(6) (Professional) Pati	ents ar	nd family	y memi	bers	are com	forta	able to r	emain at	home k	nowing
6.19(6) (Professional) Pati that they can access supp	ents ar	nd family any time	y memi e	bers	are com	forta	able to r	emain at	home ki	nowing
6.19(6) (Professional) Pati that they can access supp Strongly disagree	ents ar ort at a 3.0	nd family any time 1	y mem 9 0	bers 0	are com	f ort a	able to r	emain at	home k	nowing
6.19(6) (Professional) Pati that they can access supp Strongly disagree Somewhat disagree	ents ar ort at a 3.0 12.1	nd family any time 1 4	y mem 9 0 25.0	bers 0 1	are comi 0 0	forta 0 0	able to re 0 0	emain at 0 0	home k 2.1 10.6	nowing 1 5
6.19(6) (Professional) Pati that they can access supp Strongly disagree Somewhat disagree Neither agree/disagree	ents ar ort at a 3.0 12.1 45.5	nd family any time 1 4 15	y mem 0 25.0 75.0	0 1 3	are com 0 0 0	forta 0 0	able to re 0 0 28.6	emain at 0 0 2	2.1 10.6 42.6	1 5 20
6.19(6) (Professional) Pati that they can access supp Strongly disagree Somewhat disagree Neither agree/disagree Somewhat agree	ents ar ort at a 3.0 12.1 45.5 24.2	nd family any time 1 4 15 8	y mem 0 25.0 75.0 0	0 1 3 0	are com 0 0 0 100.0	forta 0 0 0 3	0 0 28.6 42.8	emain at 0 0 2 3	2.1 10.6 42.6 29.8	1 5 20 14
6.19(6) (Professional) Pati that they can access supp Strongly disagree Somewhat disagree Neither agree/disagree Somewhat agree Strongly agree	ents ar ort at a 3.0 12.1 45.5 24.2 15.2	nd family any time 1 4 15 8 5	y mem 0 25.0 75.0 0 0	0 1 3 0 0	are com 0 0 100.0 0	forta 0 0 0 3 0	0 0 28.6 42.8 28.6	emain at 0 0 2 3 2	Anne ki 2.1 10.6 42.6 29.8 14.9	1 5 20 14 7

Table 57 Question 6.20

Q6.20 Professionals - Overall are you satisfied with the access to community palliative care services provided in your location?										
Opinion	Peri-u	rban	Rur	al	Regio	nal	Metrop	olitan	Total res	ponses
	%	n	%	n	%	n	%	n	%	n
Yes	30.8	8	0	0	66.7	2	85.7	6	42.1	16
No	69.2	18	100.0	2	33.3	1	14.3	1	57.9	22
Total	68.4	26	5.3	2	7.9	3	18.4	7	100.0	38

People diagnosed with a life-limiting illness (Patient)

Table 58 Question 5.1

Q5.1 Patient - Your a	age groi	up?								
Age	Peri-urban		Rural		Regional		Metropolitan		Total responses	
-	%	n	%	n	%	n	%	n	%	n
18-34 years	8.3	1	0	0	25.0	1	33.3	1	11.1	3
35-49 years	25.0	3	12.5	1	0	0	0	0	14.8	4
50-64 years	41.7	5	37.5	3	25.0	1	66.7	2	40.7	11
65+	25.0	3	50.0	4	50.0	2	0	0	33.3	9
Total	44.4	12	29.7	8	14.8	4	11.1	3	100.0	27

Table 59 Question 5.2

Q5.2 Patient - How long have you been receiving community palliative care?													
Receiving care	Peri-u	Peri-urban		Rural		Regional		Metropolitan		sponses			
	%	n	%	n	%	n	%	n	%	n			
<1 month	30.8	4	0	0	0	0	33.3	1	17.8	5			
1-6 months	38.5	5	62.5	5	75.0	3	33.3	1	50.6	14			
6-12 months	7.7	1	12.5	1	0	0	0	0	7.1	2			
>1 year	23.0	3	28.6	2	25.0	1	33.4	1	25.0	7			
Total	46.4	13	28.6	8	14.3	4	10.7	3	100.0	28			

Table 60 Question 5.3

Q5.3 Patient - In what State/Territory do you live?													
State/Territory	Peri-u	urban	Rur	al	Regio	onal	Metro	oolitan	Total re	esponses			
	%	n	%	n	%	n	%	n	%	n			
Queensland	3.3	2	12.5	1	25.0	1	0	0	13.3	4			
South Australia	60.0	9	87.5	7	25.0	1	100.0	3	66.7	20			
NT, WA and ACT	0	0	0	0	0	0	0	0	0	0			
New South Wales	6.7	1	0	0	25.0	1	0	0	6.7	2			
Victoria	13.3	2	0	0	25.0	1	0	0	10.0	3			
Tasmania	6.7	1	0	0	0	0	0	0	3.3	1			
Total	50.0	15	26.7	8	13.3	4	10.0	3	100.0	30			

Table 61 Question 5.4

Q5.4 Patient – How would you describe the area where you live and receive care?								
	%	n						
Peri-urban	46.2	12						
Rural	30.8	8						
Regional	11.5	3						
Metropolitan	11.5	3						
Total	100.0	26						

Table 62 Question 5.6

Q5.6 (Patient) How strongly do you feel about the following statements?										
	Peri-	urban	Rur	al	Regio	onal	Metrop	olitan	Total r	esponses
	%	n	%	n	%	n	%	n	%	n
5.6(1) (Patient) I chose this	s locati	on as	health	servi	ces I ne	ed a	re close	by		
Strongly disagree	28.6	4	77.8	7	0	0	33.3	1	40.0	12
Somewhat disagree	35.7	5	11.1	1	25.0	1	33.3	1	26.7	8
Neither agree/disagree	0	0	0	0	25.0	1	0	0	3.3	1
Somewhat agree	21.4	3	11.1	1	50.0	2	33.3	1	23.3	7
Strongly agree	14.3	2	0	0	0	0	0	0	6.7	2
Total	46.7	14	30.0	9	13.3	4	10.0	3	100.0	30
5.6(2) (Patient) I chose this	s locati	on as	service	sIne	eed/ma	y nee	ed, are ir	n the plai	nning or	
development stage										
Strongly disagree	64.3	9	66.7	6	50.0	2	66.7	2	63.3	19
Somewhat disagree	7.1	1	22.2	2	25.0	1	0	0	13.4	4
Neither agree/disagree	0	0	0	0	0	0	0	0	0	0
Somewhat agree	21.4	3	11.1	1	25.0	1	33.3	1	20.0	6
Strongly agree	7.2	1	0	0	0	0	0	0	3.3	1
Total	46.7	14	30.0	9	13.3	4	10.0	3	100.0	30
5.6(3) (Patient) I chose this	s locati	on as	health	servi	ces tha	t pro	vide for	increase	d care n	eeds are
close by (eg hospice, age	d care	facility	, specia	alist (care)					
Strongly disagree	71.4	10	77.8	7	50.0	2	66.7	2	70.0	21
Somewhat disagree	7.1	1	11.1	1	25.0	1	0	0	10.0	3
Neither agree/disagree	7.2	1	0	0	0	0	0	0	3.3	1
Somewhat agree	14.3	2	11.1	1	25.0	1	33.3	1	16.7	5
Strongly agree	0	0	0	0	0	0	0	0	0	0
Total	46.7	14	30.0	9	13.3	4	10.0	3	100.0	30
5.6(4) (Patient) There was	5.6(4) (Patient) There was an expectation that health services were accessible									

Strongly disagree	0	0	0	0	0	0	0	0	0	0
Somewhat disagree	7.2	1	11.1	1	0	0	0	0	6.7	2
Neither agree/disagree	0	0	0	0	0	0	0	0	0	0
Somewhat agree	21.4	3	44.4	4	25.0	1	0	0	26.7	8
Strongly agree	71.4	10	44.4	4	75.0	3	100	3	66.6	20
Total	46.7	14	30.0	9	13.3	4	10.0	3	100.0	30

Table 63 Question 5.7

Q5.7 Patient - How did	you ideı	ntify, co	ontact or	' acce	ss servi	ices a	Ind inforn	nation?	Select all t	hat
Find information	Peri-u	rban	Ru	ral	Regional		Metropolitan		Total responses	
	%	n	%	n	%	n	%	n	%	n
Not involved in access	0	0	33.3	3	0	0	0	0	10.0	3
Health Professional	71.4	10	44.4	4	75.0	3	100.0	3	66.7	20
Referred	78.6	11	66.7	6	100	4	66.7	2	76.7	23
Friends/family	0	0	11.1	1	25.0	1	33.3	1	10.0	3
Pamphlets/booklets	0	0	0	0	0	0	0	0	0	0
Online forums	14.3	2	0	0	0	0	0	0	6.7	2
Support groups	14.3	2	11.1	1	25.0	1	0	0	13.3	4
Google	7.1	1	11.1	1	0	0	0	0	6.7	2
Knowledge/experience	0	0	0	0	0	0	0	0	0	0
Total responses	45.6	26	28.1	16	15.8	9	10.5	6	100.0	57
Total respondents	46.7	14	30.0	9	13.3	4	10.0	3	100.0	30

Table 64 Question 5.8

Q5.8 Patient - Did yo needed?	ou find i	t difficu	lt to find	d info	rmation	abou	it service/	s and i	dentify what	was
Yes/no	Peri-u	rban	Ru	ral	Regio	nal	Metrop	olitan	Total resp	onses
	%	n	%	n	%	n	%	n	%	n
Yes	46.2	6	50.0	3	75.0	3	33.3	1	50.0	3
No	53.8	7	50.0	3	25.0	1	66.7	2	50.0	13
Total	50.0	13	23.1	6	15.4	4	11.5	3	100.0	26

Table 65 Question 5.9

Q5.9 Patient - Did you find it difficult to access identified services when required?												
Yes/no	Metropolitan Total responses											
	%	n	%	n	%	n	%	n	%	n		
Yes	38.5	5	85.7	6	33.3	1	100	1	54.2	13		
No	61.5	8	14.3	1	66.7	2	0	0	45.8	11		
Total	54.2	13	29.1	7	12.5	3	4.2	1	100.0	24		

Table 66 Question 5.10

Q5.10 Patient - Please select all community services you are using/have used. Select all that apply												
Services	Peri-u	ırban	Rural		Regional		Metropolitan		Total responses			
	%	n	%	n	%	n	%	n	%	n		
Unsure	0	0	11.1	1	25.0	1	0	0	6.7	2		
Specialist Palliative Care (SPC)	71.4	10	44.4	4	50.0	2	33.3	1	56.7	17		
Community Nursing	7.1	1	44.4	4	25.0	1	0	0	20.0	6		

Personal Care	14.3	2	22.2	2	25.0	1	0	0	16.7	5
Transport Assistance	0	0	22.2	2	0	0	0	0	6.7	2
Occupational Therapy	14.3	2	33.3	3	25.0	1	0	0	20.0	6
Psychiatrist/ Psychologist	14.3	2	0	0	0	0	33.3	1	10.0	3
Social Worker	35.7	5	22.2	2	50.0	2	33.3	1	33.3	10
Physiotherapy	14.3	2	11.1	1	0	0	0	0	10.0	3
Dietician/Speech Pathology	7.1	1	0	0	0	0	0	0	3.3	3
Palliative Care Specialist	21.4	3	0	0	0	0	0	0	3.3	3
Spiritual support	0	0	0	0	0	0	0	0	0	0
Volunteers	0	0	0	0	0	0	0	0	0	0
Complementary therapy	7.1	1	0	0	0	0	0	0	3.3	1
Housework/Garden maintenance	21.4	3	11.1	1	25.0	1	0	0	16.7	5
Home maintenance/ modification	7.1	1	11.1	1	0	0	0	0	6.7	2
Equipment	28.6	4	44.4	4	0	0	0	0	26.7	8
Advance care directive assistance	14.3	2	0	0	0	0	0	0	6.7	2
Financial/legal assistance	14.3	2	0	0	25.0	1	0	0	10.0	3
Regular GP	35.7	5	11.1	1	50.0	2	0	0	26.7	8
Family/friends	28.6	4	22.2	2	75.0	3	33.3	1	33.3	10
MND Association	7.1	1	0	0	0	0	0	0	3.3	1
Breast cancer nurse	7.1	1	0	0	0	0	0	0	3.3	1
My Aged Care	0	0	11.1	1	0	0	0	0	3.3	1
Home Oxygen	7.1	1	0	0	0	0	0	0	3.3	1
None	7.1	1	22.2	2	0	0	33.3	1	13.3	4
Total responses	51.4	54	29.5	31	14.3	15	4.8	5	100.0	105
Total respondents	46.7	14	30.0	9	13.3	4	10.0	3	100.0	30

Table 67 Question 5.11

Q5.11 Patient - Do the services you receive meet your needs and expectations?												
Yes/no	Peri-u	rban	Ru	Rural		Regional		Metropolitan		ponses		
	%	n	%	n	%	n	%	n	%	n		
Yes	53.8	7	44.4	4	25.0	1	66.7	2	48.3	14		
Partially	23.1	3	33.3	3	50.0	2	0	0	27.6	8		
No	23.1	3	22.2	2	25.0	1	33.3	1	24.1	7		
Total	44.8	13	31.0	9	13.8	4	10.4	3	100.0	29		

Table 68 Question 5.12

5.12 Patient - If you are not using all services that are available, why not? Select all that apply												
Reasons	Peri-u	rban	Rural		Regional		Metropolitan		Total responses			
	%	n	%	n	%	n	%	n	%	n		
Not needed now	61.5	8	44.4	4	50.0	2	66.7	2	55.2	16		
Family provide	23.1	3	11.1	1	25.0	1	0	0	17.2	5		
Not happy with service previously	0	0	0	0	0	0	0	0	0	0		
Not when I want	0	0	11.1	1	0	0	0	0	3.4	1		
Not available as often as I want	0	0	22.2	2	0	0	0	0	6.9	2		
I am under 65 years	53.8	7	44.4	4	25.0	1	33.3	1	44.8	13		

Not available in this location	23.1	3	55.6	5	25.0	1	0	0	31.0	9
Local providers lack resources	15.4	2	44.4	4	0	0	0	0	20.7	6
Too expensive	30.8	4	22.2	2	0	0	0	0	24.1	7
Using all I knew about	0	0	0	0	0	0	0	0	0	0
Personal choice	15.4	2	22.2	2	25.0	1	0	0	13.8	4
Unsure why no access	7.7	1	22.2	2	25.0	1	0	0	13.8	4
Total responses	44.8	30	40.3	27	10.4	7	4.5	3	100.0	67
Total respondents	44.9	13	31.0	9	13.8	4	10.3	3	100.0	29

Table 69 Question 5.13

Q5.13 Patient - Please select a	ny of t	he foll	owing	that	apply					
	Peri-	urban	Rui	ral	Regio	onal	Metrop	olitan	Total respor	ises
	%	n	%	n	%	n	%	n	%	n
Q5.13(1) (Patient) I have had one or more hospital stays extended as services unavailable to support my discharge home	80.8	8	57.1	4	0	0	100.0	1	61.9	13
Q5.13(2) (Patient) I have needed to attend a clinic or hospital out- patients for medication administration as in-home services unavailable	30.0	3	14.3	1	33.3	1	100.0	1	28.6	6
Q5.13(3) (Patient) I have needed to attend a clinic or hospital out- patients for assistance with wound/catheter/stoma care as in- home services unavailable	30.0	3	14.3	1	0	0	0	0	19.0	4
Q5.13(4) (Patient) I have needed to call an ambulance for assistance with pain/symptom control as unable to contact local services	20.0	2	28.6	2	0	0	0	0	19.0	4
Q5.13(5) (Patient) Respite in aged care facility has been required as in-home services unavailable to support hospital discharge	0	0	14.3	1	0	0	0	0	4.8	1
Q5.13(6) (Patient) Will need to transfer to hospital for end-of-life care to remain in this location	20.0	2	85.7	6	66.7	2	0	0	47.8	10
Total responses	47.6	10	33.3	7	14.3	3	4.8	1	100.0	21

Table 70 Question 5.14

Q5.14 Patient - What is the longest distance you have needed to travel for healthcare services?													
Distance	Peri-u	rban	Ru	ral	Regio	nal	Metropo	olitan	Total resp	onses			
	%	n	%	n	%	n	%	n	%	n			
Services local	0	0	0	0	0	0	0	0	0	0			
10-50 km	7.1	1	0	0	25.0	1	100.0	2	13.8	4			
50-100 km	64.3	9	0	0	0	0	0	0	31.0	9			
100-150 km	28.6	4	44.4	4	0	0	0	0	27.6	8			
150-250 km	0	0	55.6	5	75.0	3	0	0	27.6	8			
Unsure	0	0	0	0	0	0	0	0	0	0			
Total responses	48.3	14	31.0	9	13.8	4	6.9	2	100.0	29			

Table 71 Question 5.15

Q5.15 Patient - What barriers to accessing services do you believe apply in your situation? Select all that apply												
Barriers	Peri-u	rban	Ru	ral	Regio	nal	Metrop	olitan	Total res	sponses		
	%	n	%	n	%	n	%	n	%	n		
No barriers to access	21.4	3	0	0	0	0	0	0	10.3	3		
I am under 65 years	57.1	8	66.7	6	25.0	1	100	2	58.6	17		
Distance for providers	28.6	4	100	9	25.0	1	0	0	48.3	14		
Providers lack resources	21.4	3	44.4	4	0	0	0	0	24.1	7		
Time taken to start	7.1	1	33.3	3	0	0	0	0	13.8	4		
Travel from	92.8	13	65.2	15	57.1	4	0	0	16.7	32		
Distance to travel to	21.4	3	77.8	7	25.0	1	0	0	37.9	11		
Diversity	0	0	11.1	1	0	0	0	0	3.4	1		
Language	0	0	11.1	1	0	0	0	0	3.4	1		
l do not want palliative care	7.1	1	33.3	3	0	0	0	0	13.8	4		
Challenging behaviours	0	0	11.1	1	0	0	0	0	3.4	1		
Financial	7.1	1	11.1	1	0	0	0	0	6.9	2		
Personal choice	7.1	1	0	0	0	0	0	0	3.4	1		
Care needs exceed local provider scope	14.3	2	44.4	4	25.0	1	0	0	24.1	7		
No need identified	21.4	3	0	0	25.0	1	50.0	1	17.2	5		
Total responses	39.1	43	50.0	55	8.2	9	2.7	3	100.0	110		
Total respondents	48.3	14	31.0	9	13.8	4	6.9	2	100.0	29		

Table 72 Question 5.16

Q5.16 Patient - Thinking about distance to health services, please select all statements that apply to you.

	Peri-u	urban	Rura	I	Regio	onal	Metrop	olitan	Total respon	ses
	%	n	%	n	%	n	%	n	%	n
Q5.16(1) Patient - The distance required to travel to some health services has influenced my decisions about attending appointments, treatment or tests	42.8	6	44.4	4	50.0	2	50.0	1	44.8	13
Q5.16(2) Patient – The distance required to travel to some health	50.0	7	11.1	1	25.0	1	0	0	3.0	9

services has influenced where I live										
Q5.16(3) Patient - Travel any distance to health services is difficult (exhausting, pain, discomfort, etc)	85.7	12	77.8	7	25.0	1	50.0	1	72.4	21
Q5.16(4) Patient – Private transport to some health services is difficult due to cost of petrol, parking, lost wages and time involved for my family	85.7	12	88.9	8	50.0	2	100	2	82.7	24
Q5.16(5) Patient – Public transport is not an option in this location	92.3	12	72.8	7	75.0	3	50.0	1	79.3	23
Q5.16(6) Patient – Community car is not available/expensive/not an option	64.3	9	44,4	4	50.0	2	66.7	2	58.6	17
Q5.16(7) Patient – Health services are provided in my home	7.1	1	11.1	1	25.0	1	0	0	10.3	3
Q5.16(8) Patient – Health services are provided locally	0	0	11.1	1	25.0	1	0	0	6.9	2
Q5.16(9) Patient – I will need to relocate to be closer to services that will provide the level of care I need/will need	78.6	11	100.0	9	50.0	2	0	0	79.3	23
Total respondents	48.3	14	31.0	9	13.8	4	6.9	2	100.0	29

Table 73 Question 5.17

Q5.17 Patient - How mu	ch do yo	ou ag	ree wit	h the	following	y stat	ements?	•		
	Peri-ur	ban	Ru	ral	Region	al	Metrop	olitan	Total r	esponses
	%	n	%	n	%	n	%	n	%	n
5.17(1) (Patient) I have b	been sup	porte	ed in m	y cho	oice to re	main	in own h	ome		
Strongly disagree	8.3	1	22.2	2	25.0	1	0	0	14.8	4
Somewhat disagree	0	0	0	0	0	0	0	0	0	0
Neither agree/disagree	16.7	2	11.1	1	0	0	100.0	2	18.5	5
Somewhat agree	25.0	3	44.4	4	0	0	0	0	25.9	7
Strongly agree	50.0	6	22.2	2	75.0	3	0	0	40.8	11
Total	44.5	12	33.3	9	14.8	4	7.4	2	100.0	27
5.17(2) (Patient) Person	al choic	es, pr	eferen	ces a	nd goals	of ca	re are be	eing resp	pected	
Strongly disagree	16.7	2	22.2	2	25.0	1	0	0	18.5	5
Somewhat disagree	0	0	0	0	0	0	0	0	0	0
Neither agree/disagree	16.7	2	11.1	1	0	0	50.0	1	14.8	4
Somewhat agree	16.7	2	44.4	4	0	0	50.0	1	25.9	7
Strongly agree	50.0	6	22.2	2	75.0	3	0	0	40.8	11
Total	44.5	12	33.3	9	14.8	4	7.4	2	100.0	27
5.17(3) (Patient) All com	municat	tion a	nd car	e pro	vided has	s bee	n cultura	ally and g	gender i	nclusive
and respectful			-	-			-	-		
Strongly disagree	8.3	1	0	0	25.0	1	0	0	7.7	2
Somewhat disagree	0	0	0	0	25.0	1	0	0	3.8	1
Neither agree/disagree	8.3	1	12.5	1	0	0	0	0	7.7	2
Somewhat agree	33.3	4	37.5	3	0	0	50.0	1	30.8	8
Strongly agree	50.0	6	50.0	4	50.0	2	50.0	1	50.0	13
Total	46.1	12	39.8	8	15.4	4	7.7	2	100.0	26
5.17(4) (Patient) I have a	access to	o sup	port 24	hou	rs a day a	and 7	days a v	veek		
Strongly disagree	8.3	1	44.4	4	0	0	0	0	18.5	5
Somewhat disagree	25.0	3	33.3	3	50.0	2	0	0	29.6	8
Neither agree/disagree	8.3	1	0	0	0	0	0	0	3.8	1
Somewhat agree	16.7	2	22.2	2	25.0	1	100.0	2	25.9	7

Strongly agree	41.7	5	0	0	25.0	1	0	0	22.2	6
Total	44.5	12	33.3	9	14.8	4	7.4	2	100.0	27
5.17(5) (Patient) My qua	lity of lif	e has	been i	impro	ved by re	emair	ning in m	y own h	ome	1
Strongly disagree	8.3	1	0	0	0	0	0	0	3.7	1
Somewhat disagree	0	0	0	0	0	0	0	0	0	0
Neither agree/disagree	16.7	2	33.3	3	0	0	0	0	18.5	5
Somewhat agree	8.3	1	11.1	1	0	0	0	0	7.4	2
Strongly agree	66.7	8	55.5	5	100.0	4	100.0	2	70.4	19
Total	44.5	12	33.3	9	14.8	4	7.4	2	100.0	27
5.17(6) (Patient) Access	to com	nunit	v servi	ices h	ave redu	ced r	ny need	for ambi	ulance tr	ansport
and hospital admission	S		-				•			•
Strongly disagree	8.3	1	33.3	3	25.0	1	50.0	1	22.2	6
Somewhat disagree	16.7	2	22.2	2	0	0	0	0	14.8	4
Neither agree/disagree	33.3	4	11.1	1	0	0	50.0	1	22.3	6
Somewhat agree	16.7	2	22.2	2	75.0	3	0	0	25.9	7
Strongly agree	25.0	3	11.1	1	0	0	0	0	25.9	4
Total	44.5	12	33.3	9	14.8	4	7.4	2	100.0	27
5.17(7) (Patient) I have I	been inc	luded	in all o	decis	ion-maki	ng ab	out care	and ser	vices pro	ovided
Strongly disagree	8.3	1	0	0	0	0	0	0	3.8	1
Somewhat disagree	0	0	33.3	3	25.0	1	0	0	14.8	4
Neither agree/disagree	0	0	33.3	3	0	0	0	0	11.1	3
Somewhat agree	41.7	5	11.1	1	25.0	1	0	0	25.9	7
Strongly agree	50.0	6	22.2	2	50.0	2	100	2	44.4	12
Total	44.5	12	33.3	9	14.8	4	7.4	2	100.0	27
5.17(8) (Patient) I am co	mfortab	e wit	h the le	evel o	of sympto	m ma	anageme	nt that is	s availab	le
Strongly disagree	16.7	2	22.2	2	0	0	0	0	7.4	2
Somewhat disagree	0	0	11.1	1	25.0	1	0	0	14.8	4
Neither agree/disagree	8.3	1	22.2	2	25.0	1	0	0	14.9	4
Somewhat agree	33.3	4	44.4	4	0	0	50.0	1	33.3	9
Strongly agree	117	5	0	Δ	50.0	2	50.0	1	20.6	0
	41.7	0	0	0	0.00	Z	0.00		29.0	0
Total	44.5	12	33.3	9	14.8	2 4	7.4	2	29.0 100.0	0 27
Total 5.17(9) (Patient) I have i	44.5 received	12 supp	33.3 ort wit	9 h adv	14.8 14.8	∠ 4 e plai	7.4 7.9, co	2 2 2	100.0 29.0 g my wil	o 27 I and grief
Total 5.17(9) (Patient) I have r counselling	44.5 received	12 supp	33.3 ort wit	9 h adv	14.8 ance car	2 4 e plai	7.4 nning, co	2 ompletin	29.0 100.0 g my wil	o 27 I and grief
Total 5.17(9) (Patient) I have r counselling Strongly disagree	44.5 received	12 supp	33.3 ort wit	9 h adv 2	14.8 vance car	2 4 e plai	7.4 nning, co	2 cmpletin	100.0 g my wil	27 I and grief
Total 5.17(9) (Patient) I have i counselling Strongly disagree Somewhat disagree	44.5 received	12 supp 2 1	33.3 ort wit 22.2 0	9 h adv 2 0	0 25.0	2 4 e plai 0 1	0 0	2 ompletin	29.0 100.0 g my wil 14.8 7.4	27 I and grief
Total 5.17(9) (Patient) I have in counselling Strongly disagree Somewhat disagree Neither agree/disagree	44.5 received 16.7 11.1 8.3	2 1 1 2	33.3 ort wit 22.2 0 22.2	9 h adv 2 0 2	0 25.0 25.0	2 4 e pla 0 1 1	0 0 0	2 cmpletin 0 0 0	29.0 100.0 g my wil 14.8 7.4 14.9	27 I and grief 4 2 4
Total 5.17(9) (Patient) I have in counselling Strongly disagree Somewhat disagree Neither agree/disagree Somewhat agree	44.5 received 16.7 11.1 8.3 33.3	2 1 1 2 1 1 4	33.3 ort wit 22.2 0 22.2 44.4	9 h adv 2 0 2 4	0 25.0 50.0	2 4 e plan 0 1 1 2	0 0 0 100	2 cmpletin 0 0 0 2	29.0 100.0 g my wil 14.8 7.4 14.9 44.4	27 I and grief 4 2 4 12
Total 5.17(9) (Patient) I have in counselling Strongly disagree Somewhat disagree Neither agree/disagree Somewhat agree Strongly agree	44.5 received 16.7 11.1 8.3 33.3 41.7	2 1 1 1 1 4 5	33.3 ort wit 22.2 0 22.2 44.4 0	9 h adv 2 0 2 4 0	0 25.0 50.0 0	2 e plan 0 1 1 2 0	0 0 0 0 100 0	2 completin 0 0 0 2 0	100.0 g my wil 14.8 7.4 14.9 44.4 18.5	27 I and grief 4 2 4 12 5
Total 5.17(9) (Patient) I have in counselling Strongly disagree Somewhat disagree Neither agree/disagree Somewhat agree Strongly agree Total	44.5 received 16.7 11.1 8.3 33.3 41.7 44.5	2 1 1 1 1 4 5 12	33.3 ort wit 22.2 0 22.2 44.4 0 33.3	9 h adv 2 0 2 4 0 9	0 25.0 50.0 0 14.8	2 4 e plan 0 1 1 2 0 4	30.0 7.4 0 0 0 0 100 0 7.4	2 completin 0 0 2 0 2 0 2	100.0 g my wil 14.8 7.4 14.9 44.4 18.5 100.0	0 27 I and grief 4 2 4 12 5 27
Total 5.17(9) (Patient) I have in counselling Strongly disagree Somewhat disagree Neither agree/disagree Somewhat agree Strongly agree Total 5.17(10) (Patient) I have	44.5 received 16.7 11.1 8.3 33.3 41.7 44.5 received	2 1 1 2 1 4 5 12 d ass	33.3 ort wit 22.2 0 22.2 44.4 0 33.3 istance	9 h adv 2 0 2 4 0 9 e as n	0 25.0 25.0 50.0 0 14.8 eeded	2 4 e plan 1 1 2 0 4 4	0 0 0 100 0 7.4 0 0 0 100 0 7.4	2 completin 0 0 2 0 2 0 2	100.0 g my wil 14.8 7.4 14.9 44.4 18.5 100.0	0 27 I and grief 4 2 4 12 5 27
Total 5.17(9) (Patient) I have in counselling Strongly disagree Somewhat disagree Neither agree/disagree Somewhat agree Strongly agree Total 5.17(10) (Patient) I have Strongly disagree	44.5 received 16.7 11.1 8.3 33.3 41.7 44.5 received 8.3	2 12 3 1 1 4 5 12 d ass 1	33.3 ort wit 22.2 0 22.2 44.4 0 33.3 istance 55.6	9 h adv 2 0 2 4 0 9 9 e as n	30.0 14.8 vance car 0 25.0 50.0 0 14.8 eeded 25.0	2 4 e plan 1 1 2 0 4 1 2	30.0 7.4 nning, co 0 0 0 0 0 7.4	2 completin 0 0 2 0 2 0 2 0 2	23.0 100.0 g my wil 14.8 7.4 14.9 44.4 18.5 100.0	0 27 I and grief 4 2 4 12 5 27
Total 5.17(9) (Patient) I have in counselling Strongly disagree Somewhat disagree Somewhat agree Strongly agree Total 5.17(10) (Patient) I have Strongly disagree Somewhat agree Strongly agree Total 5.17(10) (Patient) I have Strongly disagree Somewhat disagree Somewhat disagree	44.5 received 16.7 11.1 8.3 33.3 41.7 44.5 received 8.3 8.3	2 12 2 1 1 4 5 12 d ass 1 1	33.3 ort wit 22.2 0 22.2 44.4 0 33.3 istance 55.6 0	9 h adv 2 0 2 4 0 9 9 5 0	30.0 14.8 vance car 0 25.0 50.0 0 14.8 eeded 25.0 0 14.8 eeded 25.0	2 4 e plan 1 1 2 0 4 1 0 - - - - - - - - - - - - -	30.0 7.4 nning, co 0 0 0 100 0 7.4 0 50.0	2 completin 0 0 2 0 2 0 2 0 1 1	23.0 100.0 g my wil 14.8 7.4 14.9 44.4 18.5 100.0 25.9 7.4	0 27 I and grief 4 2 4 12 5 27 7 2 -
Total 5.17(9) (Patient) I have in counselling Strongly disagree Somewhat disagree Somewhat agree Strongly agree Total 5.17(10) (Patient) I have Strongly disagree Somewhat agree Strongly agree Total 5.17(10) (Patient) I have Strongly disagree Somewhat disagree Neither agree/disagree	41.7 44.5 received 16.7 11.1 8.3 33.3 41.7 44.5 received 8.3 8.3 41.7	2 12 2 1 1 4 5 12 d ass 1 1 5	33.3 ort wit 22.2 0 22.2 44.4 0 33.3 istance 55.6 0 22.2	9 h adv 2 0 2 4 0 9 9 5 0 2	30.0 14.8 vance car 0 25.0 50.0 0 14.8 eeded 25.0 0 14.8 eeded 25.0 0	2 4 e plan 1 1 2 0 4 1 0 1 0 1 0 1 1 0 1 1 1 2 0 4 1 1 1 2 0 4 1 1 1 2 0 1 1 1 2 0 1 1 1 1 2 0 1 1 1 1 2 0 1 1 1 1 2 0 1 1 1 1 2 0 1 1 1 1 1 1 1 1 1 1 1 1 1	30.0 7.4 nning, co 0 0 0 0 0 7.4 0 50.0 0	2 completin 0 0 2 0 2 0 2 0 1 0 1	23.0 100.0 g my wil 14.8 7.4 14.9 44.4 18.5 100.0 25.9 7.4 29.7	0 27 I and grief 4 2 4 12 5 27 7 2 8
Total 5.17(9) (Patient) I have in counselling Strongly disagree Somewhat disagree Neither agree/disagree Strongly agree Total 5.17(10) (Patient) I have Strongly disagree Somewhat agree Strongly agree Total 5.17(10) (Patient) I have Strongly disagree Somewhat disagree Neither agree/disagree Somewhat agree	41.7 44.5 received 16.7 11.1 8.3 33.3 41.7 44.5 received 8.3 8.3 41.7 16.7	12 supp 2 1 4 5 12 d ass 1 5 1 5 12 3 1 5 1 5 2	33.3 ort wit 22.2 0 22.2 44.4 0 33.3 istance 55.6 0 22.2 22.2 22.2	9 h adv 2 0 2 4 0 9 9 5 0 2 2 2	0 25.0 25.0 50.0 0 14.8 eeded 25.0 0 25.0 25.0 25.0	2 4 e plan 1 1 2 0 4 1 0 1 1 0 1 1 0 1 1 0 1 1 1 2 0 4 1 1 1 2 0 4 1 1 1 2 0 1 1 1 2 0 1 1 1 2 0 1 1 1 2 0 1 1 1 2 0 1 1 1 2 0 1 1 1 1 2 1 1 1 1 1 2 0 1 1 1 1 1 1 1 1 1 1 1 1 1	30.0 7.4 0 0 0 0 0 7.4 0 50.0 0 50.0	2 completin 0 0 2 0 2 0 2 0 1 0 1 0 1	23.0 100.0 g my wil 14.8 7.4 14.9 44.4 18.5 100.0 25.9 7.4 29.7 22.2	0 27 I and grief 4 2 4 12 5 27 7 2 8 6
Total5.17(9) (Patient) I have in counsellingStrongly disagreeSomewhat disagreeSomewhat agree/disagreeSomewhat agreeStrongly agreeTotal5.17(10) (Patient) I haveStrongly disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeStrongly disagreeSomewhat disagreeSomewhat disagreeSomewhat agreeSomewhat agreeSomewhat agreeSomewhat agreeStrongly agree	41.7 44.5 received 16.7 11.1 8.3 33.3 41.7 44.5 received 8.3 8.3 41.7 16.7 25.0	2 12 supp 2 1 1 4 5 12 d ass 1 1 5 2 3	33.3 ort wit 22.2 0 22.2 44.4 0 33.3 istance 55.6 0 22.2 22.2 0	9 h adv 2 0 2 4 0 9 9 5 0 2 2 2 0	30.0 14.8 vance car 0 25.0 50.0 0 14.8 eeded 25.0 0 14.8 eeded 25.0 0 25.0 25.0 25.0 25.0 25.0	2 4 e plan 1 2 0 4 1 0 1 1 1 1 1 1 1 1 1 1 1 1 1	30.0 7.4 nning, co 0 0 0 100 0 7.4 0 50.0 0 50.0 0 50.0 0	Image: light constraints 0 0 0 0 2 0 2 0 2 0 2 0 1 0 1 0	23.0 100.0 g my wil 14.8 7.4 14.9 44.4 18.5 100.0 25.9 7.4 29.7 22.2 14.8	0 27 1 and grief 4 2 4 12 5 27 7 2 8 6 4
Total 5.17(9) (Patient) I have in counselling Strongly disagree Somewhat disagree Somewhat agree Strongly agree Total 5.17(10) (Patient) I have Strongly disagree Somewhat agree Strongly agree Total 5.17(10) (Patient) I have Strongly disagree Somewhat disagree Neither agree/disagree Somewhat agree Strongly disagree Strongly disagree Somewhat agree Strongly agree Total	41.7 44.5 received 16.7 11.1 8.3 33.3 41.7 44.5 received 8.3 8.3 41.7 16.7 25.0 44.5	2 12 supp 2 1 1 4 5 12 d ass 1 1 5 2 3 12	33.3 ort wit 22.2 0 22.2 44.4 0 33.3 istance 55.6 0 22.2 22.2 0 33.3	9 h adv 2 0 2 4 0 9 9 5 0 2 2 0 2 0 9	30.0 14.8 vance car 0 25.0 50.0 0 14.8 eeded 25.0 0 14.8 eeded 25.0 0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 14.8	2 4 e plan 1 2 0 4 1 0 1 1 1 1 1 4	30.0 7.4 0 0 0 0 0 0 7.4 0 50.0 0 50.0 0 50.0 0 7.4 0	I I 2 0 0 0 2 0 2 0 2 0 1 0 1 0 2 2	23.0 100.0 g my wil 14.8 7.4 14.9 44.4 18.5 100.0 25.9 7.4 29.7 22.2 14.8 100.0	0 27 1 and grief 4 2 4 12 5 27 7 2 8 6 4 27
Total 5.17(9) (Patient) I have i counsellingStrongly disagreeSomewhat disagreeSomewhat disagreeSomewhat agreeStrongly agreeTotal 5.17(10) (Patient) I have Strongly disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeStrongly disagreeSomewhat disagreeSomewhat disagreeSomewhat agreeSomewhat agreeSomewhat agreeStrongly agreeTotal 5.17(11) (Patient) I am s	41.7 44.5 received 16.7 11.1 8.3 33.3 41.7 44.5 received 8.3 8.3 41.7 16.7 25.0 44.5 atisfied 1	2 12 3 1 1 4 5 12 1 2 3 1 2 3 12 that n	33.3 ort wit 22.2 0 22.2 44.4 0 33.3 istance 55.6 0 22.2 22.2 0 22.2 0 33.3 ny fam	9 h adv 2 4 0 9 9 9 5 0 2 2 2 0 9 9 11y an	30.0 14.8 vance car 0 25.0 50.0 0 25.0 50.0 0 25.0 25.0 25.0 0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 14.8 d caregive	2 4 e plan 1 2 0 4 1 0 4 1 1 1 1 4 vers a	30.0 7.4 0 0 0 0 100 0 7.4 0 50.0 0 50.0 0 50.0 0 7.4 0 50.0 0 7.4 0 50.0 0 7.4 0 50.0 0 7.4 0 7.4 0 7.4 0	I I 2 0 0 0 2 0 2 0 2 0 1 0 2 0 1 0 2 0	23.0 100.0 g my wil 14.8 7.4 14.9 44.4 18.5 100.0 25.9 7.4 29.7 22.2 14.8 100.0 support	0 27 1 and grief 4 2 4 12 5 27 7 2 8 6 4 27 8 6 4 27 they need
Total 5.17(9) (Patient) I have i counsellingStrongly disagreeSomewhat disagreeSomewhat agree/disagreeSomewhat agreeStrongly agreeTotal 5.17(10) (Patient) I have Strongly disagreeSomewhat disagreeSomewhat disagreeStrongly disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat agreeStrongly agreeTotal 5.17(11) (Patient) I am s Strongly disagree	41.7 44.5 received 16.7 11.1 8.3 33.3 41.7 44.5 received 8.3 8.3 41.7 16.7 25.0 44.5 atisfied 1 8.3	12 supp 2 1 4 5 12 3 12 3 12 3 12 13 14 15 16 17 18 19 11 12 13 14 15 16 17 18 19 10 11 12 13 14 15 16 17 17 18 19 10 10	33.3 33.3 ort wit 22.2 44.4 0 33.3 istance 55.6 0 22.2 22.2 33.3 istance 55.6 0 22.2 22.2 0 33.3 ny fam 0	9 h adv 2 0 2 4 0 9 9 as n 5 0 2 2 0 9 9 illy an 0	30.0 14.8 vance car 0 25.0 50.0 0 14.8 eeded 25.0 0 25.0 25.0 0 25.0 0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 0	2 4 e plai 1 2 0 4 1 0 4 1 1 1 1 4 vers a 0	30.0 7.4 0 0 0 0 7.4 0 50.0 0 50.0 0 50.0 0 50.0 0 7.4 0 50.0 0 7.4 0	Image: light constraints 0 0 0 0 2 0 2 0 2 0 1 0 2 ving the solution	23.0 100.0 g my will 14.8 7.4 14.9 44.4 18.5 100.0 25.9 7.4 29.7 22.2 14.8 100.0 support 3.7	0 27 I and grief 4 2 4 12 5 27 7 2 8 6 4 27 they need 1
Total 5.17(9) (Patient) I have i counsellingStrongly disagreeSomewhat disagreeSomewhat agree/disagreeSomewhat agreeStrongly agreeTotal 5.17(10) (Patient) I have Strongly disagreeSomewhat disagreeSomewhat disagreeStrongly disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat agreeStrongly agreeTotal 5.17(11) (Patient) I am s Strongly disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat disagree	41.7 44.5 received 16.7 11.1 8.3 33.3 41.7 44.5 received 8.3 41.7 16.7 25.0 44.5 atisfied 1 8.3 8.3	12 supp 2 1 4 5 12 3 12 3 12 3 12 3 12 1 1 1 1 1 1 1	33.3 33.3 ort wit 22.2 44.4 0 33.3 istance 55.6 0 22.2 22.2 0 33.3 istance 55.6 0 22.2 0 33.3 ny fam 0 22.2	9 h adv 2 0 2 4 0 9 9 6 8 as n 5 0 2 2 0 9 9 11y an 0 2	30.0 14.8 vance car 0 25.0 50.0 0 14.8 eeded 25.0 0 25.0 25.0 0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 0 0 0 0	2 4 e plai 1 2 0 4 1 0 4 1 1 1 1 4 vers a 0 0 0	30.0 7.4 0 0 0 0 0 7.4 0 50.0 0 50.0 0 50.0 0 7.4 0 50.0 0 7.4 0 50.0 0 50.0 0 50.0 0 50.0	I I 2 0 0 0 0 2 0 2 0 2 0 2 0 1 0 2 /ing the solution 1 0 1 1 1 1 1 1 1 1 1 1 1 1 1	23.0 100.0 g my will 14.8 7.4 14.9 44.4 18.5 100.0 25.9 7.4 29.7 22.2 14.8 100.0 support 3.7 14.8	0 27 I and grief 4 2 4 12 5 27 7 2 8 6 4 27 they need 1 4
Total 5.17(9) (Patient) I have i counsellingStrongly disagreeSomewhat disagreeSomewhat agree/disagreeSomewhat agreeStrongly agreeTotal 5.17(10) (Patient) I have Strongly disagreeSomewhat disagreeSomewhat disagreeStrongly disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat agreeStrongly agreeTotal 5.17(11) (Patient) I am s Strongly disagreeSomewhat disagreeNeither agree/disagreeSomewhat disagreeNeither agree/disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeNeither agree/disagree	41.7 44.5 received 16.7 11.1 8.3 33.3 41.7 44.5 received 8.3 41.7 16.7 25.0 44.5 atisfied 1 8.3 8.3 16.7	12 supp 2 1 4 5 12 4 5 12 3 12 3 12 3 12 3 12 that m 1 2	33.3 ort wit 22.2 0 22.2 44.4 0 33.3 istance 55.6 0 22.2 22.2 0 33.3 ny fam 0 22.2 55.6	9 h adv 2 0 2 4 0 9 e as n 5 0 2 2 0 9 ily an 0 2 5	30.0 14.8 vance car 0 25.0 50.0 0 14.8 eeded 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 14.8 d caregiv 0 25.0	2 4 e plan 0 1 1 2 0 4 1 0 1 1 1 1 1 4 vers a 0 0 1 1 1 1 1 1 1 1 1 1 1 1 1	30.0 7.4 0 0 0 0 100 0 7.4 0 50.0 0 50.0 0 7.4 0 50.0 0 7.4 0 50.0 0 7.4 0 50.0 0 7.4 0 50.0 0 50.0 0	Image: light state Image:	25.0 100.0 g my wil 14.8 7.4 14.9 44.4 18.5 100.0 25.9 7.4 29.7 22.2 14.8 100.0 support 3.7 14.8 29.7	0 27 I and grief 4 2 4 12 5 27 7 2 8 6 4 27 they need 1 4 8
Total 5.17(9) (Patient) I have i counsellingStrongly disagreeSomewhat disagreeSomewhat agreeStrongly agree Total5.17(10) (Patient) I have Strongly disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat agreeSomewhat agreeSomewhat agreeStrongly agree Total5.17(11) (Patient) I am s Strongly disagreeSomewhat agreeSomewhat disagreeSomewhat agreeSomewhat agree	41.7 44.5 received 16.7 11.1 8.3 33.3 41.7 44.5 received 8.3 8.3 41.7 16.7 25.0 44.5 atisfied 1 8.3 8.3 16.7 16.7 16.7	12 supp 2 1 4 5 12 d ass 1 5 2 3 12 that n 1 2 2 3 12 2 2 3 1 1 2 2 2	33.3 33.3 ort wit 22.2 0 22.2 44.4 0 33.3 istance 55.6 0 22.2 0 33.3 ny fam 0 22.2 0 33.3 ny fam 0 22.2 55.6 22.2	9 h adv 2 0 2 4 0 9 e as n 5 0 2 2 0 9 illy an 0 2 5 2 2	30.0 14.8 vance car 0 25.0 50.0 0 14.8 eeded 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 14.8 d caregiv 0 25.0 75.0	2 4 e plan 1 1 2 0 4 1 0 4 vers a 0 0 1 3	30.0 7.4 nning, co 0 0 0 0 7.4 0 50.0 0 50.0 0 7.4 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0	I I 2 0 0 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 1 0 2 0 1 0 1 0 1 0 1 0 1 1	25.0 100.0 g my wil 14.8 7.4 14.9 44.4 18.5 100.0 25.9 7.4 29.7 22.2 14.8 100.0 support 3.7 14.8 29.7 29.6	0 27 I and grief 4 2 4 12 5 27 7 2 8 6 4 27 they need 1 4 8 8 8
Total 5.17(9) (Patient) I have i counsellingStrongly disagreeSomewhat disagreeSomewhat agreeSomewhat agreeStrongly agree Total5.17(10) (Patient) I have Strongly disagreeSomewhat disagreeSomewhat disagreeSomewhat agreeStrongly disagreeSomewhat disagreeSomewhat agreeStrongly disagreeSomewhat agreeStrongly agree Total5.17(11) (Patient) I am s Strongly disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeStrongly disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeStrongly disagreeSomewhat agreeSomewhat agreeSomewhat agreeStrongly agreeStrongly agree	41.7 44.5 received 16.7 11.1 8.3 33.3 41.7 44.5 received 8.3 41.7 16.7 25.0 44.5 atisfied 1 8.3 16.7 25.0 44.5 atisfied 1 8.3 16.7 50.0	12 supp 2 1 4 5 12 d ass 1 5 2 3 12 that n 1 2 6	33.3 33.3 ort wit 22.2 0 22.2 44.4 0 33.3 istance 55.6 0 22.2 0 33.3 istance 55.6 0 33.3 ny fam 0 22.2 55.6 22.2 0 32.3 ny fam 0 22.2 0 22.2 0 22.2 0	9 h adv 2 0 2 4 0 9 e as n 5 0 2 2 0 9 illy an 0 2 5 2 0 0 9 0 9 0 9 0 9 0 9 0 1 2 0 2 0 0 2 2 0 0 2 2 0 0 2 4 0 0 2 0 1 2 0 0 2 0 1 2 1 1 2 0 1 2 1 2	30.0 14.8 vance car 0 25.0 50.0 0 14.8 eeded 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 14.8 d caregiv 0 0 25.0 75.0 0	2 4 e plan 1 2 0 4 1 2 0 4 1 1 1 1 1 4 vers a 0 0 1 3 0	30.0 7.4 0 0 0 100 0 7.4 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0	I I 2 0 0 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 1 0 2 0 1 0 1 0 1 0 1 0	25.0 100.0 g my wil 14.8 7.4 14.9 44.4 18.5 100.0 25.9 7.4 29.7 22.2 14.8 100.0 support 3.7 14.8 29.7 29.6 22.2	0 27 I and grief 4 2 4 12 5 27 7 2 8 6 4 27 they need 1 4 8 6 6
Total 5.17(9) (Patient) I have i counsellingStrongly disagreeSomewhat disagreeSomewhat disagreeSomewhat agreeStrongly agree Total5.17(10) (Patient) I have Strongly disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeStrongly disagreeSomewhat disagreeSomewhat disagreeStrongly agree Total5.17(11) (Patient) I am s Strongly disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeStrongly disagreeSomewhat disagreeStrongly disagreeStrongly disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat agreeStrongly disagreeSomewhat agreeSomewhat agreeStrongly agreeTotal	44.5 received 16.7 11.1 8.3 33.3 41.7 44.5 received 8.3 8.3 41.7 16.7 25.0 44.5 atisfied 1 8.3 16.7 25.0 44.5 atisfied 1 8.3 16.7 50.0 44.5	12 supp 2 1 4 5 12 d ass 1 5 2 3 12 that n 1 2 6 12	33.3 33.3 ort wit 22.2 44.4 0 33.3 istance 55.6 0 22.2 22.2 0 33.3 istance 55.6 0 22.2 0 33.3 ny fam 0 22.2 55.6 22.2 0 33.3 ny fam 0 22.2 0 33.3	9 h adv 2 0 2 4 0 9 e as n 5 0 2 2 0 9 illy an 0 2 5 2 0 9 9 illy an 9 9 9 1 9 9 9 1 9 9 9 9 1 9 9 9 9 9 9	30.0 14.8 vance car 0 25.0 50.0 0 14.8 eeded 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 14.8 d caregiv 0 25.0 75.0 0 14.8	2 4 e plan 1 2 0 4 1 2 0 4 1 1 1 1 1 1 1 4 vers a 0 0 1 3 0 4	30.0 7.4 0 0 0 0 0 7.4 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 7.4	I I 2 0 0 0 2 0 2 0 2 0 2 0 2 0 2 0 1 0 2 0 1 0 1 0 1 0 1 0 1 0 2 2	25.0 100.0 g my wil 14.8 7.4 14.9 44.4 18.5 100.0 25.9 7.4 29.7 22.2 14.8 100.0 support 3.7 14.8 29.7 29.6 22.2 100.0	0 27 1 and grief 4 2 4 12 5 27 7 2 8 6 4 27 11 4 8 6 27 27 2 8 6 27 27
Total 5.17(9) (Patient) I have i counsellingStrongly disagreeSomewhat disagreeSomewhat disagreeSomewhat agreeSomewhat agreeStrongly agreeTotal 5.17(10) (Patient) I have Strongly disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat agreeStrongly agreeTotal 5.17(11) (Patient) I am s Strongly disagreeSomewhat disagreeSomewhat disagreeSomewhat agreeStrongly disagreeSomewhat agreeStrongly disagreeSomewhat agreeStrongly disagreeSomewhat agreeStrongly agreeTotal 5.17(12) (Patient) All the	41.7 44.5 received 16.7 11.1 8.3 33.3 41.7 44.5 received 8.3 8.3 41.7 16.7 25.0 44.5 atisfied 1 8.3 16.7 50.0 44.5 service	12 supp 2 1 4 5 12 d ass 1 1 5 12 3 12 3 12 3 12 3 12 3 12 3 12 6 12 6 12 6 12 5	33.3 33.3 ort wit 22.2 44.4 0 33.3 istance 55.6 0 22.2 22.2 22.2 0 33.3 ny fam 0 22.2 0 33.3 o 22.2 0 33.3 o 22.2 0 33.3 o 22.2 0 33.3	9 h adv 2 0 2 4 0 9 9 as n 5 0 2 2 0 9 illy an 0 2 5 2 0 9 9 ve bee	30.0 14.8 vance car 0 25.0 50.0 0 25.0 25.0 0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 14.8 d caregiv 0 25.0 75.0 0 14.8 an provid	2 4 e plan 0 1 2 0 4 1 0 4 vers a 0 0 1 3 0 4 ed pr	30.0 7.4 0 0 0 0 7.4 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 7.4 omptly	1 2 0 0 2 0 2 0 2 0 1 0 2 0 1 0 1 0 1 0 1 0 1 0 2	23.0 100.0 g my wil 14.8 7.4 14.9 44.4 18.5 100.0 25.9 7.4 29.7 22.2 14.8 100.0 support 3.7 14.8 29.7 29.6 22.2 100.0	0 27 I and grief 4 2 4 12 5 27 7 2 8 6 4 27 7 2 8 6 1 4 8 6 27 they need 1 4 8 6 27
Total 5.17(9) (Patient) I have i counsellingStrongly disagreeSomewhat disagreeSomewhat disagreeSomewhat agreeStrongly agree Total5.17(10) (Patient) I have Strongly disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat agreeStrongly agree Total5.17(11) (Patient) I am s Strongly disagreeSomewhat disagreeSomewhat disagreeSomewhat disagreeSomewhat agreeSomewhat agreeStrongly agreeTotal 5.17(12) (Patient) All the Strongly disagree	41.7 44.5 received 16.7 11.1 8.3 33.3 41.7 44.5 received 8.3 41.7 16.7 15.0 44.5 8.3 8.3 41.7 16.7 25.0 44.5 8.3 16.7 50.0 44.5 service 8.3	12 supp 2 1 4 5 12 d ass 1 4 5 12 d ass 1 1 5 2 3 12 that n 1 2 6 12 6 12 6 12 6 12 13	33.3 33.3 ort wit 22.2 44.4 0 33.3 istance 55.6 0 22.2 22.2 0 33.3 ny fam 0 22.2 0 33.3 o 22.2 0 33.3 o 22.2 0 33.3 o 33.3	9 h adv 2 0 2 4 0 9 4 0 9 2 2 0 9 9 10 2 2 0 9 9 10 2 2 0 9 9 10 2 2 0 9 9 9 10 10 10 10 10 10 10 10 10 10	30.0 14.8 vance car 0 25.0 50.0 0 25.0 50.0 0 25.0 25.0 25.0 25.0 25.0 25.0 14.8 d caregiv 0 25.0 14.8 d caregiv 0 14.8 an provid 0	2 4 e plan 1 2 0 4 1 2 0 4 1 1 1 1 1 1 1 1 1 4 vers a 0 0 1 3 0 4 ed pr	30.0 7.4 nning, co 0 0 0 0 7.4 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 7.4 omptly 0	1 2 0 0 0 2 0 2 0 2 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 2 0 1 0 2 0 1 0 2 0 1 0 2 0 1 0 2 0 1 0 2 0 2 0 2 0 2 0	23.0 100.0 g my wil 14.8 7.4 14.9 44.4 18.5 100.0 25.9 7.4 29.7 22.2 14.8 100.0 support 3.7 14.8 29.7 22.2 14.8 29.7 29.6 22.2 100.0 14.8	0 27 I and grief 4 2 4 12 5 27 7 2 8 6 4 27 7 2 8 6 4 8 6 27 they need 1 4
Total 5.17(9) (Patient) I have i counselling Strongly disagree Somewhat disagree Somewhat agree/disagree Somewhat agree Strongly agree Total 5.17(10) (Patient) I have Strongly disagree Somewhat disagree Strongly disagree Somewhat disagree Somewhat disagree Somewhat agree/disagree Somewhat agree Strongly agree Total 5.17(11) (Patient) I am s Strongly disagree Somewhat disagree Somewhat disagree Somewhat disagree Somewhat agree Strongly disagree Somewhat agree Strongly agree Total 5.17(12) (Patient) All the Strongly disagree Somewhat disagree Strongly disagree Somewhat disagree Somewhat disagree Strongly disagree Somewhat disagree	41.7 44.5 received 16.7 11.1 8.3 33.3 41.7 44.5 received 8.3 41.7 44.5 received 8.3 41.7 16.7 25.0 44.5 atisfied 1 8.3 8.3 16.7 50.0 44.5 eservice 8.3 33.3	12 supp 2 1 4 5 12 d ass 1 4 5 12 d ass 1 5 2 3 12 3 12 3 12 3 12 3 12 3 12 3 12 3 12 6 12 6 12 6 12 6 14	0 33.3 33.3 ort wit 22.2 0 22.2 44.4 0 33.3 istance 55.6 0 22.2 22.2 0 33.3 0 22.2 0 33.3 0 22.2 0 33.3 0 22.2 0 33.3 33.3 22.2 0	9 h adv 2 0 2 4 0 9 8 as n 5 0 2 2 0 9 9 11y an 0 2 5 2 0 9 9 9 9 9 9 9 9 9 9 9 9 9	30.0 14.8 vance car 0 25.0 50.0 0 25.0 50.0 0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 14.8 d caregiv 0 25.0 75.0 0 14.8 en provid 0 25.0	2 4 e plan 1 2 0 4 1 2 0 4 1 1 1 1 1 1 1 4 vers a 0 0 1 3 0 4 ed pr 1 3 0 1 1 1 1 2 0 4 1 1 1 2 0 4 1 1 1 1 1 1 1 1 1 1 1 1 1	30.0 7.4 nning, co 0 0 0 0 7.4 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0 0 50.0	I I 2 0 0 0 0 2 0 2 0 2 0 2 0 1 0 2 ving the solution 1 0 1 0 2 0 1 0 2 0 1 0 2 0 1 0 2 0 1 0 1 1 0 2 0 1 1 0 1 0 1	23.0 100.0 g my wil 14.8 7.4 14.9 44.4 18.5 100.0 25.9 7.4 29.7 22.2 14.8 100.0 support 3.7 14.8 29.7 22.2 14.8 29.7 29.6 22.2 100.0 14.8 29.6 22.2 100.0	0 27 I and grief 4 2 4 12 5 27 7 2 8 6 4 27 they need 1 4 8 6 27 they need 1 4 8 6 27 4 8 8 6 27 4 8
Total 5.17(9) (Patient) I have i counselling Strongly disagree Somewhat disagree Somewhat agree Strongly agree Total 5.17(10) (Patient) I have Strongly agree Total 5.17(10) (Patient) I have Strongly disagree Somewhat disagree Somewhat disagree Somewhat agree/disagree Somewhat agree Strongly agree Total 5.17(11) (Patient) I am s Strongly disagree Somewhat disagree Somewhat disagree Somewhat disagree Somewhat agree Strongly disagree Somewhat agree Strongly agree Total 5.17(12) (Patient) All the Strongly disagree Somewhat disagree Somewhat disagree Strongly disagree Somewhat disagree Neither agree/disagree Somewhat disagree Somewhat disagree Somewhat disagree Somewhat dis	41.7 44.5 received 16.7 11.1 8.3 33.3 41.7 44.5 received 8.3 41.7 16.7 25.0 44.5 atisfied 1 8.3 8.3 16.7 25.0 44.5 atisfied 1 8.3 8.3 16.7 50.0 44.5 eservice 8.3 33.3 8.3	12 supp 2 1 4 5 12 4 5 12 3 12 3 12 3 12 3 12 3 12 3 12 3 12 6 12 6 12 6 14 4 1	33.3 33.3 ort wit 22.2 44.4 0 33.3 istance 55.6 0 22.2 0 22.2 0 33.3 o 22.2 0 33.3 0 22.2 0 33.3 c2.2 0 33.3 22.2 0 33.3 22.2 0 33.3 22.2 0 33.3 22.2 0 33.3 22.2 11.1	9 h adv 2 0 2 4 0 9 9 9 9 2 2 0 9 9 1 1 2 0 9 9 1 2 0 9 9 1 2 0 9 9 1 2 0 9 9 1 2 0 2 2 0 9 9 1 2 1 2 0 2 2 4 0 2 2 4 0 2 2 4 0 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	30.0 14.8 vance car 0 25.0 25.0 50.0 0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 25.0 75.0 0 25.0 75.0 0 14.8 en provid 0 25.0 50.0	2 4 e plai 1 2 0 4 1 2 0 4 1 1 1 1 1 1 1 4 vers a 0 0 1 3 0 4 ed pr 1 3 0 1 1 1 2 0 4 1 1 1 2 0 4 1 1 1 2 0 4 1 1 1 1 1 1 1 1 1 1 1 1 1	30.0 7.4 0 0 0 0 0 0 0 50.0 0	I I 2 0 0 0 0 2 0 2 0 2 0 2 0 1 0 2 ving the : 0 1 0 2 0 1 0 2 0 1 0 2 0 1 0 2 0 1 0 2 0	23.0 100.0 g my will 14.8 7.4 14.9 44.4 18.5 100.0 25.9 7.4 29.7 22.2 14.8 100.0 support 3.7 14.8 29.7 29.6 22.2 100.0 14.8 29.6 22.2 100.0	0 27 I and grief 4 2 4 12 5 27 7 2 8 6 4 27 they need 1 4 8 6 27 they need 1 4 8 6 27 4 8 4 4 4 4 4 4 4
Total 5.17(9) (Patient) I have i counselling Strongly disagree Somewhat disagree Neither agree/disagree Somewhat agree Strongly agree Total 5.17(10) (Patient) I have Strongly disagree Somewhat disagree Strongly disagree Somewhat disagree Somewhat disagree Somewhat agree/disagree Somewhat agree Strongly agree Total 5.17(11) (Patient) I am s Strongly disagree Somewhat disagree Somewhat disagree Somewhat disagree Somewhat agree/disagree Somewhat agree Strongly agree Total 5.17(12) (Patient) All the Strongly disagree Somewhat disagree	41.7 44.5 received 16.7 11.1 8.3 33.3 41.7 44.5 received 8.3 41.7 46.5 received 8.3 41.7 16.7 25.0 44.5 atisfied 8.3 16.7 16.7 50.0 44.5 eservice 8.3 33.3 8.3 33.3 8.3 33.3 8.3	12 supp 2 1 4 5 12 d ass 1 5 1 5 1 5 1 1 5 2 3 12 that n 1 2 6 12 5 1 2 6 1 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	33.3 33.3 ort wit 22.2 0 22.2 44.4 0 33.3 istance 55.6 0 22.2 0 33.3 istance 55.6 0 22.2 0 33.3 pt fam 0 22.2 0 33.3 22.2 0 33.3 pt fam 0 22.2 0 33.3 pt fam 0 22.2 0 33.3 pt fam 33.3 22.2 0 33.3 pt fam 33.3 22.2 11.1 33.3	9 h adv 2 0 2 4 0 9 e as n 5 0 2 2 0 9 illy an 0 2 5 2 0 9 9 ve bee 3 2 1 3	30.0 14.8 vance car 0 25.0 50.0 0 25.0 50.0 0 14.8 eeded 25.0 25.0 25.0 25.0 25.0 14.8 d caregiv 0 25.0 75.0 0 14.8 en provid 0 25.0 50.0 25.0	2 4 e plan 1 1 2 0 4 1 0 4 vers a 0 0 1 1 4 vers a 0 0 1 3 0 4 ed pr 1 3 0 1 1 1 1 1 2 1 1 1 2 1 1 1 1 1 1 1 1 1 1 1 1 1	30.0 7.4 nning, co 0 0 0 0 50.0	I I 2 0 0 0 2 0 2 0 2 0 2 0 2 0 2 0 1 0 2 0 1 0 1 0 1 0 2 0 1 0 2 0 1 0 2 0 1 0 2 0 1 0 1 0 1 0 1 0 1 0 1 0	23.0 100.0 g my wil 14.8 7.4 14.9 44.4 18.5 100.0 25.9 7.4 29.7 22.2 14.8 29.7 22.2 14.8 29.7 29.6 22.2 100.0 14.8 29.7 29.6 22.2 100.0 14.8 29.6 22.2 100.0	0 27 I and grief 4 2 4 12 5 27 7 2 8 6 4 2 8 6 27 they need 1 4 8 6 27 4 8 4 8 4 8 4 6

					1					1
Total	44.5	12	33.3	9	14.8	4	7.4	2	100.0	27
5.17(13) (Patient) I am a	ble to ac	cess	the me	edica	tion that I	need	l easily	-		
Strongly disagree	16.7	2	22.2	2	0	0	0	0	14.8	4
Somewhat disagree	16.7	2	33.3	3	50.0	2	0	0	25.9	7
Neither agree/disagree	8.3	1	22.2	2	0	0	0	0	11.2	3
Somewhat agree	16.7	2	22.2	2	50.0	2	100.0	2	29.6	8
Strongly agree	41.7	5	0	0	0	0	0	0	18.5	5
Total	44.5	12	33.3	9	14.8	4	7.4	2	100.0	27
5.17(14) (Patient) My pe	rsonal c	hoice	s for e	nd-of	-life care	inclu	ding hos	spice, ho	me deat	h and
voluntary assisted dyin	g, have l	been	suppor	rted a	nd I have	beer	n able to	make in	formed of	decisions
Strongly disagree	8.3	1	37.5	3	0	0	0	0	15.4	4
Somewhat disagree	16.7	2	25.0	2	25.0	1	0	0	19.2	5
Neither agree/disagree	41./	5	25.0	2	50.0	2	100.0	2	42.4	11
Somewhat agree	8.3	1	12.5	1	25.0	1	0	0	11.5	3
Strongly agree	25.0	3	0	0	0	0	0	0	11.5	3
Total	46.1	12	39.8	8	15.4	4	7.7	2	100.0	26
5.17(15) (Patient) All sei	rvices I r	need	are affo	ordab	le					
Strongly disagree	8.3	1	55.5	5	25.0	1	0	0	25.9	7
Somewhat disagree	41.7	5	11.1	1	25.0	1	0	0	25.9	7
Neither agree/disagree	16.7	2	0	0	0	0	50.0	1	11.2	3
Somewhat agree	8.3	1	33.3	3	50.0	2	50.0	1	25.9	7
Strongly agree	25.0	3	0	0	0	0	0	0	11.1	3
Total	44.5	12	33.3	9	14.8	4	7.4	2	100.0	27
5.17(16) (Patient) I belie	ve the st	taff p	rovidin	g my	care hav	e the	training	, knowle	dge and	expertise
to provide the quality of	f care l n	eed			-	-				
Strongly disagree	8.3	1	0	0	0	0	0	0	3.7	1
Somewhat disagree	0	0	0	0	25.0	1	0	0	3.7	1
Neither agree/disagree	8.3	1	44.4	4	0	0	0	0	18.6	5
Somewhat agree	25.0	3	33.3	3	50.0	2	0	0	29.6	8
Strongly agree	58.3	7	22.2	2	25.0	1	100	2	44.4	12
Total	44.5	12	33.3	9	14.8	4	7.4	2	100.0	27
5.17(17) (Patient) I have	choice of	of pro	ovider a	and/o	r staff wh	o atte	end my ł	nome		
Strongly disagree	16.7	2	44.4	4	0	0	0	0	22.2	6
Somewhat disagree	22.2	2	0	0	25.0	1	50.0	1	14.8	4
Neither agree/disagree	41.7	5	11.1	1	50.0	2	50.0	1	33.4	9
Somewhat agree	8.3	1	22.2	2	25.0	1	0	0	14.8	4
Strongly agree	33.3	4	0	0	0	0	0	0	14.8	4
Total	44.5	12	33.3	9	14.8	4	7.4	2	100.0	27
5.17(18) (Patient) Family	y/caregiv	ver st	ress ar	nd bu	rnout has	s beei	n reduce	d with se	ervice pr	ovision
Strongly disagree	8.3	1	22.2	2	0	0	0	0	11.1	3
Somewhat disagree	8.3	1	0	0	25.0	1	50.0	1	11.1	3
Neither agree/disagree	16.7	2	55.5	5	25.0	1	0	0	29.7	8
Somewhat agree	16.7	2	22.2	2	50.0	2	50.0	1	25.9	7
Strongly agree	50.0	6	0	0	0	0	0	0	22.2	6
Total	44.5	12	33.3	9	14.8	4	7.4	2	100.0	27
5.17(19) (Patient) I have	a good	relati	onship	with	health pr	ofess	sionals i	nvolved	in my ca	re
Strongly disagree	8.3	1	0	0	0	0	0	0	3.7	1
Somewhat disagree	0	0	0	0	25.0	1	0	0	3.7	1
Neither agree/disagree	0	0	22.2	2	0	0	0	0	7.4	2
Somewhat agree	25.0	3	55.6	5	0	0	0	0	29.6	8
Strongly agree	66.7	8	22.2	2	75.0	3	100.0	2	55.6	15
Total	44.5	12	33.3	9	14.8	4	7.4	2	100.0	27

Table 74 Question 5.17 (Descriptive Statistics)

go. If I allent - now mach do you agree with	i the	10110	ng state	illenits :		Varianaa	Confidence
Statement	n	min	max	mean	SD	variance	Interval
5.17(1) (Patient) I have been supported in my choice to remain in own home	27	1	5	3.8	1.4	1.9	0.3
5 17(2) (Patient) Personal choices		-	-				
proferences and goals of care are being							
respected	27	1	5	3.7	1.5	2.1	0.3
5.17(3) (Patient) All communication and							
care provided has been culturally and							
gender inclusive and respectful	26	1	5	4.1	1.2	1.4	0.2
5.17(4) (Patient) I have access to support							
24 hours a day and 7 days a week	27	1	5	3.0	1.5	2.2	0.3
5.17(5) (Patient) My quality of life has							
been improved by remaining in my own							
home	27	1	5	4.4	1.0	1	0.2
5.17(6) (Patient) Access to community							
services have reduced my need for							
ambulance transport and hospital							
admissions	27	1	5	2.9	1.9	1.4	0.3
5.17(7) (Patient) I have been included in							0.0
all decision-making about care and							
services provided	27	1	5	39	12	15	0.2
5 17(8) (Patient) I am comfortable with the	21			0.0	1.2	1.0	0.2
level of symptom management that is							
available	27	1	5	36	13	16	0.2
5 17(0) (Patient) I have received support	21		5	5.0	1.5	1.0	0.2
with advance care planning completing							
with advance care planning, completing	27	1	5	3 1	1 2	17	0.2
5 47(10) (Patient) have received	21	- 1	5	5.4	1.5	1.7	0.2
5.17(10) (Patient) I have received	27	1	F	2.0	1 1	1.0	0.2
assistance as needed	21	1	5	2.9	1.4	1.9	0.3
5.17(11) (Patient) I am satisfied that my							
family and caregivers are receiving the	27	1	F	2 5	1 1	10	0.2
Support they need	21	1	5	3.5	1.1	1.2	0.2
5.17(12) (Patient) All the services I need	07	4	F	2.0	4.4	1.0	0.2
nave been provided promptly	21	1	Э	3.0	1.4	1.9	0.3
5.17(13) (Patient) I am able to access the	~7		-	0.4		1.0	0.0
medication that I need easily	21	1	5	3.1	1.4	1.9	0.3
5.17(14) (Patient) My personal choices for							
end-of-life care including nospice, nome							
death and voluntary assisted dying, have							
been supported and I have been able to	~~		-	0.0	10		0.0
make informed decisions	20	1	5	2.8	1.2	1.4	0.3
5.17(15) (Patient) All services I need are	~ 7		-	0.7		4.0	
affordable	27	1	5	2.7	1.4	1.9	0.3
5.17(16) (Patient) I believe the staff							
providing my care have the training,							
knowledge and expertise to provide the	~-		_				
quality of care I need	27	1	5	4.1	1.1	1.1	0.2
5.17(17) (Patient) I have choice of provider	07		-		4.0	4.0	0.0
and/or stan who attend my nome	21	1	5	2.9	1.3	1.8	0.3
5.17(18) (Patient) Family/caregiver stress							
and burnout has been reduced with	0 -		_	~ .			
service provision	27	1	5	3.4	1.3	1.6	0.2
5.17(19) (Patient) I have a good							
relationship with health professionals			_				-
involved in my care	27	1	5	4.3	1.0	1.0	0.2

Q5.17 Patient - How much do you agree with the following statements?

Table 75 Question 5.18

Q5.18 Patient - How do you receive palliative care services in your home? Select all that apply												
Methods	Peri-urban		Ru	Rural		nal	Metrop	olitan	Total res	ponses		
	%	n	%	n	%	n	%	n	%	n		
Face-to-face	91.7	11	66.7	6	75.0	3	100.0	2	81.5	22		
Phone	83.3	10	77.8	7	75.0	3	0	0	74.1	20		
Video	8.3	1	0	0	25.0	1	0	0	7.4	2		
Email	8.3	1	0	0	0	0	0	0	3.7	1		
Text	8.3	1	0	0	0	0	50.0	1	7.4	2		
Apps	8.3	1	0	0	0	0	0	0	3.7	1		
Hospital/clinic	8.3	1	11.1	1	0	0	0	0	7.4	2		
Total respondents	44.5	12	33.3	9	14.8	4	7.4	2	100.0	27		
Total responses	52.0	26	28.0	14	14.0	7	6.0	3	100.0	50		

Table 76 Question 5.19

Q5.19 Patient - Have yo apply	u exper	ienceo	d any pr	obler	ns with (digita	l health c	ptions?	Select all	that
Problems	Peri-u	rban	Ru	ral	Regio	nal	Metrop	olitan	Total res	ponses
	%	n	%	n	%	n	%	n	%	n
No problems	66.7	8	0	0	50.0	2	50.0	1	40.7	11
Lack access/skill	16.7	2	33.3	3	50.0	2	0	0	25.9	7
Not used digital options	0	0	55.5	5	0	0	0	0	18.5	5
Poor mobile coverage	8.3	1	77.8	7	0	0	0	0	29.6	8
Unreliable/no internet	8.3	1	66.7	6	25.0	1	0	0	29.6	8
Not offered digital option	8.3	1	11.1	1	0	0	50.0	1	11.1	3
Total	44.4	12	33.3	9	14.8	4	7.5	2	100.0	27

Table 77 Question 5.20

Q5.20 Patient - How much do you agree with the following statements?											
	Peri-urban		Rural		Regional		Metropolitan		Total responses		
	%	n	%	n	%	n	%	n	%	n	
5.20(1) (Patient) Digital health is an important tool in ensuring I have regular communication with my palliative care team											
Strongly disagree	8.3	1	0	0	0	0	0	0	3.7	1	
Somewhat disagree	0	0	0	0	0	0	0	0	0	0	
Neither agree/disagree	41.7	5	55.6	5	50.0	2	50.0	1	48.1	13	
Somewhat agree	25.0	3	33.3	3	25.0	1	50.0	1	29.6	8	
Strongly agree	25.0	3	11.1	1	25.0	1	0	0	18.6	5	
Total	44.5	12	33.3	9	14.8	4	7.4	2	100.0	27	
5.20(2) (Patient) I have the resources and confidence to use digital health effectively											
Strongly disagree	0	0	0	0	0	0	0	0	0	0	
Somewhat disagree	0	0	33.3	3	50.0	2	0	0	18.5	5	
Neither agree/disagree	16.7	2	0	0	0	0	50.0	1	11.1	3	
Somewhat agree	25.0	3	55.6	5	25.0	1	0	0	33.3	9	
Strongly agree	58.3	7	11.1	1	25.0	1	50.0	1	37.1	10	
Total	44.4	12	33.3	9	14.8	4	7.5	2	100.0	27	
5.20(3) (Patient) Digital health consultations are an acceptable alternative to face-to-face consultations with medical professionals											
Strongly disagree	0	0	0	0	0	0	0	0	0	0	
Somewhat disagree	16.7	2	0	0	25.0	1	0	0	11.5	3	
Neither agree/disagree	16.6	2	62.5	5	25.0	1	50.0	1	34.7	9	
Somewhat agree	50.0	6	25.0	2	50.0	2	50.0	1	42.3	11	

Strongly agree	16.7	2	12.5	1	0	0	0	0	11.5	3	
Total	46.1	12	30.8	8	15.4	4	7.7	2	100.0	26	
5.20(4) (Patient) Digital health means that I can access medical support that I would not be able to											
access due to distance											
Strongly disagree	0	0	0	0	0	0	0	0	0	0	
Somewhat disagree	8.3	1	0	0	25.0	1	0	0	7.7	2	
Neither agree/disagree	8.3	1	25.0	2	0	0	0	0	11.6	3	
Somewhat agree	66.7	8	62.5	5	25.0	1	100.0	2	61.5	16	
Strongly agree	16.7	2	12.5	1	50.0	2	0	0	19.2	5	
Total	46.1	12	30.8	8	15.4	4	7.7	2	100.0	26	
5.20(5) (Patient) I am comfortable to remain at home knowing that I can access support at any time											
Strongly disagree	7.7	1	0	0	0	0	0	0	3.8	1	
Somewhat disagree	23.1	3	0	0	25.0	1	0	0	15.4	4	
Neither agree/disagree	7.7	1	57.1	4	25.0	1	0	0	23.1	6	
Somewhat agree	23.1	3	42.9	3	25.0	1	100.0	2	34.6	9	
Strongly agree	38.4	5	0	0	25.0	1	0	0	23.1	6	
Total	50.0	13	26.9	7	15.4	4	7.7	2	100.0	26	

Table 78 Question 5.21

Q5.21 Patients - Overall are you satisfied with the access to community palliative care services in your location?											
Opinion	Peri-u	Peri-urban		Rural		Regional		Metropolitan		Total responses	
	%	n	%	n	%	n	%	n	%	n	
Yes	66.7	8	50.0	3	75.0	3	0	0	60.9	14	
No	33.3	4	50.0	3	25.0	1	33.3	1	39.1	9	
Total	52.2	12	26.1	6	17.4	4	4.3	1	100.0	23	